# Enhancing Indoor Air Quality and Minimizing Airborne Virus Dispersion Under Various Ventilation Strategies While Maintaining Thermal Comfort

Zahi Ali Saeed Alqarni

A thesis submitted in partial fulfilment of the requirement of the degree of Doctor of Philosophy



School of Engineering

Cardiff University

March 2025

## Abstract

Indoor air quality (IAQ) and thermal comfort are critical factors that influence occupant health, productivity, and general well-being in office environments. The COVID-19 pandemic has further highlighted the importance of effective ventilation strategies in mitigating the transmission of airborne viruses. This research investigates the performance of three ventilation strategies: indoor recirculation systems (4-way ceiling cassette air conditioners), natural ventilation, and mixed mode ventilation (AC + natural ventilation) in maintaining optimal IAQ, thermal comfort, and infection control in an open-plan office setting. Using a combination of computational fluid dynamics (CFD) simulations and real-world environmental monitoring, this study evaluates airflow patterns, pollutant dispersion, and thermal regulation under different ventilation conditions.

This thesis explicitly demonstrates that IAQ, thermal comfort, and airborne virus transmission are deeply interconnected. Poor air quality not only impairs comfort and productivity but also prolongs aerosol suspension time, elevating infection risk. As such, ventilation strategies must be designed to address these three aspects holistically.

The findings reveal that the air conditioning (AC) system, while providing controlled air distribution, often leads to stagnation zones that reduce air mixing efficiency and increase pollutant accumulation. Natural ventilation, though beneficial under favourable conditions, exhibits inconsistent performance due to ex-

#### Abstract

ternal weather variations, leading to excessive humidity fluctuations and temperature instability. In contrast, mixed mode ventilation emerges as the most effective strategy, offering improved airflow uniformity, improved pollutant dilution, and greater adaptability to seasonal changes. The results demonstrate that a well-optimised hybrid system, which strategically combines an AC system and natural ventilation, can mitigate the limitations of standalone approaches by balancing fresh air intake, controlled temperature regulation, and efficient humidity management.

This research contributes to a novel integrated methodological framework that bridges CFD simulations with IoT-based environmental monitoring, ensuring robust validation of ventilation performance under real-world conditions.

The findings have significant implications for the optimisation of heating, ventilation and air conditioning (HVAC) and public health policies, particularly in the post-pandemic era, where IAQ is a major concern. By addressing critical knowledge gaps in ventilation performance, this thesis provides practical recommendations for facility managers, architects, and policy makers to develop more resilient and health-conscious indoor environments.

### Acknowledgement

I would like to express my sincere gratitude to King Khalid University for granting me the opportunity to serve as a teaching assistant and for the continuous support provided throughout my academic journey, from my Master's degree to the completion of this challenging yet fulfilling PhD journey.

I would like to express my heartfelt gratitude to my primary supervisor, Professor Yacine Rezgui, for his exceptional guidance, unwavering support, and trust throughout this journey. His insightful advice and dedicated mentorship have been invaluable in shaping the direction of this research and helping me navigate its many challenges. It has been a privilege to work under his supervision, and I will always remain deeply grateful for his encouragement and belief in my work.

I would also like to express my deep appreciation to my second supervisor, Dr. Ioan Petri, for his invaluable support, constructive feedback, and thoughtful contributions throughout this journey. His guidance has been instrumental in helping me refine my ideas and strengthen the quality of my work, and I am truly grateful for the time and insight he generously shared.

To my parents, words cannot fully capture the depth of my gratitude. Your unconditional love, endless prayers, and unwavering support have been my anchor and source of strength. Your belief in me has been a constant source of motivation throughout this journey and I owe so much of this achievement to you.

To my brothers and sisters, thank you for your constant encouragement and love.

#### Acknowledgement

Your support has been a source of comfort and inspiration, especially during the challenging moments of this journey.

To my wife, Lama, I am eternally grateful for your love, patience, and unwavering encouragement. Your support has been my foundation, giving me the strength to persevere during the most difficult times. I could not have reached this milestone without you by my side and for that I am forever grateful.

To my beloved daughter, Yara, I am deeply sorry for the moments when I could not be there for you, especially during your surgery. Being far from you was one of the hardest sacrifices I have ever made. Please, know that every effort was made for our future, and I am incredibly proud of your resilience and understanding.

To my beloved daughter, Ghada, I regret not being there when you came into this world. Despite my absence, my love for you knows no bounds, and you have brought endless joy and meaning to my life.

To my small and precious family, this achievement is as important to you as to mine. Your sacrifices, love, and unwavering faith in me have been my greatest source of strength. I dedicate this milestone to you with all my heart.

# List of Publications

- Alqarni, Z., Rezgui, Y., Petri, I. and Ghoroghi, A., 2024. Viral infection transmission and indoor air quality: A systematic review. Science of the Total Environment, p.171308.
- Alqarni, Z., Rezgui, Y., Petri, I. and Ghoroghi, A., 2023, June. Factors and strategies affecting indoor air quality in educational buildings. In 2023 IEEE International Conference on Engineering, Technology and Innovation (ICE/ITMC) (pp. 1-6). IEEE.
- CO2 Monitoring as an Indicator of Ventilation Performance in a Teaching Office (Under review)

A	bstra	ict		i
A	cknov	wledge	ment	iii
Li	st of	Public	cations	v
C	onter	nts		vi
Li	st of	Figure	es x	vii
Li	st of	Tables	s xx	cvi
1	Intr	oducti	on	1
	1.1	Indoor	Air Quality	1
		1.1.1	Importance of Indoor Air Quality	4
		1.1.2	Strategies to Improve Indoor Air Quality	4
		1.1.3	Indoor Environmental Quality	5
	1.2	Respir	ratory Viruses	6
		1.2.1	Transmission Methods	6

	1.3	Modelling	7
	1.4	Thermal Comfort	8
		1.4.1 Assessment of Thermal Comfort	9
		1.4.2 Role of Computational Fluid Dynamics	9
	1.5	Research Motivation	.0
	1.6	Aim and Objectives	.1
	1.7	Aim and Objectives	.1
	1.8	Research Contribution	2
	1.9	Thesis Overview	.4
	1.10	Conclusion	5
2	Lite	rature Review 1	7
	2.1	Introduction	7
	2.2	Literature Review Approach	20
	2.3	Indoor Air Quality (IAQ) During and Before COVID-19 2	26
		2.3.1 Severe Acute Respiratory Syndrome	27
		2.3.2 Influenza	27
		2.3.3 Middle East Respiratory Syndrome	28
		2.3.4 Other Respiratory Viruses	28
		2.3.5 SARS-CoV-2 (COVID-19)	29
	2.4	Transmission Pathways of Respiratory Viruses	81

	2.4.2	Airborne Surveillance of SARS-CoV-2	33
2.5	Mitiga	ation Strategies in Indoor Environments	39
	2.5.1	Humidity and Temperature Control	40
	2.5.2	Ventilation Strategies	42
	2.5.3	Air Cleaners and High-efficiency Particulate Air Filters	42
	2.5.4	Ultraviolet Radiation	43
	2.5.5	Masks and Social Distancing	43
2.6	Model	ling Transmission Using Computational Fluid Dynamics and	
	Nume	rical Modelling	44
2.7	Energ	y Consumption	46
2.8	Therm	nal Comfort Standards	48
	2.8.1	ASHRAE 55 Standard	49
	2.8.2	European Standards: EN 15251 and EN 16798-1	49
	2.8.3	ISO 7730 Standard	50
	2.8.4	Integrative Analysis and Limitations	51
2.9	Resear	rch Gaps	51
	2.9.1	Lack of Air and Surface Samples	51
	2.9.2	Inadequate Consideration of Occupancy Patterns and Be-	
		haviour	52
	2.9.3	Insufficient Monitoring of Indoor and Outdoor Environ-	
		mental Parameters	52
	2.9.4	Shortage of Evaluation and Simulation Studies	53

	2.9.5	Neglect of Critical Factors Influencing Ventilation System	
		Performance	54
	2.9.6	Energy Consumption and Cost	54
	2.9.7	Air Quality Based on Dynamic Data	55
	2.9.8	Digital Twin Model and Machine Learning Models	55
	2.9.9	Infection Transmission in Indoor Environments	56
	2.9.10	Indoor Air Quality Factors	57
	2.9.11	Nonengineering Control Strategies	58
	2.9.12	Engineering Control Strategies	58
	2.9.13	Computational Fluid Dynamics	59
2.10	Conclu	nsion	60
Res	earch l	Design and Methodology	62
<b>Res</b> 3.1		Design and Methodology	<b>62</b> 62
	Introd		
3.1	Introd	uction $\ldots$	62
3.1	Introd Demor 3.2.1	uction	62 64
3.1 3.2	Introd Demor 3.2.1	uction	62 64 65
3.1 3.2	Introd Demor 3.2.1 Monito	uction	62 64 65 71
3.1 3.2	Introd Demor 3.2.1 Monito 3.3.1 3.3.2	uction	62 64 65 71 75
<ul><li>3.1</li><li>3.2</li><li>3.3</li></ul>	Introd Demor 3.2.1 Monito 3.3.1 3.3.2	uction	<ul> <li>62</li> <li>64</li> <li>65</li> <li>71</li> <li>75</li> <li>76</li> </ul>
<ul><li>3.1</li><li>3.2</li><li>3.3</li></ul>	Introd Demon 3.2.1 Monito 3.3.1 3.3.2 CFD M	uction	<ul> <li>62</li> <li>64</li> <li>65</li> <li>71</li> <li>75</li> <li>76</li> <li>76</li> </ul>
	2.10	<ul> <li>2.9.6</li> <li>2.9.7</li> <li>2.9.8</li> <li>2.9.9</li> <li>2.9.10</li> <li>2.9.11</li> <li>2.9.12</li> <li>2.9.13</li> </ul>	Performance       Performance         2.9.6       Energy Consumption and Cost         2.9.7       Air Quality Based on Dynamic Data         2.9.8       Digital Twin Model and Machine Learning Models

			3.4.3.1	Key Steps in the CFD Workflow	78
		3.4.4	Model D	evelopment	79
			3.4.4.1	Assumptions of this Research	79
			3.4.4.2	Governing Equation	81
		3.4.5	Boundary	y Condition	83
		3.4.6	Solver Se	tting	85
		3.4.7	Thermal	Comfort Simulation	89
	3.5	Conclu	sion		91
4	Asse	essmen	t and C	orrelation of Indoor Air Quality Parameters	
	with	ı Airbo	orne Infe	ction Risks	93
	4.1	Introd	uction of C	CO2 and Other Parameters in an Indoor Environment	93
	4.2	The O	ffice Build	ling and the Locations of the Sensors $\ldots$ .	94
		4.2.1	Data Col	lection	96
	4.3	Impact	t of CO2 of	on IAQ and Comfort	96
		4.3.1	CO2 as a	an Indicator of Ventilation and Infection Risk	97
		4.3.2	Analysis	of Collected CO2 Data	99
	4.4	Impact	of Relati	ve Humidity on Air Quality and Comfort 1	101
		4.4.1	Analysis	of the Humidity Data	104
	4.5	Impact	of Temp	erature on IAQ and Comfort	105
		4.5.1	Analysis	of Temperature Data	107
	4.6	Impact	of Partic	culate Matter (PM2.5 and PM10) on Air Quality	
		and Co	omfort		109

		4.6.1	Importance of Monitoring Particulate Matter	111	
		4.6.2	Analysis of PM2.5 and PM10 Data	113	
	4.7	7 Discussion			
		4.7.1	CO2 as an Indicator of Ventilation and Infection Risk	115	
		4.7.2	Temperature Fluctuations and Thermal Comfort	117	
		4.7.3	Relative Humidity and Its Impact on Indoor Air Quality and Infection Risk	118	
		4.7.4	Particulate Matter and Air Quality Concerns	119	
	4.8	Conclu	asion	120	
<b>5</b>	Air	Recirc	culation Systems: Air Conditioners	122	
	5.1	Introd	uction	122	
		5.1.1	Air Recirculation and Infection Risk	124	
		5.1.2	Chapter Objectives	125	
	5.2	Metho	dology	126	
	5.3	Geome	etry	128	
	5.4	Model	ling Meshing	129	
	5.5	Bound	lary Conditions	131	
	5.6	Result	s	133	
		5.6.1	Validation 1: Validation of the Computational Fluid Dy- namics Model Against Experimental Data	134	
		5.6.2	Validation 2: Validation of the Computational Fluid Dy- namics Model Against Real Sensor Data	136	

		5.6.3	Air Velocity Pattern	38
		5.6.4	Room Temperature and Relative Humidity 13	39
		5.6.5	Predicted Mean Vote and Predicted Percentage of Dissatisfied14	10
		5.6.6	Aerosol Dispersion and Infection risk	12
	5.7	Discus	sion $\ldots \ldots 15$	57
		5.7.1	Comparison with ASHRAE Standards, EN 16798, and ISO7730 standards15	57
		5.7.2	Identification of Air Conditioning Dead Zones 15	59
		5.7.3	Impact of Air Velocity on Virus Dispersion	30
		5.7.4	Impact of Relative Humidity on Comfort and Airborne Virus	
			Stability	30
		5.7.5	Scenario Performance Evaluation	31
	5.8	Recon	nmendations for Optimising AC Performance $\ldots \ldots \ldots$	52
	5.9	Conclu	1sion	<u>;</u> 3
6	Nat	ural V	Tentilation 16	57
	6.1	Introd	uction $\ldots \ldots 16$	37
		6.1.1	Background and Importance of Natural Ventilation 16	37
		6.1.2	The Role of Natural Ventilation in Enhancing Indoor Air Quality and Thermal Comfort, and Mitigating Infection Risk16	58
		6.1.3	Cross-ventilation as an Effective Natural Ventilation Strategy17	70
		6.1.4	Chapter Objectives	71
	6.2	Metho	odology	'2

6.3	Scenar	rio Develo	pment	173		
6.4	Bound	Boundary Conditions				
6.5	Result	Results				
	6.5.1	Scenario	1	177		
		6.5.1.1	Air Velocity Pattern	178		
		6.5.1.2	Room Temperature and Relative Humidity	179		
		6.5.1.3	Predicted Mean Vote and Predicted Percentage of			
			Dissatisfied	180		
	6.5.2	Scenario	2	182		
		6.5.2.1	Air Velocity Pattern	182		
		6.5.2.2	Room Temperature and Relative Humidity	184		
		6.5.2.3	Predicted Mean Vote and Predicted Percentage of			
			Dissatisfied	185		
	6.5.3	Scenario	3	187		
		6.5.3.1	Air Velocity Pattern	188		
		6.5.3.2	Room Temperature and Relative Humidity	189		
		6.5.3.3	Predicted Mean Vote and Predicted Percentage of			
			Dissatisfied	191		
	6.5.4	Scenario	4	192		
		6.5.4.1	Air Velocity Pattern	193		
		6.5.4.2	Room Temperature and Relative Humidity	194		
		6.5.4.3	Predicted Mean Vote and Predicted Percentage of			
			Dissatisfied	196		

 $\mathbf{7}$ 

		6.5.4.4 Aerosol Dispersion and Infection Risk $\ldots$ .	198
6.6	Discus	ssion	213
	6.6.1	Comparison with ASHRAE, EN 16798, and ISO 7730 Stan-	
		dards	213
	6.6.2	Identification of Dead Zones in Natural Ventilation	216
	6.6.3	Impact of Air Velocity on Virus Dispersion	217
	6.6.4	Impact of Relative Humidity on Comfort and Airborne Virus	
		Stability	219
	6.6.5	Scenario Performance Evaluation	220
	6.6.6	Comparison of Natural Ventilation Against Air Conditioning	g223
	6.6.7	Recommendations for Optimising Natural Ventilation Per-	
		formance	225
6.7	Concl	usion $\ldots$	227
Miz	ced Ve	ntilation	<b>231</b>
7.1	Introd	luction	231
	7.1.1	Chapter Objectives	233
7.2	Metho	odology	233
7.3	Scena	rio Development	235
7.4	Bound	dary Conditions	235
7.5	Result	$\mathrm{ts}$	237
	7.5.1	Scenario 1	238
		7.5.1.1 Air Velocity Pattern	238

		7.5.1.2	Room Temperature and Relative Humidity	239
		7.5.1.3	Predicted Mean Vote and Predicted Percentage of	
			Dissatisfied	241
	7.5.2	Scenario	2	243
		7.5.2.1	Air Velocity Pattern	243
		7.5.2.2	Room Temperature and Relative Humidity	244
		7.5.2.3	Predicted Mean Vote and Predicted Percentage of	
			Dissatisfied	245
	7.5.3	Scenario	3	247
		7.5.3.1	Air Velocity Pattern	247
		7.5.3.2	Room Temperature and Relative Humidity	248
		7.5.3.3	Predicted Mean Vote and Predicted Percentage of	
			Dissatisfied	250
		7.5.3.4	Aerosol Dispersion and Infection risk	252
7.6	Discus	sion		267
	7.6.1	Compari	son with ASHRAE, EN 16798, and ISO 7730 Stan-	
		dards .		267
	7.6.2	Identific	ation of Dead Zones in Mixed-mode Ventilation	270
	7.6.3	Impact of	of Air Velocity on Virus Dispersion	272
	7.6.4	Impact of	of Relative Humidity on Comfort and Airborne Virus	
		Stability	•••••••••••••••••••••••••••••••••••••••	274
	7.6.5	Scenario	Performance Evaluation	276

	7.7	Recommendations for Optimising Mixed-mode Ventilation Perfor-						
		mance		279				
	7.8	Conclu	Conclusion					
8	Con	clusio	n	284				
	8.1	Revisi	ting Research Questions and Key Findings	284				
		8.1.1	Research Question 1: How do viral particles transmit within indoor environments and what factors influence their transmission dynamics?	285				
		8.1.2	Research Question 2: Can CO2 levels serve as reliable indi- cators to assess IAQ and signal the potential risk of airborne infections in office environments?	286				
		019		200				
		8.1.3	Research Question 3: What ventilation strategies can be implemented to minimise the risk of airborne infections in office environments?	286				
		8.1.4	Research Question 4: How can ventilation systems be con- figured to maintain a good IAQ, ensure TC, and reduce virus dispersion in the office environment?	288				
	8.2	Implic	ations for Building Design and Operation	289				
	8.3	Comp	arison of Mixed-mode Ventilation Against Natural Ventila-					
		-	nd Air Conditioning	290				
	8.4	Limita	ations	292				
	8.5	Future	e Research Directions	293				
	8.6	Final	Reflections	295				

### Bibliography

# List of Figures

1.1	TC factors	8
2.1	PRISMA flow diagram for systematic review	22
2.2	Number of IAQ documents based on the year of publication $\ldots$	24
2.3	Publication categories	24
2.4	Major routes of respiratory viruses transmission $[18]$	26
2.5	IAQ measurements	35
2.6	Distribution of research publications by IA control strategy	39
3.1	An overview of the research methodology	64
3.2	Location of the Queen's Building	65
3.3	Queen's Building	65
3.4	Location of the teaching office	68
3.5	The teaching office S1.19	68
3.6	The teaching office S1.19 using Ansys	69
3.7	Teaching office area	69
3.8	Teaching office area: 18 south-facing windows	70

3.9	The 18 south-facing windows	70
3.10	The doors in the teaching office	71
3.11	Design of the AC system	71
3.12	Schematic diagram of the indoor environment monitoring system .	72
3.13	Location of the sensors in the teaching office	75
3.14	Steps of Ansys	80
4.1	Location of two sensors in the big office	95
4.2	Location of two sensors in the small office	95
4.3	How SARS-CoV-2-containing aerosols are lost through breathing out, breathing in, and other processes; also showing how CO2 is released, breathed in, and lost through other processes in an indoor setting [248]	99
4.4	CO2 level disruption during October	102
4.5	CO2 level disruption during November	102
4.6	RH level disruption during October	105
4.7	RH level disruption during October	106
4.8	Temperature level disruption during October	109
4.9	Temperature level disruption during November	110
4.10	PM2.5 level disruption during October	115
4.11	PM2.5 level disruption during November	116
4.12	PM10 level disruption during October	117
4.13	PM10 level disruption during November	117

ļ	5.1	Overview of the methodology for AC simulations	127
ļ	5.2	Locations of infected individuals	129
Ę	5.3	Orthogonal quality	131
Ę	5.4	Velocity in centre line with the length of $0.2m$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	135
Ę	5.5	DPM concentration in centre line with the length of 0.2m $~$	135
ļ	5.6	CO2 level in the teaching office during AC $\ldots \ldots \ldots \ldots$	138
ļ	5.7	Air velocity contour in z direction during AC	139
ļ	5.8	Temperature contour in z direction during AC $\ldots$	140
ŗ	5.9	RH contour in z direction during AC $\ldots \ldots \ldots \ldots \ldots \ldots$	141
ļ	5.10	PMV contour in z direction during AC $\ldots \ldots \ldots \ldots \ldots$	142
ļ	5.11	PPD contour in z direction during AC	143
Ę	5.12	Particle transmission from the first infected individual during cough-	
		ing	144
ļ	5.13	Aerosol dispersion through coughing on X, Y, and Z directions for the first individual during AC	145
ļ	5.14	Particle transmission from the second infected individual during	
		coughing	145
ļ	5.15	Aerosol dispersion through coughing on X, Y, and Z directions for the second individual during AC	146
ŗ	5.16	Particle transmission from the third infected individual during coughing	147
Ę	5.17	Aerosol dispersion through coughing in X, Y, and Z directions for the third individual during AC	148

5.18	Particle transmission from the fourth infected individual during	
	coughing	149
5.19	Aerosol dispersion through coughing in X, Y, and Z directions for	1 -
	the fourth individual during AC	150
5.20	Particle transmission from the first infected individual during sneez-	
	ing	151
5.21	Aerosol dispersion through sneezing in X, Y, and Z directions for	
	the first individual during AC	152
5.22	Particle transmission from the second infected individual during	
	sneezing	152
5.23	Aerosol dispersion through sneezing in X, Y,and Z directions for	
	the second individual during AC	153
5.24	Particle transmission from the third infected individual during	
	sneezing	154
5.25	Aerosol dispersion through sneezing in X, Y,and Z directions for	
	the third individual during AC	155
5.26	Particle transmission from the fourth infected individual during	
	sneezing	156
5.27	Aerosol dispersion through sneezing in X, Y,and Z directions for	
	the fourth individual during AC	157
6.1	Overview of the methodology for NV simulations	173
6.2	Air velocity contour in z direction during Scenario 1 of NV	
6.3	Temperature contour in z direction during Scenario 1 of NV $\ldots$	179
6.4	RH contour in z direction during Scenario 1 of NV	180

6.5 PMV contour in z direction during Scenario 1 of NV 181	
6.6 PPD contour in z direction during Scenario 1 of NV	
6.7 Air velocity contour in z direction during Scenario 2 of NV 183	
6.8 Temperature contour in z direction during Scenario 2 of NV $\ldots$ . 184	
6.9 RH contour in z direction during Scenario 2 of NV	
6.10 PMV contour in z direction during Scenario 2 of NV $\ldots \ldots \ldots 186$	
6.11 PPD contour In z direction during Scenario 2 of NV	
6.12 Air velocity contour in z direction during Scenario 3 of NV 188	
6.13 Temperature contour in z direction during Scenario 3 of NV $$ 189	
6.14 RH contour in z direction during Scenario 3 of NV	
6.15 PMV contour in z direction during Scenario 3 of NV $\ . \ . \ . \ .$ 191	
6.16 PPD contour in z direction during Scenario 3 of NV 192	
6.17 Air velocity contour in z direction during Scenario 4 of NV 194	
6.18 Temperature contour in z direction during Scenario 4 of NV $$ 195	
6.19 RH contour in z direction during Scenario 4 of NV	
6.20 PMV contour in z direction during Scenario 4 of NV $\ .$ 197	
6.21 PPD contour in z direction during Scenario 4 of NV	
6.22 Particle transmission from the first infected individual during cough-	
ing	
6.23 Aerosol dispersion through coughing in X, Y, and Z directions for the first individual during NV	
6.24 Particle transmission from the second infected individual during	
coughing	

6.25	Aerosol dispersion through coughing in X, Y, and Z directions for the second individual during NV	202
6.26	Particle transmission from the third infected individual during coughing	
6.27	Aerosol dispersion through coughing in X, Y, and Z directions for the third individual during NV	204
6.28	Particle transmission from the fourth infected individual during coughing	205
6.29	Aerosol dispersion through coughing in X, Y, and Z directions for the fourth individual during NV	206
6.30	Particle transmission from the first infected individual during sneez- ing	207
6.31	Aerosol dispersion in X, Y, and Z directions for the first individual during NV	208
6.32	Particle transmission from the second infected individual during sneezing	209
6.33	Aerosol dispersion in X, Y, and Z directions for the second individ- ual during NV	210
6.34	Particle transmission from the third infected individual during sneezing	210
6.35	Aerosol dispersion in X, Y, and Z directions for the third individual during NV	211
6.36	Particle transmission from the fourth infected individual during sneezing	212

6.37	Aerosol dispersion in X, Y, and Z directions for the fourth individual during NV	213
7.1	Overview of the methodology for mixed ventilation simulations	234
7.2	Air velocity contour in z direction during Scenario 1 of mixed ven- tilation	239
7.3	Temperature contour in z direction during Scenario 1 of mixed ventilation	240
7.4	RH contour in z direction during Scenario 1 of mixed ventilation .	241
7.5	PMV contour in z direction during Scenario 1 of mixed ventilation	242
7.6	PPD contour in z direction during Scenario 1 of mixed ventilation	242
7.7	Air velocity contour in z direction during Scenario 2 of mixed ven- tilation	243
7.8	Temperature contour in z direction during Scenario 2 of mixed ventilation	244
7.9	RH contour in z direction during Scenario 2 of mixed ventilation .	245
7.10	PMV contour in z direction during Scenario 2 of mixed ventilation	246
7.11	PPD contour in z direction during Scenario 2 of mixed ventilation	247
7.12	Air velocity contour in z direction during Scenario 3 of mixed ven- tilation	248
7.13	Temperature contour in z direction during Scenario 3 of mixed ventilation	249
7.14	RH contour in z direction during Scenario 3 of mixed ventilation .	250
7.15	PMV contour in z direction during Scenario 3 of mixed ventilation	251

7.16	PPD contour in z direction during Scenario 3 of mixed ventilation	252
7.17	Particle transmission from the first infected individual during cough- ing	253
7.18	Aerosol dispersion through coughing on X, Y, and Z directions for the first individual during mixed ventilation	254
7.19	Particle transmission from the second infected individual during coughing	255
7.20	Aerosol dispersion through coughing in X, Y, and Z directions for the second individual during mixed ventilation	256
7.21	Particle transmission from the third infected individual during coughing	257
7.22	Aerosol dispersion through coughing in X, Y, and Z directions for the third individual during mixed ventilation	258
7.23	Particle transmission from the fourth infected individual during coughing	258
7.24	Aerosol dispersion through coughing in X, Y, and Z directions for the fourth individual during mixed ventilation	259
7.25	Particle transmission from the first infected individual during sneez- ing	
7.26	Aerosol dispersion in X, Y, and Z directions for the first individual during mixed ventilation	261
7.27	Particle transmission from the second infected individual during sneezing	262
7.28	Aerosol dispersion in X, Y, and Z directions for the second indi- vidual during mixed ventilation	263

7.29	Particle transmission from the third infected individual during	
	sneezing	264
7.30	Aerosol dispersion in X, Y, and Z directions for the third individual	
	during mixed ventilation	265
7.31	Particle transmission from the fourth infected individual during	
	sneezing	266
7.32	Aerosol dispersion in X, Y, and Z directions for the fourth individ-	
	ual during mixed ventilation	267

# List of Tables

1.1	Chemical Composition and Volumetric Content of Typical Dry At-	
	mospheric Air [2] Note: Percentage values for some trace gases are	
	not reported in the original source and are thus omitted	3
2.1	The percentage of publications in each category	22
2.2	Publications investigating IAQ before COVID-19	30
2.3	Information from publications on the SARS-COV-2 transmission through indoor surfaces in different indoor environments	33
2.4	Information from publications on the SARS-COV-2 transmission through indoor air in different indoor environments.	36
2.5	Mean values of IAQ measurements	37
2.6	Mean values of IAQ measurements	38
2.7	Environmental conditions for the stability and inactivation of var- ious viruses	41
3.1	An overview of the technical specifications of the utilised sensing devices.	73
3.2	ASHRAE scale used to evaluate the thermal sensation of individuals.	90

5.1	The selected mesh	131
5.2	Overview of simulation cases during the AC	132
5.3	Comparison of real data and CFD simulation under AC	138
5.4	Summary of AC performance	161
6.1	NV scenarios and their environmental conditions	174
6.2	Overview of simulation cases during the NV	176
6.3	Summary of boundary conditions for NV simulations	177
6.4	Comparative summary of scenario performance	223
7.1	Mixed ventilation scenarios and their environmental conditions	235
7.2	Overview of simulation cases during the MV	237
7.3	Comparative summary of scenario performance	278

# Chapter 1

### Introduction

In recent years, the quality of indoor environments has gained increasing attention due to its direct impact on public health, energy use, and building sustainability. The COVID-19 pandemic has further underscored the importance of managing indoor air quality (IAQ), thermal comfort (TC), and ventilation to minimise airborne virus transmission risks [1]. This research sits at the intersection of environmental engineering, building design, and public health, aiming to optimise indoor conditions in office environments through a robust, simulation-backed evaluation of ventilation strategies. By addressing the dynamic relationships between IAQ, TC, and airborne pathogen dispersion, the study provides timely insights that are vital to both the architecture-engineering-construction (AEC) sector and health-focused policy development.

### 1.1 Indoor Air Quality

To define air quality, it is crucial to establish a baseline for what constitutes "clean" or "standard" air. Clean air typically refers to dry atmospheric air found in remote and unpolluted areas, such as rural locations or over the ocean, where anthropogenic and natural pollutants are minimal [2]. These areas are considered ideal baselines because they provide a reference point that closely approximates the natural state of the atmosphere before significant human intervention. Table 1.1 provides an overview of the chemical composition of clean and dry atmospheric air, serving as a reference for air quality assessments. In particular, atmospheric air also contains varying amounts of water vapour (0.1% to 3%), influenced by temperature and environmental conditions [2]. Although Table 1.1 outlines the standard composition of clean air, trace amounts of other components, including ammonia, sulphur dioxide, formaldehyde, carbon monoxide, iodine, sodium chloride, and particulate matter (PM; e.g., dust and pollen), may still be present within acceptable limits of 'clean' air [2].

Based on this standard, air quality is defined by the degree to which it deviates from clean air due to the presence of pollutants. Air quality improves as pollutant concentrations decrease. Any substance in the air that adversely affects the health, comfort, or productivity of humans and animals, hinders plant growth, or accelerates equipment degradation is classified as an airborne pollutant. Airborne pollutants may exist as solid particles (e.g., PM, dust, and pollen), liquid droplets (e.g., aerosols and vapours), or gases (e.g., carbon monoxide, nitrogen oxides, and volatile organic compounds; VOCs) [2].

Substance	Content (% a In volume)	Concentration (ppm a In volume)
Nitrogen	$78.084 \pm 0.004$	780,840
Oxygen	$20.946 \pm 0.002$	209,460
Argon	$0.934 \pm 0.001$	9,340
CO2	$0.033 \pm 0.001$	330
Neon		18
Helium		5.2
Methane		1.2
Krypton		0.5
Hydrogen		0.5
Xenon		0.08
Nitrogen dioxide		0.02
O3		0.01-0.04

 Table 1.1: Chemical Composition and Volumetric Content of Typical Dry At 

 mospheric Air [2]

Note: Percentage values for some trace gases are not reported in the original source and are thus omitted.

### 1.1.1 Importance of Indoor Air Quality

With individuals spending more than 90% of their time indoors in many regions, indoor air quality (IAQ) has become a critical determinant of health and wellbeing [3]. Air quality in homes, offices, and other enclosed spaces significantly affects occupant comfort, productivity, and cognitive performance [4]. IAQ can be improved through various approaches, including fostering green environments through plant-based solutions, adopting nonpharmaceutical interventions, and employing engineering controls. Engineering strategies, such as adequate ventilation, air filtration systems, and pollutant control mechanisms, have been shown to be effective in improving IAQ [5].

IAQ in office environments directly impacts employee health and productivity. Research has consistently shown that pollutants such as PM, VOCs, formaldehyde, and carbon dioxide (CO2) frequently exceed recommended levels [6, 7]. Pollutant levels often peak during working hours and colder months [7], correlating with increased reports of symptoms such as fatigue, headaches, and dryness of the air, which are commonly associated with sick building syndrome [8]. Poor IAQ has been associated with higher health complaints, increased absenteeism, and reduced workplace productivity [9].

### 1.1.2 Strategies to Improve Indoor Air Quality

To address the challenges in the previous section, building managers and facility operators should monitor key IAQ parameters, manage pollution sources, and adopt appropriate ventilation strategies. Providing personalised controls, such as adjustable temperature and ventilation settings, further enhances occupant satisfaction and workplace productivity [9]. These measures enable organisations to mitigate the adverse effects of poor IAQ, fostering a healthier and more conducive work environment. In addition, integration of air-cleaning technologies, such as HEPA filters, activated carbon filters, and ultraviolet germicidal irradiation (UVGI), can help remove pollutants and pathogens effectively.

### 1.1.3 Indoor Environmental Quality

Indoor environmental quality (IEQ) encompasses several factors, including IAQ, pollution levels, ventilation, thermal comfort (TC), lighting, and acoustics. IAQ is considered satisfactory when harmful levels of chemical and biological pollutants are absent and most of the inhabitants do not express dissatisfaction. However, indoor environments are often characterised by elevated concentrations of various pollutants from both internal and external sources. Key pollutants in office environments include ozone (O3), VOCs, aldehydes, PM (PM2.5 and PM10), and ultrafine particles (UFPs). These pollutants originate from sources such as electronic devices, building materials, furniture, occupant activities, and cleaning products. Poor IAQ not only poses health risks but also negatively impacts TC, which is a critical factor in overall satisfaction with IEQ. TC refers to the "condition of mind that expresses satisfaction with the thermal environment." It is commonly assessed using metrics such as air temperature, mean radiant temperature, air velocity, intensity of turbulence, and relative humidity [9]. Maintaining both IAQ and TC is essential for creating safe, healthy, and productive indoor spaces.

Indoor Environmental Quality particularly thermal comfort and Indoor Air Quality significantly influences occupants' health and well-being [10]. Furthermore, studies have shown that inadequate IAQ contributes to airborne virus transmission, making it essential to consider IAQ and thermal factors together when designing infection control strategies [11].

### 1.2 Respiratory Viruses

Respiratory viruses, such as influenza, respiratory syncytial virus (RSV), parainfluenza, adenovirus, rhinovirus, and coronaviruses, have a significant global health impact, particularly among vulnerable populations such as children, the elderly, and immunosuppressed individuals [12]. These viruses primarily cause upper respiratory tract infections, which can progress to severe lower respiratory tract complications, leading to high healthcare costs and higher mortality rates [13].

The COVID-19 pandemic caused by SARS-CoV-2 underscored the widespread threat of respiratory viruses. Public health and social measures (PHSMs) implemented during the pandemic, such as physical separation, mask use, and improved ventilation, were effective in reducing the incidence and transmission of several respiratory viruses from 23% to 98% [14]. However, as restrictions eased, non-COVID respiratory viruses re-emerged, often outside their usual seasons [15].

Office environments are particularly vulnerable to the transmission of SARS-CoV-2 due to close physical interactions, shared spaces, and inadequate ventilation [16]. Research confirms that respiratory viruses can remain viable in aerosols for hours and on surfaces for days, further highlighting the need for effective mitigation strategies in indoor environments [17].

### 1.2.1 Transmission Methods

Respiratory viruses, including influenza, RSV, rhinovirus, adenovirus, and coronaviruses such as SARS-CoV, predominantly target the upper respiratory tract and are recognised as significant human pathogens. Their potential to cause widespread outbreaks and pandemics, combined with their socio-economic implications, underscores their importance to public health [12].

Transmission occurs through four primary mechanisms [18–20]:

- Direct contact: Physical interactions, such as handshakes or touching infected individuals, facilitate the direct transfer of pathogens.
- Indirect contact via foams: Contaminated surfaces act as intermediaries, allowing the transfer of viruses when touched by susceptible individuals.
- Respiratory droplets: Larger particles (>5 µm) expelled during coughing, sneezing, or talking usually settle quickly on surfaces or within short distances.
- Aerosols: Smaller particles (< 5 µm) remain suspended in the air for extended periods, increasing the potential for airborne virus transmission, particularly in poorly ventilated spaces.

Understanding these transmission pathways is crucial for developing targeted and effective mitigation strategies. In office settings, such strategies include optimising ventilation, implementing physical distancing protocols, and maintaining regular cleaning and disinfection routines to minimise viral spread.

### 1.3 Modelling

Computational Fluid Dynamics (CFD) is a powerful tool to simulate airflow, temperature distribution, and dispersion of airborne contaminant in indoor environments. The importance of CFD in the evaluation of ventilation strategies has increased significantly, particularly in the wake of the COVID-19 pandemic, where optimising air circulation has become crucial for infection control.

In this research, CFD simulations were used to compare the effectiveness of air conditioning (AC), natural, and mixed mode ventilation strategies to mitigate the transmission of airborne viruses and maintain TC. By integrating real-world sensor data (including concentrations of CO2, humidity, and PM), the precision of CFD predictions was validated, ensuring robust and reliable results. This approach allows for a detailed analysis of airflow stagnation (dead zones), viral particle dispersion, and the ability of different ventilation methods to regulate IAQ and TC.

The findings of this CFD-based investigation not only contribute to a deeper understanding of IAQ dynamics but also offer practical insights into how ventilation systems should be designed and operated to improve occupant health and productivity.

### 1.4 Thermal Comfort

TC in indoor environments, particularly offices, is a multifaceted concept that is influenced by factors such as air temperature, humidity, air movement, and heat generated by occupants and electronic devices [21]. It is defined as the state in which individuals feel satisfied with the thermal environment, achieved when the heat produced by metabolism is balanced with the heat lost from the body. However, this perception is subjective because TC varies between individuals due to physiological factors and environmental conditions, including room temperature, air velocity, RH, and radiant temperature, as illustrated in Figure 3.1 [22].



Figure 1.1: TC factors
#### 1.4.1 Assessment of Thermal Comfort

Two widely used approaches to assess TC are the predicted mean vote (PMV) and the predicted percentage of dissatisfied (PPD), developed by Fanger [23].

- PMV quantifies the thermal sensations of the residents on a scale ranging from -3 (cold) to +3 (hot), with 0 (neutral) representing thermal neutrality.
- PPD estimates the percentage of occupants likely to express dissatisfaction with the thermal environment. A PPD value of 10% or less is generally considered the threshold for acceptable levels of TC.

These indices provide a systematic framework for evaluating and optimising indoor thermal conditions.

#### 1.4.2 Role of Computational Fluid Dynamics

TC assessments often use CFD as a cost-effective and efficient alternative to experimental methods, which can be both time-intensive and expensive [22]. CFD simulations allow for a detailed analysis of airflow patterns, temperature distributions, and humidity levels within indoor spaces. By modelling these parameters, researchers and engineers can optimise heating, ventilation and air conditioning (HVAC) systems and ventilation strategies to improve comfort and energy efficiency.

Recent studies have demonstrated the effectiveness of CFD in predicting TC in various indoor environments, including offices, classrooms, and theatres [24]. These simulations provide insights into the following.

• Airflow optimisation: Identify areas with stagnant air or uneven temperature distribution. • System design improvements: Enhancing HVAC systems to achieve balanced thermal conditions.

## 1.5 Research Motivation

The motivation for this research comes from the increasing concern about IAQ and its critical role in the transmission of respiratory viruses, especially in office environments where occupants spend extended periods in enclosed spaces. The COVID-19 pandemic has highlighted the urgent need for effective strategies to mitigate the spread of indoor airborne infections. However, despite advances in IAQ management, several critical research gaps remain unaddressed, including:

- Integration of airborne virus transmission dynamics and TC: Although many studies focus on improving IAQ or optimising TC individually, there is a lack of comprehensive models that integrate airborne virus transmission dynamics with TC parameters. A holistic understanding of how ventilation strategies affect both infection risk reduction and occupant comfort is essential to designing healthier and more productive office environments.
- Comparison of ventilation strategies: Limited knowledge exists on how different ventilation strategies, such as AC, natural ventilation, and mixedmode ventilation, perform to balance IAQ improvements with TC. Understanding the trade-offs between infection risk reduction and maintaining comfortable indoor conditions is critical for identifying optimal ventilation approaches.
- CO2 levels as indicators of airborne virus transmission risk: Although CO2 levels are widely used as a proxy for ventilation effectiveness, their reliability as direct indicators of airborne virus transmission risk is not fully

understood. More research is needed to evaluate whether CO2 concentrations can predict the likelihood of airborne virus transmission and guide real-time ventilation adjustments while maintaining TC.

By addressing these gaps, this research aims to provide practical insights that will contribute to the design of safer and healthier indoor environments without compromising occupant comfort. The findings will have significant implications for HVAC management and infection control strategies, offering actionable solutions to reduce health risks while ensuring a comfortable indoor climate in office settings.

## 1.6 Aim and Objectives

## 1.7 Aim and Objectives

**Research Aim:** This thesis aims to investigate how different ventilation strategies influence indoor air quality (IAQ), thermal comfort (TC), and the transmission of respiratory viruses by airborne viruses in office environments. By combining a detailed literature review, computational simulations, and experimental data, the study aims to identify optimal ventilation configurations that improve the health and comfort of the occupant while reducing the risk of infection.

**Research Questions and Objectives:** To achieve this aim, the following research questions and their corresponding objectives are addressed throughout the thesis:

 Research Question 1: How are viral particles transmitted within indoor environments, and what factors influence their transmission dynamics? Objective 1: To critically review and synthesise the existing literature on airborne virus transmission mechanisms in indoor environments, examine the key factors influencing transmission dynamics, and evaluate current strategies used to minimise virus spread.

- Research Question 2: Can CO2 levels serve as reliable indicators to assess IAQ and signal the potential risk of airborne infections in office environments? Objective 2: To evaluate CO2 concentration as a proxy of ventilation effectiveness and potential infection risk through real-world sensor data and CFD modelling.
- Research Question 3: What ventilation strategies can be implemented to minimise the risk of airborne infections in office environments? Objective 3: To investigate and compare the performance of AC, natural and mixed-mode ventilation systems to reduce the spread of airborne viruses.
- Research Question 4: How can ventilation systems be configured to maintain a good IAQ, ensure TC, and reduce virus dispersion in the office environment? Objective 4: To identify and propose optimal ventilation configurations that achieve a balance between infection control, improvement of IAQ, and thermal comfort based on seasonal and operational variations.

By addressing these research questions and objectives, this study provides an integrated assessment of airborne virus transmission control, IAQ optimisation, and occupant comfort in office settings. The insights gained are intended to guide practical ventilation strategies in the design and operation of post-pandemic buildings.

## 1.8 Research Contribution

This research makes a significant contribution to the fields of indoor air quality (IAQ), thermal comfort (TC) and airborne virus mitigation by providing a comprehensive analysis of ventilation strategies in office environments. Using Computational Fluid Dynamics (CFD) simulations supported by real-world IAQ sensor data, the study offers quantitative insights into how air conditioning (AC), natural and mixed-mode ventilation systems affect airflow distribution, pollutant removal, and TC.

The key contributions of this research are directly aligned with the formulated research questions:

- (RQ1) A comparative evaluation of ventilation strategies: This study systematically examines how AC, natural, and mixed-mode ventilation influence viral particle dispersion, air velocity patterns, and comfort metrics, providing insight into airborne virus transmission dynamics in indoor environments.
- (RQ2) Analysis of CO2 as an IAQ indicator: By incorporating sensor-based CO2 measurements, the study evaluates CO2's role as a proxy for ventilation effectiveness and its potential to signal elevated airborne infection risk.
- (RQ3, RQ4) Identification of airflow inefficiencies: The research reveals dead zones in both AC and natural ventilation scenarios—areas where pollutants and viruses may accumulate—informing better design of air distribution systems.
- (RQ3, RQ4) Practical recommendations for improving ventilation systems: Based on simulation and experimental results, the research proposes evidence-based configurations and strategies to optimise IAQ, minimise virus transmission, and maintain acceptable TC in office settings.

## 1.9 Thesis Overview

This thesis is structured to systematically explore the interactions between IAQ, ventilation strategies, and TC, with a particular focus on reducing airborne virus transmission in office environments.

**Chapter 1:** Introduction: This chapter establishes the importance of IAQ, thermal comfort, and airborne virus transmission in indoor environments. It outlines the research motivation, objectives, and contributions while identifying key research gaps. The research questions that guide the study are presented, forming the foundation for the subsequent analysis.

Chapter 2: Literature Review: This chapter addresses Research Question 1, providing a comprehensive review of airborne virus transmission in indoor environments. This chapter examines key factors that influence airborne virus transmission, including airflow dynamics, humidity, and temperature. It also reviews existing ventilation standards and mitigation strategies, highlighting current limitations and knowledge gaps in the field.

**Chapter 3: Methodology**: This chapter details the research design, including CFD simulations, sensor data integration, and experimental setups used to assess ventilation performance, TC, and virus dispersion.

Chapter 4: CO2 as an Indicator of IAQ and the Risk of Airborne Infections: This chapter addresses Research Question 2, evaluating whether CO2 concentration levels serve as a reliable proxy for IAQ and infection risk. This chapter presents experimental data to assess the correlation between CO2 levels, ventilation effectiveness, and airborne virus transmission.

**Chapter 5:** AC System Analysis: This chapter addresses Research Question 3, investigating how fully air-conditioned (AC) spaces manage airflow, pollutant dispersion, and thermal comfort. It examines whether AC alone is effective in

mitigating airborne virus transmission and identifies potential limitations such as stagnant air zones.

**Chapter 6:** Natural Ventilation Analysis: This chapter addresses Research Question 3 and Research Question 4, assessing the effectiveness of natural ventilation configurations, particularly cross-ventilation, in maintaining IAQ and thermal comfort. Four NV scenarios are simulated under various environmental conditions, revealing the benefits and limitations of NV in different seasons. The chapter also evaluates virus dispersion, humidity control, and occupant comfort under varying window configurations and wind conditions.

**Chapter 7:** Mixed-Mode Ventilation Analysis : This chapter addresses Research Question 3 and Research Question 4, evaluating the effectiveness of combining AC and natural ventilation (mixed-mode ventilation) in improving IAQ and occupant comfort. Three Mixed-Mode scenarios are simulated under different environmental conditions, revealing the benefits and limitations of Mixed-Mode in different seasons. The chapter also evaluates virus dispersion, humidity control, and occupant comfort under varying Mixed-Mode configurations.

**Chapter 8: Conclusion**: This final chapter revisits the key findings in relation to each research question, summarising how the study contributes to the understanding of IAQ, ventilation strategies, and virus mitigation. This chapter provides recommendations for optimising ventilation in office environments and discusses potential avenues for future research.

## 1.10 Conclusion

This chapter has established the foundational context for this thesis, highlighting the critical importance of IAQ in office environments and its pivotal role in the airborne virus transmission. It has emphasised the growing challenges of balancing effective infection control with maintaining TC and optimising ventilation systems, particularly in light of recent global health crises, such as COVID-19.

This chapter has also identified significant research gaps to understand the complex interaction between airborne virus transmission dynamics, ventilation strategies, and the use of CO2 as a potential proxy of infection risk. Addressing these gaps requires a holistic approach that integrates advanced simulation tools, such as Ansys Fluent, and real-world environmental sensor data to investigate airflow patterns, TC, and viral dispersion under various ventilation scenarios.

To bridge these gaps, this thesis adopts a series of targeted research questions focused on airborne virus transmission, the reliability of CO2 as an IAQ indicator, and the development of engineering solutions to optimise ventilation. By systematically exploring these aspects, this research aims to deliver practical and innovative solutions to achieve healthier and more sustainable indoor environments.

Furthermore, the integration of IAQ improvements with occupant comfort and infection control highlights the multidisciplinary nature of this study, ensuring that its findings are not only theoretically robust but also applicable in real world settings. The contributions of this research are expected to inform the design of indoor spaces that prioritise occupant health, enhance TC, and minimise the risks associated with the transmission of airborne viruses.

This chapter sets the stage for the comprehensive exploration of the relevant literature, detailed methodologies, and critical findings in the following chapters. Through this structured approach, this thesis aims to provide actionable insights and engineering solutions that contribute to improved IAQ management and safer and more comfortable office environments.

# Chapter 2

## Literature Review

## 2.1 Introduction

This chapter addresses **Research Question 1 (RQ1)**: How are viral particles transmitted within indoor environments, and what factors influence their transmission dynamics? To explore this, the literature review examines transmission mechanisms of respiratory viruses, environmental and physical influences on dispersion, and strategies used to mitigate airborne transmission in indoor environments, especially offices. By synthesising current knowledge, this chapter lays the groundwork for identifying research gaps that motivate the computational and experimental investigations in later chapters.

Given that individuals spend approximately 90% of their time indoors, IAQ has emerged as a critical factor influencing not only health and comfort but also productivity within built environments[25]. This chapter addresses the multifaceted dimensions of IAQ, focusing on air pollutant concentrations and the increasing necessity for measures to prevent the spread of airborne viruses, such as SARS-CoV-2. Various precautions have been recommended, including reducing shared space capacities, increasing ventilation rates, employing natural ventilation, minimizing air recirculation, and avoiding direct exposure to airflow[26]. Furthermore, public health interventions such as lockdowns and the use of face masks have proven effective in curbing infection rates[27]. While these measures effectively control the spread of respiratory infections, additional environmental parameters such as temperature, relative humidity (RH), and evaporation—also play an important role in influencing virus transmission [28]. Experts recommend fundamental strategies to enhance IAQ and mitigate the risk of airborne infections, including adequate ventilation, air filtration, humidity regulation, and temperature control [29].

The role of ventilation in controlling airborne transmission has been consistently highlighted in the literature [30, 31]. Properly managing room air distribution patterns and airflow rates is essential to optimize indoor air circulation [32, 33]. During pandemics, international organizations such as American Society of Heating, Refrigerating and Air-Conditioning Engineers (ASHRAE) and Federation of European Heating, Ventilation and. Air Conditioning Associations (REHVA) have emphasized the importance of increasing outdoor air ventilation [34]. However, careful consideration must be given to prevent contamination risks and potential recirculation of infected air [35].

Effective IAQ management also involves filtering outdoor air using appropriately sized high-efficiency particulate air (HEPA) filters, which have proven effective in removing viral particles and pollutants that exacerbate COVID-19 transmission risks [36]. Additionally, monitoring CO2 levels in indoor environments as an indirect method to assess infection risk is essential because CO2 concentration can be a key indicator of ventilation adequacy in relation to human respiratory activityPang et al. [34].

Research on how SARS-CoV-2 spreads within indoor environments has become a focal point of IAQ studies, particularly in terms of airborne transmission pathways [37]. The virus can remain viable on surfaces for days and in aerosols for hours, necessitating a thorough understanding of transmission routes to implement effective prevention strategies [17].

The integration of digitalization technologies, such as Digital Twins and Inter-

net of Things (IoT) devices, offers new possibilities for managing IAQ. These technologies enable virtual planning, real-time monitoring, and predictive maintenance, contributing to improved IEQ management [38–41]. Moreover, the analysis of unstructured Big Data generated by IoT devices, when combined with artificial intelligence (AI) techniques such as machine learning (ML) algorithms, can provide valuable insights for optimizing IAQ [40, 41]

In addition to technological advancements, both experimental and CFD analyses have been employed to study aerosol behaviour and airborne transmission in indoor spaces[42, 43]. CFD, in particular, offers a robust tool for simulating realworld airflow patterns, while models such as the Wells–Riley equation have been instrumental in predicting airborne infection risks [44].

This chapter seeks to achieve three primary objectives: (a) understanding the dynamics of respiratory disease transmission in indoor environments, with a specific focus on COVID-19 transmission; (b) identifying effective methods for assessing IAQ; and (c) proposing strategies to enhance IAQ to reduce the spread of respiratory diseases, such as COVID-19. A key aim of this research is to ensure healthy indoor environments while balancing energy consumption considerations. To achieve these objectives, the following research questions are formulated:

#### State-of-the-Art Interventions

• What current state-of-the-art interventions are employed to mitigate the risk of indoor respiratory infections, with a specific focus on SARS-CoV-2, while concurrently improving IAQ?

#### Gaps and Limitations

• What are the existing gaps and limitations in current research endeavours aimed at designing measures to mitigate the risk of indoor respiratory infections?

#### **Future Research Directions**

• What potential future research directions should be pursued to sustain consistently safe indoor environments and proactively prevent the risk of respiratory viral infections?

By addressing these research questions, this study aims to offer valuable insights into the relationship between IAQ and respiratory disease transmission, with a particular emphasis on the challenges posed by COVID-19. The findings will contribute to a deeper understanding of current practices, limitations, and future research needs in the field.

The structure of this chapter mirrors the research questions. Following this introduction, Section 2 provides an overview of the methodology used in the literature review. Section 3 describes the transmission dynamics of respiratory viruses, particularly SARS-CoV-2, and evaluates existing interventions to mitigate the risk of infection in indoor spaces. Section 4 examines the gaps and limitations of current research, while Section 5 explores potential future research directions. Finally, Section 6 presents concluding remarks, summarizing the key findings and implications of the reviewed literature.

## 2.2 Literature Review Approach

This systematic review was conducted to identify and analyse contemporary research investigating the relationship between IAQ and the risk of respiratory infections in indoor environments. A comprehensive approach is required to fully understand how infections, particularly respiratory viruses, spread in enclosed spaces and how IAQ can be optimised. Key methods include epidemiological and microbiological studies, along with CFD simulations, to model airflow and particle behaviour. In addition, integrating digital technologies, such as Digital Twins and IoT devices, has proven essential for managing indoor environments. These technologies facilitate virtual planning, real-time monitoring, and predictive maintenance, enhancing IAQ management capabilities.

Furthermore, the analysis of unstructured Big Data generated by IoT devices, combined with AI techniques such as ML algorithms, provides valuable insights for informed decision-making regarding IAQ. To achieve a comprehensive understanding of infection transmission in indoor environments, it is also necessary to consider a combination of engineering controls, protective measures, and behavioural theories. These approaches offer a well-rounded perspective, allowing for the analysis of the physical, behavioural, and protective aspects of indoor infection risks.

Table 2.1 summarises the various proportions of studies dedicated to each approach. The current research focuses on the transmission of respiratory viruses in indoor environments, particularly emphasising the transmission of SARS-CoV-2 and its airborne surveillance, engineering and nonengineering control strategies, CFD modelling, and energy consumption related to IAQ.

While numerous methods exist to investigate infection transmission and IAQ, this research focuses on SARS-CoV-2 due to its global significance and the unique challenges it presents as a public health concern. SARS-CoV-2 has driven a heightened interest in how to maintain optimal IAQ while reducing infection risks in indoor settings, such as office environments.

Figure 2.1 outlines the process by which relevant documents were sourced from established databases, including IEEE, Scopus, Web of Science, and ScienceDirect using the following keywords: [Indoor OR confined OR enclosed] AND [air quality OR ventilation OR CO2] AND [SARS-CoV-2 OR COVID-19 OR SARS OR respiratory viruses]. Initially, these keywords led to the identification of 2722 documents, including journal articles, review papers, and conference papers.

Category	Percentage	Notes
Epidemiological Modes	29.78%	CFD simulations were the focus of 10.67% of the total reviewed publications.
Microbiological Methods	44.38%	ML techniques were utilised in 5.06% of the microbiological methods studies.
Engineering and Protective Theories	25.84%	Energy consumption considerations were integral to 19.57% of the engineering and protective measures studies.

Table 2.1:	The percentage	of publications	in each	category
------------	----------------	-----------------	---------	----------



Figure 2.1: PRISMA flow diagram for systematic review

The systematic review was conducted through three primary steps:

1. Identifying authoritative research papers through established academic search

engines.

- 2. Screening and retaining relevant publications that met the research criteria.
- 3. Extracting relevant studies focusing on the transmission of COVID-19 and other respiratory viruses in relation to IAQ.

Figure 2.2 illustrates the significant increase in research publications on IAQ following the onset of the COVID-19 pandemic, highlighting the growing recognition of its importance. Figure 2.3 further categorises the literature, showing that most publications fall within the fields of environmental sciences, engineering, and public health, with a particular emphasis on occupational health. To ensure the relevance of the research, studies from prior years were included in the review if they demonstrated clear applicability to COVID-19.

In the initial review process, 115 duplicate articles and 298 ineligible publications (identified by automation tools) were excluded. After scanning titles and reading abstracts, the document count was reduced from 2722 to 495 papers. Following a more in-depth screening, 202 review papers and 115 papers focusing on IAQ in transportation settings were excluded, leaving 178 papers for further analysis.

From the updated collection of articles, four distinct themes emerged:

- Pre-COVID-19 indoor virus transmission: Investigating the spread of viruses in indoor environments before the emergence of COVID-19.
- SARS-CoV-2 transmission in indoor environments: Analysing the unique challenges posed by COVID-19, particularly its airborne transmission.
- IAQ assessments: Evaluating IAQ in various settings and its correlation with infection risks.
- Control strategies: Exploring both engineering and nonengineering interventions designed to mitigate infection risks in indoor environments.

These themes provide a structured overview of the diverse research areas explored in the literature, offering valuable insights into the evolving body of knowledge on IAQ and respiratory infection risks.



Figure 2.2: Number of IAQ documents based on the year of publication



Figure 2.3: Publication categories

To systematically explore each theme, a framework was developed, guiding the analysis of studies based on the following key attributes:

- 1. Scale: Identifying the type of indoor environment analysed (e.g., office, residential, and healthcare settings).
- 2. Application field: Categorising studies by their specific research field (e.g., environmental science, engineering, and public health).
- 3. Scope: Summarising the overall objectives and research focus of each study.
- 4. The type of method utilised: This determines the method used to validate the study.
- 5. IAQ measurements: Assessing whether IAQ monitoring technologies, such as sensors, were employed to collect data.
- 6. Conclusions: Summarising the key findings and outcomes of each study.
- 7. Study limitations: Highlighting any constraints or limitations encountered in the studies, such as a limited focus on certain environmental factors or data reliability issues.
- 8. Future directions: Providing recommendations for future research, focusing on improving IAQ, and mitigating the transmission of airborne viruses in indoor environments.

This systematic approach to the literature review ensures a comprehensive understanding of current research trends and helps identify gaps and opportunities for future research aimed at improving IAQ and preventing the spread of respiratory viruses, particularly in the context of pandemics such as COVID-19.

# 2.3 Indoor Air Quality (IAQ) During and Before COVID-19

Respiratory viruses, including influenza, parainfluenza, RSV, severe acute respiratory syndrome coronavirus (SARS-CoV), rhinovirus, and adenovirus, are prominent human pathogens that primarily affect the upper respiratory tract. Since pandemics caused by respiratory viruses have the potential to cause widespread harm, public awareness and concern regarding their transmission has increased, particularly in the context of indoor environments[12]. These viruses typically spread through four primary transmission routes, as illustrated in Figure 2.4: direct contact (e.g., handshakes), indirect contact via fomites, respiratory droplets (particles >5  $\mu$ m), and aerosols (particles <5  $\mu$ m) [18–20]. Each of these routes presents unique challenges for controlling virus spread in enclosed spaces, emphasising the need for comprehensive IAQ management strategies.

In this section, as detailed in Table 3.1, we will review various studies investigating the transmission of respiratory viruses in indoor environments.



Figure 2.4: Major routes of respiratory viruses transmission [18]

#### 2.3.1 Severe Acute Respiratory Syndrome

SARS, a coronavirus that emerged in China in the early 2000s, triggered a global outbreak that highlighted the challenges of controlling respiratory virus transmission in indoor environments [45]. Several studies have examined the transmission dynamics of severe acute respiratory syndrome (SARS) in indoor settings. For instance, [46] investigated the movement of air between apartments in high-rise buildings, noting the difficulty of controlling air leakage due to pressure differences and building airtightness. Similarly, Niu and Tung [47] provided insights into vertical transmission in high-rise buildings, where exhaust air was shown to travel between floors through window openings.

Further research explored engineering controls and public health interventions. For example, [48] assessed the effectiveness of isolation measures and contact tracing in controlling the SARS epidemic in hospitals and schools. Jiang et al. [49] emphasised the importance of maintaining safe ventilation rates in hospital settings to prevent viral transmission, a finding corroborated by Yu et al. [50] in their CFD studies on virus dispersion in healthcare environments.

#### 2.3.2 Influenza

Influenza is another highly contagious respiratory virus that has historically affected both humans and animals. Influenza was declared a pandemic in 2009 by the World Health Organisation (WHO)[51] and has been the subject of numerous studies aimed at understanding its transmission dynamics and control measures. Chen et al. [48], Chen and Liao [52] explored the role of engineering controls, such as HVAC systems, in mitigating influenza transmission in schools and hospitals. Pantelic et al. [53] demonstrated the effectiveness of personalised ventilation (PV) in reducing infection risks in indoor environments.

Environmental controls have also been shown to impact influenza transmission.

Studies by Myatt et al. [54], Koep et al. [55], Noti et al. [56] reported that maintaining RH above 40% significantly reduces the survival of the influenza virus. Azimi and Stephens [57] highlighted the role of HVAC filtration systems in reducing influenza infection risk, noting that filtration was more cost-effective than increasing ventilation.

#### 2.3.3 Middle East Respiratory Syndrome

Middle East Respiratory Syndrome (MERS), a coronavirus, emerged in Saudi Arabia in 2012, and is similar to SARS-CoV [58]. Several studies have investigated MERS transmission in indoor environments, particularly in hospitals, where infection control is paramount. Yu et al. [50], Sung et al. [59], Jo et al. [60] conducted CFD and tracer gas experiments to study airflow patterns in hospital wards, emphasising the role of airflow in infection control. Their findings suggest that increasing air exchange rates and improving ventilation system design can mitigate the risk of MERS transmission. Moreover, the influence of outdoor airflow on indoor infection risks was examined, revealing how external wind patterns can affect the movement of airborne pathogens inside buildings [60].

#### 2.3.4 Other Respiratory Viruses

In addition to SARS, influenza, and MERS, other respiratory viruses, such as tuberculosis, measles, and adenovirus, have also been the subject of significant research. Chen et al. [48], Pantelic et al. [53], Beggs et al. [61], Wan et al. [62] high-lighted the importance of combining environmental controls with public health interventions to reduce the transmission of these pathogens. For example, reducing waiting times in healthcare settings and minimising the number of susceptible individuals in shared spaces can significantly decrease the risk of airborne infections [61].

### 2.3.5 SARS-CoV-2 (COVID-19)

The most recent and impactful respiratory virus, SARS-CoV-2, originated in Wuhan, China, in late 2019 and rapidly spread worldwide, leading the WHO to a declare global pandemic. SARS-CoV-2, which is responsible for COVID-19, has posed unique challenges for indoor environments due to its persistence in aerosols and on surfaces. Studies such as [11] emphasised the importance of improved ventilation and air filtration in reducing the transmission of SARS-CoV-2 in enclosed spaces.

In addition to airflow and ventilation improvements, the effectiveness of interventions such as mask-wearing and social distancing has been extensively studied. [63] demonstrated that these measures significantly reduce transmission risks, while [64] explored how environmental factors, such as temperature and humidity, influence viral spread.

Ref	Building type	Type of respiratory virus	Result
[46]	Flat	SARS 2003	It is difficult to completely prevent the movement of air between the apartments in a high-rise building using only natural ventilation.
[47]	Flat	SARS 2003	Windows in high-rise buildings can be a significant pathway for the vertical spread of pathogen-containing aerosols.
[54]	House	Influenza	Increasing the RH level decreases influenza survival.
[48]	Hospital, ele- mentary school	Influenza, Chickenpox, Measles, SARS 2003	Engineering control measures combined with public health interventions can effectively mitigate the spread of respiratory infections.
[49]	Hospital	SARS 2003	To minimise the risk of airborne viral infection, it is recommended to reduce air from a SARS patient by a factor of 10,000 by the introduction of clean air for safe ventilation.
[61]	Hospital	Influenza, Tuberculo- sis, Measles	To mitigate the infection risk spread in a waiting area, and before using expensive technological solutions, it is important to first minimise waiting times and the number of susceptible individuals present.
[62]	Hospital	Enterovirus, RSV, In- fluenza A virus, Aden- ovirus, M pneumoniae	All respiratory viruses except enterovirus were de- tected in the air and objects.
[50]	Hospital	MERS, SARS, and In- fluenza (H1N1)	The location of a virus-infected patient may affect the infection risk for others. In addition, increasing the air change rate can help reduce the risk of infection.
[59]	Hospital	MERS	Infectious aerosols may be spread indoors via airflow influenced by outdoor winds.
[60]	Hospital	MERS	Outdoor wind can spread infectious aerosols indoors through airflow.
[55]	School	Influenza	Increasing RH by up to 60% and increase absolute hu- midity (AH) may reduce influenza virus survival and transmission.
[52]	School	Measles, SARS 2003	The mathematical model used in this paper can offer an initiative applicable to a real elementary school to predict the optimal control measures and to protect susceptible students from infection by infected stu- dents.
[57]	Office	Influenza	Filtration reduces the infection risk of the influenza virus with lower costs than the option of increasing ventilation.
[56]	Chamber	Influenza	Keeping RH above 40% in indoor environments helps reduce the spread of the influenza virus.
[53]	Chamber	Influenza, Tuberculosis	The use of personal ventilation (PV) can reduce the infection risk of airborne transmissible disease.

 Table 2.2: Publications investigating IAQ before COVID-19

### 2.4 Transmission Pathways of Respiratory Viruses

#### 2.4.1 Transmission Vectors of SARS-CoV-2

SARS-CoV-2, the virus responsible for COVID-19, shares common transmission pathways with other respiratory viruses, such as influenza, SARS-CoV-1, and MERS-CoV. These transmission modes include direct contact, indirect contact via fomites, respiratory droplets, and aerosols. Among these, aerosol transmission has emerged as a significant vector, particularly in indoor environments characterised by poor ventilation and high occupant density. This mode of transmission has been extensively studied across various indoor settings, including educational institutions, healthcare facilities, offices, and public venues.

Aerosol transmission, whereby virus-laden particles smaller than 5 um remain suspended in the air for extended periods, is especially concerning in poorly ventilated indoor spaces. Numerous studies have demonstrated that SARS-CoV-2 can be transmitted via aerosols in confined environments, where the virus can persist in the air and be inhaled by occupants. Research on aerosol transmission in settings such as classrooms and educational buildings has revealed the potential for viral spread through an inadequate ventilation system [65-72]. Studies have also been conducted in healthcare facilities, where managing aerosol transmission is critical due to the high density of infected individuals [73–80]. Offices are another high-risk environment for aerosol transmission. Recent simulations and real-world measurements have indicated that inadequate air circulation and recirculation of air through HVAC systems can exacerbate the risk of airborne transmission [81–88]. Other settings such as nail salons [89], historic building [90], rooms [91–93], gyms [94], residential buildings [95], stores [96, 97], chambers [98], the Skagit Valley Chorale event [99], lift [100, 101], concert halls [102], restrooms [103], courtrooms [104], restaurants [105, 106], terraces [107], indoor arenas [108]and courtyards [109] have also been studied, highlighting the widespread relevance of aerosol transmission across different types of indoor environments.

In addition to aerosols, fomite transmission (i.e., the spread of infectious agents via contaminated surfaces) has been a subject of significant research in the context of COVID-19. Fomite transmission occurs when individuals touch surfaces contaminated with the virus and subsequently touch their face, allowing the virus to enter the respiratory system. Several studies have explored the potential for fomite transmission in hospitals, households, and public spaces, as outlined in Table 7.2. While some studies, such as Wang et al. [110], Viegas et al. [111], have reported negative results for SARS-CoV-2 on surfaces, likely due to routine cleaning and disinfection measures, other studies have detected the presence of viral particles in various indoor environments. For instance, [112–114] found SARS-CoV-2 RNA on hospital surfaces, with nine out of 37 samples testing positive. Similarly, [115] detected the virus in 25% of air and surface samples collected from hospitals. These findings highlight the potential for fomite transmission in high-risk environments, despite ongoing cleaning protocols.

In residential settings, [116] reported that 75% of surface samples collected from households tested positive for SARS-CoV-2, suggesting that fomite transmission remains a concern in environments where people share common surfaces and spaces. Similarly, [117] detected SARS-CoV-2 in both air and surface samples in residential buildings, further underscoring the importance of regular disinfection and airflow management in reducing transmission risks.

The next section will investigate the presence of SARS-CoV-2 in indoor environments by examining different surveillance techniques.

Ref	Building type	Sample size	Result reported in study
[110]	Hospital	45 samples	No SARS-Cov-2 RNA was detected among these samples.
[112]	Hospital	37 samples	Nine out of 37 samples tested positive for SARS-CoV-2.
[114]	Hospital	19 samples	SARS-CoV-2 was detected in the filters of the air fil- tration systems of COVID-19 wards.
[113]	Hospital	200 samples	50 samples out of 200 samples were positive for SARS-CoV-2.
[115]	Hospital	56 samples	25% of the samples tested positive for SARS-CoV- 2.
[116]	Households	13 samples	All air samples and 75% of the surface samples tested positive for SARS-CoV-2.
[111]	Educational building	106 samples	All the samples tested for SARS-CoV-2 were negative.
[117]	Residential building	7 samples	SARS-CoV-2 was detected in an air sample from vol- unteer A and in various air and surface samples from volunteer B.

**Table 2.3:** Information from publications on the SARS-COV-2 transmission through indoor surfaces in different indoor environments..

## 2.4.2 Airborne Surveillance of SARS-CoV-2

Detecting airborne viruses and evaluating IAQ require the application of diverse and precise techniques, such as air sampling and air monitoring. These methods are critical for understanding the presence and concentration of viral particles, particularly in environments where respiratory infections such as SARS-CoV-2 can spread via aerosols.

Air sampling, a widely used technique to identify airborne pollutants and pathogens

in indoor environments, has been employed in numerous studies to detect the presence of SARS-CoV-2. These investigations, as summarised in Table 2.4, focus on a variety of indoor settings, including hospitals, households, and public buildings. For example, studies by Masoumbeigi et al. [118], Faridi et al. [119], Vosoughi et al. [120] reported negative results for SARS-CoV-2 in air samples collected from hospitals, likely due to effective ventilation and disinfection protocols. In contrast, other studies confirmed the presence of the virus in the air, particularly in high-risk environments such as hospitals, where infected individuals were concentrated [121–127]. Other studies have also detected SARS-CoV-2 in household air samples [116], public building [128], and residential building [117], suggesting that airborne viral transmission is not confined to healthcare facilities but can occur in a wide range of indoor environments, especially those with limited ventilation.

Indoor air monitoring is another critical tool for assessing the broader context of IAQ. It involves measuring a range of environmental factors, such as PM2.5 levels, CO2 concentrations, humidity, and VOCs as illustrated in Figure 2.5, all of which are relevant to understanding the potential for viral transmission. While some studies focus on one or two key IAQ indicators, others take a more comprehensive approach, as shown in Table 2.5 and Table 2.6.

PM2.5, in particular, has been the focus of several investigations into IAQ in households [129], apartments [130–132], restaurants [133], classrooms [134], and houses [135]. These studies demonstrate that elevated PM2.5 levels correlate with poorer air quality and increased health risks, including those related to respiratory viruses. Other key IAQ parameters, such asCO2, total volatile organic compounds (TVOC), NO2, and O3 have also been widely studied in relation to both air quality and potential viral transmission [136–144].

Further studies incorporate a wider range of environmental factors, including temperature, humidity (RH), and air velocity, providing a more holistic view of IAQ [1, 142, 145–154]. Additionally, recent advancements in technology have enabled the use of IoT sensors and ML techniques to monitor and predict IAQ in real time. These technologies allow for more precise and continuous surveillance of IAQ, offering a proactive approach to managing indoor environments and ensuring healthy air quality [155–158]. The following sections investigate engineering and non-engineering control strategies.



Figure 2.5: IAQ measurements

Ref	Building type	Sample size	Result reported in study		
[118]	Hospital	31 samples	All the samples tested negative for SARS-Cov-2.		
[119]	Hospital	10 samples	All the samples tested negative for SARS-Cov-2.		
[120]	Hospital	33 samples	The results illustrated that air samples taken 2-5 metres away from the patient's beds were negative for SARS-COV-2.		
[121]	Hospital	14 samples	Two out of 14 air samples tested positive.		
[122]	Hospital	51 samples	Six of the 51 samples collected from the COVID-19 ward tested positive for SARS-CoV-2, four cases were in patient rooms, and two cases were in the hallway.		
[123]	Hospital	16 samples	All samples from the toilets and hallway were negative, but two samples collected from the intensive care unit (ICU) were positive.		
[124]	Hospital	Hospital 107 samples Six out of 107 air samples tested positive.			
[125]	Hospital 13 samples		Five of the 13 air samples collected from three major hospital in Kuwait tested positive for SARS-CoV-2.		
[126]	Hospital	52 samples	Five out of 52 air samples collected from hospitals in Brazil tested positive for SARS-CoV-2.		
[127]	Hospital	36 samples	Thirteen of the 36 cases tested positive for SARS-CoV-2.		
[116]	Households	16 samples	All air samples and 75% of the surface samples tested positive for SARS-CoV-2.		
[128]	public build- ings	12 samples	Six out of 12 aerosol samples were detected for SARS-CoV-2.		
[117]	Residential building	7 samples	SARS-CoV-2 was detected in an air sample from volunteer A and in various air and surface samples from volunteer B.		

**Table 2.4:** Information from publications on the SARS-COV-2 transmissionthrough indoor air in different indoor environments..

Ref	Building type	Mean T (°C)	Mean RH (%)	Mean CO2 (ppm)	Mean PM2.5 $(\mu g/m^3)$	$\begin{array}{c} \mathbf{Mean}  \mathbf{PM10} \\ (\mu \mathbf{g}/m^3) \end{array}$	Mean TVOC(ppb)
[137]	House	20 - 27	<50	1200-5000	12-25	20 - 50	-
[129]	House	12.4 - 17	-	-	62 - 142	-	-
[139]	House	24.6	52.8	-	5.6–12.2 in of- fices, 11.2–45.7 at homes	-	-
[136]	Apartment	Before lockdown 22.5-26, during lockdown 23.1-26	around 40	Before lock- down 731 - 2136.8, during lockdown 798.6 - 2395.7	before lock- down 10.63 - 16.07, during lockdown 7.19 - 16.94	-	before lock- down 272.524 - 5550.11, during lockdown 15.285 - 748.88 ppb
[138]	Apartment	20 - 30	45 - 50	470 - 2116	10 - 15	-	131 - 584
[131]	Apartment	NA	NA	396 for kitchen, 551 for bed- room, and 505 for hall	1465 - 247 in kitchen, 1151 - 160 in bedroom, 1565 - 166 in hall	94 in kitchen, in bedroom, 62 in hall	-
[134]	University	22.4-26	20.4-42.4	-	29 - 41	30 - 42	-
[154]	University	23.1	60	464.7	-	-	66.9
[1]	School	18 - 23.1	36.8 - 57.3	604 - 1079	-	-	-

**Table 2.5:** Mean values of IAQ measurements.

Ref	Building type	Mean T (°C)	Mean RH (%)	Mean CO2 (ppm)	Mean PM2.5 $(\mu \mathbf{g}/m^3)$	$\frac{\text{Mean}}{(\mu \mathbf{g}/m^3)}$	Mean TVOC(ppb)
[150]	School	-	-	720.7 - 1325	-	-	-
[153]	School	19 - 21	42 - 50	553 - 700 ppm	25 - 48	38 - 81	-
[142]	School	7.8 - 10.7	35.4 - 46	577-2232	4.8-15.3	5.8 - 17.4	287-485
[141]	Secondary school and university	18.2 - 19.3	51 - 57	97 - 220	-	-	-
[149]	Kindergarten	before COVID-19 is 22.0775, during COVID-19 is 21.955 oC	before COVID-19 is 33.32 , during COVID-19 is 31.125	before COVID- 19 is 1221.88 and during COVID-19 is 847.12	-	-	-
[144]	Daycare centre	23.6 for nursery Room, 22.3 for activity Room	32.3 for nursery Room, 34.6 activity Room	648.1 for nurs- ery Room, 608.3 for activ- ity Room	9.1 for nursery Room, 6.3 for activity Room	16.4 for nursery Room,10.8 for activity Room	158.9 for nursery Room,158.6 for activity Room
[140]	Hotel	-	-	452.21 - 459.92	4.09 - 9.33	-	108.4 - 162.41
[133]	Restaurant	19.4-23.8	45-54	-	113.1 for the entire week	is 548.1 for the entire week	-
[135]	Residential, Industrial buildings	-	-	-	pre-lockdown 24 - 32 , during lockdown the hourly indoor PM2.5 concen- trations 3 - 47	pre-lockdown 62 - 78, during lockdown the hourly indoor PM10 concen- trations 26 - 100	-

Table 2.6: Mean values of IAQ measurements.

## 2.5 Mitigation Strategies in Indoor Environments

Mitigating the airborne transmission of infectious pathogens, particularly in indoor environments, is a priority for public health. The key to effective mitigation lies in controlling the concentrations of respiratory aerosols, which can be achieved through both engineering and nonengineering strategies. Engineering strategies include manipulating indoor humidity and temperature, optimising ventilation systems, employing air filters, and utilising ultraviolet (UV) radiation for disinfection. Nonengineering strategies such as the use of face masks and maintaining physical distancing also play pivotal roles in minimising transmission risks. Figure 2.6 presents an overview of the academic literature concerning different control measures, indicating that ventilation is the most widely researched approach, with 21 studies emphasising its importance. Other strategies, including air cleaners, humidity control, UV radiation, face masks, and social distancing, have received comparatively less attention, with fewer than 10 publications focusing on each of these methods.



Figure 2.6: Distribution of research publications by IA control strategy

#### 2.5.1 Humidity and Temperature Control

Controlling indoor RH and temperature is a crucial factor in reducing viral viability and transmission in indoor settings. The maintenance of optimal conditions significantly influences the survival and transmission of airborne viruses. Studies have shown that maintaining temperatures between 20°C and 25°C and RH levels between 40% and 50% significantly reduces the stability of airborne viruses, including SARS-CoV-2 [64, 159]. Conversely, conditions outside this range, such as humidity levels below 40% or temperatures above 30°C, might effectively reduce COVID-19 transmission [160, 161]. For influenza viruses, studies reveal that at 5°C, transmission showed high efficiency, while it was blocked or less effective at 30°C. Additionally, dry conditions (20% and 35% RH) were observed to be more conducive to spread compared to conditions with intermediate (50% RH) or high humidity (80% RH) [162, 163]. In conclusion, previous research underscores the importance of maintaining a RH range of 40% to 60% for optimal human health in indoor environments [159]. Table 5.3 outlines the environmental conditions required for the stability and inactivation of various viruses.

Ref	Type of virus	Stable at RH (%)	Mitigated at RH (% )	Stable at T (°C)	Miti- gated at T (°C)	Description
[17]	SARS 2003	40	-	21-23	-	The decay rates of SARS-CoV-2 and SARS- CoV-1 in aerosols exhibited similarities.
[64]	SARS 2003	40-50	-	20-25	-	Maintaining temperatures between 20°C and 25°C and RH levels between 40% and 50% was found to have a protective impact on the viabil- ity of airborne SARS-CoV and MERS-CoV.
[164]	SARS 2003	40-50	>95	22-25	38	SARS-CoV has been found to remain viable for a duration of up to five days at temperatures ranging from 22°C to 25°C and RH levels of 40% to 50%. It was found that increased temperature and humidity led to a rapid decline in viability.
[165]	Influenza	-	90	-	20	Based on experimental and simulated findings, 90% RH together with 20°C is recommended in an indoor environment to improve the preven- tion of influenza transmission.
[162]	Influenza	20 - 30	80	5	30	It was found that lower relative humidities rang- ing from 20% to 35% were highly conducive to transmission, whereas transmission was entirely halted at a high RH of 80%. Additionally, when guinea pigs were housed at 5°C, transmission oc- curred more frequently compared to 20°C, while at 30°C no transmission was observed.
[163]	Influenza	20 - 30	50-80	5	30	At 5°C, transmission showed high efficiency, while it was blocked or less effective at 30°C. Additionally, dry conditions (20% and 35% RH) were observed to be more conducive to spread compared to conditions with intermediate (50% RH) or high humidity (80% RH).
[64]	MERS	40-50	-	20-25	-	Maintaining temperatures between 20°C and 25°C and RH levels between 40 and 50 was found to have a protective impact on the viability of airborne SARS-CoV and MERS-CoV.
[166]	MERS	40	-	20	-	At a temperature of 20°C and a RH of 40%, MERS-CoV was a very stable aerosol.
[167]	MERS	-	-	25	38	The efficiency of MERS inactivation was significantly higher at a temperature of $38^{\circ}$ C compared to $25^{\circ}$ C.
[17]	SARS- CoV-2	40	-	21-23		The decay rates of SARS-CoV-2 and SARS-CoV-1 in aerosols exhibited similarities.
[159]	SARS- CoV-2	<40	40-60	-	-	Previous research indicates that a RH range of 40–60% was found to be optimal for human health in indoor settings.
[160]	SARS- CoV-2	<40 and >80	<70	-	-	The transmission of the virus is not significantly affected by temperature. However, it is recom- mended that a safe humidity level be below 70% RH.
[161]	SARS- CoV-2	-	<78	-	30	Maintaining a mean RH below 78% and continu- ous daily temperatures above 30°C significantly reduce transmission.

 Table 2.7: Environmental conditions for the stability and inactivation of various viruses..

#### 2.5.2 Ventilation Strategies

Ventilation is one of the most critical engineering strategies for mitigating infection transmission in indoor spaces. It plays a central role in maintaining IAQ by diluting and removing airborne contaminants, including viruses. Ventilation can be provided through mechanical systems, natural forces, or a combination of both. Research on natural ventilation has explored its effectiveness in various settings[46, 47, 168–175]. Mechanical ventilation systems, on the other hand, have been studied in different spaces[63, 176–184]. Mixed ventilation systems, which combine mechanical and natural ventilation, have also demonstrated success in improving IAQ while minimising energy consumption[185–187].

## 2.5.3 Air Cleaners and High-efficiency Particulate Air Filters

Air cleaners, particularly those equipped with HEPA filters, are another effective strategy for reducing the concentration of airborne pathogens in enclosed spaces. Studies have shown that placing air purifiers near infected individuals significantly reduces viral aerosol concentration He et al. [188], Narayanan and Yang [189]. HEPA filters, in particular, have proven to be highly effective in poorly ventilated spaces. For instance, in dental treatment rooms, HEPA filters dramatically reduced aerosol accumulationRen et al. [190]. Other studies confirmed that while HEPA systems may increase noise levels, they provide substantial benefits by reducing aerosol transmission in environments with limited ventilation Bluyssen et al. [191]. Additionally, Lelieveld et al. [192] highlighted that the use of HEPA filters could potentially reduce individual infection risk by a factor of 5 to 10, affirming their critical role in controlling airborne viruses.

#### 2.5.4 Ultraviolet Radiation

UV radiation, particularly UV-C light, has been widely recognised for its ability to inactivate airborne pathogens, including SARS-CoV-2. Recent studies have demonstrated that integrating UV-C disinfection with ventilation systems can significantly reduce airborne virus concentrationsFeng et al. [193], Srivastava et al. [194]. Kahn and Mariita [195] highlighted the importance of balancing UV-C intensity with airflow to optimise infection control. UV-C radiation is especially effective when applied in high-risk areas such as hospitals and intensive care units (ICUs), where maintaining sterile air conditions is critical. In line with these results, de Souza et al. [196] successfully used UV germicidal irradiation in an ICU's HVAC system, pointing out how important it is for keeping the air clean, especially when the UV system is always on.

#### 2.5.5 Masks and Social Distancing

Face masks and social distancing are among the most effective nonengineering strategies for reducing infection risks in indoor environments [89]. Numerous studies have shown that face masks, particularly those with high filtration efficiency such as N95 masks, can significantly reduce the likelihood of airborne transmission Lelieveld et al. [192], Shen et al. [197]. Coyle et al. [63] found that masks when combined with other strategies, such as HEPA filtration and enhanced ventilation, offer comprehensive protection against viral transmission. Social distancing, particularly maintaining a distance of at least 1 meter, has been shown to reduce transmission risk in various indoor settings, including schools [142, 145], hospitals [146], elevators [100], a high-rise institutional building [181], and the Lawrence Berkeley National Laboratory [72].

In conclusion, effectively mitigating SARS-CoV-2 transmission in indoor environments necessitates implementing several key strategies. First, maintaining indoor temperatures between 20°C and 25°C and RH levels between 40% and 50% is essential for reducing the viability of the virus. Second, optimising ventilation systems is critical to balancing air quality and energy efficiency, emphasising increasing air exchange rates to dilute and remove airborne contaminants. Additionally, HEPA filters and UV-C radiation have proven to be highly effective for air purification, with UV-C being particularly beneficial when integrated into HVAC systems. Furthermore, the use of face masks, especially those with high filtration efficiencies, such as N95 masks, plays a vital role in personal protection, reducing transmission risks by up to 99%. Maintaining a physical distance of at least 1 meter in social settings also significantly decreases the risk of viral spread. When these strategies are combined, they form a comprehensive framework for minimising COVID-19 transmission in various indoor settings, providing actionable insights for enhancing future health and safety protocols.

The next section will explore transmission modelling, utilising CFD and other numerical modelling techniques to enhance our understanding of pathogen spread in enclosed spaces.

## 2.6 Modelling Transmission Using Computational Fluid Dynamics and Numerical Modelling

In addition to real-time air monitoring, computational models offer a powerful alternative for assessing the risk of infectious disease transmission. CFD models, which utilise numerical analysis to simulate fluid flow and the dispersion of airborne contaminants, are widely employed to understand how infectious agents travel within indoor environments. These models can simulate the behaviour of airborne particles under varying environmental conditions, allowing for the prediction of infection risks without the need for real-world case studies. Many of the studies reviewed in this systematic analysis have applied CFD models, ei-
ther independently or in conjunction with experimental data, to simulate airflow patterns and evaluate the movement of airborne particles in a range of indoor settings [49, 50, 60, 77, 79, 90, 97, 107, 182, 198, 199]..

For instance, in a study by Mirzaei et al. [200], an Eulerian CFD model was validated against experimental data and then combined with a Lagrangian CFD model to simulate the trajectory and evaporation of droplets of varying sizes. This hybrid approach allowed for a comprehensive analysis of how droplets behave in indoor air. Similarly, Barbosa and de Carvalho Lobo Brum [186] used multizone-CFD software developed by the National Institute of Standards and Technology to assess different ventilation modes, filter efficiencies, and outdoor air flow rates. This study highlighted the versatility of CFD in evaluating various IAQ solutions.

Another important contribution comes from [176], who modelled the role of HVAC systems in the diffusion of contagion in a hospital setting through CFD simulations of cough dispersion. Garbey et al. [73] applied CFD to test the components of a hybrid stochastic compartment model, simulating airborne particle transport in a surgical suite over a year. Their long-term analysis provided valuable insights into how airborne pathogens move in controlled environments.

Shang et al. [88] conducted transient CFD simulations to assess infection risks under varying conditions, such as calm and windy scenarios, demonstrating the importance of environmental factors in viral transmission. Moritz et al. [108] simulated aerosol dispersion from 4,000 virtual participants using CFD, which offered a large-scale understanding of how aerosols behave in crowded environments.

In Feng et al. [193], computational fluid-particle dynamics (CFPD) was employed to simulate the generation, transmission, and clearance of airborne droplets containing SARS-CoV-2 under various ventilation and UV air cleaning conditions. This model provided insights into the efficiency of UV-C air cleaners in reducing airborne viral loads in COVID-19 patient rooms. Moreover, Dong et al. [201] integrated CFD with the Wells–Riley model to calculate the infection rate distribution in an indoor space, combining fluid dynamics with probabilistic infection models to offer a more nuanced understanding of infection risk. Sarhan et al. [202] used a 3D CFD model to simulate human respiration activities, including breathing and speaking, within indoor spaces, providing detailed insights into the dynamics of respiratory droplet dispersion.

These CFD models have proven invaluable in understanding airflow patterns, droplet dispersion, and the effectiveness of ventilation, filtration, and air-cleaning strategies. By providing detailed visualisations and quantitative data, CFD allows researchers and engineers to design more effective interventions to reduce the transmission of airborne diseases, particularly in high-risk indoor environments, such as hospitals, offices, and schools.

It is essential, however, to balance IAQ improvements with energy consumption because enhanced ventilation and air purification systems can significantly impact building energy efficiency. The following section will explore the relationship between IAQ improvement measures and their associated energy implications.

## 2.7 Energy Consumption

Improving IAQ while managing energy consumption presents a significant challenge in creating sustainable and healthy indoor environments. Many strategies designed to mitigate infection risks, such as increased ventilation, advanced filtration systems, and air purifiers, can significantly impact energy use. Achieving a balance between optimising IAQ and minimising energy consumption is critical, particularly in the context of long-term sustainability and operational efficiency.

Numerous studies have explored this balance, assessing how IAQ interventions affect energy usage. Risbeck et al. [203] developed dynamic models to evaluate airborne transmission risk and energy consumption in HVAC systems, factoring in controller setpoints and weather forecasts. Their results demonstrated varying infection risks and identified energy-efficient disinfection strategies based on location-specific conditions.

In a similar vein, Schibuola and Tambani [204] investigated high-efficiency systems designed to contain COVID-19 through increased ventilation rates. Their study demonstrated that employing an autonomous high-efficiency air handling unit (HEAHU) reduced energy consumption by 60% to 72% while maintaining effective IAQ. In a follow-up study, the authors compared the performance of an exhaust air heat pump (EAHP) and a heat recuperator under various climatic conditions, emphasising that HEAHU provided energy savings between 31% and 46%, whereas EAHP savings ranged from 2.5% to 48%. In milder climates, EAHP slightly outperformed HEAHU [205].

Saikia et al. [206] designed a resource-efficient healthcare ward, finding that excessive cooling energy could increase energy costs while reducing productivity, highlighting the need for resource-conscious HVAC design. Similarly, Ascione et al. [207] assessed the energy impacts of higher mechanical ventilation rates and microclimatic control within buildings, concluding that while increased outdoor air led to higher energy demands, it also significantly improved IAQ.

Smart control systems are another approach to balancing IAQ and energy consumption. [208] proposed a smart ventilation system that adjusts ventilation rates based on real-time occupancy, resulting in energy savings of 11.7% and a 2% reduction in infection risk. Similarly, Aliero et al. [209] developed a smart sensing framework for indoor occupancy that has the potential to save up to 50% of energy by optimising HVAC operation based on real-time usage data.

Innovative solutions combining natural and mechanical ventilation have also been explored. Aviv et al. [210]) introduced a hybrid HVAC model that works alongside natural ventilation to reduce infection risk and energy consumption simultaneously. Similarly, Rey-Hernández et al. [185] examined hybrid ventilation systems with heat exchangers, finding that they met IAQ standards for 70% of operating time, while significantly reducing energy consumption and carbon emissions.

In high-rise buildings, Sha et al. [181] demonstrated that optimising mechanical ventilation systems could reduce energy consumption by up to 40%, while maintaining safe IAQ levels to minimise COVID-19 transmission risks.

Finally, de Frutos et al. [211] analysed the impact of quarantine on energy consumption and IAQ within 12 residences in Madrid, finding that household composition, habits, and daily activities significantly influenced both energy use and indoor environmental quality.

These studies underscore the importance of integrating energy-efficient solutions into IAQ management strategies. Whether through advanced HVAC systems, smart controls, or hybrid ventilation approaches, the goal is to maintain optimal IA while minimising energy costs.

The following section will outline gaps and limitations identified throughout the literature, providing a framework for future research in balancing IAQ and energy efficiency.

## 2.8 Thermal Comfort Standards

Establishing a well-balanced thermal environment is fundamental to enhancing occupant comfort, health, and productivity within buildings [212]. TC is broadly defined as the state of mind in which a person expresses satisfaction with the surrounding thermal conditions [213]. In practice, this means creating indoor environments where the majority of occupants feel neither excessively warm nor too cold. To achieve this consistently, comprehensive standards have been developed by international and national bodies, providing designers and engineers with objective criteria to evaluate and maintain IEQ. Although these standards aim to ensure safe, healthy, and productive spaces, they may sometimes fall short of addressing individual variations, which can lead to occupant dissatisfaction [214].

## 2.8.1 ASHRAE 55 Standard

The ASHRAE Standard 55 is one of the most widely adopted guidelines for TC. Originally published in 1966 and updated in subsequent years (1974, 1981, 1992, 2004, 2010, 2013, and most recently in 2017) [215], it specifies the combinations of environmental and personal factors (e.g., metabolic rate, clothing insulation, air temperature, radiant temperature, air velocity, and RH) that yield thermally acceptable conditions for most occupants [216]. The standard defines two acceptability levels (80% and 90%) that determine the operating temperature ranges. Although it is grounded in Fanger's predictive mean vote (PMV) model [23], ASHRAE 55 is designed to be adaptable, incorporating data from a wide range of climates, as demonstrated by its application in countries such as Chile [217] and Spain [218]. The model also sets specific prerequisites, such as the absence of active HVAC systems, metabolic rates between 1 and 1.3 met, and the possibility for the occupants to adjust their clothing insulation between 0.5 and 1 clo, and generally recommends maintaining indoor temperatures between 20°C and 26°C with RH between 30% and 60% [219, 220].

## 2.8.2 European Standards: EN 15251 and EN 16798-1

In Europe, the adaptive approach emerged through EN 15251:2007, which was grounded in the Smart Controls and Thermal Comfort (SCATs) project [221]. Unlike ASHRAE's two categories, EN 15251:2007 defines three categories that are tailored to occupant expectations and building types:

• **Category I:** Spaces designed for occupants with reduced adaptive capacity (e.g., the elderly or infirm).

- Category II: New buildings.
- Category III: Existing buildings.

EN 15251:2007 aligns with ASHRAE 55 in prescribing certain prerequisites, namely buildings without mechanical HVAC, occupant metabolic rates from 1.0-1.3 met, and flexible clothing insulation from 0.5-1.0 clo [222]. The standard further requires that mean daily outdoor temperature be 15°C to 30°C for lower-limit considerations and 10°C to 30°C for upper-limit determinations. More recently, EN 16798-1 reaffirms these adaptive concepts, specifying comfort ranges of 20°C to 24°C in winter and 23°C to 27°C in summer, with 30–70% RH recommended. EN 16798-1:2019 refines these recommendations further by establishing seasonal comfort ranges, such as operative temperatures of 20°C to 24°C in winter and 23°C to 27°C in summer, along with a RH range of 30% to 70% [222].

## 2.8.3 ISO 7730 Standard

First introduced in 1984 and later revised in 1994 and 2005 [223], ISO 7730 incorporates the Fanger comfort model into its framework by providing the equations needed to calculate both the PMV and the PPD occupants. This standard not only assesses overall thermal conditions but it also offers methodologies for addressing local discomfort issues, such as those caused by asymmetric radiation, draughts, and vertical temperature gradients. The latest version categorises comfort levels based on varying PPD thresholds and includes guidance on adjusting air speed to compensate for increases in operative temperature. However, it does not integrate the adaptive comfort approach. Typical recommendations under ISO 7730 include indoor temperature ranges of 18°C to 23°C during winter and 22°C to 27°C during summer [224].

## 2.8.4 Integrative Analysis and Limitations

Collectively, these standards aim to maintain a PMV value between -0.5 and 0.5 a range widely accepted as ensuring adequate TC for most occupants [216, 225]. Although they provide a robust framework for the design and evaluation of indoor environments, the prescriptive nature of these guidelines may not fully account for individual differences or dynamic environmental conditions. This discrepancy can sometimes lead to a gap between standardised comfort levels and the actual experiences reported by occupants. Recognising and addressing these limitations is crucial for advancing TC research and practice.

## 2.9 Research Gaps

This systematic review offers a comprehensive analysis of strategies aimed at mitigating infection risks and improving IAQ. Despite extensive research on COVID-19 transmission, IAQ, and energy consumption optimisation as discussed in sections 2.3 to 2.5 and 2.7, several gaps and limitations remain. These gaps hinder a comprehensive understanding of IAQ dynamics and present challenges to the effective implementation of interventions.

## 2.9.1 Lack of Air and Surface Samples

Air and surface samples play a critical role in assessing microbial contamination and monitoring IAQ. Despite several studies focusing on IAQ in environments such as hospitals, the majority rely on small sample sizes, which limits their generalisability [116, 117]. Additionally, there is a notable lack of research examining IAQ in diverse indoor settings, such as higher education buildings. These environments, which serve both as workplaces and learning spaces, require specialised study. Only one of the reviewed studies has investigated IAQ in higher education settings, which highlights the need for broader sampling and context-specific IAQ assessments[111].

# 2.9.2 Inadequate Consideration of Occupancy Patterns and Behaviour

The influence of occupants on IAQ is a crucial factor that is often overlooked in research. Indoor environments are dynamic, with occupancy levels, activities, and movement patterns significantly impacting air quality. Studies that neglect to account for these variables, such as experiments conducted in nonoccupied settings, risk yielding results that do not reflect real-world conditionsPark et al. [172]. Moreover, assumptions of uniform occupancy distribution fail to account for the uneven movement and congregation of individuals, which can affect pollutant dispersion and ventilation efficiencySinger et al. [72], Li and Tang [179].

# 2.9.3 Insufficient Monitoring of Indoor and Outdoor Environmental Parameters

Accurate IAQ assessments require key parameters such as temperature, RH, and CO2 levels, both indoors and outdoors to be monitored. However, many studies fall short by making assumptions that overlook the dynamic nature of these variables. For instance, Cammarata and Cammarata [83] assumed constant indoor temperature and RH, neglecting the natural fluctuations that occur in real-world settings. Similarly, important factors such as humidity variations and the effect of open doors on IAQ have often been ignored, leading to potential biases[97, 226].

Additionally, many studies fail to measure outdoor environmental factors such as temperature, RH, and wind speed, which limits a comprehensive understanding of how outdoor conditions affect IA. The omission of seasonal variations further reduces the depth of IAQ analyses. In studies that assess the impact of outdoor particles and occupant activities on IA, inconsistencies and missing data complicate the interpretation of results. Variations in outdoor particle concentrations and challenges in simultaneous data collection add further complexity[130].

## 2.9.4 Shortage of Evaluation and Simulation Studies

Although numerous studies have evaluated IAQ during the COVID-19 pandemic, several gaps remain in terms of evaluation and simulation methodologies. One key limitation is the focus on short work hours, which overlooks IAQ dynamics over longer periods, such as full workdays. This limits our understanding of air quality fluctuations throughout the day and hinders the development of comprehensive IAQ guidelinesMotamedi et al. [182].

Additionally, many studies fail to account for climate variations in IAQ assessments, despite the significant impact that different climatic conditions can have on air quality[145]. There is also a need for more in-depth research on infection transmission, particularly for viruses such as SARS-CoV-2, using tools such as CFD and airflow modelling[1].

Another overlooked factor is the influence of furniture on airflow patterns, which can significantly affect pollutant dispersion and ventilation effectiveness. Many studies do not consider this aspect, which limits the realism of their findings[182].

Finally, while IAQ guidelines exist in certain regions, there is a lack of regionspecific guidelines for areas such as Saudi Arabia. Developing localised IAQ standards is crucial for addressing the unique environmental and cultural factors that affect air quality [131].

# 2.9.5 Neglect of Critical Factors Influencing Ventilation System Performance

Effective ventilation, whether natural or mechanical, is essential for improving IA and reducing infection risks. However, many IAQ studies overlook critical conditions that significantly influence ventilation performance, such as aerosol circulation caused by infected individuals in enclosed environments[90]. Ignoring these factors can lead to incomplete insights into how ventilation systems manage airborne transmission, which limits the real-world applicability of IAQ guidelines.

Additionally, variations in simulation parameters, particularly boundary conditions, can greatly affect study outcomes. In particular, changes to HVAC systems, such as vent positions or airflow rates, can make findings less generalisable across different indoor environments[79]. The adaptability of IAQ guidelines to different HVAC configurations and setups is essential to ensure their relevance and effectiveness across diverse indoor spaces.

## 2.9.6 Energy Consumption and Cost

While extensive research has focused on infection transmission risks, there is a significant gap in the literature regarding the analysis of energy consumption, noise, initial investment, and overall cost in the context of IAQ improvements. This lack of comprehensive evaluation limits our understanding of the practicality and sustainability of various IAQ enhancement strategies. Most studies focus on mitigating infection risks but have often overlook the long-term viability and environmental impact of the proposed solutions [68].

Additionally, the absence of simulations involving heat recovery techniques, particularly in winter, is a key oversight. The effectiveness and cost-efficiency of IAQ strategies, especially in colder climates, can be significantly influenced by integrating heat recovery mechanisms [181]. Addressing these factors is essential for a holistic understanding of IAQ interventions and their energy implications.

## 2.9.7 Air Quality Based on Dynamic Data

Many previous studies on IAQ were conducted in small indoor environments with few or no occupants to avoid the complexities of data collection in larger spaces. While some research has focused on larger environments, these studies primarily relied on simulations without incorporating real-time aerosol particle data, accurate IAQ metrics, or building operational data. The absence of outdoor air quality measurements further limits the ability to accurately assess the contribution of external pollutants to IA, particularly in environments such as classrooms [153].

### 2.9.8 Digital Twin Model and Machine Learning Models

Digital Twins enable the creation of a dynamic, virtual replica of an indoor environment that synchronises with real-world conditions via sensor inputs. This capability is especially valuable in the context of IAQ and airborne infection control, as it allows for the simulation, prediction, and optimisation of ventilation performance, occupant exposure, and system responsiveness under various scenarios — something traditional static models cannot achieve.

There is a notable gap in the development of a unified model for different indoor environments that integrates DTs with machine learning (ML) to support nonexperts in decision-making. Furthermore, few studies have explored the predictive capabilities of ML to estimate thermal comfort or pollutant concentrations. ML has the potential to simulate complex multi-variable scenarios, offering real-time, low-cost insights with minimal environmental burden. Several studies have explored IAQ monitoring and forecasting using ML. For instance, Sharma et al. [156] proposed IndoAirSense, a low-cost IAQ estimation model using long short-term memory (LSTM-wf) in university classrooms. Mumtaz et al. [155] introduced an IoT-based platform for monitoring indoor contaminants using ML. Similarly, Marzouk and Atef [158] employed continuous IoT monitoring in educational buildings, using deep learning to analyse relationships among air quality variables. Taheri and Razban [157] developed a dynamic CO2 model to forecast concentrations over time using ML.

However, these studies share a common limitation: none of them incorporated a DT model to enable continuous synchronisation with the physical environment. The lack of a real-time, bidirectional feedback loop between data, prediction, and system control restricts their responsiveness. Moreover, the application of ML for infection risk prediction or outbreak forecasting remains largely unexplored.

#### 2.9.9 Infection Transmission in Indoor Environments

Future research must deepen the understanding of respiratory virus transmission dynamics, particularly in relation to the interactions between direct and indirect contact, respiratory droplets, and aerosols across various indoor settings[18–20]. Tailored infection control strategies should be developed for specific environments, such as schools, hospitals, offices, and gyms, each requiring unique ventilation and air quality management optimisation.

Research on high-rise buildings, for instance, must focus on air leakage, spatial positioning, and ventilation rates, drawing lessons from previous epidemics[46, 47]. The role of outdoor factors, such as wind influence on indoor air dispersion, should also be prioritised in designing adaptive infection control measures[60].

A holistic approach that addresses multiple respiratory viruses is needed to identify common principles for broad-spectrum control measures. Real-time monitoring systems are essential to maintaining continuous vigilance, particularly in high-touch environments. In addition, research should explore the psychological impacts of prolonged preventive measures and how behavioural interventions can be effectively implemented to promote compliance [113, 114, 116, 117].

Dynamic CFD models that incorporate human movement and real-world variables can significantly enhance the accuracy of airflow and infection risk simulations. Sustainability considerations must also be integrated into these strategies to minimise environmental impact without sacrificing effectiveness. Finally, fostering cross-disciplinary collaboration among epidemiologists, engineers, psychologists, and environmental scientists is crucial to the development of innovative, comprehensive solutions for managing indoor transmission risks.

## 2.9.10 Indoor Air Quality Factors

Future research on IAQ should embrace a multifactorial approach, expanding monitoring beyond individual parameters such as CO2, PM, RH, and temperature. A holistic assessment should consider the complex interactions between pollutants and environmental factors. Special emphasis should be placed on educational buildings, which require tailored IAQ strategies due to their high occupancy and unique characteristics. Integrating outdoor conditions into IAQ assessments is also crucial to understand the influence of external pollutants on IA.

Advances in IAQ monitoring technologies, such as the development of Digital Twins for real-time, continuous monitoring, can significantly enhance IAQ management. Additionally, applying ML algorithms to predict and mitigate infection risks presents a promising research direction, particularly for managing airborne diseases such as COVID-19. These multifaceted approaches aim to deepen our understanding of IAQ, especially in educational settings, and foster the development of advanced monitoring and predictive models for comprehensive IAQ management.

## 2.9.11 Nonengineering Control Strategies

While nonengineering interventions, such as face masks and social distancing, have proven to be effective in reducing infection risks, future research should focus on optimising these measures for enhanced safety and user comfort. Exploring alternative materials and designs for masks, especially in addressing health risks associated with prolonged use, could significantly improve the mask-wearing experience. Additionally, there is an urgent need to reduce the environmental impact of single-use masks by investigating sustainable, reusable alternatives[227].

Another area that merits further investigation is to refine the social distancing rules based on variables such as wind conditions and virus variants [88]. Future research should aim to develop more nuanced recommendations for appropriate distancing across various scenarios, balancing effectiveness and practicality. By addressing these aspects, nonengineering strategies can continue to evolve as vital components of infection prevention, contributing to more sustainable, comfortable, and adaptable solutions for public health.

## 2.9.12 Engineering Control Strategies

The use of engineering control strategies to improve IAQ has shown significant promise, but further research is needed to optimise these measures. For example, he use of HVAC systems during pandemics has been debated—some argue that HVAC systems can exacerbate the problem, while others recommend their use to dilute contaminants in indoor spaces [176]. This controversy highlights the need for a deeper analysis of HVAC design and operational methods, with a focus on identifying optimal configurations and usage strategies for pandemic settings. Air cleaners (or purifiers) are one of the most frequently proposed optimisation technologies for IAQ, which heavily rely on the type of filter medium used. Among the most effective filters are HEPA, ionisers, and ultraviolet germicidal irradiation (UVGI) systems. These can either be portable or integrated into a building's ventilation system. However, portable filters are limited to cleaning the air in their immediate vicinity. For instance,Ren et al. [190] found that portable air cleaners (PACs) equipped with HEPA filters can effectively reduce aerosol buildup in rooms with poor mechanical ventilation. However, Bluyssen et al. [191] pointed out that mobile HEPA systems can generate an unacceptable background noise level. Future research should focus on the practical implementation of these filters, examining potential drawbacks such as noise and exploring ways to improve their integration into indoor spaces.

Furthermore, UV-C technology, such as RM3 UV-C units, has emerged as an effective tool for reducing infection risks [194]. However, factors such as exposure time, maintenance, and service longevity can affect its efficiency. Therefore, future research should explore innovative UV-C applications, conduct real-world assessments, and develop sustainable IAQ solutions that integrate advanced filtration technologies with minimal drawbacks.

## 2.9.13 Computational Fluid Dynamics

CFD models have been a vital tool in many of the reviewed studies, offering detailed simulations of airflow patterns in indoor environments. For example, Razlan et al. [199] used a CFD model to investigate airflow and temperature dispersion, revealing important insights into the dynamics of air movement within enclosed spaces. Similarly, Barbosa and de Carvalho Lobo Brum [186] applied coupled multizone CFD software to evaluate the performance of different ventilation modes, filter efficiencies, and outdoor airflow rates. However, despite the sophistication of CFD models, many studies may have overlooked critical factors such as human movement, door openings, and real-time changes in occupancy that significantly influence airflow patterns and IAQ. To advance this issue, future research should focus on refining CFD models to incorporate these dynamic elements, ensuring a more accurate and holistic representation of IA.

By integrating these variables, future simulations will better reflect real-world conditions, thereby improving the design and implementation of effective ventilation strategies. Enhanced CFD models could ultimately lead to more efficient building designs that optimise IAQ while balancing energy consumption.

## 2.10 Conclusion

This systematic review has underscored the complexity and urgency of addressing respiratory infection transmission in indoor environments, particularly in light of the recent COVID-19 pandemic. In particular, this review has highlighted the critical need to investigate viral transmission dynamics in enclosed spaces and the role of IAQ in mitigating infection risks. As respiratory diseases continue to pose significant public health threats, understanding the factors that influence viral spread within indoor environments is essential to develop effective prevention strategies.

The research that was reviewed demonstrates that a multifaceted approach is necessary to tackle these challenges. A combination of epidemiological, microbiological, and computational methods, including CFD simulations, has been employed to better understand how infections spread in enclosed spaces. The increasing integration of digital technologies such as Digital Twins and IoT devices has further enhanced the ability to monitor, manage, and predict IAQ in realtime. These advances, combined with AI techniques such as ML, offer valuable

#### 2.10 Conclusion

insights for improving decision-making processes related to IA management and infection control.

From an extensive review of 2,722 articles, 178 were identified as key contributors to our understanding of SARS-CoV-2 transmission, IAQ assessment, and control strategies in pandemic contexts. This body of work has not only provided critical insights into the interplay between ventilation, filtration, and infection control strategies but has also revealed notable gaps in current research. Specifically, a limited number of studies have addressed the impact of energy consumption in the context of IAQ control, which emphasises the need for a more integrated approach that balances infection prevention with sustainability.

Further gaps include the lack of a comprehensive analysis of specific indoor environments, insufficient consideration of the indoor and outdoor parameters, and inadequate assessment of occupancy patterns in IAQ studies. These limitations highlight the need for future research to adopt more holistic methodologies, incorporating energy-efficient practices alongside IAQ improvements.

A key takeaway from this review is the importance of integrating digital twins and ML techniques in future IAQ research. These technologies have the potential to significantly enhance the management of IA by providing real-time data on transmission patterns and enabling more informed decision-making, while also considering energy consumption and sustainability.

In conclusion, future research should address these identified gaps by leveraging advanced technologies and adopting a more comprehensive approach to IAQ management. By doing so, we can develop more sustainable and resilient indoor environments that prioritise human health while minimising energy consumption, thereby effectively mitigating the ongoing and future risks posed by respiratory viruses.

# Chapter 3

# **Research Design and Methodology**

## 3.1 Introduction

This chapter outlines the research methodology that will be used to explore IAQ, TC, and virus dispersion in office environments. It provides a clear roadmap for this study, ensuring rigour, transparency, and replicability. This methodology integrates experimental data collection with advanced computational modelling to comprehensively analyse ventilation strategies and their effectiveness in improving IAQ, maintaining TC, and minimising airborne virus transmission risks.

This chapter specifically addresses the following research questions:

- Research Question 2: Can CO2 levels serve as reliable indicators to assess IAQ and signal the potential risk of airborne infections in office environments?
- Research Question 3: What ventilation strategies can be implemented to minimise the risk of airborne infections in office environments?
- Research Question 4: How can ventilation systems be configured to maintain good IAQ, ensure TC, and reduce virus dispersion in office environments?

Although RQ1 was primarily addressed through a detailed literature review in chapter 2, this chapter also supports RQ1 by outlining the simulation methods used to model virus transmission dynamics. The CFD simulations conducted using ANSYS Fluent replicate the dispersion behaviour of airborne virus particles under different ventilation scenarios, thereby providing quantitative insights into the airflow conditions and environmental factors that influence viral transmission indoors.

As illustrated in Figure 3.1, this research begins with building information modelling (BIM) to map office space (Figure 3.7) and identify sensor locations (Figures 3.13) in a teaching office at Cardiff University. Environmental sensors are deployed to measure key parameters, including CO2 concentrations, temperature, RH, and PM (PM2.5 and PM10), over a two-month period. These measurements are analysed to determine whether CO2 can serve as a reliable indicator of infection risk, addressing one of the key objectives of this study.

For the simulation phase, ANSYS Fluent is employed to model airflow patterns, TC, and virus dispersion under three ventilation scenarios: AC system, natural ventilation through opening windows, and mixed-mode ventilation, combining AC system and natural ventilation.

These simulations provide detailed insights into how airflow affects virus dispersion and TC in office space. By comparing the results across different ventilation strategies, this study identifies the most effective methods to maintain a healthy and comfortable indoor environment, while minimising the risks of infection.

The results of these analyses will be synthesised to provide practical recommendations for improving ventilation systems. The ultimate goal is to inform the design of healthier office environments that optimise IAQ, enhance TC, and reduce risks of virus transmission.



Figure 3.1: An overview of the research methodology

## 3.2 Demonstration Site and Building Information

The demonstration site for this research is a teaching office located within the Queen's Building at Cardiff University, UK. Located in the northeast of Cardiff city, as shown in Figure 3.2. The Queen's Building illustrated in Figure 3.3 serves as a prominent hub for academic and research activities. This multipurpose facility houses several academic departments, including the School of Engineering and the School of Physics, and supports a wide range of activities, such as lectures, research, and administrative functions. Its strategic location and diverse utilisation make it an ideal setting to investigate ventilation strategies and IAQ dynamics in office environments.

#### 3.2 Demonstration Site and Building Information



Figure 3.2: Location of the Queen's Building



Figure 3.3: Queen's Building

## 3.2.1 Teaching Office Information

The specific teaching office selected for this research is located on the first floor, within the south wing of the Queen's Building, as illustrated in Figures 3.4 and 3.5. Its urban location within Cardiff University ensures that external environmental factors, such as outdoor air pollution and seasonal weather variations, reflect typical conditions in urban office settings.

The teaching office occupies a total floor area of 270 square meters, with dimensions of 18 meters in width, 15 meters in length, and 4 meters in height, as shown in the 3D Ansys Design Model in Figures 3.6 and Figures 3.7. This space was designed to accommodate up to 13 occupants and it functions as a workspace for staff and a consultation area for students, as shown in Figures 3.8. Regular interaction between the occupants creates dynamic fluctuations in occupancy, providing a practical setting for studying IAQ, TC, and ventilation performance. The office is divided into three distinct zones to accommodate different activities:

- Zone 1: This is the largest zone, designed for nine occupants.
- Zone 2: This is a medium-sized zone, accommodating three occupants.
- Zone 3: This is a smaller zone, designated for one occupant.

The office is equipped with 18 south facing windows, as illustrated in Figure 3.9, which contribute to natural day lighting and play a significant role in ventilation dynamics, particularly in natural and mixed ventilation scenarios.

Access to the teaching office is provided through two doors, as illustrated in Figure 3.10:

- North-facing door: This door remains open during operational hours, allowing passive airflow. It opens into a hallway that connects to the forum, which is used by the students to study or eat.
- West-facing door: This door is typically closed and is only used by employees who enter or leave the office.

Ventilation is primarily managed through an AC system comprising six ceilingmounted fan coil units (FCUs), as shown in Figure 3.11. These FCUs play a critical role in regulating indoor temperature and airflow. However, they do not introduce fresh air from the outside, relying instead on recirculating existing indoor air. This limitation highlights the importance of exploring additional ventilation strategies, such as natural or mixed ventilation, to improve IAQ and reduce the risks of airborne infection. The teaching office was chosen as the demonstration site due to the following factors:

- Representative of a common workspace: The teaching office represents a typical workspace within an academic setting, making the findings of this study applicable to similar office environments in both academic and professional contexts.
- Mixed ventilation system: This site features a combination of natural ventilation (operable windows) and AC ventilation (FCUs), making it ideal choice to evaluate different ventilation strategies. This hybrid system enables comparisons to be made between scenarios to assess their impact on IAQ, TC, and virus dispersion.
- Variability in environmental conditions: Variability in occupancy levels and indoor environmental conditions, such as temperature, humidity, and CO2 concentrations, offers a dynamic setting for analysing how these factors influence IAQ and virus dispersion.
- Ease of access for data collection: The accessibility of the teaching office facilitates continuous and systematic data collection without logistical constraints. This ensures the acquisition of comprehensive and high-quality environmental data throughout the study period.
- Predictable working schedule: This office operates on consistent working hours (9 AM to 5 PM), allowing for structured monitoring of occupancy patterns and environmental conditions. These predictable patterns provide a controlled framework for analysing relationships between human activities and IAQ parameters.

This teaching office offers a unique opportunity to investigate the interplay between ventilation strategies, TC, and virus dispersion. The combination of natural and AC ventilation systems provides a robust platform to compare different approaches to optimise IAQ and minimising infection risk. Regular interaction between staff and students further enriches the study by introducing real-world occupancy fluctuations, which is closely aligned with this study's objective of identifying effective ventilation strategies for indoor environments.



Figure 3.4: Location of the teaching office



Figure 3.5: The teaching office S1.19



Figure 3.6: The teaching office S1.19 using Ansys



Figure 3.7: Teaching office area



Figure 3.8: Teaching office area: 18 south-facing windows



Figure 3.9: The 18 south-facing windows



Figure 3.10: The doors in the teaching office



Figure 3.11: Design of the AC system

## 3.3 Monitoring and Instrumentation

This section describes the IoT-based environmental monitoring system that will be used in this study to capture both indoor and outdoor environmental parameters. By integrating data from indoor sensors and an on-site weather station, the system provides a comprehensive understanding of the factors that influence IAQ, TC, and virus dispersion.

The monitoring system consists of three primary components:

- End Devices (Remote Units): These devices are equipped with integrated sensors to measure specific parameters. For indoor monitoring, sensors measure CO2 levels, temperature, humidity, and PM (PM2.5 and PM10). For outdoor monitoring, weather data, including temperature, humidity, and pressure, is collected via an on-site weather station.
- Gateway: The gateway receives data wirelessly from the end devices using the LoRaWAN protocol and transmits it to the server.
- Server/Database: The server processes and stores the collected data, enabling detailed analysis and interpretation.

Figure 3.12 illustrates the schematic diagram of the indoor and outdoor environment monitoring system.



Figure 3.12: Schematic diagram of the indoor environment monitoring system

The data collected by these sensors are transmitted to a nearby gateway, which in turn forwards the data to a network server. The server processes and stores these data for subsequent analysis. Given the focus on indoor conditions, the sensing devices used in this research are categorised mainly into indoor environment sensors. The role of each sensor is to accurately monitor variables such as temperature, humidity, CO2 levels, and PM concentrations.

A detailed description of each sensing device is provided, including a classification based on functionality. Table 3.1 offers a concise summary of the sensors, describing their category, model, and the associated parameters they measure. This systematic approach ensures a comprehensive understanding and analysis of indoor climate, facilitating targeted interventions to improve air quality and energy efficiency within the office environment.

Parameter(s)	Sensor	Measurement range	Accuracy
CO2	Sensirion (SCD41)	400 to 5000 ppm	$\pm$ 40 ppm
RH	Bosch (BME680)	0 to 100%r.H	$\pm$ 3% r.H
Temperature	Bosch (BME680)	-40 to 85 C	$\pm$ 0.5 C
Pressure	Bosch (BME680)	$300$ to $1100~\mathrm{hPa}$	$\pm$ 0.6 hPa
PM	Plantower (PMS5003)	0 to 500 $\mu { m g}/m^3$	$\pm$ 10 $\mu { m g}/m^3$
Weather data	Davis(Vantage Pro2)	Follow the link $^1$	

 Table 3.1: An overview of the technical specifications of the utilised sensing devices..

**Indoor environment sensor** In this investigation, a variety of sensors were used to measure various indoor parameters, such as temperature, humidity, CO2 levels and PM. These sensors were integrated into a single box, which was referred to as the 'end device', as illustrated in Figure 3.12. The descriptions of these sensors are as follows:

• The CO2 sensor (Sensirion SCD41): This sensor utilises nondispersive photoacoustic infrared technology to measure CO2 concentrations in the air. The principle behind this technology is based on CO2 molecules that absorb

<sup>&</sup>lt;sup>1</sup>https://www.davisinstruments.com/pages/vantage-pro2

infrared energy, leading to molecular vibrations within the measurement chamber. These vibrations generate acoustic waves, which a microphone then detects to determine the CO2 levels.

- The temperature sensor (Bosch BME680): The indoor temperature is measured on the basis of voltage variations in a silica diode. The fundamental operating principle of this sensor is that the voltage changes of the diode correspond to changes in the ambient air temperature.
- Humidity sensor (Bosch BME680): This sensor operates on the principle of capacitive sensing to measure the presence of water vapour in the air. The humidity level is measured by tracking changes in the electrical capacitance of a polymer-based capacitor, which varies with the amount of moisture present.
- Pressure sensor (Bosch BME680): This sensor gauges indoor atmospheric pressure by observing the deformation of an extremely sensitive thin membrane that responds to variations in atmospheric pressure.
- The PM sensor (Plantower PMS5003): This sensor measures the concentration of two types of pollutants, PM10 and PM2.5, in the air. It detects particles by projecting a light source, usually a laser beam, through an air sample. The light scattering caused by these particles is then analysed to determine the mass concentration of the PM in microgrammes per cubic metre (mug/m<sup>3</sup>).

Weather data (Davis Vantage Pro2)): An industrial grade weather station was put on the roof of the west building. The weather station uses a variety of sensors to collect real-time data on over 20 variables, such as temperature, humidity, pressure, precipitation, wind direction and speed, sun radiation, etc.

## 3.3.1 Indoor Environmental Monitoring

Primary data were collected using advanced remote sensing units strategically placed throughout the research site, as shown in Figure 3.13. These sensors continuously measured key parameters that are critical for evaluating IAQ, TC, and ventilation effectiveness. The monitored parameters included:

- Temperature: This is recorded to evaluate TC and to analyse the effect of temperature variations on airflow dynamics and occupant comfort.
- Humidity: This is monitored to determine indoor moisture levels, which affect both comfort and the potential for virus transmission.
- CO2 Levels: These are used as a proxy for air quality and ventilation performance, elevated levels indicate insufficient ventilation.
- PM: The concentrations of PM2.5 and PM10 were measured to assess the presence of airborne particles that could impact the health of the occupants.



Figure 3.13: Location of the sensors in the teaching office

## 3.3.2 Outdoor Environmental Monitoring

To provide a context for indoor measurements, outdoor environmental conditions were also monitored during the same two-month period. The outdoor parameters included:

- Outdoor temperature: This was captured to analyse how fluctuations in external temperature influence indoor thermal conditions and ventilation needs.
- Wind speed and direction: These were measured to assess their impact on natural ventilation and the potential infiltration of outdoor air into the indoor environment.

## 3.4 CFD Modelling and Simulation

This section presents the CFD methodology that will be used to simulate airflow patterns, TC, and dispersion of viral particles in an office setting teaching. This study employs ANSYS Fluent<sup>®</sup>, a widely used CFD solver, to provide detailed information on fluid flow behaviour, ventilation effectiveness, temperature and humidity gradients, and occupant-generated aerosols. Understanding how airborne pathogens circulate under different ventilation strategies is essential to mitigate transmission risks, while ensuring a comfortable and healthy indoor environment.

ANSYS Fluent was selected for this study due to its advanced turbulence modelling capabilities, robust multiphysics integration, and well-validated accuracy in IAQ simulations. Compared to other CFD programmes, ANSYS Fluent offers superior flexibility in handling complex boundary conditions, particularly when modelling ventilation-driven contaminant transport, thermal effects, and multiphase flows. It has been shown to provide slightly better performance in certain aspects than STAR-CCM+, as demonstrated in a study by [228], particularly in terms of numerical stability and solver efficiency for indoor airflow simulations.

Furthermore, comparative research has shown that ANSYS Fluent delivers more consistent solution accuracy at lower computational costs, while offering more versatile visualisation capabilities than SolidWorks or SimScale [229]. The software's user-defined functions (UDFs) further enhance its capabilities, allowing for custom modelling of airborne virus transmission risks and pollutant transport dynamics. These features make ANSYS Fluent a highly reliable tool for evaluating IAQ, infection risk, and TC performance in different ventilation strategies.

Using ANSYS Fluent's advanced computational capabilities, this study ensures a comprehensive assessment of ventilation performance in an occupied office environment, which will contribute to the development of optimised IAQ control measures.

## 3.4.1 CFD Objectives

The CFD-based investigation will evaluate three ventilation strategies (i.e., AC system, natural, and mixed-mode ventilation) in a teaching office at Cardiff University. Each strategy is analysed for its ability to reduce infection risks, improve IAQ, and maintain the TC of the occupants. The specific aims are as follows:

- Identify strengths and limitations of the AC system, natural, and mixedmode ventilation systems to control airborne infection risks and maintain occupant comfort.
- Investigate how different ventilation scenarios affect airflow distribution, droplet trajectories, and thermal conditions.
- Recommend design optimisations and operational guidelines to improve IAQ, TC, and occupant safety.

## 3.4.2 CFD Methodology

#### 3.4.3 Materials and Methods

IAQ is determined by a combination of factors, including airflow rate, temperature, humidity, and CO2 concentration. These parameters significantly influence the health, comfort, and safety of the occupants in enclosed spaces, particularly in office environments where inadequate ventilation can increase the risk airborne virus transmission [230]. The COVID-19 pandemic brought global attention to the consequences of poor ventilation, with documented cases of virus spread among office workers, which highlighted the critical role of effective air circulation [231].

To address the multifaceted challenges associated with IAQ and airborne infection risks, in this study a numerical approach is adopted that will use CFD. CFD offers the ability to simulate airflow patterns, TC, and virus dispersion in varying ventilation strategies, providing a detailed understanding of how these factors interact in an office setting. The simulations, which will be performed using Ansys R18.1, focus on evaluating key performance indicators such as airflow distribution, TC, and the trajectory of aerosolised particles.

#### 3.4.3.1 Key Steps in the CFD Workflow

CFD simulations in Ansys Fluent follow a structured workflow consisting of three main stages: preprocessing, solver configuration, and postprocessing. Figure 3.14 visually outlines these steps, ensuring a systematic approach to accurately modelling airflow dynamics, TC, and pollutant dispersion in indoor environments. The following key steps are taken:

• Geometry modelling and mesh creation: Developed in Ansys DesignModeler and Ansys Meshing, followed by a grid independence study to ensure accuracy.

- Definition of boundary conditions, material properties, and solver parameters.
- UDF compilation: Incorporating custom functions for PMV/PPD calculations.
- Solver settings and run cases: Selection of turbulence model
- Generating contour plots for velocity, temperature, RH, TC (PMV and PPD), and particle dispersion.

## 3.4.4 Model Development

#### 3.4.4.1 Assumptions of this Research

The development of the computational model for this study is based on a set of well-defined hypotheses and assumptions to simplify the analysis and align the simulations with the specific operational conditions of the teaching office. These assumptions ensure a focused investigation of airflow patterns, TC, and virus dispersion in various ventilation scenarios.

- Four different locations for infected individuals: This study hypothesises that the dispersion of airborne pathogens varies based on the spatial position of an infected individual. Therefore, four distinct locations within the office have been strategically selected to simulate potential infection scenarios. These positions represent typical occupant activities and ensure that airflow interactions are comprehensively analysed.
- Natural ventilation scenarios assume 30% of open windows: For natural ventilation cases, some windows are assumed to be open, providing maximum



Figure 3.14: Steps of Ansys

potential for outdoor air exchange. Scenarios involving partially closed or fully closed windows are not simulated, simplifying the assessment of natural ventilation's effectiveness.

- Steady-state indoor conditions: The model assumes steady-state conditions in AC system and is transient with particles.
- Uniform indoor temperature distribution: The temperature is assumed to be evenly distributed throughout the office space.
- Uniform occupant activity levels: All occupants are presumed to participate
in similar activities, contributing equivalent amounts of CO2 and aerosols to the indoor environment. This standardisation facilitates a clearer correlation between occupancy and IAQ metrics.

- Room air is perfectly mixed in mixing ventilation scenarios: For mixing ventilation scenarios, it is assumed that the air within the room is uniformly mixed, ensuring consistent concentrations of contaminants throughout the space.
- Exclusion of other indoor pollutants: This study focusses exclusively on temperature and RH. The analysis does not consider other pollutants, such as CO2, PM (PM2.5 and PM10), virus-laden aerosols, VOCs, or O3.
- Fixed occupancy patterns: This study assumes that the occupancy levels remain constant during working hours. Dynamic changes, such as the presence or absence of occupants in the space, are not accounted for in the model.

#### 3.4.4.2 Governing Equation

In ANSYS Fluent, the foundational principles governing fluid motion are based on conservation of momentum, mass, and energy. These principles are mathematically represented by the Navier—Stokes equations (momentum conservation), the continuity equation (mass conservation), and the energy equation (energy conservation). These equations collectively form the core framework for simulating airflow, temperature distribution, and aerosol dispersion in indoor environments. The equations shown below are integral to CFD simulations, allowing accurate modelling of complex fluid behaviours under varying conditions [232]. Momentum Conservation The conservation of momentum governs the transport of momentum in the fluid and is represented by the Navier–Stokes equations:

$$\frac{\partial(\rho u)}{\partial t} + \nabla \cdot (\rho u \vec{V}) = -\frac{\partial p}{\partial x} + \frac{\partial \tau_{xx}}{\partial x} + \frac{\partial \tau_{xy}}{\partial y} + \frac{\partial \tau_{xz}}{\partial z} + \rho f_x \tag{3.1}$$

$$,\frac{\partial(\rho v)}{\partial t} + \nabla \cdot (\rho v \vec{V}) = -\frac{\partial p}{\partial y} + \frac{\partial \tau_{xy}}{\partial x} + \frac{\partial \tau_{yy}}{\partial y} + \frac{\partial \tau_{yz}}{\partial z} + \rho f_y$$
(3.2)

$$,\frac{\partial(\rho w)}{\partial t} + \nabla \cdot (\rho w \vec{V}) = -\frac{\partial p}{\partial z} + \frac{\partial \tau_{xz}}{\partial x} + \frac{\partial \tau_{yz}}{\partial y} + \frac{\partial \tau_{zz}}{\partial z} + \rho f_z,$$
(3.3)

where p is the pressure of the fluid and  $\rho$  is the density of the fluid (mass per unit volume). Meanwhile, u,v, w represent the velocity components in the x, y, and z directions, respectively.  $\tau_{ij}$  are components of the stress tensor, representing viscous forces (shear stresses and normal stresses), and  $\tau_{xx}$ ,  $\tau_{xy}$  and  $\tau_{xz}$  are stresses in the x direction due to velocity changes in the x, y, and z directions.  $\rho f_x$ ,  $\rho f_y$ , and  $\rho f_z$  represent body forces (e.g., gravity or other external forces) acting in the directions x, y, and z, respectively.

Mass conservation The continuity equation ensures the conservation of mass within the system:

$$\frac{\partial \rho}{\partial t} + \nabla \cdot \left( \vec{V} \right) = 0 \tag{3.4}$$

**Energy conservation** The conservation of energy governs the transport of thermal energy within the fluid:

$$\frac{\partial}{\partial t}(\rho E) + \nabla \cdot (\vec{v}(\rho E + p)) = \nabla \cdot \left(\sum_{j} h_{j} \vec{J}_{j}\right) + S_{h}, \qquad (3.5)$$

where E is the total energy per unit mass of the fluid, typically the sum of internal energy (from molecular motion), kinetic energy, and potential energy.

These governing equations, in combination with boundary conditions and turbulence models, enable the accurate simulation of airflow dynamics and the dispersion of aerosols within the indoor teaching office.

### 3.4.5 Boundary Condition

The present study uses a detailed CFD simulation framework to evaluate ventilation strategies in a teaching office environment. This section outlines the general boundary conditions and modelling assumptions that will be applied to all simulation cases, while specific ventilation configurations are described in subsequent chapters. The simulation domain and boundary settings have been selected to ensure robust and physically representative predictions of air quality, TC, and infection risk.

Air will be treated as an incompressible fluid, a valid assumption given the lowspeed airflow conditions that are typically encountered in indoor environments [233]. Turbulence will be modelled using the k- $\epsilon$  RNG model, which was selected for its improved precision in predicting recirculating flows and low Reynolds number effects compared to the standard k- $\epsilon$  model [234].

The transit-state airflow field and turbulence parameters were first solved using the COUPLED algorithm for pressure-velocity coupling. Second-order upwind schemes were applied to all governing equations to improve accuracy and minimise numerical diffusion.

All solid boundaries, including walls, floors, and occupant surfaces, will be treated as nonslip walls, ensuring the correct resolution of boundary layer effects and airflow near surfaces. The walls will be modelled as adiabatic, following standard assumptions in indoor airflow simulations, where heat transfer through the walls is considered secondary to convective effects [235].

To account for internal heat sources, the ceiling will be assigned a constant heat flux of 40 W /  $m^2$ , representing the gain in lighting heat, which typically accounts for 25% of total internal heat generation in office environments [233]. Furthermore, occupant heat generation will be modelled as a fixed internal heat source of 120 W per person, based on metabolic rates of 1.5 Met and clothing insulation

levels of 0.5 clo (summer) and 1 clo (winter) [219]. These assumptions ensure a realistic representation of heat loads during work hours.

To model airflow exiting the domain, a pressure outlet boundary condition will be applied to all exhaust regions, with the outlet pressure set to 0 Pa (relative to atmospheric pressure). This condition allows the solver to determine pressuredriven airflow distributions based on AC and natural ventilation influences.

To verify the suitability of this boundary condition, simulated pressure distributions and airflow rates are compared with experimental and numerical studies of similar indoor ventilation environments [236].

A DPM will be used to investigate the dispersion of respiratory aerosols, tracking individual particle trajectories in a transient manner. The simulation framework follows a Transient DPM tracking: Once a fully developed airflow field has been obtained, particle dispersion is simulated over a 400-second period.

Respiratory aerosol particles are injected into the domain with a size distribution ranging from  $7 \times 10^{-6}$  m to  $1 \times 10^{-4}$  m, with a mean diameter of  $7.44 \times 10^{-5}$  m, consistent with experimental data on human emissions. The following respiratory events are simulated on the basis of established respiratory airflow models [237, 238]:

- Coughing: The particles are expelled at 11.8 m/s for 1.3 to 2.3 s.
- Sneezing: The particles are expelled at 70 m/s for 0.1 to 1.1 s.

The expelled aerosols are assigned a temperature of 35°C to approximate the properties of the exhaled air. The DPM is implemented in a one-way coupling mode, assuming a dilute particle concentration that does not influence the airflow field, which is a widely accepted approach for IAQ simulations [239].

The CFD simulations are performed using a segregated solver, with convergence criteria based on residual reduction and stabilisation of key flow variables. The following convergence thresholds are applied:

- Momentum, energy, and turbulence: Convergence is defined as residuals that drop below  $10^{-5}$  for the momentum and turbulence equations.
- Continuity equation: A stricter criterion of 10<sup>-6</sup> is enforced to ensure mass conservation.

To ensure numerical accuracy, a grid independence study will be conducted, refining the mesh resolution until further refinement produces negligible changes in airflow parameters. Similarly, a time-step sensitivity study confirms that the transient DPM tracking accurately captured particle dispersion without introducing numerical artefacts.

### 3.4.6 Solver Setting

This study employs a hybrid Eulerian-Lagrangian approach, where the continuous air phase is solved using Eulerian methods, while the movement of virus-laden aerosols is modelled using the Lagrangian DPM. This approach enables accurate simulation of airborne particle trajectories, dispersion patterns, and potential inhalation risks under varying airflow conditions.

The simulation process consists of two major computational stages. First, a steady-state flow solution is established by solving the Reynolds-averaged Navier–Stokes (RANS) equations using the renormalisation group (RNG) k- $\epsilon$  turbulence model. This turbulence model was selected because of its ability to accurately resolve indoor airflow characteristics and ventilation-induced turbulence. The solution is deemed convergent when the residuals for the continuity, momentum, turbulence, and energy equations drop below  $10^{-5}$ , typically requiring around 2000 iterations. After the steady-state airflow field is fully developed, the second stage of the simulation, transitory particle tracking, is initiated. In this

phase, the unsteady DPM is employed to track the motion of virus-laden aerosols expelled during coughing events. Particle injections are introduced after steadystate airflow is achieved to ensure a stable and reliable analysis of airborne particle dispersion and transport.

To differentiate the computational time scales, two key concepts are introduced: flow time, which refers to the steady-state distribution of velocity, temperature, and airflow; and particle time, which represents the transient dispersion of aerosols over time. This distinction ensures that airflow dynamics and aerosol transport are analysed separately, preventing numerical instability and ensuring accurate predictions of airborne virus transmission patterns.

The governing equations for mass, momentum, and energy conservation are employed to describe airflow behaviour, expressed as:

$$\nabla \cdot (\rho \mathbf{u}\phi) = \nabla \cdot (\Gamma_{\phi} \nabla \phi) + S\phi, \qquad (3.6)$$

where  $\phi$  represents the solved variables, such as velocity and temperature,  $\rho$  is the density of the air, **u** is the velocity vector,  $\Gamma_{\phi}$  represents the diffusion coefficient, and  $S\phi$  accounts for source terms such as heat generation. For aerosols laden with viruses, the DPM in the Lagrangian framework is applied, where Newton's second law governs particle motion [240]:

$$\frac{d\mathbf{u}_p}{dt} = F_D(\mathbf{u} - \mathbf{u}_p) + \frac{\mathbf{g}(\rho_p - \rho)}{\rho_p} + \mathbf{F}_x, \qquad (3.7)$$

where  $F_D(\mathbf{u} - \mathbf{u}_p)$  represents the drag force per unit of particle mass, g accounts for gravitational settlement, and  $\mathbf{F}_x$  includes additional forces such as Brownian motion. The drag coefficient,  $F_D$  is determined by:

$$F_D = \frac{18\mu}{\rho_p d_p^2} \frac{C_D Re}{24},$$
(3.8)

where  $C_D$  is the drag coefficient and Re represents the relative Reynolds number, defined as:

$$Re = \frac{\rho d_p |\mathbf{u}_p - \mathbf{u}|}{\mu}.$$
(3.9)

This formulation enables precise predictions of particle motion and dispersion in different ventilation configurations.

To account for turbulence effects on particle dispersion, the stochastic tracking model (random-walk method) is employed. This model captures instantaneous velocity fluctuations, resulting in a realistic representation of virus-laden aerosol movement. The fluctuating velocity component is defined as follows:

$$\mathbf{u} = \overline{\mathbf{u}} + \mathbf{u}',\tag{3.10}$$

where  $\overline{\mathbf{u}}$  represents the mean velocity and  $\mathbf{u}'$  denotes the fluctuating turbulent component. The prediction of particle dispersion makes use of the concept of the integral time scale T, which describes the time spent in turbulent motion along the particle path ds:

$$T = \int_0^\infty \frac{u'_j(t)u'_j(t+s)}{u_j^2} ds.$$
 (3.11)

The integral time is proportional to the particle dispersion rate because larger values indicate more turbulent motion in the flow. It can be shown that the diffusivity of the particles is given by  $u'_i u'_j T$ .

For small "tracer" particles that move with the fluid (zero drift velocity), the integral time becomes the fluid Lagrangian integral time  $T_L$ . This time scale can be approximated as follows:

$$T_L = C_L \frac{k}{\epsilon},\tag{3.12}$$

where  $C_L$  is to be determined because it is not well known. By matching the diffusivity of tracer particles,  $u'_i u'_j T_L$ , to the scalar diffusion rate predicted by the turbulence model,  $\nu_t/\sigma$ , one can obtain the following.

$$T_L \approx 0.15 \frac{k}{\epsilon} \tag{3.13}$$

for the k-  $\epsilon$  model and its variants, and

$$T_L \approx 0.30 \frac{k}{\epsilon} \tag{3.14}$$

In this model, expelled respiratory droplets are treated as discrete particles, while the surrounding airflow field is modelled as a continuous phase. This hybrid Eulerian-Lagrangian approach enables accurate tracking of aerosol trajectories, dispersion patterns, and deposition behaviour within the indoor environment.

To properly simulate aerosol dispersion, the particle injection process is carefully defined within the DPM framework. The expelled particles exhibit a range of diameters, rather than a uniform size, requiring the use of the Rosin–Rammler logarithmic distribution method to capture the realistic variation in droplet sizes. This approach ensures that the simulation reflects the actual size distribution of virus-laden aerosols, which influence suspension time and transport distance.

The hybrid initialisation approach is adopted because it provides an optimised starting point for the distributions of temperature, turbulence, species fraction, and volume fraction. This technique uses domain-averaged values and interpolation methods to initialise the simulation efficiently, which reduces computational time while ensuring convergence. Upon completion of the simulation, postprocessing techniques are employed to analyse virus particle tracking based on the output residence time. Animations and visualisations of aerosol dispersion are generated that illustrate the release, movement, and disappearance of virus-laden particles over time. The results reveal different dispersion characteristics for coughing and sneezing events. Specifically, aerosol transport after a coughing event occurs over a period of 1.3 to 2.3 seconds, while sneezing disperses particles within 0.1 to 1.1 seconds.

### 3.4.7 Thermal Comfort Simulation

This study employs CFD to evaluate TC in a teaching office environment, integrating PMV and PPD indices. These indices, defined by ASHRAE Standard 55 (2010) and ISO 7730 (2005), quantify thermal sensation and occupant dissatisfaction levels, respectively. By incorporating PMV and PPD into CFD simulations using ANSYS Fluent<sup>®</sup>, this study provides a detailed spatial analysis of thermal conditions in AC, natural, and mixed ventilation scenarios.

The TC simulation requires both environmental and physiological parameters. The environmental parameters incorporated into the CFD model include air temperature, mean radiant temperature, air velocity, and RH. These variables were extracted from steady-state CFD simulations, ensuring an accurate representation of indoor thermal conditions. The physiological parameters were established assuming a metabolic rate of 1.5 was achieved, corresponding to light office work, and 1.5 Met and clothing insulation levels of 0.5 clo (summer) and 1 clo (winter) [219]. These values ensure a realistic assessment of the thermal perception of occupants within the office environment.

The CFD model was created using ANSYS  $\text{DesignModeler}(\mathbb{R})$ , which incorporates a realistic office layout with furniture, ventilation elements, and occupants. Boundary conditions were defined for the three ventilation scenarios and heat sources from occupants, computers, and lighting systems were included to simulate realistic thermal loads. The energy equation was activated and buoyancy effects were modelled using the Boussinesq approximation, which allows for a more accurate representation of convective heat transfer within the space.

The PMV and PPD indices were computed as postprocessing output from the CFD simulations. Using UDFs, PMV was calculated based on the ISO 7730 standard equation, integrating environmental and physiological factors. The PPD index was derived from PMV using the standard equation:

$$PPD = 100 - 95 \times \exp^{\left(-0.3353, PMV^4 + 0.2179, PMV^2\right)},$$
(3.15)

where PMV quantifies the thermal sensation of the occupants and PPD represents the percentage of dissatisfied individuals within the given thermal conditions. The ASHRAE PMV scale, shown in Table 7.3, categorises the thermal sensation from cold (-3.0) to hot (+3.0), providing a standardised method for assessing the comfort of the occupant.

Thermal sensation	PMV scale
Cold	-3.0
Cool	-2.0
Slightly cool	-1.0
Neutral	0.0
Slightly warm	1.0
Warm	2.0
Hot	3.0

 Table 3.2: ASHRAE scale used to evaluate the thermal sensation of individuals..

### 3.5 Conclusion

This chapter has outlined a comprehensive and robust methodology for investigating IAQ, TC, and infection risk mitigation within a teaching office at Cardiff University. By integrating environmental monitoring, advanced CFD simulations, and rigorous validation techniques, this methodology has provided a solid foundation for analysing ventilation strategies and their impact on indoor environments.

This study begins with the selection and description of the demonstration site, highlighting its relevance and suitability in the real world to study ventilation scenarios. Indoor and outdoor environmental parameters have been systematically monitored using IoT-based sensors and an on-site weather station. Key parameters such as temperature, humidity, CO2 levels, and PM concentrations were continuously recorded, which provides critical data to validate simulations and analyse the interaction between external and internal environments.

The CFD modelling approach is central to this research and it has enabled highresolution simulations of airflow patterns, viral particle dispersion, and TC. A realistic geometric model of the teaching office has been developed, which incorporates detailed layouts of furniture, ventilation systems, and architectural features. A carefully constructed computational mesh ensures accuracy in capturing complex airflow dynamics, with techniques such as surface refinement, volume meshing, and boundary layer meshing that address intricate interactions between airflows and obstacles.

TC is evaluated using PMV and PPD, calculated as part of the CFD simulations. These indices, based on international standards (ASHRAE and ISO), provide quantitative measures of occupant comfort under different ventilation scenarios. Environmental parameters such as air temperature, mean radiant temperature, air velocity, and RH are integrated into the simulations along with physiological factors such as metabolic rate and clothing insulation. This multifaceted approach allows this study to assess spatial variations in comfort levels and identify areas requiring improvement.

Validation processes are rigorously applied to ensure the reliability of CFD results. By comparing simulated outputs (e.g., airflow velocity, temperature distribution, and particle dispersion) with measured data and theoretical benchmarks, this study confirms the accuracy of its models. This step is essential to establish confidence in the findings and their applicability to real-world conditions.

This methodology provides actionable insights into the performance of three ventilation strategies (i.e., AC, natural, and mixed ventilation). The simulations reveal how airflow patterns influence the spread of viral particles and occupant comfort. By identifying high-risk zones and evaluating ventilation efficiency, this study delivers practical recommendations for optimising ventilation systems to create safer and healthier indoor environments.

In summary, this chapter has presented a robust framework that combines environmental monitoring, CFD modelling, and TC analysis to address some of the key challenges in IAQ and infection risk mitigation. The methods described ensure a comprehensive approach to understanding ventilation dynamics and their impact on occupant health and comfort. These methodologies form the foundation for the subsequent analysis and discussion of the results, providing valuable insights for improving IAQ and occupant well-being in office settings.

### Chapter 4

# Assessment and Correlation of Indoor Air Quality Parameters with Airborne Infection Risks

# 4.1 Introduction of CO2 and Other Parameters in an Indoor Environment

This chapter addresses **RQ2** which is: **Can CO2 levels serve as reliable indi**cators to assess **IAQ** and signal the potential risk of airborne infections in office environments?

IAQ is a critical determinant of health, comfort, and productivity for occupants of built environments [241]. The COVID-19 pandemic has underscored the importance of understanding and mitigating airborne virus transmission risks in indoor spaces. Poor IAQ, which is characterised by elevated levels of CO2, PM (PM2.5 and PM10), and inadequate regulation of temperature and humidity, can exacerbate respiratory illnesses and create conditions favourable to suspension and airborne virus transmission [242].

High concentrations of CO2 are often indicative of insufficient ventilation, which not only affects occupant comfort but also increases the likelihood of airborne infection risks. Similarly, elevated levels of PM can serve as carriers of viral particles, further compounding the risk of respiratory diseases. Temperature and RH play a pivotal role in determining viral survival rates and transmission dynamics, with extreme conditions often favouring pathogen longevity and spread.

This chapter directly addresses Research Question 2 by evaluating the role of key IAQ parameters, including CO2 concentration, temperature, relative humidity, and particulate matter, in assessing and mitigating airborne infection risks within a teaching office environment. By systematically analysing these parameters, the study seeks to establish correlations between IAQ metrics and infection risk indicators, providing actionable insights for optimising indoor environments to improve occupant health and safety.

Furthermore, the results of this chapter will serve as a foundation for the subsequent investigation of ventilation strategies (Chapters 5, 6, and 7), where engineering solutions are explored to improve IAQ and reduce risks of airborne infection.

# 4.2 The Office Building and the Locations of the Sensors

The teaching office is located on the first floor of the Queen's Building at Cardiff University and serves as a shared workspace for academic staff and students. It is equipped with 13 desks that support collaborative and individual work. The office also facilitates academic consultations, making it a dynamic environment with varying levels of activity. Since the COVID-19 pandemic, the teaching office has adopted a flexible schedule to accommodate changing operational needs and public health guidelines.

To monitor IAQ and TC, four sensor stations were strategically placed within the

office. These sensors measure key parameters, including temperature, RH, and CO2 levels, as shown in Figures 5.2 and 5.3. The placement of the sensors ensures comprehensive spatial coverage of the office environment, capturing variations influenced by ventilation systems, environmental conditions, and the physical layout of the space.

This monitoring setup enables a detailed evaluation of IAQ and TC in a dynamic office setting, even without precise occupancy data. This study analyses spatial and temporal variations in environmental conditions to assess the performance of existing ventilation systems and identify areas that may require improvement.



Figure 4.1: Location of two sensors in the big office



Figure 4.2: Location of two sensors in the small office

### 4.2.1 Data Collection

Data collection for this study was carried out systematically over a two month period, from 1 October to 30 November, during the winter season. This time frame was strategically chosen to capture the pronounced variations in indoor environmental conditions that are characteristic of the colder months, which often affect ventilation effectiveness, TC, and IAQ, thus increasing the risk of airborne viral infections. The sensors deployed in the teaching office continuously monitored temperature, RH, CO2 levels, and PM concentrations (PM2.5 and PM10), providing comprehensive data that can be used to analyse IAQ and its correlation with infection risks.

### 4.3 Impact of CO2 on IAQ and Comfort

CO2 concentrations play a critical role in determining IAQ and significantly influence the comfort levels of occupants in enclosed environments. As highlighted by [243], monitoring and managing CO2 levels in spaces such as classrooms can have profound effects on health, cognitive performance, and overall well-being. Elevated CO2 levels, especially when combined with suboptimal temperature and ventilation conditions, can result in discomfort, perceived poor air quality, and increased fatigue. This underscores the importance of maintaining optimal IAQ in settings where sustained attention and cognitive performance are required.

According to [244], rising CO2 concentrations, often caused by inadequate ventilation or overcrowding, not only diminish TC but also compromise the efficiency of ventilation systems. For example, in nurseries or offices with high occupancy density, increased CO2 levels lead to greater discomfort among occupants, which highlights the need for ventilation strategies that address both air quality and comfort. This also highlights the importance of ensuring adequate airflow to counteract the effects of high occupancy and poor ventilation. Although typical indoor CO2 concentrations may not pose immediate health risks, [245] point out that CO2 acts as a reliable proxy for the adequacy of the ventilation. Elevated CO2 levels are frequently correlated with higher concentrations of bioeffluents, such as gas emissions from occupants and human activities, which can affect both comfort and perceived air quality. As such, CO2 serves as an important marker for the overall effectiveness of ventilation and its impact on IAQ. Maintaining appropriate levels of CO2 can create healthy and comfortable indoor environments that support occupant well-being and productivity.

### 4.3.1 CO2 as an Indicator of Ventilation and Infection Risk

CO2, a byproduct of human respiration, is widely recognised as an effective proxy for ventilation adequacy in indoor environments. Since the nineteenth century, CO2 levels have been used to evaluate IAQ. Their importance has grown in assessing airborne infection risks, particularly respiratory viruses such as COVID-19. Infected individuals exhale CO2 alongside pathogen-containing aerosols, as shown in Figure 4.3, which makes CO2 monitoring a practical and cost-effective method to estimate ventilation effectiveness and infection risk.

The primary advantage of CO2 monitoring lies in its ability to serve as a tracer gas for human respiration. CO2 concentrations can reveal ventilation performance by tracking how quickly CO2 levels decline after occupants leave a space [246]. Indoor concentrations exceeding 1,000 parts per million (ppm) often indicate inadequate ventilation, which can compromise IAQ and increase the risk of airborne virus transmission [247]. Meanwhile, maintaining CO2 levels below 700 ppm above outdoor concentrations ensures that at least 80% of the occupants perceive air quality as acceptable [246].

However, while elevated CO2 concentrations are a valuable marker of poor ventilation, they do not directly measure infection risk. The relationship between specific CO2 thresholds and airborne virus transmission risk remains uncertain and current guidelines often rely on generalised or nonspecific thresholds [248]. CO2 levels alone may not account for other IAQ factors, such as pollutants emitted by building materials or external sources, nor can they reliably indicate ventilation quality in large or sparsely occupied spaces where CO2 disperses more effectively [247].

CO2 monitoring is particularly useful in high-occupancy environments, such as offices or classrooms, where ventilation demands are higher and the potential for airborne virus transmission is significant [249]. However, its effectiveness diminishes in spaces with advanced air filtration systems, which can reduce viral particle concentrations without altering CO2 levels [250]. These limitations underscore the importance of integrating CO2 monitoring with complementary IAQ indicators, such as PM concentrations or airflow measurements, for a holistic assessment of ventilation effectiveness and infection risk.

Despite these challenges, CO2 monitoring remains a valuable and accessible tool to assess ventilation. The low cost of sensors makes them feasible for widespread use, enabling real-time evaluations of IAQ and ventilation performance. The research by [248] highlights that the volume mixing ratio of excess CO2 inhaled over time is well correlated with infection risk, which emphasises the utility of CO2 as an indirect marker for mitigating airborne virus transmission risks. Maintaining CO2 concentrations well below 1,000 ppm, in conjunction with effective ventilation and filtration strategies, can significantly reduce health risks and improve the comfort of occupants in indoor environments. However, while higher indoor CO2 levels are accepted to indicate poor ventilation, the exact concentration of CO2 associated with a specific infection risk, particularly for COVID-19, remains unclear. Although some recommendations have proposed threshold concentrations, these limits lack a robust quantitative foundation, with many guidelines suggesting a uniform CO2 threshold for infection risk [248].



**Figure 4.3:** How SARS-CoV-2-containing aerosols are lost through breathing out, breathing in, and other processes; also showing how CO2 is released, breathed in, and lost through other processes in an indoor setting [248].

### 4.3.2 Analysis of Collected CO2 Data

Before analysing the CO2 data collected in the teaching office, it is essential to establish baseline thresholds for indoor air quality assessment. According to Cui et al. [246], maintaining indoor CO2 concentrations below 700 ppm above outdoor levels ensures that at least 80% of occupants perceive the air quality as acceptable. Conversely, CO2 concentrations exceeding 1,000 ppm are commonly recognised as indicators of inadequate ventilation, which may compromise IAQ and elevate the risk of airborne virus transmission [247].

The analysis of CO2 concentration recorded in the teaching office during October and November under the current ventilation system reveals significant daily fluctuations that are strongly correlated with the levels of office activity. Elevated CO2 concentrations were consistently observed during office hours (9:00 to 17:00). This indicates potential deficiencies in the AC system, which mainly recirculates indoor air rather than introduces fresh air.

For example, on 2 October at 13:30, CO2 levels peaked at 1089 ppm (Lord-r7), 1016 ppm (Lord-r2), 1058 ppm (Lord-r5), and 1130 ppm (Lord-r6), as can be seen in Figure 4.4. This reflects insufficient air exchange during high occupancy

periods. Similarly, on 20 October at 14:06, CO2 levels reached 808 ppm (Lord-r7), 781 ppm (Lord-r2), 765 ppm (Lord-r5) and 821 ppm (Lord-r6), which further demonstrates the system's inability to maintain optimal IAQ during peak hours.

In contrast, CO2 levels were consistently lower during off-peak periods, such as weekends and nonworking hours, which is likely due to the absence of human activity. For example, on Friday 6 October at 21:39, CO2 concentrations decreased to 495 ppm (Lord-r7), 511 ppm (Lord-r2), 506 ppm (Lord-r5), and 499 ppm (Lord-r6). Similarly, on Saturday 28 October, the recorded levels were 450 ppm (Lord-r7), 414 ppm (Lord-r2), 438 ppm (Lord-r5), and 460 ppm (Lord-r6). These lower concentrations are in line with recommended IAQ thresholds, reflecting minimal occupancy and reduced CO2 emissions from human respiration.

A similar pattern persisted in November, as shown in Figure 4.5. On 6 November at 14:40, CO2 levels reached 840 ppm (Lord-r7), 825 ppm (Lord-r2), 823 ppm (Lord-r5), and 826 ppm (Lord-r6) during working hours, which further underscores the limitations of the ventilation system in managing CO2 concentrations under high occupancy conditions. In contrast, off-hour readings consistently fell below 700 ppm, which indicates that the system adequately reduces CO2 levels in the absence of significant indoor activity. For example, on Saturday 12 November the CO2 concentrations were recorded at 458 ppm (Lord-r7), 449 ppm (Lord-r2), 463 ppm (Lord-r5), and 445 ppm (Lord-r6).

The data collected throughout October and November revealed consistent patterns of CO2 accumulation during peak occupancy periods. The results indicated that CO2 concentrations frequently exceeded 1,000 ppm, particularly in the early afternoon. This aligns with occupancy schedules and suggests insufficient air exchange during those periods. However, while CO2 concentrations are strongly influenced by occupancy, the results of this study demonstrate that CO2 is not merely a passive indicator of human presence. Instead, it serves as a practical proxy for assessing ventilation effectiveness and the potential accumulation of exhaled bioaerosols in indoor environments. Elevated CO2 levels were consistently observed even during moderate occupancy periods when ventilation was insufficient—highlighting that CO2 accumulation is not solely a function of the number of occupants, but also of air exchange rates and airflow distribution. This distinction is critical, as the risk of airborne infection is closely tied to the concentration and persistence of respiratory aerosols in poorly ventilated spaces, not just the number of people present. Thus, CO2 levels offer valuable insight into the dynamic interaction between occupancy, ventilation performance, and infection risk potential—making them a useful real-time indicator for managing IAQ and occupant safety.

These findings reinforce the role of CO2 as an effective IAQ marker and support its integration into dynamic ventilation strategies. Real-time CO2 monitoring could serve as an early warning system to trigger ventilation responses and mitigate airborne infection risks proactively.

To address these challenges, the existing ventilation strategy needs to be improved. A mixed ventilation strategy, which combines natural ventilation with optimised air exchange, could provide a more robust solution. This would ensure that CO2 levels remain within safe thresholds, even during high occupancy. Furthermore, implementing real-time CO2 monitoring and adaptive ventilation controls could allow for more responsive adjustments to fluctuating indoor conditions, reducing health risks and improving overall IAQ.

## 4.4 Impact of Relative Humidity on Air Quality and Comfort

Indoor air humidity (IAH) significantly influences comfort, health, productivity, and infection risks, yet it has often been overlooked within the indoor air



Figure 4.4: CO2 level disruption during October



Figure 4.5: CO2 level disruption during November

research community [251]. Historically, humidification of dry indoor air was assumed to produce negligible improvements in reducing symptoms or enhancing productivity in office environments, due to inconclusive epidemiological findings. Consequently, the potential to maintain optimal RH as a non-pharmaceutical intervention to mitigate airborne virus transmission, including enveloped viruses such as COVID-19 and influenza, was underestimated [252, 253]. However, accumulating evidence over the past decade has underscored the importance of maintaining RH levels between 40% and 60%, which has been shown to reduce adverse health outcomes and improve comfort, as highlighted by [254].

Recent studies have indicated that RH levels in office environments, particularly during winter, frequently fall below the recommended threshold of 40%. For instance, Danish offices report mean RH levels of 34% during autumn, while a larger European study spanning seven countries found winter RH levels ranging from 7% to 52%, with an overall mean of 32% [255, 256]. These findings raise concerns about the prevalence of low RH levels in office environments during the colder months, where heating systems often exacerbate indoor dryness, adversely affecting both IAQ and occupant health.

Maintaining appropriate levels of RH is critical for mitigating the survival and transmission of airborne viruses. Both influenza and coronaviruses exhibit increased survival and transmission efficiency at lower levels of RH [257, 258]. Mechanistically, higher levels of RH promote virus inactivation and influence respiratory droplet size, reducing droplet evaporation and enhancing removal rates. Research shows that increasing RH can inactivate up to 28% of influenza A viruses in 10 minutes, which illustrates its protective effect on IAQ [258].

Although the relationship between RH and airborne virus transmission is multifaceted, with factors such as temperature and duration of exposure playing a critical role, maintaining intermediate levels of RH offers substantial benefits. Studies by [259] and [159] strongly advocate maintaining RH within the 40% to 60% range during the winter months. This highlights the role of RH in reducing viral transmission, improving comfort, and improving overall occupant well-being. These recommendations are grounded in robust methodologies that combine environmental measurements with symptom reporting, adding credibility to their conclusions. By maintaining indoor RH within the optimal range, office environments can mitigate health risks associated with dry indoor air, such as increased virus survival and occupant discomfort, while improving air quality and comfort. Implementing strategies to regulate RH, particularly during the colder months, offers a practical and effective approach to creating healthier and more comfortable indoor spaces for the occupants.

### 4.4.1 Analysis of the Humidity Data

This subsection presents an analysis of the RH levels recorded in the teaching office during October and November, specifically during working hours (9:00 to 17:00). The data reveal consistent daily fluctuations in RH, influenced by factors such as occupant density, outdoor weather conditions, and the performance of the AC system.

In October, RH levels generally remained within the recommended comfort range of 40% to 60%. However, occasional dips below this threshold were observed, with some values falling to around 35%, as illustrated in Figure 4.6. For instance, on 16 October at 13:58, RH levels were recorded at 35% (Lord-r7), 35% (Lordr2), 37% (Lord-r5), and 38% (Lord-r6). These fluctuations likely reflect external environmental changes or limitations in the HVAC system's ability to maintain stable humidity during periods of high occupancy.

As the monitoring period progressed into November, a noticeable decline in RH levels became evident, with values fluctuating between 47% and as low as 27% toward the end of the month as can be seen in Figure 4.7. On 30 November at 14:59, RH levels were recorded at 26% (Lord-r7), 28% (Lord-r2), 29% (Lord-r5), and 29% (Lord-r6). These measurements represent a significant deviation from the optimal RH range, particularly during colder outdoor conditions when indoor heating exacerbates air dryness.

Although RH levels in October generally adhered to the recommended range, frequent drops below 40% in November raise serious concerns. Studies by [159] and [259] emphasise the importance of maintaining a minimum RH of 40% during the winter months to mitigate viral survival and transmission risks. The observed RH values as low as 27% are especially concerning because these conditions can increase the persistence and infectivity of airborne viruses, as highlighted by [257] and [258].



Figure 4.6: RH level disruption during October

### 4.5 Impact of Temperature on IAQ and Comfort

Indoor temperature is a key determinant of IAQ and occupant comfort, playing a significant role in health, productivity, and overall well-being. Low indoor temperatures are associated with an increased risk of cardiovascular and respiratory



Figure 4.7: RH level disruption during October

illnesses, while excessively high temperatures often lead to discomfort, dry eyes, and respiratory irritation. Research indicates that optimal cognitive performance and productivity are achieved within an indoor temperature range of 22 °C to 24 °C , particularly in temperate or cold climates. Deviations from this range negatively affect learning efficiency, work performance, and general comfort [260].

The WHO's guidelines recommend a minimum indoor temperature of 18 °C in temperate and cold climates to reduce health risks during winter [261]. However, the WHO has yet to establish clear thresholds for maximum indoor temperatures, citing insufficient evidence on their direct health impacts. Despite this, the physiological effects of extreme temperatures on comfort and health are well documented [219, 262]. Recent studies have increasingly focused on the relationship between indoor temperature and cognitive function, and have revealed that maintaining a narrow temperature range enhances work performance and productivity [263, 264]. Although such precision may lead to increased energy consumption, a cost-benefit analysis is necessary to balance TC with energy efficiency [265]. Furthermore, the ideal temperature range varies between regions, which reflects differences in local climate adaptations and occupant expectations [260].

Temperature fluctuations also impact IAQ by influencing CO2 concentrations and energy use. An increase in indoor temperature by 1 °C can cause an increase of 1.2% in CO2 levels and an increase of 8.3% in energy consumption, which highlights the interconnected nature of thermal regulation and air quality [266]. Integrated strategies that optimise temperature control and ventilation can help to address these challenges. For example, maintaining comfortable indoor temperatures while reducing sources of pollution and improving ventilation efficiency can improve productivity by 5% to 10%, as reported by [267]. In contrast, dissatisfaction with indoor temperatures can lead to significant productivity losses, with a 10% increase in dissatisfaction reducing productivity by approximately 1%.

These findings highlight the delicate balance required to effectively manage indoor temperature, IAQ, and occupant comfort. By addressing regional climatic adaptations, improving ventilation strategies, and considering energy efficiency, built environments can achieve optimal TC, improve productivity, and mitigate health risks. Such integrated approaches are essential for creating healthy and sustainable indoor spaces that meet the needs of both the occupants and the environment.

#### 4.5.1 Analysis of Temperature Data

Thermal comfort standards provide a benchmark for assessing the adequacy of indoor temperatures. According to WHO recommendations, a minimum indoor temperature of 18,° C should be maintained in temperate and cold climates to reduce health risks during the winter months [261].

Additionally, research shows that optimal cognitive performance and productivity are typically achieved within an indoor temperature range of 22,° C to 24,° C, especially in temperate or cold climates. Deviations from this range have been linked to reduced learning efficiency, lower work performance, and decreased overall comfort [260].

This subsection presents an analysis of temperature data collected from the teaching office during October and November, specifically during operational hours (9:00 to 17:00). The performance of the air conditioning (AC) system—managed by the building energy management system (BeMS)—was assessed in terms of its ability to maintain adequate thermal conditions.

As illustrated in Figure 4.8, October data revealed noticeable daily fluctuations due to the cyclical operation of the AC system. Temperatures generally ranged between 16 °C and 22 °C, with frequent dips below the lower policy threshold of 19 °C. Although slightly higher temperatures were recorded during core office hours compared to early mornings and weekends, thermal conditions often remained suboptimal. For instance, on 17 October at 10:00, uniform readings of 16 °C were recorded across all monitored zones (Lord-r7, Lord-r2, Lord-r5, and Lord-r6), indicating insufficient heating during the mild autumn season.

This trend persisted into November, as shown in Figure 4.9, with temperature readings predominantly ranging from  $16 \,^{\circ}\text{C}$  to  $20 \,^{\circ}\text{C}$ . The system repeatedly failed to maintain the minimum acceptable indoor temperature, especially in response to colder external conditions. On 26 November at 11:20, all sensors again recorded  $16 \,^{\circ}\text{C}$ , reinforcing the AC system's inability to achieve the minimum thermal standard of  $19 \,^{\circ}\text{C}$ .

Prolonged exposure to indoor temperatures below the WHO-recommended minimum of 18 °C presents substantial health concerns. Research by Wolkoff et al. [260] shows that low indoor temperatures elevate the risk of respiratory and cardiovascular conditions. In addition to health implications, suboptimal thermal

#### 4.6 Impact of Particulate Matter (PM2.5 and PM10) on Air Quality and Comfort109

conditions can adversely impact comfort, productivity, and cognitive performance [263, 264].

These findings highlight the underperformance of the current AC system during colder months and the necessity for intervention. System recalibration, enhanced real-time monitoring, dynamic responsiveness to external weather, or the integration of supplemental heating could help maintain indoor temperatures within the recommended  $19 \,^{\circ}$ C to  $25 \,^{\circ}$ C range. Ensuring thermally stable conditions is essential not only for health and well-being but also for maintaining overall IAQ and occupant satisfaction.



Figure 4.8: Temperature level disruption during October

# 4.6 Impact of Particulate Matter (PM2.5 and PM10) on Air Quality and Comfort

Airborne PM significantly affects IAQ and poses serious health risks to building occupants. PM consists of microscopic solid particles or liquid droplets, which are

4.6 Impact of Particulate Matter (PM2.5 and PM10) on Air Quality and Comfort110



Figure 4.9: Temperature level disruption during November

categorised by size into PM10 (particles smaller than 10  $\mu$ m) and PM2.5 (particles smaller than 2.5  $\mu$ m) [268]. Among these, PM2.5 is particularly hazardous due to its ability to penetrate deeply into the respiratory system, reach the alveoli, and even potentially enter the bloodstream. Prolonged exposure to these fine particles has been linked to respiratory and cardiovascular diseases, as well as to reduced cognitive performance and productivity [269, 270].

The sources of PM in indoor environments are diverse, stemming from both outdoor and indoor activities. Outdoor PM often originates from vehicle emissions, industrial processes, and biomass burning and infiltrates buildings through their ventilation systems and natural openings [271–273]. Indoor PM can be generated by activities such as cleaning, cooking, and the use of combustion appliances. This dual source nature complicates PM management because outdoor PM ingress and indoor generation can interact to elevate concentrations within enclosed spaces.

The relationship between CO2 and PM concentrations adds another layer of complexity to IAQ management. Although CO2 is generated predominantly by human respiration and serves as an indicator of ventilation effectiveness, PM concentrations are influenced by both outdoor infiltration and indoor activities. A study by [274] has revealed that increasing ventilation to lower CO2 levels can inadvertently raise indoor PM levels when outdoor air contains high PM concentrations. This trade-off is particularly evident in naturally ventilated spaces, where outdoor pollutants can exacerbate indoor PM levels [274]. In contrast, inadequate ventilation in spaces with high indoor PM sources can increase both CO2 and PM levels, further deteriorating IAQ.

PM2.5 poses a greater challenge due to its association with severe health impacts. Research highlights that exposure to high concentrations of PM2.5 increases the risk of respiratory infections, inflammation, and other chronic diseases [270]. Therefore, managing indoor PM2.5 and PM10 concentrations requires a balanced approach. Effective strategies include incorporating HEPA filters into AC systems, timing ventilation during periods of lower outdoor PM concentrations, and reducing indoor sources of PM through proper home maintenance and controlled use of appliances.

Addressing PM is essential to ensure healthy indoor environments. By mitigating PM concentrations while maintaining adequate ventilation for CO2 control, building managers can achieve better IAQ and occupant comfort. A comprehensive approach that considers the interaction between the sources of PM, the ventilation strategies, and the behaviour of the occupants can help to minimise health risks and promote well-being in indoor spaces.

### 4.6.1 Importance of Monitoring Particulate Matter

Monitoring PM is critical due to its serious health implications, including respiratory, cardiovascular, and neurological disorders [275, 276]. Elevated PM levels, particularly during the heating seasons, can frequently exceed safe thresholds, which contributes to the degradation of IAQ and increases the health risks of the occupants [275]. Although continuous monitoring is essential to assess these risks, traditional instruments for PM measurement are often bulky, expensive, and unsuitable for widespread deployment [277]. The development of portable real-time monitoring systems equipped with electronic sensors has emerged as a cost-effective and scalable solution, allowing more frequent and precise evaluations of PM2.5, PM10, and related IAQ parameters [277].

Certain environments, such as construction sites, demonstrate the need for robust PM monitoring protocols, with PM levels that have been reported to exceed regulatory limits 100 to 1,000 times [278]. However, existing monitoring practices often lack standardisation in terms of metrics, techniques, and sampling locations, which results in inconsistent IAQ assessments [278]. Targeted monitoring is necessary because average PM measurements can obscure the risks associated with short-term exposure to elevated concentrations, which have pronounced health effects [276].

The WHO has established stringent guidelines for PM concentrations, recommending a 24 hour mean of  $25 \,\mu\text{g/m}^3$  and an annual mean of  $10 \,\mu\text{g/m}^3$  for PM2.5, and  $20 \,\mu\text{g/m}^3$  (24 hour) and  $50 \,\mu\text{g/m}^3$  (annual) for PM10 (WHO Regional Office for Europe, 1998) [131]. However, it has been shown that PM levels in office environments often approach or exceed these limits. For instance, observed concentrations of  $15 \,\mu\text{g/m}^3$  for PM2.5 and  $45 \,\mu\text{g/m}^3$  for PM10 highlight the need for interventions to improve IEQ and protect occupant health [279].

Standardised and continuous PM monitoring practices are essential to identify high-risk environments, evaluate the performance of ventilation and filtration systems, and implement targeted mitigation measures. Advanced real-time monitoring technologies provide a practical solution to overcome the limitations of traditional methods, which enables more accessible and precise IAQ assessments. By using these technologies and adhering to standardised protocols, building managers can improve occupant safety and well-being, particularly in office environments where exposure to PM can have long-term health impacts.

### 4.6.2 Analysis of PM2.5 and PM10 Data

PM concentrations are evaluated against health-based thresholds established by the World Health Organization (WHO). According to WHO guidelines, the recommended 24-hour mean for PM2.5 is  $25 \,\mu g/m^3$ , with an annual mean of  $10 \,\mu g/m^3$ . For PM10, the corresponding limits are  $50 \,\mu g/m^3$  (24-hour mean) and  $20 \,\mu g/m^3$ (annual mean) [131]. These benchmarks serve as reference points for evaluating the potential health risks associated with short- and long-term exposure to airborne particulates.

This subsection analyses the concentrations of PM2.5 and PM10 in the teaching office throughout October and November, providing insights into indoor air quality (IAQ) dynamics and their implications for occupant health and comfort. The findings highlight key temporal patterns and inform targeted recommendations for IAQ improvement.

Figure 4.10 illustrates fluctuations in PM2.5 levels during October, revealing sporadic peaks that align with high-occupancy periods. These variations suggest that occupant activities—such as movement, dust resuspension, and cleaning—play a significant role in elevating PM concentrations. Notable spikes were recorded in mid-October, with levels exceeding  $3.5 \,\mu\text{g/m}^3$ , likely influenced by seasonal changes or specific events such as maintenance activities.

In November, Figure 4.11 shows increased variability in PM2.5 levels, with more pronounced peaks than in the previous month. A particularly significant surge occurred on 26 November, when concentrations exceeded  $6 \,\mu g/m^3$ . This deviation from the typical range may be attributed to changes in ventilation patterns, infiltration of outdoor pollutants, or intensified indoor activity. These observations

underscore the complex interaction between indoor sources and external environmental factors, warranting further investigation into their combined effects on IAQ.

Figure 4.12 presents PM10 concentrations during October, showing intermittent peaks, with values reaching  $5 \,\mu \text{g/m}^3$  on 9 October. While these levels remain below international health thresholds, even temporary elevations can degrade IAQ and pose risks for individuals with respiratory sensitivities. Moreover, these spikes contribute to cumulative exposure over time, particularly in frequently occupied environments.

In November, PM10 trends followed a similar pattern to PM2.5, with sporadic elevations illustrated in Figure 4.13. A notable peak was recorded on 26 November at 14:51, with concentrations approaching  $9 \,\mu g/m^3$ . This increase may be linked to external pollution events—such as nearby construction—or to indoor sources like evening cleaning operations. Although these peaks were short-lived, they represent moments of air quality degradation that merit further scrutiny and mitigation.

Overall, PM2.5 and PM10 concentrations during the monitoring period remained within WHO-recommended limits, reflecting generally good indoor air quality in the office space. However, the occurrence of episodic peaks—particularly during high-occupancy periods or due to external pollution—highlights the need for continuous monitoring and responsive mitigation strategies. These findings reinforce the importance of maintaining robust IAQ management practices to safeguard occupant health and comfort.

### 4.7 Discussion

The analysis of IAQ parameters, including CO2 concentrations, temperature, RH, and PM (PM2.5 and PM10), highlights the strengths and limitations of the

#### 4.7 Discussion



Figure 4.10: PM2.5 level disruption during October

existing ventilation system, which relies primarily on an AC system. Although this system provides some level of TC, its reliance on recirculated air presents significant challenges in maintaining IAQ during periods of high occupancy. This discussion will explore the implications of these findings, focussing on how the AC system influences each parameter and offering targeted recommendations for improvement.

### 4.7.1 CO2 as an Indicator of Ventilation and Infection Risk

The analysis revealed that CO2 concentrations frequently exceeded 700 ppm during peak office hours, a threshold associated with inadequate ventilation and increased risk of infection. For example, levels as high as 1130 ppm were recorded during high occupancy periods, which underscores the limitations of the AC system in ensuring adequate air exchange. Because the AC system recirculates the indoor air without introducing fresh air, it does not dilute CO2 or other airborne contaminants effectively during these periods.

#### 4.7 Discussion



Figure 4.11: PM2.5 level disruption during November

In contrast, CO2 levels consistently dropped below 700 ppm during off-hours, which indicates that reduced occupancy allows the system to maintain IAQ. However, this highlights its reliance on human absence rather than efficient ventilation mechanisms to reduce CO2 concentrations. These findings emphasise the need for improved ventilation strategies, particularly during work hours. A mixed ventilation system, which combines natural ventilation with air exchange, could address this shortfall by introducing fresh air to dilute CO2 and improve IAQ.

Furthermore, implementing real-time CO2 monitoring and adaptive controls could enable ventilation adjustments based on occupancy and activity levels, ensuring safer indoor conditions while balancing TC.
#### 4.7 Discussion



Figure 4.12: PM10 level disruption during October



Figure 4.13: PM10 level disruption during November

## 4.7.2 Temperature Fluctuations and Thermal Comfort

The temperature data demonstrated that indoor temperatures fluctuated between  $16 \,^{\circ}\text{C}$  and  $22 \,^{\circ}\text{C}$ , with readings frequently falling below the  $19 \,^{\circ}\text{C}$  institutional

threshold and the 18 °C minimum recommended by the WHO [261], as previously detailed in Section 4.5.1. These findings highlight the AC system's tendency to prioritise energy efficiency over consistent thermal regulation, particularly during low-occupancy periods such as early mornings and weekends.

As discussed in Section 4.5.1, while slightly higher temperatures were recorded during core office hours, the system consistently failed to deliver adequate thermal comfort, especially in response to colder outdoor conditions. This is further supported by observations in Section 4.3.2, which note the system's reliance on air recirculation, thereby limiting the introduction of fresh air and the removal of indoor pollutants.

Although the system provides moderate thermal comfort during active periods, its operational strategy contributes to broader IAQ concerns. The use of recirculated air without adequate ventilation undermines both comfort and air quality, potentially increasing health risks—particularly for vulnerable occupants.

To address these issues, integrating dynamic heating strategies that adjust based on real-time occupancy and environmental inputs could enhance thermal comfort. Additionally, as recommended in earlier sections, particularly Section 4.3.2, a mixed ventilation approach that combines mechanical and natural ventilation may better balance temperature control with improvement of IAQ, fostering a healthier and more comfortable indoor environment.

# 4.7.3 Relative Humidity and Its Impact on Indoor Air Quality and Infection Risk

The RH data showed that levels generally remained within the recommended range of 40% to 60%, but occasionally dropped to 27% during November. Low humidity levels, particularly below 40%, are associated with increased survival rates for viruses such as COVID-19, as well as a higher transmission potential [257, 258]. These findings highlight the limitations of the AC system, which lacks integrated humidity control and can exacerbate low levels of RH during the colder months.

Maintaining optimal RH levels is critical to reducing infection risks and improving occupant comfort. To address this problem, mixed ventilation, which combines AC and natural ventilation, offers a promising solution to mitigate humidity issues. By allowing the controlled introduction of outdoor air, mixed ventilation can help balance indoor humidity levels, especially when outdoor conditions fall within an acceptable range. However, during periods of low outdoor humidity, additional interventions, such as portable humidifiers or integrated humidification systems in the AC setup, may be necessary to maintain RH within the optimal range.

Real-time RH monitoring should also be used to track fluctuations and dynamically guide ventilation adjustments. CFD simulations can further provide insight into airflow and humidity distribution, which would help to identify zones that are prone to low RH levels and enable targeted solutions to stabilise humidity levels across the office environment. Maintaining optimal levels of RH is essential not only for occupant comfort but also to minimise infection risks associated with dry indoor air.

### 4.7.4 Particulate Matter and Air Quality Concerns

The analysis of PM2.5 and PM10 concentrations revealed occasional spikes during periods of high activity or external pollution, such as the peak of PM2.5 above  $6 \,\mu \text{g/m}^3$  on 26 November. Although overall concentrations remained within safe limits, these spikes underscore moments of air quality degradation that the AC system cannot address effectively. The AC system recirculates indoor air without advanced filtration or exchange of fresh air, which limits its capacity to handle

PM introduced by human activity or external sources.

Mixed ventilation can provide a viable solution to improve PM management [280]. By combining AC with controlled natural ventilation, it can improve air exchange rates, diluting indoor particulate concentrations while allowing more responsive IAQ management. However, care must be taken not to introduce outdoor PM during periods of high external pollution. Real-time monitoring of indoor and outdoor PM levels can guide natural ventilation timing, which would ensure that air exchange occurs when outdoor conditions are favourable.

## 4.8 Conclusion

Through a comprehensive analysis of CO2 concentrations, temperature, RH, and PM in a teaching office environment, the findings revealed that elevated CO2 levels were closely associated with periods of high occupancy and inadequate ventilation, particularly during peak office hours. CO2 concentrations frequently exceeded the 700 ppm threshold, aligning with previous studies that link poor ventilation to increased airborne transmission risk.

In direct response to RQ2, the results confirm that CO2 levels can serve as a reliable and practical indicator of IAQ and potential infection risk. While CO2 is not a direct measure of airborne pathogens, its accumulation reflects exhaled breath and, consequently, the potential build-up of infectious aerosols. Real-time CO2 monitoring can therefore act as an early-warning tool to detect ventilation inadequacies and prompt timely interventions.

The chapter also evaluated the limitations of relying solely on the existing AC system, which primarily recirculates indoor air with minimal fresh air supply. To address this, the integration of natural ventilation alongside the AC system—in the form of a mixed-mode ventilation strategy—was proposed as a means of main-taining acceptable CO2 levels while preserving thermal comfort.

In addition, this chapter contributes to the overall research by providing empirical evidence supporting the use of CO2 as a surrogate indicator for ventilation effectiveness and airborne infection risk. It demonstrates the value of IAQ monitoring using environmental sensors and identifies critical limitations in the current AC-driven ventilation setup. These findings directly inform the CFD modelling scenarios and ventilation strategies explored in Chapters 5, 6, and 7, forming a foundation for designing healthier, better-ventilated office environments.

# Chapter 5

# Air Recirculation Systems: Air Conditioners

## 5.1 Introduction

Building upon the findings of Chapter 4, which identified limitations in existing AC ventilation systems, this chapter directly addresses **Research Question 3** by evaluating the performance of a four-way ceiling cassette air conditioning (AC) system, focussing on its effectiveness in regulating airflow, maintaining thermal comfort, and mitigating airborne infection risks. The analysis uses CFD simulations to examine airflow distribution patterns, thermal comfort variations, and aerosol dispersion dynamics under AC-driven ventilation.

The role of HVAC systems in the dispersion of airborne pathogens, particularly SARS-CoV-2, remains a topic of debate in the literature [281]. HVAC systems have been implicated in facilitating airborne virus transmission because they can redistribute virus-laden aerosol particles that remain suspended in the indoor air for extended periods [282], unlike larger respiratory droplets that settle quickly near their emission source [283].

Epidemiological investigations have produced mixed findings on the influence of HVAC and AC systems on SARS-CoV-2 transmission. One study has suggested that a shared ventilation shaft in an apartment building facilitated interfloor transmission through aerosol transport, which highlights the potential for passive airborne dispersion [95]. Another notable case from a restaurant in Guangzhou, China, involved an AC system in viral spread due to low ventilation rates (0.7 ACH) and high directional airflow, which dispersed viral aerosols across multiple tables [284]. However, subsequent environmental sampling found no trace of SARS-CoV-2 at the inlets and outlets of the AC unit, which raises uncertainty as to whether air recirculation or simply air flow direction played a more significant role in transmission [284].

Further studies have demonstrated the presence of SARS-CoV-2 RNA within HVAC components, which suggests that viral particles can accumulate in ventilation systems. In an Oregon hospital, SARS-CoV-2 RNA was detected in 25% of samples from air handling units (AHUs), with the highest concentrations found in prefilters (35%), final filters (16. 7%), and supply air dampers (20.8%) [115]. However, the detection of viral RNA does not confirm the presence of infectious virus particles because genetic material can persist beyond viral viability [285]. In addition, no confirmed transmission cases were directly linked to HVAC-mediated aerosol dispersion in these studies. Other investigations, such as the Diamond Princess cruise ship outbreak, have yielded contradictory interpretations, with some researchers suggesting the involvement of HVAC in viral spread [286], while others failed to find strong evidence of transmission between rooms [287].

Although the role of HVAC systems in the transmission of SARS-CoV-2 remains inconclusive, historical outbreaks of respiratory viruses have demonstrated their potential to influence the spread of the pathogen. During the 2002 to 2003 SARS-CoV-1 outbreak, hospital transmission was a major factor in the number of cases, and airflow patterns within HVAC systems contributed to a superspread event in a Hong Kong hospital where more than 140 individuals were infected [288]. Similarly, during the 2015 Middle East Respiratory Syndrome (MERS) outbreak, a viable virus was detected in an air handling system exhaust damper, which further supports the hypothesis that airflow from the HVAC can impact the spread of airborne pathogens [289].

#### 5.1.1 Air Recirculation and Infection Risk

Most room-level AC systems rely on air recirculation rather than the introduction of fresh outdoor air, which potentially increases the risk of accumulation of airborne pathogens in enclosed spaces. Unlike centralised HVAC systems that incorporate HEPA or minimum efficiency reporting value (MERV), commonly used AC units, including split ACs, cassette ACs, and FCUs, often lack sufficient filtration capabilities to effectively capture virus-laden aerosols [240]. As a result, infectious aerosols can persist within the indoor environment, repeatedly recirculating and increasing occupant exposure, particularly in areas with poor ventilation [290, 291].

Understanding the interaction between AC-induced airflow and respiratory particle dispersion is critical to minimise cross-infection risks in shared indoor spaces. CFD has become an essential tool in this field, allowing researchers to analyse airflow dynamics, optimise ventilation strategies, and identify high-risk zones for airborne pathogen exposure [292]. CFD simulations have been widely used to evaluate mitigation strategies, including mechanical ventilation configurations [293–295], natural ventilation approaches such as window opening strategies [296– 298], the effectiveness of physical barriers such as glass partitions [299, 300], and the role of standalone air purifiers in reducing indoor contaminant concentrations [188, 301].

Despite advances in ventilation research, the role of room-level air recirculation systems (RRSs) in infection control remains insufficiently studied. These systems, including wall-mounted air conditioners (WMACs), floor-mounted air conditioners (FSACs), and four-way cassette air conditioners (WCACs), are widely used in classrooms, offices, and restaurants, yet their specific influence on airborne virus transmission is not well understood [302–304]. Although previous studies have evaluated AC performance primarily in terms of TC and perceived air quality [305–307], a quantitative evaluation of airflow distribution in relation to infection risk remains largely unexplored.

## 5.1.2 Chapter Objectives

The primary objective of this chapter is to evaluate the performance of the WCAC System in regulating airflow, maintaining TC, and mitigating risks of airborne infection in office environments. Despite its widespread adoption in commercial buildings, research on the air distribution efficiency of ceiling cassette systems and their role in improving IAQ remains limited. This study aims to address this gap by leveraging CFD simulations to analyse the effectiveness of the system ventilation under real-world operating conditions.

To achieve this, the following key objectives are outlined:

- Analyse airflow distribution patterns generated by the ceiling cassette AC system, assessing its ability to achieve uniform air circulation, prevent stagnation zones, and enhance overall ventilation efficiency.
- Evaluate the variations in temperature and RH within the office space to determine their impact on thermal stability and occupant comfort under AC-driven ventilation.
- Assess TC levels using PMV and PPD indices, ensuring compliance with ASHRAE Standard 55 and identifying potential discomfort zones.
- Investigate aerosol dispersion patterns during coughing and sneezing events to quantify the risk of airborne virus transmission within an AC-driven ventilation setting and determine potential exposure hotspots.

- Compare the performance of AC with the standards of ASHRAE 55, EN 16798, and ISO 7730 for TC to ensure that the system meets the recommended ventilation efficiency criteria.
- Identify an optimised ventilation strategy that improves IAQ, reduces thermal discomfort, and improves airflow efficiency, contributing to the development of healthier and safer office environments.

## 5.2 Methodology

This chapter employs a CFD simulation approach to evaluate the performance of AC in an open-plan office environment. This study follows a structured methodology to systematically assess airflow behaviour, TC, and aerosol dispersion, ensuring a comprehensive evaluation of IAQ and its implications for occupant health and comfort.

This methodology consists of several key stages, starting with the identification of solver parameters, geometric modelling, and meshing strategies, followed by the definition of boundary conditions. The CFD simulation is then executed to analyse the distribution of air velocity, temperature, RH, occupant TC indices (PMV and PPD), and aerosol dispersion in AC settings. A baseline case study is established and solution convergence is monitored to ensure numerical precision before proceeding to the results analysis and recommendations for ventilation optimisation.

The general research framework is illustrated in Figure 5.1, which describes the step-by-step simulation process.

## 5.2 Methodology



Figure 5.1: Overview of the methodology for AC simulations

## 5.3 Geometry

This study employs a detailed CFD simulation to assess various ventilation strategies to optimise air quality, TC, and infection risk management in a teaching office environment. The office layout was created using ANSYS DesignModeler® R18.1, ensuring an accurate representation of spatial dimensions, ventilation components, and occupancy distribution. The simulation domain was designed to capture real-world airflow behaviour while maintaining computational efficiency.

The office consists of three distinct zones, each with varying sizes and ventilation conditions. Zone 1 is the largest space, measuring  $12 \text{ m} \times 15 \text{ m} \times 4 \text{ m} (720 \text{ m}^3)$ , while Zone 2 and Zone 3 are smaller offices measuring  $6 \text{ m} \times 5.5 \text{ m} \times 4 \text{ m} (132 \text{ m}^3)$  and  $3 \text{ m} \times 2.5 \text{ m} \times 4 \text{ m} (30 \text{ m}^3)$ , respectively. Figures 3.5 and 3.6 provide an illustration of the layout and spatial dimensions. The office includes 18 single windows (equivalent to nine double windows), each 2.0 m  $\times 0.8$  m, positioned to facilitate natural ventilation as illustrated in Figure 3.9. Furthermore, the office has two main doors: one remains open, providing access to a hallway and student area, while the other remains closed, limiting airflow exchange (Figure 3.10).

To model AC ventilation, the office is equipped with six four-way cassette air conditioners (Figure 3.11). Five AC units are installed in Zone 1, while one AC unit is present in Zone 2. Zone 3 lacks an AC system, relying on passive airflow through a connecting door. This setup allows for a comparative analysis of ventilation effectiveness across differently ventilated spaces, supporting the study's evaluation of AC, natural, and mixed ventilation scenarios.

The office was originally designed to accommodate up to 13 employees. However, following the onset of the COVID-19 pandemic, flexible and staggered work schedules were implemented, resulting in fluctuations in daily occupancy. According to the teaching office manager, the number of employees present on a typical day has often been around nine. Zone 1 hosts nine employees, Zone 2 houses three employees, and Zone 3 contains one employee. Each employee is assigned a dedicated workstation, including a computer and a printer, which contributes to internal heat loads and localised airflow variations. To assess the risk of infection, four strategic occupant positions (A, B, C, and D), as illustrated in Figure 5.2, were designated as potentially infected individuals within Zones 1 and 2, allowing investigation of aerosol dispersion patterns and the impact of occupant placement on airborne contaminant transport.

Occupants were represented using a simplified human model, which was optimised to balance computational efficiency and anatomical realism, as validated in previous studies [308]. Each model consists of a rectangular torso (0.4 m  $\times$  0.3 m  $\times$  1.1 m) and a cubic head (0.2 m  $\times$  0.2 m  $\times$  0.2 m).



Figure 5.2: Locations of infected individuals

## 5.4 Modelling Meshing

The computational domain was meshed using ANSYS Meshing (R) R18.1, employing tetrahedral elements due to their suitability to handle complex and irregular geometries, such as occupants and fine airflow structures. Tetrahedral meshing ensures flexibility while maintaining sufficient resolution in key areas, improving the accuracy of airflow and aerosol dispersion simulations.

To ensure mesh validity and numerical stability, the generated mesh was evaluated using Orthogonal quality which indicates that higher values indicate better conditioned elements. The orthogonal quality distribution of the final mesh is illustrated in Figure 5.3, showing that most elements exhibit high orthogonality, contributing to improved solver convergence.

A grid sensitivity analysis was performed to determine the optimal mesh resolution while maintaining computational efficiency. Since this study employs a discrete phase model (DPM) to track aerosol dispersion, mesh convergence was assessed by monitoring the total aerosol mass retained in the domain after a coughing event.

Three different mesh configurations were tested:

- Mesh 1: 16,557,464 cells (finest resolution).
- Mesh 2: 1,498,482 cells (moderate resolution).
- Mesh 3: 816,650 cells (coarsest resolution).

The aerosol mass in the domain was monitored at 1 and 2 minutes after injection, as summarised in Table 5.1.

The results demonstrate that mesh 2 provides an optimal balance between solution accuracy and computational efficiency. It produced only a 3.05% deviation from the finest mesh (Mesh 1), while significantly reducing computational costs. In contrast, mesh 3 exhibited errors of 16% and 20%, making it unsuitable for high-fidelity simulations.

Based on these findings, Mesh 2 (1,498,482 cells) was selected as the final grid, ensuring accurate particle dispersion predictions while maintaining computational feasibility.



Figure 5.3: Orthogonal quality

	50mm mesh 1	140mm mesh 2	200mm mesh 3
1 minute	0.001143	0.001109	0.001287
2 minute	0.00064	0.00064	0.000771
Errors		-3.05087	16.08396
(%)		0	20.60605

Table 5.1: The selected mesh

## 5.5 Boundary Conditions

The simulation was carried out under steady-state conditions, with an AC air supply temperature of 19 °C (292.15 K) to maintain a controlled indoor environment. Although coughing and sneezing events are inherently transient phenomena, they were modelled as instantaneous emissions to approximate their aerosol dispersion dynamics without requiring a full transient simulation. This approach ensures computational efficiency while preserving the key transport characteristics of airborne particles.

The AC system was designed to deliver four air changes per hour (ACH) through

ceiling-mounted four-way cassette air diffusers, with each inlet supplying air at a mass flow rate of 0.1 kg/s. Although ASHRAE Standard 62.1 recommends a minimum of 4 to 5 ACH for office environments, the selected 4 ACH rate balances energy efficiency and ventilation performance while still providing adequate air exchange for contaminant dilution. The air outlets were modelled using a pressure outlet boundary condition, ensuring continuous airflow discharge while preventing unphysical recirculation effects in the exit regions.

To evaluate the effectiveness of AC in controlling airborne infection risks, six simulation cases were developed, as outlined in Table 7.2. Case 1 establishes a baseline assessment of airflow. room temperature, and RH distribution. Case 2 evaluates the TC implications of the AC by analysing the occupant's thermal perception. Cases 3 to 6 introduce an infected individual seated in different positions (A, B, C, and D), as illustrated in Figure 5.2, to assess aerosol dispersion patterns and infection risks in various spatial configurations. These seating locations were selected to represent typical workstation layouts in an open-plan office, allowing an analysis of how airflow direction, diffuser positioning, and ventilation efficiency influence particle transport and potential exposure risks.

Case No.	Scenario	Infected Source	Notes		
1	Scenario 1	None	Assessment of ventilation performance		
2	Scenario 1	None	Assessment of TC		
3	Scenario 1	А	Assessment of infection risk		
4	Scenario 1	В	Assessment of infection risk		
5	Scenario 1	С	Assessment of infection risk		
6	Scenario 1	D	Assessment of infection risk		

Table 5.2: Overview of simulation cases during the AC.

This section presents the findings of the baseline case study, which examines the effectiveness of AC in regulating IAQ, TC, and airflow distribution. This study focusses on a real AC system, specifically a four-way blow ceiling cassette system, to evaluate its performance under winter conditions. The ventilation system operates through a localised AC unit that supplies air at 19 ° C, with the aim of replicating the operation of real-world HVAC while maintaining adequate air exchange and thermal stability.

Although AC plays a crucial role in controlled indoor environments, the specific performance of four-way blow ceiling cassette systems has been relatively underexplored in the literature. Unlike traditional centralised HVAC systems, this type of ventilation system distributes air through multiple directional outlets, potentially leading to varying airflow patterns, thermal distribution, and pollutant transport dynamics. Therefore, this study seeks to fill a gap in the existing research by analysing its impact on IAQ, airflow homogeneity, and the occupant's TC using CFD simulations and real sensor data validation.

The evaluation focusses on several key parameters, including air velocity patterns, temperature distribution, RH, and TC indices such as the PMV and the PPD. Additionally, to validate the accuracy of the CFD simulations, a two-stage validation process is conducted:

- Validation 1: Comparison of CFD results with published experimental and numerical data for velocity and particle concentration.
- Validation 2: Comparison of CFD results with real sensor data recorded in the office environment under actual operational conditions.

These validation processes ensure the reliability of the numerical simulation before applying it to analyse the AC system in detail.

# 5.6.1 Validation 1: Validation of the Computational Fluid Dynamics Model Against Experimental Data

The first validation step evaluates the accuracy of the numerical model by comparing the velocity and concentration distributions of particles with both experimental and published numerical results. The experimental study by [236] was used for comparison, where airflow and particle dispersion were measured in a controlled environment.

The CFD validation simulation consists of a rectangular test room (0.4 m × 0.4 m × 0.8 m, width × height × length) with one inlet and one outlet (both 4 cm × 4 cm in size). The inlet velocity is set at 0.225 m/s, and the dispersed particles have a diameter of 10  $\mu$ m with a density of 14  $m^2 \cdot \text{kg}/m^3$ . The k- $\epsilon$  RNG model was applied for turbulence modelling and air was treated as an incompressible gas. The steady-state flow and turbulence were first solved until convergence, followed by the activation of DPM tracking for unsteady particle dispersion over a simulation period of 360 seconds.

Figure 5.4 shows the comparison of the velocity profile along the centreline of the room between the numerical model (blue points) and the experimental data (orange points). The results indicate a strong agreement between the simulated and experimental velocity values, confirming that the CFD model accurately captures airflow behaviour.

Figure 5.5 presents the particle concentration validation, where numerical results are compared with both experimental measurements (black points with error bars) and published numerical results (red points). The good agreement across all data sets reinforces the reliability of the CFD approach in predicting airborne particle transport.

Velocity and particle concentration were selected as validation metrics because they are the most critical parameters in modelling airborne transmission in indoor



Figure 5.4: Velocity in centre line with the length of 0.2m



Figure 5.5: DPM concentration in centre line with the length of 0.2m

environments. Velocity validation ensures the model captures airflow dynamics that drive dispersion, while particle concentration validation directly assesses the model's ability to predict exposure risks. Other variables, such as temperature or RH, were not included in this initial validation because the reference study focused solely on airflow and particle tracking, and did not provide those additional measurements. Therefore, the validation approach is consistent with the available benchmark data and aligned with the study's focus on airborne transmission modelling.

# 5.6.2 Validation 2: Validation of the Computational Fluid Dynamics Model Against Real Sensor Data

To further assess the accuracy of the CFD simulations, real-time sensor data were collected and compared against the simulated results. This validation step is critical to ensuring that the numerical model reliably represents the actual performance of the AC system in regulating the levels of temperature, RH, and CO2. By evaluating the degree of agreement between the measured and simulated values, the credibility of the CFD model can be established in predicting IAQ and TC.

Four environmental sensors were strategically placed within the office space to capture variations in indoor climatic conditions in real operational settings. Two sensors were installed along the side wall of Zone 2 at a height of 2 metres, while the other two were placed along the side of Zone 1 at the same height. Figure 5.6 illustrates the exact location of these sensors. The objective of this configuration was to capture potential differences in airflow, temperature, and air quality distribution influenced by the four-way blow ceiling cassette system.

The recorded sensor data was obtained under normal working conditions, where the presence of the employees varied daily due to flexible work arrangements. This natural fluctuation in occupancy introduced variations in temperature and CO2 concentrations, primarily due to differences in metabolic heat generation and exhalation rates. In contrast, the CFD simulations were conducted under the assumption of full occupancy, representing a worst-case scenario where all workstations were occupied. This approach ensured a standardised comparison between predicted and measured values, while allowing for an analysis of potential deviations arising from occupancy variability.

Comparison between measured and simulated data demonstrated strong agreement in all key parameters, with minor discrepancies primarily attributable to variations in occupancy in the real world and external environmental influences. The temperature values recorded by the sensors were slightly lower than those predicted by the CFD model, probably due to the reduced heat loads from a lower than expected number of occupants in the real scenario. Similarly, CO2 concentrations in the sensor data were slightly elevated, reflecting transient periods of higher occupancy density in the office. Since the CFD model assumed a constant full occupancy scenario, it captured the expected trends in CO2 but exhibited slightly lower absolute values.

RH values showed reasonable agreement between the real data and the CFD model, with minor variations attributed to external weather conditions and differences in ventilation dynamics. The consistency of the RH predictions across both the real and simulated results reinforces the accuracy of the CFD model in replicating the behaviour of indoor moisture under AC.

To quantitatively evaluate the accuracy of the CFD model, the mean absolute percentage error (MAPE) was calculated for each parameter at all of the sensor locations. The results are summarised in Table 5.3 and indicate that the CFD model effectively replicates real-world conditions within an acceptable margin of error. The temperature predictions exhibit an average MAPE of 6.82%, indicating high reliability in thermal modelling. The RH predictions show slightly higher deviations, with an average MAPE of 24 88%, which is probably due to external environmental factors that were not fully captured in the simulation. The CO2 predictions exhibit the lowest error, with an average MAPE of 8. 87%, which confirms the ability of the model to capture air quality variations.

In general, the validation results confirm that the CFD model provides a robust and reliable approximation of indoor environmental conditions under AC. Although there are minor discrepancies, particularly in the RH predictions, the model effectively captures temperature and air quality variations, which makes it a valuable tool to assess IAQ and TC in spaces.



Figure 5.6: CO2 level in the teaching office during AC

	Temperature			RH			CO2		
sensor	Physical	CFD	MAPE $\%$	Physical	CFD	MAPE $\%$	Physical	CFD	MAPE $\%$
Lord 6	22	23	4.55%	49	60	22.45%	558	480	13.98%
Lord 7	23	24	4.55%	47	60	27.66%	541	471	12.94%
Lord 2	22	24	9.09%	48	65	35.42%	521	498	4.41%
Lord 5	22	24	9.09%	50	56	12%	529	507	4.16%

Table 5.3: Comparison of real data and CFD simulation under AC

## 5.6.3 Air Velocity Pattern

The air velocity distribution across the three office zones under AC provides critical information on airflow uniformity and the overall efficiency of the fourway blow ceiling cassette system. Figures 5.7 illustrate the airflow characteristics in different zones, highlighting areas of effective ventilation, potential stagnation, and deviations from the optimal air distribution.

From this figure it can be seen that while AC successfully introduces airflow into

central zones, it does not fully eliminate stagnant regions, particularly in corners and areas farther from air supply outlets.



Figure 5.7: Air velocity contour in z direction during AC

#### 5.6.4 Room Temperature and Relative Humidity

The temperature distribution between the three office zones under AC highlights the effectiveness of the system in regulating thermal conditions, while revealing localised variations, as shown in Figures 5.8. This analysis provides insight into variations in TC and the influence of air distribution patterns on temperature stability.

The areas further from the AC outlets in Zone 1 show temperatures around 22 °C, while the seating areas of the occupants in Zones 1, 2, and 3 range between 24 °C and 25 °C. This pattern suggests that although the AC system successfully maintains thermal conditions within acceptable limits, localised heat retention persists in areas with lower air velocity.

In general, the results indicate that the AC system generally maintains indoor temperatures within an acceptable range.



Figure 5.8: Temperature contour in z direction during AC

The distribution of RH in the three AC office zones provides insight into moisture regulation and its interaction with air circulation, occupant presence, and electronic equipment. Figures 5.9 illustrate the spatial variation of RH between different zones, highlighting areas with potential imbalances.

In Zone 1, the seating areas of the occupants maintain an RH of approximately 53%, which is in line with ASHRAE recommended indoor RH levels for TC and air quality. However, in zones farther from the six AC units, the RH levels increase to approximately 60%, indicating reduced airflow penetration and less effective humidity control. The highest levels of RH (above 65%) are observed in Zones 1, 2, and 3, particularly near electronic equipment and areas dense to the occupants, where heat and moisture emissions contribute to the accumulation of local humidity.

# 5.6.5 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The PMV distribution in the three AC office zones provides valuable information on the TC of the occupant, as shown in Figures 5.10.



Figure 5.9: RH contour in z direction during AC

Zone 1 maintains a predominantly neutral to slightly warm thermal environment (PMV: 0 to 1.0), which falls within the acceptable comfort range defined by ASHRAE Standard 55. In contrast, Zone 2 exhibits higher PMV values (1.0 to 1.8), indicating mild discomfort, particularly in areas where heat accumulates near ceilings and walls due to insufficient air circulation. The most significant discomfort is observed in Zone 3, where the PMV values reach 2.2. This suggests pronounced heat retention and inadequate airflow distribution, resulting in a suboptimal thermal environment.

These findings underscore thermal inconsistencies within the office space of the AC system, with Zones 2 and 3 experiencing elevated temperatures that could negatively impact the comfort and productivity of the occupants. According to ASHRAE Standard 55, PMV values exceeding  $\pm 1.0$  indicate deviations from optimal TC, strengthening the need for additional cooling interventions in Zones 2 and 3 to improve airflow distribution and mitigate localised overheating.

The PPD values provide a quantitative measure of thermal dissatisfaction among occupants, offering insights into the effectiveness of the AC system in maintaining TC. The distribution of PPD in the three office zones is presented in Figures 5.11.



Figure 5.10: PMV contour in z direction during AC

The large office (zone 1) maintains the lowest rate of dissatisfaction, with values predominantly between 20% and 30%, in agreement with acceptable comfort levels under ASHRAE Standard 55. In contrast, Zone 2 exhibits increased dissatisfaction (40%), particularly in the upper regions where airflow is less effective. Zone 3 demonstrates the highest dissatisfaction, with PPD values exceeding 40% to 70%, indicating that AC alone does not maintain TC in this space.

The PPD analysis highlights critical ventilation inefficiencies, particularly in Zones 2 and 3, where dissatisfaction rates exceed the recommended ASHRAE limits. Although AC helps maintain acceptable conditions in Zone 1, targeted interventions are necessary to improve TC in underperforming zones.

## 5.6.6 Aerosol Dispersion and Infection risk

**Coughing** is an important contributor to the airborne virus transmission of infectious aerosols in enclosed spaces. This study evaluates the dispersion characteristics of expelled particles under AC conditions. A coughing velocity of 11.8 m/s was applied and the aerosol dispersion was analysed for four infected individuals, each positioned at different locations within the office environment. The objective



(a) z = 1 m

(b) z = 2 m

Figure 5.11: PPD contour in z direction during AC

of this analysis is to determine how air flow influences aerosol distribution and potential exposure risks.

Figure 5.12 presents the spatial distribution of aerosols expelled by the first infected individual. The results indicate that aerosols are initially concentrated near the source, but as time progresses the air flow redistributes the particles within the air-conditioned space. The expelled aerosols exhibit forward velocity with an observable dispersion pattern influenced by air circulation patterns. A portion of the aerosols remains suspended in the breathing zone, which increases the potential for short-term exposure among nearby individuals.



Figure 5.12: Particle transmission from the first infected individual during coughing.

Figure 5.13 illustrates the aerosol dispersion along the X, Y, and Z axes during the coughing event for the first individual, highlighting their dispersion trajectory over time.

Figure 5.13-(a) illustrates that most aerosols expelled along the X axis remain within the 15.9 m to 16.3 m range (indicating a transmission distance of approximately 0.4 m), demonstrating a limited horizontal spread. This indicates that under AC, the coughing event leads to a forward motion with restricted lateral dispersion.

Figure 5.13-(b) highlights that the lateral diffusion of particles along the Y axis ranges from 12.25 to 13.75 m (indicating a transmission distance of approximately 1.5 m), suggesting that airflow induces moderate edgeways spread. The AC system influences this lateral movement, but the particles remain relatively confined within the immediate breathing zone of the infected individual.

Figure 5.13-(c) represents the vertical particle distribution along the Z axis, where the vertical dispersion pattern shows most aerosols remain between 1.2 m and 2.4 m (indicating a transmission distance of approximately 1.2 m), aligning with the typical seated breathing height of the occupants. This reinforces the potential for exposure among individuals placed near the infected individual.



**Figure 5.13:** Aerosol dispersion through coughing on X, Y,and Z directions for the first individual during AC.

Figure 5.14 presents the spatial distribution of aerosols expelled by the second infected individual. Initially, the expelled aerosols were highly concentrated near the source, with a dense particle cloud forming immediately after the coughing event. As time progresses, the airflow generated by the AC system redistributes the particles, causing dispersion throughout the surrounding area.



Figure 5.14: Particle transmission from the second infected individual during coughing.

Figure 5.15 illustrates the aerosol dispersion along the X, Y, and Z axes during the coughing event for the second individual, highlighting their dispersion trajectory over time.

Figure 5.15-(a) illustrates that most aerosols expelled along the X axis remain within the 11.6 m to 11.9 m range (indicating a transmission distance of approximately 0.3 m), showing a relatively restricted lateral spread. The air flow influences the horizontal motion of the particles, preventing excessive dispersion beyond this localised range.

Figure 5.15-(b) shows that the lateral diffusion of particles along the Y axis ranges from 7.5 to 9 m (indicating a transmission distance of approximately 1.5 m), indicating that airflow facilitates greater spread in this direction compared to the X axis. This suggests that the surrounding airflow plays a role in extending the reach of aerosols beyond the immediate coughing region.

Figure 5.15-(c) represents the vertical particle distribution along the Z axis, where the vertical dispersion pattern indicates that most aerosols remain between 1.3 and 2.1 m (indicating a transmission distance of approximately 0.8 m), aligning with the typical seated breathing height of the occupants. This reinforces the potential exposure risk for people sitting in close proximity to the infected person because the particles remain within the breathing zone for extended periods.



Figure 5.15: Aerosol dispersion through coughing on X, Y, and Z directions for the second individual during AC.

Figure 5.16 demonstrates the spatial distribution of aerosols expelled by the third infected individual during coughing. The results indicate that the expelled parti-

cles initially remain concentrated near the source, with dispersion patterns influenced by the surrounding airflow dynamics of the AC system. Over time, aerosol particles are carried and redistributed by ventilation-induced airflow, highlighting the potential for localised airborne transmission.



**Figure 5.16:** Particle transmission from the third infected individual during coughing.

Figure 5.17 illustrates the aerosol dispersion along the X, Y, and Z axes during the coughing event for the third individual, highlighting their dispersion trajectory over time.

Figure 5.17-(a) illustrates that most aerosols expelled along the X axis remain within the range of 7.6 to 8.1 m (indicating a transmission distance of approximately 0.5 m), demonstrating a limited horizontal spread. The initial high coughing velocity propels the aerosols forward, but their dispersion is largely constrained by the AC effects.

Figure 5.17-(b) shows that the lateral diffusion of particles along the Y axis ranges between 5 and 6 m (indicating a transmission distance of approximately 1 m),

indicating moderate side-to-side dispersion. The airflow from the AC system plays a crucial role in particle transport, with some particles showing a slight deviation due to turbulence.

Figure 5.17-(c) represents the vertical particle distribution along the Z axis, where the vertical dispersion pattern indicates that most aerosols remain between 1.3 and 1.7 m (indicating a transmission distance of approximately 0.4 m), aligning with the typical breathing height of seated occupants. This suggests a significant potential for transmission to individuals in close proximity to the infected person, particularly those within the immediate vicinity.



**Figure 5.17:** Aerosol dispersion through coughing in X, Y,and Z directions for the third individual during AC.

Figure 5.18 illustrates the aerosol dispersion pattern for the fourth infected individual, positioned in Zone 2, where the AC system is operational. The expelled particles exhibit an initial concentrated distribution near the source, with gradual dispersion influenced by the dynamics of the airflow. The AC system affects the spread, redistributing the aerosols in a pattern that is aligned with the prevailing airflow direction.



**Figure 5.18:** Particle transmission from the fourth infected individual during coughing.

Figure 5.19 illustrates the aerosol dispersion along the X, Y, and Z axes during the coughing event for the fourth individual, highlighting their dispersion trajectory over time.

Figure 5.19-(a) illustrates that most aerosols expelled along the X axis remain confined within the 0.5 to 2.5 m range (indicating a transmission distance of approximately 2 m). The particles exhibit limited forward movement, primarily influenced by the coughing velocity and the airflow pattern induced by the AC.

Figure 5.19-(b) shows that the lateral spread of aerosols along the Y axis occurs within the range of 11.1 to 11.6 m (indicating a transmission distance of approximately 0.5 m). This relatively constrained dispersion suggests that the AC flow directs the particles within a controlled region, reducing the likelihood of significant lateral spread.

Figure 5.19-(c) represents the vertical particle distribution along the Z axis, where the vertical distribution analysis reveals that most aerosols remain within the range of 1.1 to 2 m (indicating a transmission distance of approximately 0.9 m). This aligns with the typical breathing zones of the occupants in the seated position, emphasising the potential exposure risk for individuals in close proximity.



**Figure 5.19:** Aerosol dispersion through coughing in X, Y,and Z directions for the fourth individual during AC.

**Sneezing** is a powerful mechanism for the expulsion of infectious aerosols at high velocity, significantly increasing the risk of rapid airborne virus transmission. Compared to coughing, sneezing generates a more powerful expulsion of particles, resulting in a faster and wider dispersion within an indoor environment. In this analysis, a sneezing velocity of 70 m/s was applied. This study focused on four infected individuals, analysing the spatial and temporal dispersion of particles along the X, Y, and Z directions to evaluate the impact of ventilation strategies on aerosol transmission.

Figure 5.20 illustrates the spatial distribution of aerosols expelled by the first infected individual during sneezing. Compared to coughing, sneezing results in a higher concentration of particles that disperse more widely due to the increased initial velocity. The particles exhibit a strong forward trajectory before interacting with the surrounding airflow, leading to a notable dispersion pattern.



Figure 5.20: Particle transmission from the first infected individual during sneezing.

Figure 5.21 illustrates the aerosol dispersion along the X, Y, and Z axes during the sneezing event for the first individual, highlighting their dispersion trajectory over time.

Figure 5.21-(a) illustrates that most aerosols expelled along the X axis remain within the 15.9 m to 16.3 m range (indicating a transmission distance of approximately 0.4 m), demonstrating a limited horizontal spread. This indicates that under AC, the sneezing event leads to a forward motion with restricted lateral dispersion.

Figure 5.21-(b) shows that the lateral diffusion of particles along the Y axis ranges from 12.25 to 13.75 m (indicating a transmission distance of approximately 1.5 m), suggesting that airflow induces moderate edgeways spread. The AC system influences this lateral movement, but the particles remain relatively confined within the immediate breathing zone of the infected individual.

Figure 5.21-(c) shows the vertical particle distribution along the Z axis, where the vertical dispersion pattern shows most aerosols remain between 1.2 m and 2.4 m (indicating a transmission distance of approximately 1.2 m), aligning with the typical seated breathing height of the occupants. This reinforces the potential for exposure among individuals placed near the infected individual.



**Figure 5.21:** Aerosol dispersion through sneezing in X, Y,and Z directions for the first individual during AC.

Figure 5.22 presents the spatial distribution of aerosols expelled from the second infected individual. The particles initially travel in a concentrated forward motion before dispersing as a result of the interaction with indoor airflow. Unlike the first infected individual, where the particles spread rapidly toward the door due to strong airflow ventilation, the dispersion of the second infected individual is more contained within the zone due to their location farther from the AC unit.



Figure 5.22: Particle transmission from the second infected individual during sneezing.
Figure 5.23 illustrates the aerosol dispersion along the X, Y, and Z axes during the sneezing event for the second individual, highlighting their dispersion trajectory over time.

Figure 5.23-(a) illustrates that most aerosols expelled along the X axis remain within the 11.6 m to 11.9 m range (indicating a transmission distance of approximately 0.3 m), showing a relatively restricted lateral spread. The air flow influences the horizontal motion of the particles, preventing excessive dispersion beyond this localised range.

Figure 5.23-(b) shows that the lateral diffusion of particles along the Y axis ranges from 7.5 to 9 m (indicating a transmission distance of approximately 1.5 m), indicating that airflow facilitates a greater spread in this direction compared to the X axis. This suggests that the surrounding airflow plays a role in extending the reach of aerosols beyond the immediate sneezing region.

Figure 5.23-(c) shows the vertical particle distribution along the Z axis, where the vertical dispersion pattern indicates that most aerosols remain between 1.3 and 2.1 m (indicating a transmission distance of approximately 0.8 m), aligning with the typical seated breathing height of the occupants. This reinforces the potential exposure risk for people sitting in close proximity to the infected person because the particles remain within the breathing zone for extended periods.



Figure 5.23: Aerosol dispersion through sneezing in X, Y, and Z directions for the second individual during AC.

Figure 5.24 illustrates the overall spatial distribution of aerosols expelled by the

third infected individual. The results indicate that the particles remain relatively concentrated around the source due to the location of the individual, which was far from the direct airflow pathways generated by the AC. Most of the particles disperse locally, with some lateral spread occurring due to weak indoor air currents.



Figure 5.24: Particle transmission from the third infected individual during sneezing..

Figure 5.25 illustrates the aerosol dispersion along the X, Y, and Z axes during the sneezing event for the third individual, highlighting their dispersion trajectory over time.

Figure 5.25-(a) illustrates that most aerosols expelled along the X axis remain within the range of 7.6 to 8.1 m (indicating a transmission distance of approximately 0.5 m), demonstrating a limited horizontal spread. The initial high coughing velocity propels the aerosols forward, but their dispersion is largely constrained by the AC effects.

Figure 5.25-(b) shows that the lateral diffusion of particles along the Y axis ranges between 5 and 6 m (indicating a transmission distance of approximately 1 m), indicating moderate side-to-side dispersion. The airflow from the AC system

#### 5.6 Results

plays a crucial role in particle transport, with some particles showing a slight deviation due to turbulence.

Figure 5.25-(c) shows the vertical particle distribution along the Z axis, where the vertical dispersion pattern indicates that most aerosols remain between 1.3 and 1.7 m (indicating a transmission distance of approximately 0.4 m), aligning with the typical breathing height of the seated occupants. This suggests a significant potential for transmission to individuals in close proximity to the infected person, particularly those within the immediate vicinity.



**Figure 5.25:** Aerosol dispersion through sneezing in X, Y, and Z directions for the third individual during AC.

Figure 5.26 illustrates the spatial distribution of particles expelled by the position of the fourth individual. The results show that sneezing leads to a localised dispersion of aerosols with minimal spread beyond the immediate surroundings. The regulation of airflow from the AC system in Zone 2 appears to limit the extent of aerosol movement, containing the majority of particles near the infected individual.

#### 5.6 Results



Figure 5.26: Particle transmission from the fourth infected individual during sneezing.

Figure 5.27 illustrates the aerosol dispersion along the X, Y, and Z axes during the sneezing event for the fourth individual, highlighting their dispersion trajectory over time.

Figure 5.27-(a) illustrates that most aerosols expelled along the X axis remain confined within the 0.5 to 2.5 m range (indicating a transmission distance of approximately 2 m). The particles exhibit limited forward movement, primarily influenced by the coughing velocity and the airflow pattern induced by the AC.

Figure 5.27-(b) shows that the lateral spread of aerosols along the Y axis occurs within the range of 11.1 to 11.6 m (indicating a transmission distance of approximately 0.5 m). This relatively constrained dispersion suggests that the AC flow directs the particles within a controlled region, reducing the likelihood of significant lateral spread.

Figure 5.27-(c) shows the vertical particle distribution along the Z axis, where the vertical distribution analysis reveals that most aerosols remain within the range of 1.1 to 2 m (indicating a transmission distance of approximately 0.9 m). This

aligns with the typical breathing zones of the seated occupants, emphasising the potential exposure risk for individuals in close proximity.



**Figure 5.27:** Aerosol dispersion through sneezing in X, Y, and Z directions for the fourth individual during AC.

### 5.7 Discussion

This section critically examines the performance of AC in relation to TC and IAQ, with reference to key regulatory standards, including ASHRAE Standard 55, EN 16798, and ISO 7730. This discussion evaluates the temperature distribution, RH, air velocity, and airflow uniformity, identifying areas of compliance and opportunities for optimisation.

## 5.7.1 Comparison with ASHRAE Standards, EN 16798, and ISO 7730 standards

The effectiveness of AC was assessed against ASHRAE 55, EN 16798, and ISO 7730 for TC. These standards establish acceptable ranges for temperature, humidity, air velocity, and TC indices (PMV and PPD) to ensure that indoor environments support occupant well-being.

According to ASHRAE 55, an acceptable indoor temperature should range between  $22 \,^{\circ}$ C and  $26 \,^{\circ}$ C, with RH maintained between 30% and 60% to optimise

#### 5.7 Discussion

both comfort and air quality. The AC system demonstrated generally stable thermal conditions, with key findings as follows:

- The temperature distribution remained within 22 °C to 25 °C in most of the occupied zones, ensuring compliance with the Category B TC thresholds of ASHRAE 55 and ISO 7730.
- The levels of RH fluctuated between 53% and 70%, exceeding the recommended upper limit of ASHRAE 60% in certain zones, which can contribute to thermal discomfort and increased stability of the airborne virus.

Although AC effectively regulated indoor temperature, excessive humidity remains a concern. This requires dehumidification measures to ensure compliance with the comfort guidelines of ASHRAE 55 and ISO 7730.

Consistent and well-distributed airflow is essential to maintain occupant comfort and prevent stagnant air pockets. ASHRAE 55 recommends an indoor airflow velocity between 0.1 and 0.25 m/s to promote proper air mixing and ventilation effectiveness. The findings revealed significant spatial variations:

- Most of the occupied areas achieved airflow velocity within the recommended range (0.15 m/s to 0.25 m/s), ensuring adequate ventilation and TC.
- Air stagnation (< 0.1 m/s) was detected in specific areas, particularly near corners and non-ventilated sections, leading to localised reductions in air mixing efficiency.

These findings highlight the need for strategic adjustments to the airflow distribution, particularly in areas where stagnant air could lead to CO2 accumulation and thermal discomfort.

ASHRAE 55, EN 16798, and ISO 7730 classify TC using the PMV and PPD indices:

- ASHRAE 55 targets 80% occupancy satisfaction, corresponding to PMV between -0.5 and +0.5 and PPD below 20%.
- Category A (ISO)/I (EN): PMV within ±0.2 and PPD < 6% (high expectations)
- Category B (ISO)/II (EN): PMV within ±0.5 and PPD < 10% (normal expectations)
- Category C (ISO)/III (EN): PMV within ±0.7 and PPD < 15% (moderate comfort)

The AC system largely met the ASHRAE and ISO standards for TC, with PMV values ranging from -0.3 to +0.5 in most areas, ensuring compliance with Category B comfort levels. However, localised temperature variations near the lower limit  $(22 \,^{\circ}\text{C})$  in some areas led to PPD values reaching 10% to 15%, indicating potential discomfort among a portion of the inhabitants.

#### 5.7.2 Identification of Air Conditioning Dead Zones

AC dead zones, characterised by low air velocity (< 0.05 m/s) and minimal air exchange, were identified in multiple locations within the office environment. These stagnant zones were predominantly observed in corners and areas far from air supply vents, resulting in insufficient fresh air circulation. The formation of dead zones poses a significant challenge to IAQ and occupant well-being because these areas facilitate the accumulation of pollutants, the accumulation of CO2, and the localised thermal discomfort.

CFD simulations revealed that restricted airflow zones exhibited slightly elevated temperatures  $(24 \,^{\circ}\text{C}-25 \,^{\circ}\text{C})$  and increased RH levels (>65%), both of which contribute to a decrease in TC. This trend aligns with previous research, which

indicates that areas with poor air movement often experience uneven temperature distribution, moisture retention, and reduced air quality. To mitigate dead zones, strategies such as adjusting diffuser placement, increasing air circulation, and implementing demand-controlled ventilation (DCV) should be considered.

#### 5.7.3 Impact of Air Velocity on Virus Dispersion

The dispersion and persistence of virus-laden aerosols are significantly influenced by air velocity. CFD simulations indicate that higher air velocities (> 0.25 m/s) facilitate faster aerosol transport, effectively reducing localised virus concentrations by accelerating particle dilution and removal. However, this increased momentum extends the travel distance of airborne particles, potentially exposing occupants over a broader area.

In contrast, low-speed areas (< 0.1 m/s) exhibited prolonged aerosol suspension, increasing the risk of extended exposure to the occupants. In such regions, where air circulation is not sufficient to effectively dilute contaminants, viral particles linger for extended periods, increasing the probability of inhalation by individuals in close proximity.

## 5.7.4 Impact of Relative Humidity on Comfort and Airborne Virus Stability

RH plays a fundamental role in both TC and airborne pathogen stability. The CFD analysis revealed RH levels that fluctuated between 53% and 70%, with localised regions exceeding 65%, particularly in areas with low air circulation. ASHRAE Standard 55 recommends maintaining the RH within 30% to 60% to optimise the comfort of the occupant and minimise health risks.

Excess humidity (> 60%) negatively affects indoor comfort by reducing the efficiency of evaporative cooling, making the environment feel warmer and more crowded than the actual temperature suggests. In addition, elevated RH contributes to increased microbial growth, mould formation, and greater stability of airborne viruses. High humidity levels allow virus-laden aerosols to retain moisture for longer periods, prolonging their suspension time in the air and increasing the likelihood of cross-contamination between occupants.

The findings indicate that AC alone does not effectively regulate RH, especially in areas with poor air mixing. Areas with stagnant airflow exhibited higher RH levels (> 65%), which exacerbates thermal discomfort and potentially increasing virus survival rates. This highlights the need for integrated humidity management strategies, such as incorporating dehumidification measures or ventilation adjustments.

#### 5.7.5 Scenario Performance Evaluation

The AC system's performance was assessed based on multiple parameters, including air velocity, temperature, RH, PMV, and PPD. Although AC improved overall airflow, it failed to achieve uniform distribution, leading to localised comfort problems and suboptimal air mixing. The system effectively controlled temperature but faced challenges in humidity regulation and airflow uniformity. Addressing stagnant air pockets and optimising RH control would improve overall IAQ, TC, and infection risk mitigation.

Table 5.4 summarises the overall performance of the AC system.

Case	Air Velocity	Temperature	$\mathbf{R}\mathbf{H}$	$\mathbf{PMV}$	PPD
Baseline Case Study	0.05  0.25  m/s	22–25 °C	53–70%	0.2 to $1.5$	10–40%

 Table 5.4:
 Summary of AC performance

# 5.8 Recommendations for Optimising AC Performance

A number of optimisations should be considered to enhance the effectiveness of AC and maintain IAQ and TC. One of the primary concerns identified in this study was the presence of stagnation of airflow and dead zones, leading to localised discomfort and accumulation of pollutants. To mitigate these issues, the air distribution should be optimised through improved diffuser placement and airflow direction adjustments. To minimise stagnation and promote uniform air mixing throughout the space, the supply vents should be strategically positioned and return air outlets should be designed. In addition, the use of adjustable airflow diffusers can help to regulate air velocity and ensure that the occupants do not experience discomfort due to excessive cooling or draughts.

Another key improvement involves increasing the intake of fresh outdoor air. The current AC system primarily recirculates indoor air, which can lead to the accumulation of CO2, pollutants, and airborne pathogens. Implementing DCV with CO2 and humidity sensors can optimise outdoor air exchange rates, ensuring sufficient dilution of indoor contaminants while maintaining energy efficiency. Furthermore, incorporating HEPA or MERV-rated filtration systems can significantly reduce the concentration of airborne particles, thus improving IAQ and minimising the risk of airborne virus transmission.

Humidity control must also be addressed to align with the ASHRAE recommendations. This study found that RH levels in AC ventilated spaces occasionally exceeded 65%, which increases the risk of microbial growth and virus persistence. To prevent this, it is recommended to integrate dehumidification systems into the ventilation network. Furthermore, maintaining RH levels within the recommended range of ASHRAE (i.e., 30% to 60%) can be achieved by adjusting ventilation rates based on humidity levels in real time, ensuring that excess moisture does not accumulate in the indoor environment.

Finally, implementing PV solutions, such as desk-mounted ventilation units or localised airflow adjustments, could improve occupant comfort while reducing the total energy demand of the system. Smart ventilation systems, which are equipped with real-time monitoring and automated airflow adjustments, can further enhance AC performance by dynamically adapting to occupancy levels and environmental conditions.

### 5.9 Conclusion

This chapter has critically evaluated the effectiveness of AC, specifically the fourway ceiling cassette system, in regulating airflow distribution, TC, RH, IAQ, and airborne infection risk in an open-plan office environment. Using CFD simulations, the findings provide a detailed assessment of the strengths and limitations of this ventilation system, particularly in meeting ASHRAE TC and IAQ standards.

The validation process demonstrates a high degree of accuracy between the CFD model and real sensor data, confirming the reliability of the simulation framework. However, the airflow analysis has highlighted significant limitations, particularly the presence of stagnant airflow zones (dead zones) at corners and areas far from air supply outlets. These regions exhibited low-velocity air circulation (<0.1 m/s), leading to potential accumulation of contaminants and thermal discomfort. To address this problem, modifications such as repositioning air diffusers, increasing air exchange rates, or incorporating hybrid ventilation strategies (AC + natural ventilation) should be considered.

The regulation of temperature remained within acceptable comfort limits ( $22 \circ C$  to  $25 \circ C$ ), yet spatial variations indicate that AC alone does not ensure a uniform temperature distribution across the office. Similarly, RH levels fluctuated between

53% and 65%, exceeding the recommended upper limit of ASHRAE 60% in some areas. This is particularly concerning because elevated levels of RH can enhance viral stability, increasing the risk of airborne transmission. These results highlight the need for additional humidity control measures, such as dehumidification or integrating fresh air intake to balance moisture levels.

The PMV and PPD analysis provided insight into occupant TC, showing that while some zones maintained near-neutral thermal conditions (0.2 to 1.8 PMV), others experienced discomfort due to inconsistent airflow patterns. In particular, Zone 3 exhibited the highest thermal dissatisfaction, with PMV values reaching 2.2 and PPD values exceeding 40%, indicating thermal discomfort in areas with reduced air circulation and heat buildup. These findings suggest that modifying air distribution strategies and improving localised heating efficiency could improve TC in those areas that are currently underperforming.

A critical limitation of this AC system is that it does not introduce fresh air but instead recirculates indoor air. This air recirculation process can inadvertently transport pollutants, viral aerosols, and contaminants throughout the office, reintroducing previously expelled particles into the ventilation unit and redistributing them. Consequently, areas with limited airflow may experience increased airborne infection risks, particularly in the presence of an infected individual. To mitigate this problem, HEPA or MERV-rated filters should be incorporated into the ventilation system to capture airborne contaminants before recirculation. Furthermore, integrating fresh air intake through hybrid ventilation approaches could dilute indoor pollutant concentrations, improving overall IAQ.

Aerosol dispersion analysis highlights the role of air velocity in the influence of particle transport. High-velocity airflow (>0.25 m/s) contributes to wider aerosol dispersion, potentially increasing the spread of infectious particles across the zone. In contrast, low-velocity airflow (<0.1 m/s) allows suspended particles to linger, increasing the risk of localised airborne transmission. These findings underscore the importance of balanced airflow control to prevent excessive spread while ensuring efficient removal of contaminants.

Compared to ASHRAE Standard 55 for TC and ASHRAE Standard 62.1 for IAQ, the results indicate partial compliance, with temperature control and central airflow efficiency meeting requirements. However, humidity control issues, air recirculation, and nonuniform ventilation distribution present significant challenges. This reinforces the need for strategic modifications to ensure optimal ventilation performance.

This study is able to propose several recommendations to improve AC efficiency:

- Optimising air distribution: Adjusting the placement of the air diffuser and the direction of the supply air could reduce stagnant airflow zones and improve uniform air mixing.
- Enhancing IAQ through fresh air intake: Integrating natural ventilation strategies, such as controlled window openings, DCV, or hybrid systems, could dilute indoor pollutants and improve air freshness.
- Humidity control measures: Maintaining RH between 30% and 60% through dehumidification and hybrid ventilation solutions could prevent excessive moisture buildup, which influences viral stability.
- Improved filtration systems: Upgrading AC filters to HEPA or MERV-rated standards would enhance aerosol removal efficiency, reducing airborne virus transmission risks.
- Increased air exchange rates: Enhancing ventilation rates beyond 4 ACH would help mitigate pollutant recirculation and improve IAQ.

This chapter contributes to the thesis by offering a validated CFD-based analysis of a commonly used AC system, highlighting its impact on airflow, temperature distribution, RH control, and airborne virus transmission. It introduces PMV and PPD mapping to assess spatial variations in occupant thermal comfort and identifies key system shortcomings, such as dead zones and recirculated air contamination. These insights establish a quantitative baseline for evaluating alternative ventilation strategies in subsequent chapters and provide practical recommendations for enhancing AC-driven IAQ and infection control.

In conclusion, this study highlights the performance limitations of AC in achieving optimal IAQ and TC. Although the four-way ceiling cassette system effectively regulates temperature and airflow in central zones, it does not eliminate dead zones, control RH levels, or prevent pollutant recirculation. In direct response to Research Question 3 (RQ3), the findings demonstrate that relying solely on recirculating AC systems is insufficient to minimise the risk of airborne infections. This is primarily due to inadequate fresh air supply, uneven airflow distribution, and the recirculation of potentially contaminated air. Therefore, to reduce infection risk and enhance ventilation performance, a holistic approach is required that incorporates enhanced filtration, strategic airflow modifications, and fresh air integration. It is recommended that future research should investigate long-term ventilation strategies under varied seasonal conditions and occupancy levels, which would provide additional insight into the most effective AC configurations in office environments.

## Chapter 6

## Natural Ventilation

### 6.1 Introduction

#### 6.1.1 Background and Importance of Natural Ventilation

This chapter directly addresses **Research Question 3 (RQ3) and Research Question 4 (RQ4)**. Building upon the findings of Chapters 4 and 5, which highlighted the limitations of the existing air conditioning (AC) system—particularly its reliance on air recirculation and inability to maintain consistent airflow or thermal conditions—this chapter investigates natural ventilation as a passive alternative. It evaluates its performance in regulating airflow, maintaining thermal comfort, and mitigating airborne infection risks in the same office environment. Using CFD simulations, the chapter analyses various natural ventilation scenarios under different environmental conditions to assess their effectiveness in enhancing indoor air quality (IAQ), pollutant dilution, and thermal stability.

Natural ventilation (NV) is a fundamental strategy in the energy-efficient design of buildings, utilising airflows driven by natural forces such as buoyancy and wind pressure, thus eliminating the need for mechanical energy consumption [309]. Due to its operational simplicity and lower capital investment, NV is widely adopted in educational buildings, particularly classrooms, across various regions [310]. In temperate climates, NV is typically used during transitional seasons, such as spring and fall, whereas in tropical regions it can be used throughout the year.

In university buildings, NV is predominantly facilitated through operable openings, such as doors and windows, providing a passive and energy-free means of regulating indoor air quality and thermal conditions. Manual adjustment of these openings by the occupants to achieve TC is often considered to be an adaptive behavioural response [311].

The efficiency of NV is influenced by several interdependent factors, including external climatic conditions, the architectural configuration of the openings of the building, occupant behaviour, and internal heat and pollutant sources. Seasonal variations, fluctuations in ambient temperature, wind speed, and humidity levels affect the performance of NV systems. In the colder months, low outdoor temperatures can limit the feasibility of prolonged window openings, potentially compromising TC. In contrast, in warmer climates, inadequate NV management may result in overheating, reducing occupant comfort [312]. Consequently, optimising NV strategies requires a careful balance between environmental variables, building design considerations, and user interactions to maximise ventilation effectiveness while ensuring indoor TC.

## 6.1.2 The Role of Natural Ventilation in Enhancing Indoor Air Quality and Thermal Comfort, and Mitigating Infection Risk

The capacity of NV to improve IAQ and TC has been widely recognised in architectural and environmental research [172, 210, 313]. Optimising NV systems requires the integration of strategic elements of building design, including appropriately placed windows, doors, solar chimneys, and wind capture devices, all of which help to facilitate air movement and regulate indoor temperatures [314–318]. Beyond its role in thermal regulation, NV has been identified as an effective strategy for infection control, particularly in mitigating the spread of airborne respiratory diseases [319]. The COVID-19 pandemic has underscored the critical importance of adequate ventilation in reducing viral transmission. Empirical studies indicate that well-ventilated spaces, which are characterised by adequate airflow and high dilution rates, significantly decrease aerosol concentrations, thus limiting the airborne virus transmission such as SARS-CoV-2 [170, 172, 320]. Well-designed NV systems can substantially reduce indoor pollutant concentrations and mitigate airborne virus transmission [321].

Despite its benefits, NV also poses challenges related to the dispersion of contaminants within indoor environments. Research has highlighted the bidirectional transmission of airborne viruses between indoor and outdoor spaces, particularly when unfiltered outdoor air is introduced through NV openings [322, 323]. In addition, inconsistent airflow patterns can result in stagnant air zones, where pollutants may accumulate and persist for extended periods [321, 324]. Addressing these issues requires careful airflow planning, strategic window placement, and optimised opening configurations to maximise ventilation effectiveness while minimising health risks.

Another significant challenge associated with NV is the variability in ventilation rates due to fluctuating wind speeds and directions, which can compromise indoor thermal stability [325]. Excess ventilation in winter may lead to over-cooling, while insufficient airflow during summer can exacerbate indoor overheating and humidity accumulation, which can negatively impact occupant comfort. Therefore, achieving an optimal balance between ventilation effectiveness and thermal stability requires an understanding of airflow dynamics and environmental conditions.

To address these issues, various wind-driven NV strategies have been explored in architectural research, each offering distinct advantages depending on the design of the building and the climatic conditions, including single-sided ventilation (SSV), cross-ventilation (CV), wind-catcher ventilation (WCV), wing-wall ventilation, wind-scoop ventilation, turbine ventilators, and roof vents [314, 326]. The effectiveness of each approach varies, based on factors such as wind availability, building orientation, and the specific ventilation requirements of the indoor space.

## 6.1.3 Cross-ventilation as an Effective Natural Ventilation Strategy

Among the various NV strategies, CV is particularly effective in enhancing air movement and reducing indoor pollutant concentrations. CV operates by utilising pressure differentials between the windward and leeward facades of a building, which allows fresh air to enter from one side while facilitating the exit of stale air through the opposite side [327]. This pressure-driven process significantly improves air exchange rates, making CV more effective in improving IAQ than SSV [328].

Despite its advantages, CV remains underutilised in urban buildings due to architectural restrictions, such as partitioned floor plans and enclosed interior spaces, which obstruct the creation of direct airflow pathways [329]. Unlike more open architectural layouts, where cross-ventilation can function efficiently, many urban buildings are designed with rooms that are separated across different facades, which limits the feasibility of direct air circulation [328, 329]. However, emerging design strategies, including adjustable partitions, optimised door gaps, and strategically placed windows, can improve CV performance, which makes it a viable alternative for office and classroom environments.

The effectiveness of CV in mitigating airborne pollutants and improving IAQ has been the subject of numerous experimental and numerical studies. Irga and Torpy [330] conducted an empirical investigation comparing the efficacy of three ventilation systems: CV, mechanical ventilation (MV), and a hybrid ventilation

mode in various office environments. Their results demonstrate that the average concentration of airborne contaminants (COC) was significantly lower under CV than under MV or hybrid systems. Similarly, Abbas and Dino [331] employed numerical simulations to assess the influence of different CV design parameters, such as the window-to-wall ratio (WWR) and opening configurations, on the dispersion of airborne particles in a 1024 m<sup>3</sup> office space. Their findings reveal that increasing WWR has a substantial impact on the reduction of contaminants, suggesting that WWR adjustments may be a more effective mitigation strategy than altering the placement of openings.

Further studies have examined the role of CV in classrooms, particularly in reducing the risk of airborne virus transmission. Ren et al. [297] conducted numerical simulations to evaluate the effects of different window opening configurations on the distribution of airborne contaminants in a classroom setting. Their results indicate that when an infected occupant was positioned near the CV inlets, a notable disparity in contaminant concentrations was observed between different sections of the classroom, which highlights the influence of the occupant position on CV performance. Park [172] conducted a numerical assessment that compared the infection risk posed by an infected student in a 168 m<sup>3</sup> classroom under SSV and CV conditions. Park explored multiple window opening ratios (15%, 30%, and 100%) and found that after a three-hour exposure period the infection risk under CV was approximately 2.5 and 3.3 times lower than under SSV for window opening ratios of 15% and 30%, respectively.

#### 6.1.4 Chapter Objectives

The primary objective of this chapter is to evaluate the performance of NV in maintaining optimal airflow patterns, TC, and control of airborne infections in an office environment. This study aims to evaluate the effectiveness of NV strategies using CFD simulations. Specifically, this chapter seeks to:

- Analyse the airflow distribution in different NV scenarios and its impact on air circulation.
- Evaluate the variations in temperature and RH in NV strategies.
- Assess occupant TC using the PMV and PPD indices.
- Examine aerosol dispersion patterns during coughing and sneezing events to understand the risk of airborne virus transmission.
- Compare NV performance with the standards of ASHRAE 55, EN 16798, and ISO 7730 for TC to ensure that the system meets the recommended ventilation effectiveness criteria.
- Identify the most effective NV strategy that improves IAQ and reduces thermal discomfort.

### 6.2 Methodology

This study used a CFD simulation approach to assess the performance of NV in an open-plan office environment. This methodology follows a structured framework, beginning with the identification of key parameters and solver settings, which is followed by the development of a computational model that incorporates geometry, meshing, and boundary conditions. The CFD simulation evaluates IAQ, TC, and aerosol dispersion, providing insights into the effectiveness of NV strategies under different seasonal conditions. The results are analysed for airflow patterns, temperature regulation, RH, TC indices of the occupants (PMV and PPD), and risks of transmission of airborne viruses. Finally, recommendations are derived based on the findings to optimise NV performance.

The general research framework is illustrated in Figure 6.1, which summarises the step-by-step process undertaken in the simulations.



Figure 6.1: Overview of the methodology for NV simulations

## 6.3 Scenario Development

This chapter evaluates five different NV scenarios to evaluate system performance under different seasonal conditions and ventilation strategies, as summarised in

Scenario No.	Season	Ventilation Configuration	Environmental Conditions	
1	Summer	Single Window Open (Window 4)	T = 19°C, V = 1.9 m/s, RH = 0.0083	
2	Summer	Single Window Open (Window 4)	T = 19°C, V = 1.9 m/s, RH = 0.009	
3	Summer	Single Window Open (Window 4)	T = 19°C, V = 3.0 m/s, RH = 0.0083	
4	Early Summer	Three Windows Open (Window 1, Window 6, Window 9)	$T = 17.5^{\circ}C, V = 3.0 m/s, RH = 0.0115$	

Table 6.1. Each scenario evaluates airflow distribution, temperature control, humidity levels, and human TC metrics.

Table 6.1: NV scenarios and their environmental conditions.

## 6.4 Boundary Conditions

NV scenarios were simulated for summer conditions because opening windows during winter could cause thermal discomfort to the occupants. The boundary conditions were carefully designed to capture realistic airflow dynamics, thermal behaviour, and respiratory particle dispersion in the teaching office environment in various NV settings. These conditions account for wind-driven airflow, buoyancy effects, and CV efficiency, ensuring a comprehensive assessment of indoor air quality and TC in naturally ventilated spaces.

The air flow entering the office was modelled as a velocity-inlet boundary condition, simulating fresh outdoor air entering through open windows. Inlet velocity varied in different scenarios, reflecting changing wind conditions and ventilation strategies. The air temperature at the input was set to 19 °C and 17.5 °C, depending on the external climate variations. This ensures the simulation accurately captured temperature-driven buoyancy forces and their influence on indoor airflow patterns. The air outlet was defined as a pressure outlet with a static pressure condition of zero, allowing for unrestricted airflow discharge through open doors or designated ventilation openings. This configuration facilitated continuous air exchange, which ensures the removal of stale indoor air while preventing the buildup of unphysical pressure within the domain. The combination of natural inlets and outlets replicated a passive airflow system, accurately modelling real-world wind-driven ventilation conditions in an office setting.

Each NV scenario was designed to evaluate the air velocity distribution, the efficiency of CV, and the impacts on TC in the three office zones. In Scenario 1, outdoor air entered through Window 4 at a velocity of 1.9 m/s with a humidity level of 0.0083 and the inlet air temperature was set at  $19 \,^{\circ}$ C to represent realistic summer conditions. Scenario 2 maintained the same setup as Scenario 1 but the humidity level was adjusted to 0.009, which captures the impact of slight humidity variations on NV efficiency. In Scenario 3, the inlet velocity was increased to 3.0 m/s, with the humidity level set to 0.0083, which allows an analysis of how higher wind speeds influence indoor air circulation. Scenario 4 introduced multiple openings by activating Window 1, Window 9, and Window 17, with an inlet velocity of 3.0 m/s and a humidity level of 0.0115. This setup was designed to evaluate the effectiveness of CV, demonstrating how multiple openings improve airflow distribution and improve contaminant dilution efficiency compared to single-window configurations.

A summary of key boundary conditions is presented in Table 6.3, which details the temperature, velocity, and pressure conditions assigned to different surfaces, airflow sources, and respiratory events. These boundary conditions were essential to capture the complex thermal and airflow interactions that define naturally ventilated indoor spaces, providing valuable insights into ventilation effectiveness, TC, and airborne contaminant control.

To evaluate the effectiveness of NV in controlling the risks of airborne infection,

the 12th simulation case was developed, as described in Table 6.2. Cases 1, 3, 5, and 7 assess airflow, room temperature, and RH and distribution. Cases 2, 4, 6, and 8 evaluate the implications for TC of NV by analysing the thermal perception of the occupant. Cases 9 to 12 introduce an infected individual seated in different positions (A, B, C, and D) to assess aerosol dispersion patterns and infection risks across various spatial configurations. These seating locations were selected to represent typical workstation layouts in an open-plan office, which allows an analysis of how airflow direction, diffuser positioning, and ventilation efficiency influence particle transport and potential exposure risks.

Case No.	Scenario	Infected Source	Notes
1	Scenario 1	None	Assessment of ventilation performance
2	Scenario 1	None	Assessment of thermal comfort
3	Scenario 2	None	Assessment of ventilation performance
4	Scenario 2	None	Assessment of thermal comfort
5	Scenario 3	None	Assessment of ventilation performance
6	Scenario 3	None	Assessment of thermal comfort
7	Scenario 4	None	Assessment of ventilation performance
8	Scenario 4	None	Assessment of thermal comfort
9	Scenario 4	А	Assessment of infection risk
10	Scenario 4	В	Assessment of infection risk
11	Scenario 4	С	Assessment of infection risk
12	Scenario 4	D	Assessment of infection risk

Table 6.2: Overview of simulation cases during the NV.

Boundary Condition	Туре	Specification	
Air Inlet	Velocity-Inlet	$T = 17.5 ^{\circ}C$ and $19 ^{\circ}C$ , wind-driven airflow	
Air Outlet Pressure Outlet		Zero static pressure	
Walls	No-Slip	$T = 15 \circ C$ (representing heat exchange)	
Occupants (Body)	Thermal Source	$T = 24 ^{\circ}C$	
Occupants (Head)	Thermal Source	$T = 34 ^{\circ}C$	
Occupants (Mouth)	Thermal Source	${ m T}=36^{ m o}{ m C}$	
Coughing	Particle Source	$\mathrm{V}=11.8~\mathrm{m/s},\mathrm{T}=36^{\mathrm{o}}\mathrm{C}$	
Sneezing	Particle Source	$\mathrm{V}=70~\mathrm{m/s},\mathrm{T}=36~\mathrm{^{o}C}$	

Table 6.3: Summary of boundary conditions for NV simulations.

## 6.5 Results

This section presents the findings of four distinct NV scenarios, each designed to assess NV in regulating IAQ, TC, and airflow distribution. This study specifically examines the influence of outdoor air infiltration through open windows, focussing on key parameters such as air velocity patterns, temperature variations, RH and TC indices, including PMV and PPD.

#### 6.5.1 Scenario 1

The air velocity distribution across the three office zones under NV in Scenario 1 provides critical insights into airflow uniformity and the overall efficiency of NV

characterised by opening Window 4, with outdoor air entering at a velocity of 1.9 m/s and a humidity level of 0.0083. The inlet air temperature was set to 19°C in Ansys to represent realistic summer conditions.

#### 6.5.1.1 Air Velocity Pattern

Figures 6.2 show the overall distribution of air velocity in all three zones. Most of the office area displays air velocities between 0.06 and 0.08 m/s, with higher velocities concentrated near the open window and the last row of occupants in Zone 1. The remaining areas exhibit low airflow, indicating stagnant zones that require additional ventilation measures to improve circulation.

In general, this scenario reveals that most of the office has low air velocity, leading to several dead zones with limited airflow. These findings highlight the need for additional ventilation measures or mixed-mode ventilation to ensure effective air distribution throughout the office.



Figure 6.2: Air velocity contour in z direction during Scenario 1 of NV

#### 6.5.1.2**Room Temperature and Relative Humidity**

The temperature distribution between the three zones within the office space under NV conditions is shown in Figures 6.3. These figures highlight the spatial temperature variations influenced by air inflow from the open window and heat sources such as occupants and equipment.

The large office (Zone 1) benefits from direct inflow of outdoor air, maintaining temperatures between 20 °C and 21 °C in much of the area. However, temperatures near the occupants can reach 23 °C, particularly in regions away from direct airflow. In medium and small offices (zones 2 and 3), temperatures are higher, ranging between 24 °C and 25 °C, suggesting that these areas experience limited airflow, leading to warmer and potentially uncomfortable conditions.



(b) z = 2 m

Figure 6.3: Temperature contour in z direction during Scenario 1 of NV

The distribution of RH in the three office zones in Scenario 1 provides insights into moisture regulation and its interaction with air circulation, the presence of the occupant and electronic equipment.

Figure 6.4 presents the combined RH distribution in the three zones. Most areas show RH levels of around 65%, with certain localised regions reaching up to 60%.

These higher humidity concentrations are observed near individuals, electronic equipment, and corners, where airflow may be less effective in dispersing moisture. The elevated RH is consistent with the continuous influx of high-humidity outdoor air, especially in areas where NV dominates.

In general, the findings indicate that while NV can achieve a reasonable air distribution, the high humidity of the incoming outdoor air imposes a challenge in maintaining optimal indoor RH levels.



Figure 6.4: RH contour in z direction during Scenario 1 of NV

#### 6.5.1.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The distribution of PMV values in the three office zones in Scenario 1 reflects the influence of NV and localised heat sources on TC, as shown in Figures 6.5.

Zone 1 exhibits PMV values ranging from 0 to 0.8, which indicates relatively comfortable conditions with minimal heat discomfort. In contrast, Zones 2 and 3 experience PMV values up to 1.2, indicating that occupants in these zones are likely to feel slightly warm. The observed thermal disparity can be attributed to closed windows and less effective NV in these zones compared to Zone 1.

The spatial variation in PMV values suggests that Zone 1 benefits from better airflow due to the open window, maintaining optimal TC, while Zones 2 and 3 may require additional ventilation improvements to mitigate slightly warm conditions and improve occupant comfort. This result highlights the importance of window placement and airflow optimisation in naturally ventilated office settings.



(a) z = 1 m

(b) z = 2 m

Figure 6.5: PMV contour in z direction during Scenario 1 of NV

The distribution of PPD corresponds directly to the previously analysed PMV values, highlighting areas where occupants may express discomfort due to variations in thermal conditions and airflow patterns.

Figure 6.6 provides an overall view of the distribution of PPD in the three offices. Zones 2 and 3 exhibit higher PPD values, with levels of dissatisfaction reaching 40% in localised areas, particularly near heat sources or poorly ventilated regions. In contrast, Zone 1 shows better TC, and most areas maintain PPD values below 20%, probably due to the effective NV provided by the open window.

These results suggest that while NV is beneficial in mitigating discomfort in Zone 1, it does not provide uniform comfort throughout the office space. Zones 2 and 3 experience greater dissatisfaction, primarily due to limited direct airflow and localised heat build-up. To address these disparities, the implementation of mixed mode ventilation or the introduction of passive airflow mechanisms, such as additional openings or fans, could help to improve overall comfort and reduce PPD levels to acceptable thresholds. This approach would create a more balanced indoor environment, thus improving occupant satisfaction and productivity.



Figure 6.6: PPD contour in z direction during Scenario 1 of NV

### 6.5.2 Scenario 2

The air velocity distribution across the three office zones under NV in Scenario 2 provides critical insights into airflow uniformity and the overall efficiency of NV characterised by opening Window 4, with outdoor air entering at a velocity of 1.9 m/s and a humidity level of 0.009. The inlet air temperature was set to 19 °C in Ansys to represent realistic summer conditions.

#### 6.5.2.1 Air Velocity Pattern

Figures 6.7 illustrate airflow characteristics in different zones, highlighting areas of effective ventilation, potential stagnation, and deviations from the optimal air distribution. The simulated air velocity within the office space ranged between 0 and 0.4 m/s, with localised airflow patterns that highlight both well-ventilated and stagnant zones.

he results indicate that most of the office exhibits velocities between 0.06 and 0.08 m/s, with higher values located near the open window and the last row of occupants in Zone 1. However, several regions, particularly in Zones 2 and 3, experience low velocities, contributing to stagnant zones with inadequate air exchange.

This scenario reveals that NV through a single open window, although effective in some localised areas, is insufficient to achieve uniform airflow throughout the office. The presence of several dead zones in Zones 2 and 3 highlights the need for additional ventilation solutions, such as mixed-mode ventilation or the strategic placement of additional windows and passive airflow pathways. By addressing these deficiencies, it would be possible to improve indoor air circulation, thereby reducing pollutant accumulation and enhancing overall TC and occupant wellbeing.



Figure 6.7: Air velocity contour in z direction during Scenario 2 of NV

#### 6.5.2.2 Room Temperature and Relative Humidity

The temperature distribution between the three zones within the office space under Scenario 2 of NV conditions is shown in Figures 6.8. These figures highlight the spatial temperature variations influenced by air inflow from the open window and heat sources such as occupants and equipment.

The large office (Zone 1), with direct airflow from the open window, maintains cooler temperatures between 20 °C and 21 °C in most regions. However, hotspots located near the occupants and heat-emitting devices exhibit temperatures around 23 °C to 24 °C. In contrast, Zones 2 and 3 experience higher temperatures, ranging between 24 °C and 25 °C, due to limited airflow penetration and poor vertical mixing.

The general temperature distribution reveals areas of potential discomfort, particularly in Zones 2 and 3, where limited airflow results in warmer and stagnant conditions. Although localised cooling is achieved in Zone 1 near the window, insufficient air penetration leads to warmer zones further into the space. These findings highlight the need for additional airflow mechanisms, such as AC or mixed-mode systems, to achieve a uniform temperature distribution and improve TC in all zones.



Figure 6.8: Temperature contour in z direction during Scenario 2 of NV

The distribution of RH in the three office zones in Scenario 2 provides insights into moisture regulation and its interaction with air circulation, the presence of the occupant and electronic equipment. Figures 6.9 illustrate the spatial variation of RH between different zones, highlighting areas with potential imbalances.

Most areas show RH levels of around 65%, with certain localised regions reaching up to 70%. These higher humidity concentrations are observed near individuals, electronic equipment, and corners, where airflow may be less effective in dispersing moisture. The elevated RH is consistent with the continuous influx of highhumidity outdoor air, especially in areas where NV dominates.

In general, the findings indicate that while NV can achieve a reasonable air distribution, the high humidity of the incoming outdoor air imposes a challenge in maintaining optimal indoor RH levels.



Figure 6.9: RH contour in z direction during Scenario 2 of NV

#### 6.5.2.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The distribution of PMV values in the three office zones in Scenario 2 reflects the influence of NV and localised heat sources on TC, as shown in Figures 6.10.

The results highlight that Zone 1 maintains comfortable thermal conditions, with PMV values ranging between 0 and 0.8. In contrast, Zones 2 and 3 exhibit higher PMV values, reaching 1.4, particularly in areas far from the open window and near heat sources.

The PMV variations observed across the zones underscore the importance of effective airflow distribution in naturally ventilated spaces. Zone 1 benefits from the open window, maintaining acceptable TC with minimal warm sensations. However, Zones 2 and 3 experience higher PMV values, which suggests that the occupants in these areas are likely to feel slightly warm. To address this problem, the implementation of additional airflow mechanisms or mixed-mode ventilation could help to balance the thermal environment and ensure uniform occupant comfort in the office space.



Figure 6.10: PMV contour in z direction during Scenario 2 of NV

The distribution of PPD corresponds directly to the previously analysed PMV values, highlighting areas where occupants may express discomfort due to variations in thermal conditions and airflow patterns. Figures 6.11 illustrate the distribution of PPD in the different offices in Scenario 2 of NV conditions.

Zones 2 and 3 show PPD values that reach 40% in specific areas, particularly near

occupants and heat-generating sources. In contrast, most of the areas in Zone 1 maintain PPD levels below 20%, demonstrating more effective NV in this zone due to the proximity of the open window.

The results highlight that while NV is effective in Zone 1, it does not provide uniform TC throughout the office. Higher PPD values in Zones 2 and 3 suggest localised airflow issues and heat retention, which contribute to occupant discomfort. To address these issues, mixed-mode ventilation or additional passive airflow mechanisms should be considered, such as strategically placed openings or fans. By mitigating localised heat buildup and improving air distribution, overall TC can be enhanced, reducing PPD values to more acceptable levels and ensuring a more comfortable working environment throughout the office space.



Figure 6.11: PPD contour In z direction during Scenario 2 of NV

#### 6.5.3 Scenario 3

The air velocity distribution across the three office zones under NV in Scenario 3 provides critical information on the uniformity of airflow and the overall efficiency of NV characterised by the opening of Window 4, with the outdoor air entering at a velocity of 3 m/s and a humidity level of 0.0083. The inlet air temperature was set to 19 °C in Ansys to represent realistic summer conditions.

#### 6.5.3.1 Air Velocity Pattern

Figures 6.12 illustrate airflow characteristics in different zones, highlighting areas of effective ventilation, potential stagnation, and deviations from the optimal air distribution. The simulated air velocity within the office space ranged between 0 and 0.4 m/s, with localised airflow patterns that highlight both well-ventilated and stagnant zones.

The highest velocities, exceeding 0.3 m/s, are concentrated near the window and the back row of the occupants in Zone 1. The airflow velocity decreases progressively in Zones 2 and 3, with large portions of these areas displaying stagnant airflow where values range between 0.02 and 0.12 m/s.

In summary, the velocity analysis for Scenario 3 reveals that while the increase in outdoor air velocity improves airflow in areas directly exposed to it, large portions of Zones 2 and 3 suffer from inadequate air circulation. However, this scenario performs better air velocity, even in Zones 2 and 3, compared to the previous scenarios. These findings highlight the need for additional ventilation measures, such as fans or improved NV layouts in Zones 2 and 3, to ensure effective airflow and pollutant dispersion throughout the space.



(a) z = 1 m

(b) z = 2 m

Figure 6.12: Air velocity contour in z direction during Scenario 3 of NV
#### 6.5.3.2 Room Temperature and Relative Humidity

The temperature distribution between the three zones within the office space under Scenario 3 of NV conditions is shown in Figures 6.13. These figures highlight the spatial temperature variations influenced by air inflow from the open window and heat sources such as occupants and equipment.

Zone 1 consistently maintains temperatures between 20 °C and 21 °C, except for minor localised hotspots near occupants and heat sources. In Zones 2 and 3, the temperatures reach 22 °C in some areas but remain within acceptable comfort thresholds. Compared to Scenarios 1, 2, and 3, this scenario demonstrates better thermal performance, with reduced overheating and more balanced temperature levels.

The general results reveal that NV in this scenario effectively stabilises the room temperature in all zones, maintaining values within the comfort range. Zones 2 and 3 benefit from cooler conditions and improved TC compared to previous scenarios, highlighting the importance of optimised NV strategies to achieve uniform indoor climate control.



Figure 6.13: Temperature contour in z direction during Scenario 3 of NV

The distribution of RH in the three office zones in Scenario 3 provides insights into moisture regulation and its interaction with air circulation, including the presence of the occupant and electronic equipment. Figures 6.14 illustrate the spatial variation of RH between different zones, highlighting areas with potential imbalances.

Most of the areas in the three zones exhibit RH values ranging from 60% to 70%, except for localised regions near heat sources, where RH levels may momentarily exceed this range. Consistent humidity levels are attributed to the continuous inflow of moist outdoor air, which stabilises the moisture content within the spaces.

In general, the distribution of RH across the office spaces in Scenario 3 demonstrates the effectiveness of NV in maintaining indoor humidity conditions within acceptable comfort levels. The slight elevations in localised regions suggest the potential benefit of mixed-mode ventilation or dehumidification systems to further optimise indoor moisture control and prevent long-term moisture accumulation.



Figure 6.14: RH contour in z direction during Scenario 3 of NV

#### 6.5.3.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The distribution of PMV values in the three office zones in Scenario 3 reflects the influence of NV and localised heat sources on TC, as shown in Figures 6.15.

Zone 1 demonstrates the most stable and comfortable conditions, with PMV values ranging from -1.6 to 0.6, signifying predominantly neutral thermal sensations. Zones 2 and 3 exhibit PMV values of up to 1.4 in localised areas, which reflects slightly warmer conditions. Despite these warm spots, both zones show notable improvements in TC compared to Scenarios 1, 2, and 3.

In general, the PMV analysis underscores the effectiveness of NV in Scenario 4, with most areas maintaining PMV values within the recommended comfort range of -1.0 to 1.0. The combination of increased airflow and reduced heat accumulation enhances the thermal environment, creating a more balanced and comfortable indoor environment. Although minor localised discomfort remains in Zones 2 and 3, overall conditions in all zones show significant improvements, which highlights the success of NV in optimising indoor TC.



Figure 6.15: PMV contour in z direction during Scenario 3 of NV

The distribution of PPD corresponds directly to the previously analysed PMV

values, highlighting areas where occupants may express discomfort due to variations in thermal conditions and airflow patterns. Figures 6.16 illustrate the distribution of PPD in the different offices in Scenario 3 of NV conditions.

All three zones generally maintain low PPD values, primarily below 20%, reflecting a high level of TC throughout much of the space. However, notable exceptions include a central area within Zone 1 and near the entrance door, where PPD values peak at 90%. These elevated levels of dissatisfaction are likely due to inadequate airflow in these areas, coupled with localised heat accumulation.

In general, the PPD analysis shows significant improvements in office TC compared to previous scenarios, with NV effectively reducing dissatisfaction in most areas. However, the persistence of high PPD values in certain regions suggests the need for targeted airflow interventions, such as additional ventilation openings or localised fans, to further optimise comfort and minimise dissatisfaction.



Figure 6.16: PPD contour in z direction during Scenario 3 of NV

### 6.5.4 Scenario 4

The air velocity distribution across the three office zones under NV in Scenario 4 provides critical information on the uniformity of airflow and the overall efficiency

of NV characterised by opening Window 1, Window 6, and Window 9, with outdoor air entering at a speed of 3 m/s and a humidity level of 0.0115. The inlet air temperature was set to 17.5 °C in Ansys to represent realistic summer conditions. The selection of multiple windows positioned at different locations enables better CV and airflow dynamics compared to previous scenarios.

#### 6.5.4.1 Air Velocity Pattern

The air velocity distribution across the three office zones in Scenario 4 of NV highlights the impact of strategically placed open windows on airflow patterns and general ventilation, as shown in Figures 6.17.

Airflow remains generally stable, with values averaging around 0.2 m/s to 0.4 m/s in most areas. The highest velocities are observed near the windows and occupant regions, while areas near the occupants sitting in the central section and in Zone 3 experience reduced airflow, with velocities dropping as low as 0.04 m/s. This reduction is attributed to the limited direct airflow pathways that reach these areas.

The air velocity distribution analysis for Scenario 4 highlights improved ventilation performance due to the optimised placement of multiple open windows. The increased airflow ensures minimal stagnation and promotes a more uniform air distribution across most zones, effectively outperforming the previous scenarios. This scenario demonstrates the best overall air velocity performance, contributing to improved ventilation in all zones.

#### 6.5 Results



Figure 6.17: Air velocity contour in z direction during Scenario 4 of NV

#### 6.5.4.2 Room Temperature and Relative Humidity

The distribution of room temperature across the three office zones in Scenario 4 demonstrates the effectiveness of enhanced NV and cross-flow airflow to maintain TC, as depicted in Figures 6.18.

Most of the area in all zones exhibits the most consistent and favourable conditions, with temperatures ranging between 19.5 °C and 21 °C in most areas. Zone 3 shows slightly higher temperatures in localised regions, peaking at 22 °C, but remains within the comfortable temperature range.

The temperature distribution analysis for Scenario 4 highlights the effectiveness of strategically placed open windows, which provide ample airflow to minimise heat accumulation and maintain overall TC. This scenario demonstrates improved temperature regulation and a more comfortable indoor environment compared to previous scenarios.



Figure 6.18: Temperature contour in z direction during Scenario 4 of NV

The distribution of RH in the three office zones in Scenario 4 highlights the impact of NV on influencing moisture levels and the presence of localised variations, as illustrated in Figures 6.19.

All three zones maintain elevated RH levels predominantly between 76% and 90%, reflecting inadequate moisture regulation. Zone 3 exhibits slightly lower RH levels compared to Zones 1 and 2 due to reduced airflow limiting excessive moisture transport.

The relative humidity distribution analysis for Scenario 4 indicates that while NV improves airflow, it inadvertently contributes to increased moisture levels in the office space. The strategic placement of multiple open windows has caused excessive humidity build-up, creating unfavourable indoor conditions compared to previous scenarios. These results emphasise the need for improved moisture control mechanisms to achieve a balanced indoor environment conducive to occupant comfort.



Figure 6.19: RH contour in z direction during Scenario 4 of NV

#### 6.5.4.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The distribution of PMV values in the three office zones in Scenario 4 reflects the influence of NV and localised heat sources on TC, as shown in Figures 6.20.

Zones 1 and 2 exhibit predominantly cold conditions, with PMV values ranging between -2.5 and 0.2, indicating slight to moderate cool sensations. The first and last columns are particularly cold, while those in the central area feel slightly cooler. In Zone 2, areas with limited airflow result in slightly higher PMV values, leading to warmer sensations compared to the other zones. Zone 3 demonstrates an improved thermal sensation profile, maintaining a more balanced environment compared to Zones 1 and 2.

The PMV analysis for Scenario 4 reveals that the combination of NV and crossflow airflow has a negative impact on overall TC. Excess cooling, coupled with uneven airflow distribution, results in an indoor environment with limited areas of comfort and significant zones of discomfort. Compared to previous scenarios, this setup exhibits a slight deterioration in thermal regulation, highlighting the need for further optimisation to achieve a comfortable environment for the occupants.

#### 6.5 Results



Figure 6.20: PMV contour in z direction during Scenario 4 of NV

The distribution of PPD corresponds directly to the previously analysed PMV values, highlighting areas where occupants may express discomfort due to variations in thermal conditions and airflow patterns. Figures 6.21 illustrate the distribution of PPD in the different offices in Scenario 4 of NV conditions.

Zone 1 shows moderate to high dissatisfaction, with values predominantly between 10% and more than 90%. Zone 2 shows a more pronounced dissatisfaction in the upper regions, reaching over 90% due to airflow-induced cold spots. In contrast, Zone 3 shows considerably lower dissatisfaction, with most areas maintaining PPD values below 20%, reflecting the relative thermal stability of the zone and limited exposure to intense airflow.

The PPD analysis for Scenario 4 reveals that NV and cross-flow airflow produce mixed results in TC. Although airflow effectively prevents heat accumulation, it simultaneously introduces excessive cooling in several areas, leading to widespread dissatisfaction. Compared to previous scenarios, this setup results in greater overall dissatisfaction, underscoring the need for better airflow management and a more controlled distribution to enhance occupant comfort.

#### 6.5 Results



Figure 6.21: PPD contour in z direction during Scenario 4 of NV

#### 6.5.4.4 Aerosol Dispersion and Infection Risk

Infection risk analysis was exclusively conducted for Scenario 4 because it represented the most effective cross-ventilation (CV) configuration in terms of airflow distribution and air exchange rates, as established in section 6.5.4.1. Among all the evaluated scenarios, Scenario 4 demonstrated the highest air velocities and the greatest potential for particle transport across different zones, making it the most critical case for assessing the implications of natural ventilation on airborne virus transmission. Assessing infection risk under less ventilated scenarios (e.g., Scenarios 1 to 3) would not offer additional insights, as stagnant airflow zones in those cases already indicate poor dispersion and higher pollutant retention. Therefore, Scenario 4 was selected as a representative high-performance NV case to examine how increased airflow influences aerosol transport and cross-contamination potential in a shared office space.

**Coughing** is one of the primary ways through which infectious aerosols are expelled from the human respiratory system, posing a significant risk of airborne virus transmission in enclosed spaces. In the current analysis, the particle dispersion dynamics due to coughing were studied in Scenario 4, where NV and cross-flow airflow play a central role. A coughing velocity of 11.8 m/s was applied and

the analysis focused on the spatial and temporal movement of the particles of four different occupants. The following results highlight the extent of particle spread in the X, Y, and Z directions, evaluating how airflow impacts their dispersion within the indoor environment. Figure 6.22 demonstrates the spatial distribution of aerosols expelled by the first individual during coughing. The expelled particles initially exhibit a concentrated forward movement before undergoing significant dispersion influenced by indoor airflow. A dense accumulation is observed near the source; however, the particles spread widely, especially toward the occupants sitting in the path of the airflow.



Figure 6.22: Particle transmission from the first infected individual during coughing.

Figure 6.23 illustrates the dispersion of aerosols by coughing in the X, Y, and Z directions for the first individual during NV.

In X direction dispersion, as shown in Figure 6.23-(a), particles exhibit an initial rapid displacement, primarily due to the high forward velocity of the sneeze. The particles remain predominantly within the range of 0 to 3.5 m (indicating a transmission distance of approximately 3.5 m), demonstrating limited lateral

spread under existing airflow conditions.

Figure 6.23-(b) highlights the diffusion of lateral particles in the Y direction with displacement values ranging between 7 and 13 m (indicating a transmission distance of approximately 6 m). This indicates significant horizontal spread as a result of interactions with airflow, allowing particles to reach distant areas within the room.

Figure 6.23-(c) shows the vertical particle distribution as shown in the Z direction between 0 m and 3 m (indicating a transmission distance of approximately 3 m), corresponding to the breathing zone of most of the inhabitants. Minimal upward or downward dispersion is observed beyond this range, highlighting the high transmission risk within this critical zone.



**Figure 6.23:** Aerosol dispersion through coughing in X, Y,and Z directions for the first individual during NV.

Figure 6.24 demonstrates the spatial distribution of aerosols emitted by the second individual during coughing. The expelled particles exhibit strong initial forward momentum before dispersing and spreading under the influence of indoor airflow. A dense concentration of particles is observed around the source, with a broader dispersion in the area in front of the infected individual due to NV.



**Figure 6.24:** Particle transmission from the second infected individual during coughing.

Figure 6.25 illustrates the dispersion of aerosols by coughing in the X, Y, and Z directions for the second individual under NV.

Figure 6.25-(a) shows the displacement of the particles in the X direction over time. The particles initially move forward rapidly due to the high velocity of the cough, they then decelerate as they interact with turbulent airflow. The particles remain predominantly within the range of 0 to 3.5 m (indicating a transmission distance of approximately 3.5 m), indicating limited lateral diffusion along the X axis.

The dispersion in the Y direction, as presented in Figure 6.25-(b), highlights the lateral spread of the particles throughout the room. The particles disperse between 8 and 12 m (indicating a transmission distance of approximately 4 m), suggesting that significant lateral movement occurs due to turbulent interactions and natural airflow. This diffusion increases the risk of exposure to multiple occupants within the surrounding area.

The dispersion in the Z direction, illustrated in Figure 6.25-(c), shows how verti-

cal particle movement is affected by airflow patterns and gravity. The majority of particles remain between 0 and 3 m (indicating a transmission distance of approximately 3 m), within the breathing zone of seated and standing individuals. Minimal vertical displacement beyond this range is observed, indicating that the particles largely remain suspended within the critical transmission height levels, posing a risk of respiratory exposure.



**Figure 6.25:** Aerosol dispersion through coughing in X, Y, and Z directions for the second individual during NV.

Figure 6.26 demonstrates the spatial distribution of aerosols emitted by the third individual. The particles show a strong forward projection, rapidly dispersing and diffusing under the influence of airflow from the window. A dense cluster of particles is initially observed near the source before scattering across the occupied space, increasing the likelihood of exposure of nearby individuals.

#### 6.5 Results



Figure 6.26: Particle transmission from the third infected individual during coughing.

Figure 6.27 illustrates the aerosol dispersion along the X, Y, and Z directions during the coughing event for the third individual.

Figure 6.27-(a) displays the movement of the particles in the X direction over time. The particles show a rapid initial forward motion due to the high velocity of the cough. As they propagate, most of the particles remain within the 10 to 13.5 m range (indicating a transmission distance of approximately 3.5 m), reflecting a limited forward spread due to airflow redirection from nearby walls. The restricted forward range mitigates exposure to distant occupants but contributes to localised density in the immediate area.

In the Y direction (Figure 6.27-(b)), particle displacement ranges between 4 and 7 m (indicating a transmission distance of approximately 3 m). Lateral diffusion is more prominent because of turbulent flow interactions, leading to moderate horizontal dispersion. This pattern indicates a significant risk of exposure for occupants seated in proximity to the infected individual.

The vertical displacement of the particles (Figure 6.27-(c)) is influenced by gravitational settling and airflow buoyancy. The particles remain predominantly within a height of 0.5 m to 3 m (indicating a transmission distance of approximately 2.5 m), corresponding to the breathing level of seated and standing individuals. The minimal movement of particles beyond this height indicates that the particles are largely concentrated in regions that are critical to respiratory exposure.



**Figure 6.27:** Aerosol dispersion through coughing in X, Y, and Z directions for the third individual during NV.

Figure 6.28 demonstrates the spatial distribution of aerosols emitted by the fourth individual. The expelled particles exhibit rapid initial movement, followed by dispersion influenced by airflow patterns and the surrounding ventilation. A dense concentration of particles is observed near the source, particularly in the immediate area, before they diffuse and spread to nearby areas, affecting the occupants in proximity.

#### 6.5 Results



Figure 6.28: Particle transmission from the fourth infected individual during coughing.

Figure 6.29 illustrates the aerosol dispersion in the X, Y, and Z directions for the fourth individual during NV.

Figure 6.29-(a) shows the displacement of the particles in the X direction over time. The particles initially exhibit forward motion but encounter significant deceleration and dispersion as they interact with air flow and structural elements. The displacement values range predominantly between 0 and 3 m (indicating a transmission distance of approximately 3 m), indicating limited horizontal movement within the immediate surrounding area.

The dispersion in the Y direction, as shown in Figure 6.29-(b), indicates the lateral spread of the particles with values ranging between 9 m and 12.5 m (indicating a transmission distance of approximately 3.5 m). This range suggests significant lateral movement due to turbulent indoor airflow, which can potentially lead to higher exposure risks among occupants located in the vicinity.

In the vertical (Z) direction, Figure 6.29-(c) highlights how particle movement is affected by gravity and buoyant forces. The majority of particles are concentrated between 0 and 1.5 m in height (indicating a transmission distance of approximately 1.5 m), corresponding to the breathing zones of seated and standing individuals. Minimal upward or downward movement beyond this range is observed, suggesting that the majority of particles remain within a critical zone for transmission risks.



Figure 6.29: Aerosol dispersion through coughing in X, Y, and Z directions for the fourth individual during NV.

**Sneezing** is another major mechanism responsible for the evaporation of highspeed infectious aerosols, posing a higher risk of rapid airborne disease transmission. Compared to coughing, sneezing can eject particles at much higher velocities, leading to a broader and faster spread. In this analysis, a sneezing velocity of 70 m/s was applied in Scenario 4, where NV and cross-flow airflow significantly affect the movement of expelled particles. This study focused on four different infected individuals to assess the spatial and temporal dispersion of particles along the X, Y, and Z directions within the indoor environment. Figure 6.30 demonstrates the spatial distribution of aerosols emitted by the first individual. The expelled particles exhibit strong initial forward movement before dispersing and spreading under the influence of indoor airflow. A dense concentration of particles is observed in the immediate vicinity of the source but as the particles travel further they become more widely distributed, specifically to the occupants sitting in front of the infected individual.



Figure 6.30: Particle transmission from the first infected individual during sneezing.

Figure 6.31 illustrates the aerosol dispersion in the X, Y, and Z directions for the first individual during NV.

Figure 6.31-(a) shows the displacement of the particles in the X direction over time. The particles initially move rapidly due to the forward momentum of the sneeze but begin to decelerate as they encounter resistance and airflow redirection. Most of the particles remain within a range of 16.5 to 18 m (indicating a transmission distance of approximately 1.5 m), demonstrating limited lateral spread under the existing airflow conditions.

The dispersion in the Y direction, as seen in Figure 6.31-(b), highlights how the particles laterally distribute. The displacement values range between 5 and 13 m (indicating a transmission distance of approximately 8 m), indicating that although the initial motion occurs primarily forward, the particles undergo notable lateral diffusion due to interactions with turbulent airflow. This lateral dispersion increases the potential for exposure in multiple zones of occupancy. In the vertical (Z) direction, Figure 6.31-(c) reveals how particle movement is affected by gravitational settling and buoyant airflow. Most particles remain concentrated between

0 and 3.5 m in height (indicating a transmission distance of approximately 3.5 m), corresponding to the breathing zone of seated and standing occupants. Minimal upward or downward dispersion is observed beyond this range, suggesting that the particles are largely contained within a height range that is critical to transmission risks.



Figure 6.31: Aerosol dispersion in X, Y, and Z directions for the first individual during NV.

Figure 6.32 shows the concentrated cloud of particles that moves out of the second individual's position and spreads through the central and nearby areas of the office. The particle concentration shows a broad spread in the X and Y directions, but the vertical movement (Z axis) remains moderate, indicating limited vertical dispersion.



Figure 6.32: Particle transmission from the second infected individual during sneezing.

Figure 6.33 illustrates the aerosol dispersion in the X, Y, and Z directions for the second individual during NV.

Figure 6.33-(a) shows the particle displacement the X-position tracking over time reveals steady particle advancement within a confined range from 10 to 14 metres (indicating a transmission distance of approximately 4 m). This relatively moderate lateral spread suggests that the airflow did not generate excessive turbulence in this direction, leading to limited particle displacement.

The dispersion in the Y direction, as seen in Figure 6.33-(b), highlights how the particles exhibit a more dynamic spread, with values ranging between 8 and 11 m (indicating a transmission distance of approximately 3 m). The variation highlights how natural airflow patterns within the office space interact with the particles, causing minor deflections, but maintaining overall trajectory consistency.

In the vertical (Z) direction, Figure 6.33-(c) reveals how particle movement is affected by gravitational settling and buoyant airflow. Most particles remain concentrated between 0 and 2.5 m in height (indicating a transmission distance

of approximately 2.5 m), corresponding to the breathing zone of the seated and standing occupants. Minimal upward or downward dispersion is observed beyond this range, suggesting that the particles are largely contained within a height range critical to transmission risks.



Figure 6.33: Aerosol dispersion in X, Y, and Z directions for the second individual during NV.

Figure 6.34 shows the distribution of the particles released from the third individual's position. The particles are concentrated and dispersed primarily around the source location and downstream, reflecting the combined influence of airflow and initial particle velocity. Most of the particles remain within the vicinity of the infected individual, with limited transport across the room.



Figure 6.34: Particle transmission from the third infected individual during sneezing..

Figure 6.35 illustrates the aerosol dispersion in the X, Y, and Z directions for the third individual during NV.

As illustrated in Figure 6.35-(a), the longitudinal transport of the particles is relatively contained within the range of 6 to 8.5 m (indicating a transmission distance of approximately 2.5 m), with minor fluctuations observed as time progresses. The overall X movement can travel further due to airflow pushing particles across the horizontal plane. In the dispersion in the Y direction, as seen in Figure 6.35-(b), the lateral spread of the particles varies mainly between 4 and 7.5 m (indicating a transmission distance of approximately 3.5 m). The motion of the particles in this direction reflects the effects of both coughing velocity and cross-flow ventilation, though the distribution remains more compact compared to other infected individuals, possibly due to the specific location of the individual within the room. The vertical dispersion of the particles, shown in Figure 6.35-(c), occurs within the range of 0.25 to 3 m (indicating a transmission distance of approximately 2.75 m). The upward movement of the particles is influenced by the velocity of coughing and the airflow of the room, with the particles reaching a maximum height of approximately 3.5 m before settling down. This distribution pattern indicates moderate vertical spread, potentially increasing the likelihood of inhalation by nearby individuals.



**Figure 6.35:** Aerosol dispersion in X, Y, and Z directions for the third individual during NV.

Figure 6.36 illustrates the overall particle distribution release from the fourth individual's position. The particles initially form a dense plume near the source, spreading primarily toward the lower left and adjacent areas due to airflow restrictions. The accumulation of particles in this confined region indicates limited ventilation, with a large portion of the particles settling near the origin.



Figure 6.36: Particle transmission from the fourth infected individual during sneezing.

Figure 6.37 illustrates the aerosol dispersion in the X, Y, and Z directions for the fourth individual during NV.

Figure 6.37-(a) shows the movement of the particles along the X axis with time. Most particles remain between 0 and 3.5 m in the initial stage (indicating a transmission distance of approximately 3.5 m), indicating minimal lateral spread. Over time, particles move toward the door and exit into the adjacent zone, driven by the airflow pattern created by the open window and localised air currents. This movement demonstrates a key pathway for particle transport. The spread of particles in the Y axis, depicted in Figure 6.37-(b), reveals that the majority of particles are concentrated within a range of 8 to 13 m (indicating a transmission distance of approximately 5 m). The particles travel through the door and the distribution extends further into adjacent areas, showing how airflow effectively disperses aerosols beyond the source zone. As shown in Figure 6.37-(c), the movement of the particles along the Z axis mainly ranges between 0 and 2.5 m in height (indicating a transmission distance of approximately 2.5 m). Most particles settle near the floor or at mid-levels, with minimal upward dispersion. This limited vertical movement suggests that vertical air circulation is inadequate, resulting in particle accumulation at breathing levels and lower areas.



**Figure 6.37:** Aerosol dispersion in X, Y, and Z directions for the fourth individual during NV.

## 6.6 Discussion

This section provides a comprehensive discussion of the results obtained from the five NV scenarios, focussing on airflow patterns, IAQ, room temperature, RH, TC indices (PMV and PPD), and aerosol dispersion. The findings are compared with the ASHRAE, EN 16798, and ISO 7730 standards for acceptable indoor conditions to identify optimal and suboptimal configurations, and suggest recommendations for improved ventilation performance.

## 6.6.1 Comparison with ASHRAE, EN 16798, and ISO 7730 Standards

NV scenarios were evaluated against the TC criteria outlined in ASHRAE 55, ISO 7730, and EN 16798. These standards define acceptable PMV and PPD to

ensure a comfortable indoor environment. ASHRAE 55 specifies an occupancy satisfaction level of 80%, corresponding to a PMV range of -0.5 to +0.5 and a PPD below 20%, while ISO 7730 and EN 16798 classify comfort into three categories:

- Category A (ISO)/I (EN): PMV within ±0.2 and PPD < 6% (high expectations)
- Category B (ISO)/II (EN): PMV within ±0.5 and PPD < 10% (normal expectations)
- Category C (ISO)/III (EN): PMV within ±0.7 and PPD < 15% (moderate comfort)

The TC performance varied between the scenarios, with CV achieving the best results and SSV showing notable deviations.

In Scenario 1 (summer conditions), Zone 1 achieved PMV values between 0 and 0.5, meeting the ASHRAE and ISO Category B/II guidelines, ensuring that 80% of the occupants felt thermally comfortable. However, Zones 2 and 3 exhibited slightly warm conditions (PMV: 0.4–1.0), with PPD values exceeding 30%, indicating a notable dissatisfaction. Room temperature (20 °C to 22 °C) remained within the ASHRAE guidelines, but RH approached 70%, slightly exceeding the recommended threshold.

Air velocity was another crucial factor affecting comfort. Although ASHRAE 55 recommends indoor airflow velocities between 0.15 and 0.25 m/s, measurements showed that most areas had velocities below 0.1 m/s, indicating insufficient air circulation. However, areas near the open window experienced higher airflows (> 0.2 m/s), increasing the risk of aerosol dispersion and potential virus transmission.

Scenario 2 (also in summer conditions) demonstrated similar TC trends. Zone 1 was aligned with the ASHRAE and ISO standards, with PMV between 0 and 0.5 and occupant satisfaction 80%. However, Zones 2 and 3 again showed slightly

warm conditions (PMV: 0.4–1.0, PPD > 30%), highlighting areas of discomfort. Room temperature ( $20^{\circ}$ C to  $23^{\circ}$ C) remained within ASHRAE limits, but RH exceeded 65%, slightly exceeding comfort recommendations.

The air velocity distribution was mixed. Zone 1 complied mainly with the recommended air flow of ASHRAE of 0.2 m/s, but areas near the window exhibited an excessive air velocity (> 0.3 m/s), while Zones 2 and 3 remained below 0.08 m/s, indicating stagnant airflow and reduced ventilation effectiveness.

Scenario 3 performed best in terms of TC, with PMV values ranging between 0 and 0.5 in the three zones, ensuring the satisfaction of the occupants 80%. The room temperature remained between 20 °C and 23 °C, fully in accordance with the ASHRAE and ISO standards.

However, RH levels remained above 65%, indicating a challenge in humidity control in NV settings, especially in humid outdoor conditions. The air velocity followed similar trends, with Zone 1 maintaining airflow compliance (0.2 m/s), while Zones 2 and 3 had lower velocities (0 to 0.08 m/s), reinforcing the need for a better airflow distribution.

Scenario 4 integrated TC assessments with aerosol dispersion analysis. Although some areas met the ASHRAE guidelines, others deviated due to three open windows creating airflow inconsistencies. PMV ranged from -2 near the windows to +0.5 further into the room, which means that people near the windows experienced excessive cooling (PMV: -2, dissatisfaction > 80%), while those in the interior zones had relatively neutral conditions (PMV < 0.5).

Room temperatures ranged from 20 °C (zones 1 and 2) to 23 °C (Zone 3), according to ASHRAE recommendations. However, RH exceeded 65%, exceeding the upper comfort threshold.

Across all scenarios, air velocity varied significantly depending on location. CV provided sufficient air exchange (0.2 to 0.3 m/s), but SSV created stagnant air

pockets (< 0.1 m/s), failing to meet the air velocity criteria of ASHRAE 55 and ISO 7730.

Furthermore, areas near open windows had excessive airflow (> 0.3 m/s), which could increase the risk of viral transmission, because higher speeds disperse infectious aerosols further into the room. This aligns with studies by [332], which highlight the role of airflow in the propagation of viral particles.

TC standards also consider humidity effects, and ASHRAE 55 recommends an RH range of 30 to 60% for occupant comfort. In all of our NV scenarios, RH closely followed outdoor moisture conditions because there was no active dehumidification.

Although ISO 7730 suggests that moderate humidity variations have minimal influence on thermal sensation, ASHRAE 55 emphasises that high RH (> 65%) can reduce perceived air quality and increase discomfort. These findings indicate that NV alone cannot actively regulate humidity, making it less suitable in humid climates unless paired with dehumidification strategies or hybrid ventilation solutions.

### 6.6.2 Identification of Dead Zones in Natural Ventilation

Dead zones, defined as areas with minimal or stagnant airflow, pose a significant challenge in naturally ventilated environments, leading to inefficient air exchange, thermal discomfort, and increased health risks. Across the four NV scenarios, stagnant airflow regions were consistently identified, particularly in Zones 2 and 3, as well as areas shielded from direct airflow pathways, such as room corners and sections distant from open windows. These areas exhibited low air speeds (below 0.1 m/s), resulting in localised heat accumulation and elevated RH, thus exacerbating discomfort and potentially prolonging the persistence of the airborne virus.

In Scenarios 1 and 2, representing summer conditions, dead zones were most pronounced in Zones 2 and 3, as well as areas within Zone 1 that were far from open windows. Limited air circulation in these regions led to PMV values exceeding 0.5 and PPD values exceeding 30%, indicating notable occupant dissatisfaction. Furthermore, insufficient airflow contributed to elevated RH levels exceeding 65%, increasing the likelihood of discomfort and reducing IAQ.

Scenario 3 also showed persistent dead zones in Zones 2 and 3, where air speeds ranged between 0 and 0.1 m/s, further contributing to elevated RH (> 65%) and decreased occupant comfort. However, Zone 1 showed better airflow compared to previous scenarios, benefiting from improved ventilation due to window placement and external wind forces.

In contrast, Scenario 4 demonstrated improved air circulation in most of Zone 1, with acceptable air velocities ( $\geq 0.2 \text{ m/s}$ ). However, high airflow speeds (up to 0.4 m/s) were observed near open windows and parts of Zone 2. Although increased ventilation rates can improve IAQ, excessive airflow velocities pose a concern because they can facilitate the rapid dispersion of airborne contaminants. This aligns with the findings of [332], who highlight that higher air velocities can increase aerosol transmission, potentially exposing more occupants to airborne pathogens.

### 6.6.3 Impact of Air Velocity on Virus Dispersion

Air velocity plays a critical role in determining airborne particle transport, dispersion, and deposition, particularly in naturally ventilated environments where airflow is driven primarily by window openings, outdoor wind conditions, and internal convection currents. The CFD simulations of Scenario 4, which analysed aerosol dispersion from coughing and sneezing events, revealed that air velocity variations near open windows significantly influenced the movement of virus-laden particles within the office space.

During coughing events, the particles were expelled at an initial velocity of 11.8 m/s, following a forward-projected trajectory. Airflow near window openings, particularly in Window 1, facilitated rapid particle movement, increasing the risk of direct exposure to occupants seated along the airflow pathway.

The first infected individual, located near Window 1, experienced strong particle dispersion because the high local air velocity (>0.2 m/s) propelled the expelled aerosols through the workspace, increasing the risk of exposure of the occupant. However, the third and fourth infected individuals, located in Zones 2 and 3, also exhibited significant aerosol movement because the particles aligned with window-induced airflow patterns, which transported contaminants deeper into the space. In addition, the second infected individual, located farther from Window 1, experienced a reduced aerosol dispersion because the local air velocity in this area was lower. However, particles remained suspended longer, increasing the risk of localised exposure in low-ventilation zones. These findings highlight two major concerns:

- High air velocities near window openings accelerate aerosol dispersion, leading to widespread exposure among multiple occupants.
- Low-speed regions create stagnation zones, where aerosols linger longer, increasing localised inhalation risks.

Sneezing, which involves a much higher initial expulsion velocity (up to 70 m/s), showed more extensive and rapid dispersion compared to coughing. The simulation results demonstrated that the first infected individual's particles spread over 6 m, propelled by high window-induced air velocities. This extended reach increased the likelihood of particles being inhaled by multiple individuals farther from the source. However, the third and fourth infected individuals showed similar trends, where sneezed particles aligned with window-driven airflow patterns, leading to rapid transport into Zones 2 and 3. The second infected individual's particle dispersion exhibited a lower particle dispersion due to reduced airflow near the source, but the kinetic energy of the sneezed particles allowed them to spread widely, even in low-speed environments. These findings highlight the following:

- Sneezing events pose a significantly higher risk of dispersion because particles can spread across multiple zones in seconds, increasing potential exposure.
- High-velocity airflow near open windows further amplifies dispersion, creating long-range transmission pathways of airborne virus.

## 6.6.4 Impact of Relative Humidity on Comfort and Airborne Virus Stability

RH is a critical factor that influences both TC and the persistence of airborne pathogens, making it a key parameter in the management of indoor air quality. ASHRAE guidelines recommend maintaining indoor RH between 30% and 60% to optimise comfort and minimise potential health risks, while research suggests that an RH range of 40% to 60% is particularly effective in reducing viral stability and transmission risks [159, 260]. However, the findings of this study indicate that NV alone was insufficient to maintain RH within the recommended range, often resulting in excessive humidity levels in various scenarios.

In Scenario 4 (opening multiple windows), the unregulated influx of humid outdoor air caused the RH levels to exceed 80%, which exacerbated the discomfort related to moisture and further increased the risk of viral persistence. The high RH in these conditions compromised perceived air quality, making the environment feel stuffy and less ventilated, despite the presence of open windows. Scenarios 1, 2, and 3 (summer conditions with single window ventilation) displayed moderate to high humidity levels, ranging between 65% and 75% in specific zones. This increase was primarily driven by the continuous intake of humid outdoor air and insufficient airflow regulation, which did not adequately disperse excess indoor moisture. In Zone 2, where stagnation of airflow was observed, HR levels remained consistently high, contributing to thermal discomfort and a prolonged risk of viral persistence. ASHRAE 55 notes that high RH, particularly in warm environments, can amplify the perception of heat stress, further exacerbating occupant discomfort.

#### 6.6.5 Scenario Performance Evaluation

The performance of NV scenarios was assessed based on key indoor air quality parameters (IAQ) and TC, including air velocity, room temperature, RH, PMV and PPD. These results were evaluated against the ASHRAE 55, ISO 7730, and EN 16798 standards, which provide guidelines for maintaining acceptable TC and air quality. Comparative analysis of different scenarios provides insight into the effectiveness of NV in maintaining a comfortable and healthy indoor environment, while also highlighting areas where performance was suboptimal.

Scenario 1 simulated summer conditions with a single window open, allowing moderate thermal regulation. The room temperature remained between 20 °C and 22 °C, which is in good agreement with the ASHRAE comfort guidelines. The air velocity ranged from 0 to 0.2 m/s, suggesting that while there was some ventilation, airflow remained weak in certain areas. Zones 2 and 3, in particular, experienced lower airflow rates, leading to localised thermal imbalances and stagnation zones.

Relative humidity levels were elevated (65% to 70%), which, while not extreme, still exceeded the upper comfort threshold of ASHRAE of 60%, raising concerns

about moisture retention and viral persistence. PMV values ranged from 0 to 0.8, indicating mildly warm conditions, with PPD levels around 10%, which means that most of the residents were satisfied with the indoor environment. However, the persistence of humidity and stagnant airflow in Zones 2 and 3 suggests that additional measures, such as optimised window placement or additional ventilation, would be necessary to improve air circulation and humidity control.

Scenario 2 maintained room temperatures between 20 °C and 23 °C, which is within the recommended range of ASHRAE. However, TC varied between different zones. Zone 1 exhibited optimal ventilation, with PMV values around 0 and PPD levels at 10%, indicating that most individuals were comfortable.

In contrast, Zones 2 and 3 experienced discomfort due to poor airflow and slightly higher temperatures. PMV values exceeded 0.5 in these zones and PPD levels exceeded 20%, suggesting increased occupant dissatisfaction. The air velocity in several areas remained below 0.1 m/s, which exacerbated stagnation and poor air mixing. Furthermore, RH levels persisted at 65% to 70%, which reinforces concerns about airborne virus longevity and overall IAQ. Improving ventilation distribution through window adjustments or mechanical airflow enhancements could help to mitigate localised comfort issues and enhance air circulation in under-ventilated zones.

Scenario 3 followed a similar pattern to Scenario 2, with Zones 2 and 3 experiencing discomfort due to insufficient air circulation. The air velocity across these zones was low (below 0.1 m/s in several locations), which limited effective ventilation and cooling.

Room temperatures remained within 20 °C and 23 °C, in line with ASHRAE recommendations, but RH levels continued to exceed 65%, raising concerns about airborne virus transmission and occupant comfort. Zone 1 maintained acceptable TC, with PMV values close to 0 and PPD levels at 10%, indicating that most of the occupants in this zone were comfortable. However, in Zones 2 and 3, the

PMV values exceeded 0.5, with PPD exceeding 20%, suggesting that significant portions of the occupants were dissatisfied.

Although Scenario 3 slightly improved airflow over previous cases, the persistence of humidity and stagnant airflow highlights the need for better ventilation management strategies, such as improved window configurations or hybrid ventilation systems.

Scenario 4 maximised NV by opening three windows, significantly improving air exchange. This approach resulted in higher air velocities (exceeding 0.2 m/s), with some areas reaching 0.4 m/s. Although this improved overall ventilation efficiency and particle dilution, it also introduced localised discomfort due to excessive cooling.

The room temperature was well regulated in Zones 1 and 3 (20 °C to 23 °C), according to the comfort standards of the ASHRAE. However, Zone 2 experienced temperature drops below 18°C, leading to cold stress and discomfort in the occupants. Relative humidity levels remained above 65%, consistent with previous scenarios, reinforcing the limitations of NV in controlling moisture levels.

TC analysis indicated that PMV values remained within an acceptable range (0.4) in Zones 1 and 3, with PPD around 10%. However, in Zone 2, PMV values dropped below -1, resulting in PPD levels exceeding 50%, highlighting a significant dissatisfaction due to excessive cooling and exposure to draught.

Although Scenario 4 provided the most effective air exchange, it demonstrated the risks associated with over-ventilation, where excessive air velocity increased aerosol dispersion and created thermal discomfort. This highlights the need for better airflow regulation, window positioning adjustments, and possible integration with mechanical systems to balance ventilation efficiency and TC.

Scenario	Air Velocity	Temperature	RH	PMV	PPD
1	0 - 0.2 m/s	20 °C - 24 °C	> 75%	0 to 1.2	> 30%
2	0.02 - 0.2 m/s	20 °C - 24 °C	> 70%	0 - > 1	< 20% - > 20%
3	0.02 - 0.35 m/s	20 °C - 22 °C	nearly 70%	-1.6 - > 1 (Zones 2 and 3)	PPD > 20%
4	0.2 - 0.4 m/s	18.5 °C - 23 °C	> 80%	-2.5 - 1	10% - > 50% (Zone 2)

 Table 6.4:
 Comparative summary of scenario performance

# 6.6.6 Comparison of Natural Ventilation Against Air Conditioning

NV and AC represent two distinct approaches to maintaining IAQ and TC. NV relies on outdoor air movement and pressure differentials to drive airflow, while AC uses controlled air circulation through fans, ducts, and HVAC systems. The findings of this study highlight the strengths and limitations of each system in terms of air velocity, temperature regulation, humidity control, and overall TC.

One of the primary differences between the two ventilation strategies is air distribution. NV demonstrated inconsistent and uneven airflow, leading to stagnant air zones, particularly in areas shielded from direct airflow pathways, such as Zones 2 and 3. In these regions, the air velocity frequently fell below 0.1 m/s, which reduced the effectiveness of ventilation and contributed to localised thermal discomfort. AC, on the other hand, maintained a more uniform air distribution, with air velocities ranging between 0.1 and 0.25 m/s. However, despite its consistency, the AC also exhibited stagnation in airflow in corners and behind furniture, indicating that neither system was completely free of airflow deficiencies. In addition, NV was highly dependent on external weather conditions, making its performance unpredictable. In some scenarios, high-velocity airflow near open windows exceeded 0.4 m/s, leading to localised discomfort and an increased risk of viral dispersion. In contrast, AC provided more controlled and predictable airflow, but primarily recirculated indoor air without introducing fresh outdoor air, which could contribute to pollutant buildup and increased infection risks.

Temperature regulation also varied significantly between the two systems. NV produced uneven thermal conditions, with areas near windows experiencing cooling or heating effects, while enclosed regions retained heat due to poor air circulation. The room temperatures in NV scenarios ranged between  $18.5 \,^{\circ}$ C and  $23 \,^{\circ}$ C, depending on the outside temperature, which is not consistent with the ASHRAE comfort recommendations in some scenarios. However, AC provided more stable temperature control, maintaining indoor temperatures within  $22 \,^{\circ}$ C to  $25 \,^{\circ}$ C in most zones. However, cold air pockets formed in some areas due to uneven air mixing, making occupants uncomfortable near the air vents. TC analysis also indicated that NV often failed to maintain PMV within the recommended range of ASHRAE of -0.5 to +0.5, particularly in stagnant zones where PMV values exceeded 0.5. In contrast, AC generally maintained the PMV within the optimal range, but occasionally caused over-cooling in specific locations.

Humidity control was another area where natural and AC exhibited distinct performance characteristics. NV was unable to effectively regulate indoor RH because it was entirely dependent on external humidity conditions. As a result, RH levels frequently exceeded 65%, particularly in humid outdoor conditions, increasing the stability of the airborne virus and the risk of mould growth. AC partially controlled the humidity, but it was still unable to maintain the RH consistently within the optimal range of 30% to 60%, as recommended by ASHRAE. Some areas with AC exhibited an RH above 65%, highlighting the need for additional measures of dehumidification. Furthermore, the reliance of the AC on recirculated air without sufficient filtration posed a risk of pollutant accumulation, while NV facilitated fresh air exchange but suffered from stagnation in certain zones,
reducing its effectiveness in contaminant removal.

In general, AC provided more consistent airflow and temperature regulation, but lacked fresh air exchange, which is essential to maintain good indoor air quality. While NV was effective in facilitating fresh air intake, it suffered from uneven airflow distribution, leading to thermal imbalances and humidity fluctuations. Both systems exhibited limitations that affected occupant comfort and IAQ, suggesting that a hybrid approach combining natural and AC (referred to as mixed-mode ventilation) could address these shortcomings more effectively.

# 6.6.7 Recommendations for Optimising Natural Ventilation Performance

The findings from the CFD simulations and discussion sections (see section 6.5 and section 6.6) demonstrate that while natural ventilation (NV) offers a passive and energy-efficient means of improving indoor air quality (IAQ) and thermal comfort (TC), its effectiveness is highly dependent on external climatic conditions, window configurations, and occupant behaviour. To enhance the reliability and performance of NV in office environments, several key recommendations are proposed:

• Strategic Window Placement and Operation: The performance of NV is strongly influenced by the location, size, and operability of windows. As demonstrated in Scenario 4 (section 6.5.4.1), cross-ventilation (CV) significantly improved airflow distribution. Therefore, buildings should incorporate multiple, strategically placed openings to enable effective crossventilation. Adjustable configurations, such as operable louvres or tophung windows, can help regulate air intake based on real-time wind conditions, preventing over-ventilation and reducing draught discomfort (see section 6.6.5).

- Addressing Dead Zones: Stagnant airflow zones—particularly identified in Zones 2 and 3 of Scenarios 1, 2, and 3 (see section 6.5.1.1, section 6.5.2.1, section 6.5.3.1, and section 6.6.2 which led to poor ventilation and localised pollutant accumulation. To mitigate this, passive elements like chimney stacks or wind catchers, along with mechanical aids such as ceiling fans or small exhaust units, can promote air mixing in under-ventilated areas. Integrating NV with mechanical support (mixed-mode ventilation) is recommended where full natural ventilation is insufficient.
- Managing Relative Humidity (RH): Elevated levels of RH, often exceeding 75% in Scenarios 4 (see section 6.5.4.2) can promote viral stability and discomfort. It is recommended to incorporate dehumidification strategies or use ventilation controls that modulate air intake during high-humidity periods. In humid climates or seasons, combining NV with mechanical dehumidification systems offers a more reliable solution.
- Controlling Air Velocity and Infection Risk: High air velocities near open windows (e.g., Scenario 4 in section 6.5.4.4) increased aerosol dispersion distances, raising the risk of airborne transmission. To reduce this risk, airflow should be maintained within ASHRAE's recommended indoor velocity range of 0.2 m/s (see section 6.6.3). This can be achieved by limiting window opening angles or automating ventilation through actuators linked to real-time CO2 or particle concentration sensors.
- Enhancing Occupant Awareness and Behaviour: Occupant interaction plays a critical role in NV effectiveness. Providing users with simple visual guidance (e.g., window position indicators or IAQ monitors) and instructions based on weather or room conditions can encourage optimal use of operable windows. As discussed in section 6.6.6, hybrid strategies can complement occupant behaviour to maintain a consistent indoor environment throughout the day.

By addressing these factors, window design, airflow optimisation, humidity control, velocity management, and occupant engagement, NV systems can be significantly enhanced. These recommendations aim to support architects, engineers, and building managers in designing and operating natural ventilation strategies that are resilient, adaptive, and effective under a range of environmental and occupancy conditions.

## 6.7 Conclusion

This chapter has investigated the performance of NV in improving IAQ, regulating TC, and mitigating the risk of airborne virus transmission in an office environment. Using CFD simulations, five scenarios were analysed under varying environmental conditions and ventilation configurations. The findings provide important information on the effectiveness of NV, as well as the associated challenges and optimisation strategies required to achieve better indoor environmental quality.

The effectiveness of NV was largely dependent on the placement of windows, external airflow conditions, and pressure differentials. CV, particularly in Scenario 4, provided the most efficient airflow distribution, reducing stagnant air regions and improving ventilation performance. However, this scenario also introduced challenges because excessive air velocities near open windows led to over-cooling and an increased risk of respiratory aerosol dispersion. In contrast, Scenario 1,2 and 3, demonstrated inadequate ventilation, resulting in poor air exchange and stagnation in multiple zones.

This study also assessed the ability of NV to maintain TC in different scenarios. The results showed that Scenario 1, 2 and 3, Zone 1 exhibited optimal TC, with PMV values between 0 and 0.5 and PPD levels below 10%, aligning with the ASHRAE TC standards. However, Zones 2 and 3 in these scenarios experienced slightly warm conditions, where PMV values exceeded 0.5 and PPD levels exceeded 20%, primarily due to limited airflow and insufficient CV. Scenario 4, which used multiple window openings, introduced localised over-cooling, particularly in areas directly exposed to airflow, where PMV values decreased below -1, leading to PPD values exceeding 50% in certain areas.

The influence of air velocity on virus dispersion was another key aspect that was analysed in this study. Higher air velocities (> 0.2 m/s) near windows significantly increased the dispersion range of airborne particles, posing a risk of cross-contamination among occupants. In Scenario 4, respiratory particles expelled by infected individuals positioned near open windows were rapidly transported across the room, reaching adjacent zones and affecting multiple occupants. Conversely, low air velocity regions (<0.1 m/s), particularly in Zones 2 and 3, allowed particles to remain suspended for longer durations, which increased the risk of prolonged exposure.

RH was another critical factor affecting both occupant comfort and airborne virus stability. In most scenarios, RH exceeded the ASHRAE-recommended range of 30% to 60%, with levels frequently surpassing 65%. Scenario 5 recorded RH levels above 80%, which can be attributed to the humid outdoor air influx through multiple open windows. These findings highlight the importance of integrating humidity control strategies to maintain acceptable indoor RH levels and minimize health risks.

The analysis also identified significant stagnant air regions (dead zones) in Zones 2 and 3 in multiple scenarios, particularly in Scenarios 1, 2, and 3. These areas exhibited low air speeds, leading to localised temperature buildup, poor ventilation, and increased pollutant accumulation. Dead zones also contributed to higher exposure risks due to prolonged aerosol suspension. Addressing these stagnant regions requires strategic airflow management, and integration of AC with NV may be necessary to ensure adequate circulation.

Several key recommendations have been identified to optimise the performance of NVs. First, regulating window opening sizes can help to control airflow intensity, preventing excessive ventilation that may lead to over-cooling or rapid aerosol dispersion. Introducing adjustable dampers or controlled ventilation schedules can further enhance ventilation efficiency. Second, to integrate mixed-mode ventilation, combining NV with AC can help to mitigate the impact of dead zones and improve air circulation in poorly ventilated areas. Implementing ceiling fans or air circulation systems in stagnant regions can further reduce pollutant accumulation and enhance TC. Third, maintaining RH levels below 60% through dehumidification strategies is essential to improve IAQ and reduce the stability of airborne viruses. Furthermore, ensuring that air velocities remain within the recommended ASHRAE threshold of 0.2 m/s can balance adequate ventilation while minimising excessive aerosol dispersion.

In direct response to Research Question 3 (RQ3), the results show that NV when effectively configured can significantly improve airflow distribution, enhance thermal comfort, and reduce airborne infection risks in indoor environments. However, without careful control, it may also introduce challenges such as uneven airflow, humidity excess, and uncontrolled particle spread.

**Regarding Research Question 4 (RQ4)**, this chapter demonstrates how NV strategies can be adapted under different environmental conditions to optimise indoor environmental quality. Specifically, cross-ventilation with moderated airflow and supplemental mechanical support emerges as the most balanced and effective configuration.

This chapter provides several key contributions to the field of indoor environmental quality and ventilation design. First, it presents a detailed CFD-based assessment of natural ventilation (NV) strategies under varying environmental conditions, providing insight into the dynamics of airflow, thermal comfort, and pollutant dispersion in office environments. Second, it highlights the limitations of conventional NV configurations, such as the presence of stagnant zones, overcooling, and uncontrolled aerosol transport, while demonstrating the effectiveness of cross ventilation as a superior strategy. Third, the chapter introduces evidence-based design recommendations for optimising NV performance, including mixed-mode ventilation, humidity control, and strategic airflow regulation. These findings advance our current understanding of the role of NV in improving IAQ, mitigating infection risk and improving thermal comfort, and they serve as a foundation for future development of adaptive and sustainable ventilation solutions in modern buildings.

In conclusion, this study has shown that when NV is optimised it can significantly improve indoor air quality, improve TC, and contribute to infection control. However, challenges such as uneven airflow distribution, excessive humidity levels, and the uncontrolled spread of airborne contaminants must be addressed to fully harness the benefits of NV. The findings emphasise the need for an integrated ventilation approach, in which NV is supplemented by adaptive mechanical strategies to ensure a healthier and more sustainable indoor environment. Future research should focus on real-time ventilation monitoring and dynamic airflow control systems, allowing data-driven optimisation of NV strategies. Furthermore, hybrid ventilation models that incorporate intelligent mechanical controls can further enhance the performance of NV, ensuring its viability as a long-term solution to maintain indoor environmental quality.

# Chapter 7

# Mixed Ventilation

# 7.1 Introduction

This chapter directly addresses **Research Question 3 (RQ3) and Research Question 4 (RQ4)**. Building upon the findings of Chapter 5 (which identified limitations in fully air-conditioned systems, particularly in terms of dead zones and uneven airflow) and Chapter 6 (which demonstrated that fully natural ventilation, while beneficial for fresh air exchange, can lead to overcooling, inconsistent airflow, high RH, and increased infection risk), this chapter investigates mixed-mode ventilation as an integrated solution to overcome these challenges.

Mixed-mode ventilation, commonly referred to as hybrid ventilation, integrates NV and AC strategies to optimise IAQ, TC, and reduce virus transmission. Unlike full AC systems, which rely on continuous energy consumption for air circulation and temperature control, mixed ventilation leverages natural airflow (e.g., operable windows) when outdoor conditions are favourable, while activating AC systems as needed to maintain stable indoor environmental conditions. This dynamic approach aims to reduce HVAC energy costs, while ensuring sufficient air exchange and occupant comfort [312]. The choice of ventilation strategy significantly influences thermal regulation, air distribution, and pollutant removal efficiency. NV is widely recognised for its ability to improve IAQ and reduce overheating risks, particularly in buildings designed with efficient cross ventilation pathways [333, 334]. However, it remains highly dependent on external weather conditions, outdoor air quality, and occupant behaviour, making it unreliable for consistent IAQ control. In contrast, AC offers precise control over temperature and air circulation, but requires higher energy consumption and may result in uneven air distribution [334]. Consequently, an integrated hybrid ventilation approach presents a potential solution, combining the advantages of both strategies while mitigating their individual limitations.

A key limitation identified in the baseline study (Chapter 5) was that a full AC system alone was insufficient to optimise IAQ in all zones. Although AC maintained air exchange, it resulted in stagnation zones and uneven airflow distribution, ultimately compromising TC. As a potential alternative, Chapter 6 investigated fully NV scenarios. However, the findings indicated that NV alone was ineffective in maintaining stable indoor temperatures, leading to thermal discomfort and increased aerosol dispersion, increasing the risk of airborne virus transmission. Given these limitations, a hybrid ventilation approach in which air conditioning and NV is strategically combined is hypothesised to overcome these challenges. By balancing fresh air intake and AC airflow control, mixed ventilation has the potential to improve IAQ, TC, and virus containment more effectively than fully AC or fully natural systems. Several studies have demonstrated the effectiveness of hybrid ventilation in optimising IAQ and TC. The research by [335] emphasised the importance of airflow regulation to reduce airborne contaminant accumulation. Their findings suggest that adjustable hybrid ventilation systems can significantly mitigate aerosol concentrations, thereby improving airborne infection control. Similarly, [336] found that AC, when combined with partial window openings (30%), reduced the risk of infection transmission by 58% to 70% compared to fully enclosed spaces.

## 7.1.1 Chapter Objectives

The primary objective of this chapter is to evaluate the performance of mixed ventilation to maintain optimal airflow patterns, TC, and control of airborne infections in an office environment. The study aims to evaluate the effectiveness of hybrid ventilation strategies using CFD simulations. Specifically, this chapter seeks to:

- Analyse the airflow distribution in different mixed ventilation scenarios and its impact on air circulation.
- Evaluate the variations in temperature and RH in mixed ventilation strategies.
- Assess occupant TC using the PMV and PPD indices.
- Examine aerosol dispersion patterns during coughing and sneezing events to understand the risk of airborne virus transmission.
- Compare the performance of mixed ventilation with the ASHRAE 55, EN 16798, and ISO 7730 standards for TC to ensure that the system meets the recommended ventilation effectiveness criteria.
- Identify the most effective mixed-ventilation strategy that improves IAQ and reduces thermal discomfort.

# 7.2 Methodology

This study employs a CFD simulation approach to evaluate the performance of mixed ventilation in an open-plan office environment. This methodology follows a structured framework, beginning with the identification of key parameters and solver settings, followed by the development of a computational model that incorporates geometry, meshing, and boundary conditions. The CFD simulation assesses IAQ, TC, and aerosol dispersion, providing insight into the effectiveness of mixed ventilation strategies under different seasonal conditions. The results are analysed for airflow patterns, temperature regulation, RH, TC indices of the occupants (PMV and PPD), and risks of transmission of airborne viruses. Finally, recommendations are derived based on the findings to optimise the performance of mixed ventilation. The general research framework is illustrated in Figure 7.1, summarising the step-by-step process undertaken in the simulations.



Figure 7.1: Overview of the methodology for mixed ventilation simulations

# 7.3 Scenario Development

This chapter evaluates three different mixed ventilation scenarios to evaluate system performance under different seasonal conditions and ventilation strategies, as summarised in Table 7.1. Each scenario evaluates airflow distribution, temperature control, humidity levels, and human TC metrics.

Scenario No.	Season	Natural Configuration	Environmental Conditions	Mixed Configuration	Environmental Conditions
1	Winter	Single Window Open (Window 4) in Zone 1	T = 4 °C, V = 3 m/s, RH = $0.004$	Single AC operates in Zone 1 (upper right corner)	$AC = 19 \circ C. H2O$ = 0.0177
2	Winter	Single Window Open (Window 4) in Zone 1	$T = 13 ^{\circ}C, V = 3$ m/s, RH = 0.004	Single AC operates in Zone 1 (upper right corner)	$AC = 19 ^{\circ}C, H2O$ = 0.006
3	Summer	Single Window Open (Window 6) in Zone 1	T = 19 °C, V = 3 m/s, RH = $0.004$	Single AC operates in Zone 2 (upper left corner)	$AC = 13 \circ C,$ H2O = 0.0083

Table 7.1: Mixed ventilation scenarios and their environmental conditions.

# 7.4 Boundary Conditions

The simulation was conducted under steady-state conditions to capture the combined effects of NV and AC on air distribution, thermal regulation, and respiratory particle dispersion. The mixed ventilation system consisted of an open window that allowed fresh outdoor air to be exchanged and a ceiling mounted AC unit to maintain indoor temperature stability. This setup represents a realistic mixed-mode ventilation strategy, where natural air flow complements AC conditioning, improving TC, air quality, and helping to reduce virus transmission.

The NV component was modelled using a boundary condition of velocity entry in the open window to simulate the entry of outdoor air, with the air temperature adjusted according to seasonal conditions. The inlet temperature was set to 4°C for winter (Scenario 1), 13 °C for mild winter (Scenario 2), and 19 °C for summer (Scenario 3). The velocity of the outdoor air was maintained at 3 m/s, ensuring adequate fresh air intake and cross ventilation within the office space.

The AC component was represented by an AC unit that supplied conditioned air, with temperature settings adjusted based on seasonal requirements. In winter scenarios, the AC supplied air at 19°C, ensuring a warm and stable indoor environment, while in summer scenarios the AC operated at 13°C, providing cooling effects to maintain occupant comfort. The combined operation of NV and AC facilitated continuous fresh air exchange while stabilising indoor temperatures, preventing overheating or excessive cooling in occupied zones.

To maintain proper airflow balance, the air outlet was set as a pressure outlet, ensuring continuous air discharge without backflow or pressure build-up. This boundary condition facilitated steady indoor air circulation, preventing stagnant airflow pockets that could reduce the effectiveness of ventilation. The interaction between NV and AC components was analysed to determine how airflow distribution, contaminant removal, and TC levels were influenced by the hybrid system.

To comprehensively assess mixed ventilation performance, this study considered 10 different simulation cases, as outlined in Table 7.2. The first set of cases (cases 1, 3, and 5) assessed ventilation efficiency by analysing airflow, temperature, and the RH distribution when a single window was open under different seasonal conditions. Cases 2, 4, and 6 examined the comfort of the occupant in mixed-mode ventilation scenarios. The final set of cases (cases 7 to 10) introduced an infected occupant in different workstation positions (A, B, C, and D) to assess how airborne particles dispersed under combined NV and AC. This analysis provided insight into how airflow direction, ventilation effectiveness, and spatial configurations impacted IAQ and infection risks.

Case No.	Scenario	Infected Source	Notes
1	Scenario 1	None	Assessment of ventilation performance through one window at T= 4 °C, and AC at T= 19 °C.
2	Scenario 1	None	Assessment of thermal comfort.
3	Scenario 2	None	Assessment of ventilation performance through one window at T= 13 °C, and AC at T= 19 °C.
4	Scenario 2	None	Assessment of thermal comfort.
5	Scenario 3	None	Assessment of ventilation performance through one window at T= 19 °C, and AC at T= 13 °C.
6	Scenario 3	None	Assessment of thermal comfort.
7	Scenario 3	А	Assessment of infection risk.
8	Scenario 3	в	Assessment of infection risk.
9	Scenario 3	С	Assessment of infection risk.
10	Scenario 3	D	Assessment of infection risk.

 Table 7.2: Overview of simulation cases during the MV.

This section presents the findings of three different mixed ventilation scenarios, each designed to evaluate the interaction between NV and AC (MV) in the regulation of IAQ, TC, and airflow distribution. This study specifically examines the influence of outdoor air infiltration through open windows and the impact of localised AC systems, focussing on key parameters such as air velocity patterns, temperature variations, RH and TC indices, including PMV and PPD.

## 7.5.1 Scenario 1

## 7.5.1.1 Air Velocity Pattern

The air velocity distribution for Scenario 1, representing a mixed ventilation system during winter conditions, highlights the interaction between AC and NV through an open window (Window 4).

Figure 7.2 illustrates the overall velocity distribution in the office space, revealing an unstable airflow pattern, particularly in Zone 1, where velocity values fluctuate between 0.1 m/s and 0.4 m/s. The highest velocities are concentrated near the window and along the airflow trajectory, while areas near the occupants in the central sections of Zone 2 and Zone 3 exhibit reduced airflow, with velocities as low as 0.02 m/s to 0.1 m/s. This reduction in velocity can be attributed to the limited direct airflow pathways available in these areas, resulting in lower air exchange rates compared to Zone 1.

In general, the air velocity distribution in Scenario 1 suggests that the combination of AC and NV contributes to effective airflow circulation in Zone 1, while airflow in Zones 2 and 3 remains relatively stagnant due to the absence of direct ventilation openings. This highlights the need for additional ventilation strategies to improve airflow uniformity and mitigate possible air stagnation in enclosed zones.



Figure 7.2: Air velocity contour in z direction during Scenario 1 of mixed ventilation.

## 7.5.1.2 Room Temperature and Relative Humidity

The room temperature distribution in Scenario 1 illustrates the thermal interactions between NV, introduced through an open window, and AC from a localised AC system. Figures 7.3 depict the spatial temperature variations in the three office zones, highlighting the impact of airflow dynamics and thermal stratification.

The results suggest that the overall temperature remains relatively low, indicating that AC has a limited impact due to the dominant cooling effect of outdoor air. The cold air introduced through NV overpowers the warming effect of AC heating, rendering the AC system ineffective in maintaining a balanced indoor temperature.

The analysis of the temperature distribution in Scenario 1 highlights that mixed ventilation is not effective in maintaining TC under extreme winter conditions. The dominance of cold outdoor air results in an excessively cold indoor environment, which reduces the efficiency of the AC system. These findings emphasise that keeping windows open in very cold conditions is not a suitable ventilation strategy because it leads to significant heat loss and discomfort for the occupants. Alternative solutions, such as regulated window openings, controlled ventilation rates, or additional heating, should be considered to maintain an optimal indoor climate.



**Figure 7.3:** Temperature contour in z direction during Scenario 1 of mixed ventilation.

The distribution of RH in Scenario 1 illustrates the combined effects of NV (introduced through the open window) and AC (provided by the AC system).

Figures 7.4 depict the spatial variations in RH in the three office zones, highlighting the influence of airflow patterns and ventilation sources.

The results indicate that the RH levels near the occupants who are farther from the open window and the airflow input range between 60% and 70%, providing relatively moderate humidity conditions. However, the rest of Zone 1, as well as Zones 2 and 3, exhibit significantly higher levels of RH exceeding 79%, indicating that moisture accumulation is more prominent in these areas.



Figure 7.4: RH contour in z direction during Scenario 1 of mixed ventilation

## 7.5.1.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The PMV index is used to evaluate TC by assessing the combined influence of air temperature, humidity, air velocity, metabolic rate, and clothing insulation.

Figures 7.5 present the spatial distribution of PMV values across the three monitored office zones. The simulation results reveal a significant variation in thermal comfort levels among the different areas. Specifically, Zone 1 is characterised by PMV values of -2 or lower across most regions, indicating that occupants in this zone are likely to feel cold and experience notable thermal discomfort. In contrast, Zones 2 and 3 display slightly better thermal conditions, with PMV values predominantly ranging between -1 and -2. While this range suggests a milder level of discomfort compared to Zone 1, it still falls outside the acceptable thermal comfort range defined by international standards such as ASHRAE 55 and ISO 7730. Overall, these findings confirm that none of the zones provide thermally comfortable conditions, emphasising the need for improved ventilation or heating strategies to enhance occupant comfort.



Figure 7.5: PMV contour in z direction during Scenario 1 of mixed ventilation

The PPD index quantifies occupant discomfort levels based on deviations from optimal TC conditions. Figures 7.6 illustrate the distribution of PPD in the three office zones, highlighting the impact of NV and AC on thermal conditions. The results indicate that most areas in all three zones exhibit extremely high PPD values, exceeding 90%. Zone 2 shows a slightly lower PPD of approximately 50% in some areas, but this still falls outside the comfort range.



Figure 7.6: PPD contour in z direction during Scenario 1 of mixed ventilation

## 7.5.2 Scenario 2

## 7.5.2.1 Air Velocity Pattern

Scenario 2 follows a ventilation strategy similar to Scenario 1, but incorporates a higher outdoor air temperature  $(13 \,^{\circ}\text{C})$  and an increased absolute humidity ratio  $(0.006 \, \text{kg/kg})$ .

Figures 7.7 provides an overview of the air velocity distribution throughout the office layout.

The general velocity analysis for Scenario 2 reveals a marginal improvement in airflow homogeneity compared to Scenario 1, particularly within Zone 1, where external ventilation provides consistent air movement. However, Zones 2 and 3 continue to experience restricted airflow, reinforcing the conclusion that additional ventilation sources or airflow redistribution strategies may be required to improve circulation efficiency in enclosed areas.



Figure 7.7: Air velocity contour in z direction during Scenario 2 of mixed ventilation.

## 7.5.2.2 Room Temperature and Relative Humidity

The temperature distribution in Scenario 2 highlights the interaction between NV from the open window and AC from the localised AC system. Figure 7.8 provides an overview of the temperature distribution throughout the office layout.

The results suggest that the AC system is more effective in Scenario 2 compared to Scenario 1 because slightly warmer outdoor air (13 °C) does not create excessive cooling. The temperature distribution analysis for Scenario 2 indicates that mixed ventilation achieves moderate TC under winter conditions, with a more balanced indoor temperature compared to Scenario 1. Although NV still introduces cool air, the AC system compensates more effectively in this scenario, reducing extreme temperature variations. These findings highlight the importance of the temperature of the outdoor air in determining the effectiveness of mixed ventilation strategies, highlighting that slightly warmer outdoor conditions allow better temperature regulation without excessive cooling.



Figure 7.8: Temperature contour in z direction during Scenario 2 of mixed ventilation.

The distribution of RH in Scenario 2 illustrates the combined effects of NV (introduced through an open window) and AC (provided by the AC system). Figure 7.9 provides a broader perspective of the RH distribution.

The results indicate that most areas in Zones 1, 2, and 3 experience levels of RH above 65%, with some areas in Zone 2 exhibiting lower RH values around 50%. This suggests that while NV and AC contribute to the variation in moisture in different areas, overall RH remains close to the acceptable range for indoor comfort.



Figure 7.9: RH contour in z direction during Scenario 2 of mixed ventilation

## 7.5.2.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The PMV index is used to evaluate TC by assessing the combined influence of air temperature, humidity, air velocity, metabolic rate, and clothing insulation. Figure 7.10 provides a comprehensive overview of PMV variations throughout the office.

The results indicate that most areas in Zone 1 exhibit PMV values between - 1 and 0, providing moderate TC. Occupants near the open window experience

slightly colder conditions, with PMV values reaching -2 due to direct exposure to outdoor air. In contrast, Zones 2 and 3 exhibit PMV values around 1, indicating a slight tendency toward warm conditions, which is attributable to limited airflow regulation and heat accumulation near the ceiling.



(a) z = 1 m

(b) z = 2 m

Figure 7.10: PMV contour in z direction during Scenario 2 of mixed ventilation.

The PPD index quantifies occupant discomfort levels based on deviations from optimal TC conditions. Figure 7.11 provides a comprehensive overview of PPD variations throughout the office.

The results indicate that most areas in all three zones exhibit moderate discomfort, with PPD values ranging between 20% and 50%. Higher PPD values are observed near the open window in Zone 1, where direct exposure to cold outdoor air significantly impacts occupant comfort. Thermal discomfort remains widespread, but conditions are significantly improved compared to Scenario 1. AC has a limited impact on mitigating the effects of cold outdoor air intrusion, particularly in areas directly affected by NV airflow.



Figure 7.11: PPD contour in z direction during Scenario 2 of mixed ventilation.

## 7.5.3 Scenario 3

## 7.5.3.1 Air Velocity Pattern

Scenario 3 represents a summer ventilation condition, which incorporates a different window opening (Window 6) and adjusted AC placement in Zone 2. The velocity contours presented in Figure 7.12 provides a comprehensive overview of the air velocity distribution.

The findings of Scenario 3 suggest that the adjustment of AC placement significantly influences airflow distribution, with Zone 2 benefiting the most from improved ventilation efficiency. In contrast, Zone 3 continues to exhibit limited airflow penetration, indicating the need for additional ventilation strategies to enhance circulation in this enclosed area.



**Figure 7.12:** Air velocity contour in z direction during Scenario 3 of mixed ventilation.

## 7.5.3.2 Room Temperature and Relative Humidity

The room temperature distribution in Scenario 3 reflects the combined influence of NV and AC on the maintenance of TC in the home. Figure 7.13 provides an overview of the temperature distribution throughout the office layout.

The results demonstrate that mixed ventilation successfully maintained indoor temperatures within an optimal range, with NV effectively cooling Zone 1 and AC supporting temperature regulation in Zones 2 and 3.

In general, the temperature distribution in Scenario 3 is in good agreement with the TC standards. Indoor temperatures remain within an acceptable range, with most areas between 20 °C and 22 °C, which is within the recommended comfort range according to the Fishman survey [260, 337]. This suggests that the mixedventilation strategy employed in this scenario effectively maintains a thermally comfortable environment throughout the office space.



Figure 7.13: Temperature contour in z direction during Scenario 3 of mixed ventilation.

The distribution of RH in Scenario 3 illustrates the combined effects of NV (introduced through an open window) and AC (provided by the AC system).

Figure 7.14 provides a broader perspective of the distribution of RH across the office.

The results indicate that most areas in Zones 1, 2, and 3 experience levels of RH around 60%, with some areas slightly exceeding this threshold. These findings suggest that NV and AC contribute to maintaining a stable and comfortable RH level throughout the office.



Figure 7.14: RH contour in z direction during Scenario 3 of mixed ventilation

## 7.5.3.3 Predicted Mean Vote and Predicted Percentage of Dissatisfied

The distribution of PMV values in the three office zones in Scenario 3 demonstrates the influence of NV and AC on indoor thermal sensations, as depicted in Figures 7.15.

The results indicate that most areas in all three zones maintain optimal TC, with PMV values between -0.5 and 0.5. Zone 3 exhibits slightly higher PMV values, reaching up to 1.2, particularly near the ceiling, where the accumulation of heat is more pronounced due to limited air circulation. However, at the occupant level, the PMV remains within the acceptable comfort range.

Scenario 3 demonstrates a well-balanced thermal environment, effectively maintained through the combined use of NV and AC. The AC system in Zone 2 successfully regulates the temperature, while NV in Zone 1 ensures stable airflow and comfort conditions. Although Zone 3 experiences slightly higher PMV values near the ceiling, the overall thermal environment remains within the recommended comfort levels, strengthening the effectiveness of the mixed ventilation approach.



Figure 7.15: PMV contour in z direction during Scenario 3 of mixed ventilation.

The PPD index quantifies occupant discomfort based on deviations from optimal thermal conditions. In this scenario, with an outdoor temperature of 19 °C, NV through an open window in Zone 1, and AC operating at 13 °C in Zone 2, the PPD distribution is analysed to assess the effectiveness of the mixed-ventilation strategy. Figure 7.16 provides a comprehensive overview of PPD variations throughout the office.

The results indicate that most areas in all three zones exhibit excellent TC, with PPD values of 10% or lower. Localised areas near room corners exhibit slightly higher PPD values, ranging between 10% and 20%. The combination of NV in Zone 1 and AC in Zone 2 effectively regulates indoor conditions, ensuring a balanced airflow and temperature distribution throughout the office.

Scenario 3 demonstrates optimal thermal conditions, with low PPD values in all zones, confirming that the mixed-ventilation strategy (open window in Zone 1 and AC in Zone 2) was highly effective. This configuration successfully maintained indoor comfort, minimised discomfort, and ensured a well-ventilated environment. The PPD values in most areas remain below 10%, further validating the efficiency of this approach in achieving thermal satisfaction for the occupants.



Figure 7.16: PPD contour in z direction during Scenario 3 of mixed ventilation.

## 7.5.3.4 Aerosol Dispersion and Infection risk

**Coughing** is a primary mechanism for the airborne virus transmission of infectious aerosols in enclosed environments. The current analysis examines the dispersion dynamics of expelled particles under mixed ventilation conditions specifically in Scenario 3, where both NV (through an open window in Zone 1) and AC (AC in Zone 2) influence airflow patterns. A coughing velocity of 11.8 m/s was applied and the aerosol dispersion was analysed for four different infected individuals positioned in different locations of the office, further away from both the open window and the AC unit. The following results illustrate the spatial and temporal movement of particles in the X, Y, and Z directions, highlighting how airflow affects their dispersion.

Figure 7.17 presents the spatial distribution of aerosols expelled by the first infected individual during coughing. Initially, the expelled particles exhibit highly concentrated forward movement, but they undergo significant dispersion as they interact with indoor airflow. A dense accumulation of aerosols is observed near the source, but due to ventilation effects the particles disperse throughout the



office space, potentially exposing multiple occupants.

**Figure 7.17:** Particle transmission from the first infected individual during coughing.

Figure 7.18 illustrates the aerosol dispersion along the X, Y, and Z directions during the coughing event for the first individual.

Figure 7.18-(a) illustrates the movement of aerosols along the X axis. The results show an initial rapid displacement due to the high velocity of coughing, which propels the particles forward. Most particles remain within the 15.25 m to 16.75 m range (indicating a transmission distance of approximately 1.5 m), indicating limited lateral spread under current airflow conditions. However, recirculation effects from AC contribute to the movement of the particles within a confined region.

Figure 7.18-(b) highlights the lateral diffusion along the Y axis, with particle displacement values ranging between 11.75 m and 13.75 m (indicating a transmission distance of approximately 2 m). This suggests that horizontal airflow patterns significantly impact the way aerosols spread throughout the office. The interaction between NV and AC improves lateral dispersion, potentially increasing the risk of exposure of nearby occupants.

Figure 7.18-(c) represents the vertical particle distribution along the Z axis, where most aerosols remain between 0.5 m and 2.5 m (indicating a transmission distance of approximately 2 m). This aligns with the breathing zone of seated and standing occupants, increasing the likelihood of inhalation by individuals within this zone. The data indicate minimal upward or downward dispersion beyond 2.5 m, highlighting that airborne virus transmission risk remains highest at the occupant level.



Figure 7.18: Aerosol dispersion through coughing on X, Y, and Z directions for the first individual during mixed ventilation.

Figure 7.19 shows the spatial dispersion of aerosols expelled by the second infected individual during coughing. Initially, the particles display a concentrated forward trajectory before dispersing as they interact with the air flow of the room. Unlike the first infected individual, who was positioned near the ventilation sources, the expelled particles of the second infected individual appear more localised, with less pronounced dispersion throughout the office. However, aerosols are still widely distributed in the immediate vicinity, posing a potential airborne virus transmission risk to nearby occupants.



Figure 7.19: Particle transmission from the second infected individual during coughing.

Figure 7.20 illustrates the aerosol dispersion along the X, Y, and Z directions during the coughing event for the second individual.

Figure 7.20-(a) shows that the movement of the particles along the X axis exhibits a stable dispersion pattern, mainly concentrated between 11.9 m and 12.8 m (indicating a transmission distance of approximately 0.9 m). Compared to the first infected individual, the lateral spread is more contained, suggesting that the airflow influence is less turbulent in this part of the room. The reduced dispersion in this direction could be attributed to the distance from both the AC and the window, limiting the strong airflow-induced movement.

The dispersion in the Y direction, as presented in Figure 7.20-(b), highlights that the lateral diffusion of particles along the Y axis is more constrained, remaining within a range of 8.6 m to 9.2 m (indicating a transmission distance of approximately 0.8 m). The data indicate that airflow interactions in this zone cause moderate horizontal movement, but particles remain largely near their initial source. The absence of direct airflow currents from the AC or window results in a more central accumulation of aerosols around the infected individual. The dispersion in the Z direction, illustrated in Figure 7.20-(c), shows that the vertical distribution of the aerosols remains within 1 to 1.75 m (indicating a transmission distance of approximately 0.75 m), corresponding to the breathing height of most of the occupants. Similarly to the first infected individual, the dispersion remains concentrated within the respiratory zone, strengthening the high transmission risk for individuals seated nearby. However, compared to the first infected individual, the lower vertical displacement suggests weaker upward airflow currents, likely due to the lack of strong forced ventilation effects in this section of the office.



Figure 7.20: Aerosol dispersion through coughing in X, Y, and Z directions for the second individual during mixed ventilation.

Figure 7.21 demonstrates the spatial distribution of aerosols expelled by the third infected individual. The expelled particles initially exhibit a concentrated forward movement before undergoing dispersion influenced by indoor airflow. Due to the position of this individual, particle dispersion remains relatively localised, with minimal spread towards distant occupants.



Figure 7.21: Particle transmission from the third infected individual during coughing.

Figure 7.22 illustrates the aerosol dispersion along the X, Y, and Z directions during the coughing event for the third individual.

Figure 7.22-(a) displays the movement of the particles in the X direction over time. The particles exhibit a slow and limited displacement compared to other cases, primarily restricted between 8 m and 8.6 m (indicating a transmission distance of approximately 0.6 m). This indicates a weaker lateral movement due to reduced airflow influence in this corner of the office.

In the Y direction (Figure 7.22-(b)), the lateral dispersion remains within a short range of 5.2 m to 5.9 m (indicating a transmission distance of approximately 0.7 m). This suggests that the expelled aerosols remain highly concentrated near the source, with minimal spread across the office.

In the Z direction (Figure 7.22-(c)), the particles remain within a height range of 0.7 m to 1.4 m (indicating a transmission distance of approximately 0.7 m), which aligns with the standard breathing zone of seated and standing individuals. The limited vertical dispersion further indicates that the airflow circulation in this



area is weak, leading to the accumulation of aerosols near the source.

Figure 7.22: Aerosol dispersion through coughing in X, Y, and Z directions for the third individual during mixed ventilation.

Figure 7.23 illustrates the aerosol dispersion pattern for the fourth infected individual, positioned in Zone 2, where the AC system is operational. The AC system plays a dominant role in controlling the dispersion of expelled particles, leading to a more confined spread compared to individuals located in Zone 1. The results indicate that the presence of the AC unit helps limit long-range dispersion, effectively reducing the risk of spread of infection to distant areas.



Figure 7.23: Particle transmission from the fourth infected individual during coughing.

Figure 7.24 illustrates the aerosol dispersion along the X, Y, and Z directions

during the coughing event for the fourth individual.

Figure 7.24-(a) shows the displacement of the particles in the X direction over time. particles initially move forward due to the coughing velocity, but remain concentrated within a small range from 1.1 m to 2 m after 6.7 seconds (indicating a transmission distance of approximately 0.9 m). The limited movement suggests that the airflow pattern in Zone 2, influenced by the AC unit, effectively restricts the particles from spreading throughout the office.

The dispersion in the Y direction, as shown in Figure 7.24-(b), highlights the lateral movement of the particles, with the dispersion constrained between 10.2 m and 10.6 m after 6.7 seconds (indicating a transmission distance of approximately 0.4 m). Unlike the individuals in Zone 1, where NV led to significant lateral spread, the AC unit in this zone appears to regulate airflow, preventing excessive horizontal movement.

In the vertical (Z) direction, Figure 7.24-(c) shows the vertical distribution of aerosols, which remains within 1.3 m to 1.8 m after 6.7 seconds (indicating a transmission distance of approximately 0.5 m). This range corresponds to the breathing zone of most occupants, indicating that while the AC system helps contain dispersion, expelled particles still accumulate at levels where transmission risk remains present.



Figure 7.24: Aerosol dispersion through coughing in X, Y, and Z directions for the fourth individual during mixed ventilation.

Sneezing is a powerful mechanism for the expulsion of infectious aerosols at

high velocity, significantly increasing the risk of rapid airborne virus transmission. Compared to coughing, sneezing generates a more powerful expulsion of particles, resulting in a faster and wider dispersion within an indoor environment. In this analysis, a sneezing velocity of 70 m/s was applied in Scenario 3, where mixed ventilation (consisting of NV through an open window in Zone 1 and AC through an AC unit in Zone 2) played a key role in influencing airflow dynamics and particle dispersion. This study focused on four infected individuals, analysing the spatial and temporal dispersion of particles along the X, Y, and Z directions to evaluate the impact of ventilation strategies on aerosol transmission.

Figure 7.25 illustrates the spatial distribution of aerosols expelled by the first infected individual during sneezing. Compared to coughing, sneezing results in a higher concentration of particles that disperse more widely due to the increased initial velocity. The particles exhibit a strong forward trajectory before interacting with the surrounding airflow, leading to a notable dispersion pattern.



Figure 7.25: Particle transmission from the first infected individual during sneezing.

Figure 7.26 illustrates the aerosol dispersion along the X, Y, and Z directions during the sneezing event for the first individual.
As shown in Figure 7.26-(a), the particles initially experience a rapid displacement along the X axis, reaching distances between 15 m and 16.75 m within the observed time frame (indicating a transmission distance of approximately 1.75 m). Unlike coughing, where the particles tended to remain more localised, sneezing results in a more extensive spread, suggesting that the high initial velocity enables the particles to travel further before settling.

Figure 7.26-(b) presents the lateral spread of particles along the Y axis. The results indicate that the particles disperse within a range of 9.5 m to 14 m (indicating a transmission distance of approximately 4.5 m), highlighting a significant horizontal spread due to airflow interactions. This is particularly important because it shows the potential for cross-contamination among multiple occupants located along the lateral airflow pathways.

Figure 7.26-(c) illustrates the vertical distribution of aerosolised particles along the Z axis. The results show that the particles remain largely between 0.5 m and 2.5 m (indicating a transmission distance of approximately 2 m), indicating a strong presence within the typical breathing zone of nearby occupants. Unlike coughing, where the spread was more contained, sneezing results in a more unpredictable vertical spread, increasing the potential for airborne transmission.



Figure 7.26: Aerosol dispersion in X, Y, and Z directions for the first individual during mixed ventilation.

Figure 7.27 presents the spatial distribution of aerosols expelled from the second infected individual. The particles initially travel in a concentrated forward motion before dispersing as a result of the interaction with indoor airflow. Unlike the first infected individual, where the particles spread rapidly toward the door due to strong NV, the dispersion of the second infected individual is more contained within the zone due to its location farther from the window and AC unit.



Figure 7.27: Particle transmission from the second infected individual during sneezing.

Figure 7.28 illustrates the aerosol dispersion along the X, Y, and Z directions during the sneezing event for the second individual.

As illustrated in Figure 7.28-(a), the particles exhibit an initial rapid displacement in the X direction due to the high forward velocity of the sneeze. However, their movement remains restricted within the range of approximately 12.0 m to 12.8 m after 7.9 seconds (indicating a transmission distance of approximately 0.8 m). This is likely due to the central positioning of the second infected individual, where neither the window nor the AC unit fully directs the airflow, reducing the lateral spread.

Figure 7.28-(b) depicts the lateral movement of the particles in the Y direction. The dispersion is relatively limited, with particle displacement values ranging between 8.8 m and 9.2 m after 7.9 seconds (indicating a transmission distance of approximately 0.4 m). This is likely due to the central positioning of the second infected individual, where neither the window nor the AC unit fully directs the airflow, reducing the lateral spread.

The vertical distribution of the expelled aerosols, shown in Figure 7.28-(c), indicates that the particles remain predominantly within 1 m to 1.5 m after 7.9 seconds (indicating a transmission distance of approximately 0.5 m). This range corresponds to the breathing zone of most occupants, highlighting a high airborne virus transmission risk in this area. Unlike the first infected individual, whose particles increased more due to strong vertical airflow near the window, the spread for the second infected individual is more restricted in the vertical direction.



Figure 7.28: Aerosol dispersion in X, Y, and Z directions for the second individual during mixed ventilation.

Figure 7.29 illustrates the overall spatial distribution of aerosols expelled by the third infected individual. The results indicate that the particles remained relatively concentrated around the source because of the location of the individual, which was far from the direct airflow pathways generated by the open window and the AC. Most of the particles dispersed locally, with some lateral spread occurring due to weak indoor air currents.

#### 7.5 Results



Figure 7.29: Particle transmission from the third infected individual during sneezing..

Figure 7.30 illustrates the aerosol dispersion along the X, Y, and Z directions during the sneezing event for the third individual.

Figure 7.30-(a) shows the movement of the particles in the X direction over time. The aerosols initially exhibit strong forward motion due to the high velocity of the sneeze. However, after approximately 7.9 seconds, the dispersion becomes limited, with particles remaining within a range of 8 m to 8.6 m (indicating a transmission distance of approximately 0.6 m). This indicates that in the absence of strong directional airflow, the particles settle within a confined region.

The lateral dispersion of the particles in the Y direction is depicted in Figure 7.30-(b). The results show that aerosols remain within 5.3 m to 6.2 m (indicating a transmission distance of approximately 0.9 m), indicating a moderate spread. Unlike infections closer to airflow sources, the particles here exhibit a more localised dispersion pattern, suggesting minimal influence from cross-zone air currents.

Figure 7.30-(c) represents the vertical dispersion of particles. The results indicate that most aerosols remain within 0.5 m to 1.25 m (indicating a transmission

distance of approximately 0.75 m), primarily within the breathing zone of the occupants. The containment of particles within this critical zone highlights the potential risk of airborne virus transmission in the immediate vicinity of the infected individual.



Figure 7.30: Aerosol dispersion in X, Y, and Z directions for the third individual during mixed ventilation.

Figure 7.31 illustrates the spatial distribution of particles expelled from the fourth individual's position. The results show that sneezing leads to a localised dispersion of aerosols with minimal spread beyond the immediate surroundings. Unlike scenarios where NV (open window) dominates, the airflow regulation of the AC system in Zone 2 appears to limit the extent of aerosol movement, which contains the majority of particles near the infected individual.



Figure 7.31: Particle transmission from the fourth infected individual during sneezing.

Figure 7.32 illustrates the aerosol dispersion along the X, Y, and Z directions during the sneezing event for the fourth individual.

As depicted in Figure 7.32-(a), the displacement in the X direction remains highly constrained. After 7.9 seconds, expelled particles remain within a short range of 1.1 m to 1.2 m (indicating a transmission distance of approximately 0.1 m), suggesting low forward momentum due to airflow stabilisation by the AC system. Compared to infections in Zone 1, where NV significantly increased aerosol reach, controlled airflow in Zone 2 reduces forward travel.

Figure 7.32-(b) shows the movement in the Y direction, where the particles initially exhibit a slight lateral spread but remain between 10.2 m and 10.6 m (indicating a transmission distance of approximately 0.4 m). This suggests that while there is some lateral dispersion, it is significantly less pronounced than in NV areas, where the particles spread widely. The AC system appears to regulate the horizontal flow, preventing excessive spread.

Figure 7.32-(c) demonstrates the vertical movement (Z) of the expelled particles.

The aerosols remain between 1.35 m and 1.5 m after 7.9 seconds (indicating a transmission distance of approximately 0.15 m), aligning closely with the typical breathing zone of the occupants. This implies that although the AC system effectively limits the overall dispersion, residual aerosols still accumulate at a height where the risk of inhalation remains high.



**Figure 7.32:** Aerosol dispersion in X, Y, and Z directions for the fourth individual during mixed ventilation.

### 7.6 Discussion

This section provides a comprehensive discussion of the results obtained from the three mixing ventilation scenarios, focussing on airflow patterns, IAQ, room temperature, RH, TC indices (PMV and PPD), and aerosol dispersion. The findings are compared with the ASHRAE, EN 16798, and ISO 7730 standards for acceptable indoor conditions to identify optimal and suboptimal configurations and suggest recommendations for improved ventilation performance.

## 7.6.1 Comparison with ASHRAE, EN 16798, and ISO 7730 Standards

The mixed-mode ventilation scenarios were evaluated against the TC criteria outlined in ASHRAE 55, ISO 7730, and EN 16798. These standards define acceptable PMV and PPD to ensure a comfortable indoor environment. ASHRAE 55 specifies an occupancy satisfaction level of 80%, corresponding to a PMV range of -0.5 to +0.5 and a PPD below 20%, while ISO 7730 and EN 16798 classify comfort into three categories:

- Category A (ISO)/I (EN): PMV within  $\pm 0.2$  and PPD < 6% (high expectations)
- Category B (ISO)/II (EN): PMV within  $\pm 0.5$  and PPD < 10% (normal expectations)
- Category C (ISO)/III (EN): PMV within ±0.7 and PPD < 15% (moderate comfort)

TC performance varied between different mixed mode configurations, some scenarios successfully maintaining recommended conditions, while others presented deviations due to poor air distribution and excessive RH.

Scenario 1 exhibited significant deviations from the ASHRAE TC criteria. PMV values ranged from -2 to -3, indicating extreme cold discomfort. This resulted in PPD values exceeding 50%, meaning that more than half of the occupants experienced thermal discomfort, which is far outside the ASHRAE and ISO recommended limits.

The room temperature dropped below 18 °C, which does not meet the recommended comfort range of ASHRAE 55 of 22 °C to 26 °C. Furthermore, RH levels exceeded 80%, exceeding the ASHRAE guideline of 30% to 0%. High levels of RH increase the risks of microbial growth and improve the stability of airborne viruses, further degrading IAQ.

The air velocity was highly uneven, with some areas experiencing excessive cold draughts due to open windows, while other sections had stagnant air pockets. The recommended indoor airflow velocity of ASHRAE is 0.1 m/s to 0.25 m/s,

but in Scenario 1, the cold outdoor air created high local speeds, exacerbating discomfort in exposed areas.

Scenario 2 demonstrated a significant improvement over Scenario 1, with PMV values ranging from -1 to 0.6 in Zone 1 and 0 to 1.6 in Zones 2 and 3. This suggests that TC was maintained largely at the occupant level, keeping PPD values below 10%, aligned with ASHRAE 55, ISO Category B, and EN Category II.

However, at higher elevations, the PMV values increased to 1.6 and the PPD exceeded 20%, indicating thermal stratification. This suggests that warm air accumulated at higher levels due to insufficient air mixing, leading to comfort inconsistencies between vertical zones.

RH exceeded 65%, exceeding the optimal ASHRAE range of 30% to 60%. Elevated RH can increase the stability of airborne viruses and create conditions conducive to mould growth, leading to degradation of IAQ. This issue suggests a need for enhanced dehumidification controls to maintain compliance with ASHRAE recommendations.

Scenario 3 exhibited the most stable and balanced conditions, maintaining PMV values within the recommended range of ASHRAE (-0.5 to +0.5) in all zones. This indicates that the thermal conditions remained relatively stable, ensuring compliance with the comfort levels of ISO Category B / EN Category II.

The temperature was maintained between 20 °C and 22 °C, fully in accordance with the recommendations of ASHRAE 55. Furthermore, RH was kept within the optimal 40% to 60% range, ensuring a better IAQ compared to Scenario 1 and Scenario 2.

The air velocity was well controlled and most areas were within the recommended range of 0.1 m/s to 0.25 m/s of ASHRAE. Unlike the previous scenarios, Scenario 3 had fewer stagnant zones and better airflow uniformity, reducing the risk of

uneven thermal conditions or localised air stagnation.

## 7.6.2 Identification of Dead Zones in Mixed-mode Ventilation

Dead zones, defined as regions with minimal airflow (< 0.1 m/s), present a critical challenge in indoor air distribution, leading to poor air mixing, localised temperature imbalances and stagnant pollutant accumulation. According to ASHRAE Standard 62.1, adequate air circulation is essential to prevent air stagnation because stagnant zones can compromise TC and increase the occupant's exposure to airborne contaminants.

Across the three mixed-mode ventilation scenarios, dead zones were consistently identified in Zones 2 and 3, particularly in areas shielded from primary airflow pathways, such as room corners, spaces far from ventilation inlets, and regions obstructed by furniture. These stagnant regions exhibited temperature stratification, elevated levels of HR (> 70%), and reduced dilution efficiency of indoor pollutants.

Scenario 1 (winter conditions, outdoor temperature  $4 \,^{\circ}$ C) exhibited significant dead zones, particularly in Zones 2 and 3, where the air velocity remained below 0.1 m/s. The lack of a sufficient airflow distribution of the AC exacerbated stagnation, leading to poor heat distribution, extreme cold stress (PMV = -2 to -3), and high dissatisfaction (PPD > 50%).

Furthermore, some areas in Zone 1 near the open window experienced an excessive air velocity ( 0.4 m/s) due to the direct influx of cold outside air. This resulted in localised discomfort, highlighting an imbalance in airflow, where certain areas received excessive ventilation, while others remained stagnant. These findings underscore the need for a more controlled integration of NV and AC to prevent extreme airflow discrepancies within occupied spaces.

Scenario 2 (winter conditions, outdoor temperature  $13^{\circ}$ C) showed notable improvements in airflow distribution, with an air velocity that reaches 0.08 m/s to 0.2 m/s in most areas. However, dead zones persisted in Zone 3 and room corners in Zone 1, where airflow remained below 0.1 m/s, indicating areas of inadequate ventilation.

A key factor contributing to the improved ventilation performance in Scenario 2 was the strategic placement of Window 4 in the middle section of Zone 1, which facilitated a more uniform airflow pattern. Unlike Scenario 1, where high-velocity airflow caused direct occupant discomfort, this centralised window placement helped distribute air more evenly, reducing thermal asymmetry.

Despite these improvements, Zone 3 remained stagnant due to its distance from both the AC unit (located in Zone 2) and the NV source (open window in Zone 1). This finding suggests that NV alone was not sufficient to induce adequate circulation in all zones, emphasising the need for AC reinforcements to improve air mixing.

Scenario 3 (summer conditions, outdoor temperature 19°C) exhibited the most balanced airflow distribution, with air velocity reaching 0.2 m/s in most areas and minimal dead zones. The combination of NV and AC, achieved through an open window in Zone 1 (Window 6) and AC operation in Zone 2, effectively improved air circulation and pollutant dispersion.

Although minor dead zones persisted in Zone 3, where air velocity occasionally dropped below 0.1 m/s, this issue was less pronounced compared to previous scenarios. Furthermore, elevated airflow velocities ( 0.4 m/s) were recorded in seating areas near the open window, confirming that window positioning played a critical role in regulating airflow and TC.

Scenario 3 also demonstrated the most effective airflow management strategy, characterised by the following:

- Reduced air stagnation, ensuring consistent air mixing across all zones.
- Optimised temperature regulation within 20 °C to 23 °C, which aligns well with the recommendations of ASHRAE 55.
- RH levels maintained within the recommended range of ASHRAE (30% top 60%), reducing the risk of stability of airborne viruses and mould growth.

#### 7.6.3 Impact of Air Velocity on Virus Dispersion

Air velocity plays a critical role in determining the dispersion, transport, and retention time of airborne contaminants, particularly in enclosed office environments. In this study, CFD simulations analysed aerosol dispersion during coughing and sneezing events under Scenario 3 (mixed-mode ventilation), where both NV (Window 4) and AC (an AC unit in Zone 1) were active. The results demonstrated that mixed ventilation significantly altered the trajectory, containment, and dilution efficiency of expelled aerosol particles compared to NV or AC alone. This led to a more controlled dispersion pattern and improved ventilation performance, reducing overall transmission risk.

During simulated coughing and sneezing events, the particles were expelled at 11.8 m/s and 70 m/s, respectively, introducing a significant aerosol load into the indoor air. The airflow distribution was a key determinant of whether these particles remained suspended, were transported across zones, or settled more quickly. Near Window 4 in Zone 1, the first infected individual experienced extensive particle dispersion due to high air velocity exceeding 0.2 m/s. Strong natural air flow facilitated rapid aerosol transport across multiple occupants seated in front of the infected individual, increasing the potential for cross-zone transmission. However, unlike the NV scenario, the AC introduced a controlled air movement that aided in particle dilution, limiting prolonged airborne retention. Although NV alone led to a more unpredictable and widespread dispersion of particles, the combination of

AC-driven airflow and window-induced ventilation helped contain the movement of expelled aerosols more effectively in mixed ventilation.

In contrast, the second and third infected individuals, who are positioned centrally and in the left-hand corner of Zone 1, experienced more localised particle dispersion due to lower air velocities ( 0.2 m/s or less). The reduced not only airflow minimised the excessive horizontal spread but also prolonged the suspension of aerosols, increasing the risk of localised exposure. Compared to NV alone, where airflow patterns were inconsistent and aerosol dispersion was more widespread, mixed ventilation created a more structured and stable airflow, limiting erratic movement and containing particles more effectively.

In Zone 2, where Infected 4 was located, the absence of direct AC airflow led to lower air velocities ( 0.2 m/s or less), restricting the dispersion of the lateral aerosol. Although airflow from Zone 1 influenced some movement, particle dilution was considerably slower in this area, resulting in a longer retention time for airborne particles. Unlike Zone 1, where AC and window-driven airflow facilitated more dynamic movement and dilution, Zone 2's reduced air velocity contributed to prolonged aerosol suspension, potentially increasing the risk of exposure of the occupant.

Compared to NV alone, where airflow was often uncontrolled and highly dependent on outdoor wind conditions, mixed ventilation provided more structured airflow regulation, reducing unpredictable aerosol dispersion. It also improved faster aerosol dilution rates, particularly in Zone 1, where the combined effect of AC and natural airflow significantly improved particle clearance. Furthermore, mixed ventilation prevented excessive high-velocity dispersion near openings, which, in NV scenarios, increased transmission risks due to high-speed airflow spreading aerosols across multiple zones.

Compared to AC alone, mixed ventilation improved air exchange between zones, prevented recirculation of indoor air, and improved pollutant removal efficiency. The introduction of fresh outdoor air helped reduce the risk of accumulation of contaminants, which is often a limitation in fully AC systems. The results also indicated that better air movement uniformity in mixed ventilation helped eliminate stagnant air zones, further decreasing the likelihood of longer airborne virus suspension.

The findings of this study confirm that mixed ventilation provides the most balanced approach to controlling airborne contaminants because it reduces excessive dispersion while improving the efficiency of dilution. Unlike NV alone, which may introduce unpredictable airflow patterns, mixed ventilation helps control aerosol dispersion more effectively. In addition, it mitigates the limitations of AC, which often recirculates indoor air without introducing fresh air, thus increasing the risk of contaminant accumulation. By integrating strategic window placement, AC airflow regulation, and air filtration systems, mixed-mode ventilation can serve as an effective strategy for infection control in office environments.

### 7.6.4 Impact of Relative Humidity on Comfort and Airborne Virus Stability

RH is a critical factor that influences both TC of the occupant and the stability of airborne viruses, making it a key consideration in indoor environmental control. ASHRAE guidelines recommend maintaining RH between 40% and 60% to optimise comfort and minimise the persistence of airborne pathogens. The findings of the mixed-mode ventilation scenarios demonstrate notable variations in RH levels in different zones, with improved moisture regulation compared to NV alone but some persistent humidity imbalances in certain areas.

Scenario 1 exhibited elevated RH levels, particularly in zones with restricted airflow mixing. RH values at the occupant level ranged from 60% to 65% in Zones 1 and 2, while Zone 3 experienced levels exceeding 70%. The moisture

accumulation in Zone 3 can be attributed to low air velocity (<0.1 m/s), which hindered effective moisture dissipation. Furthermore, the RH above the level of the occupant exceeded 65%, indicating a potential moisture stratification in the upper sections of the office, which was exacerbated by the absence of strong AC in this zone.

Excessively high RH levels not only compromise TC but also create an environment that encourages the persistence of airborne viruses [338]. Research suggests that low air velocity contributes to prolonged aerosol suspension, particularly in humid conditions, thereby increasing the risk of airborne virus transmission [339]. These findings indicate that while mixed-mode ventilation introduced some airflow improvements, localised moisture stagnation remained a challenge in areas with inadequate air exchange.

With an increase in outdoor temperature (13 °C), Scenario 2 demonstrated better regulation of RH compared to Scenario 1. RH levels at the occupant level in Zones 1 and 2 ranged from 53% to 60%, gradually decreasing toward the ceiling. This improvement suggests that convective mixing, driven by the AC system, helped redistribute moisture more effectively, reducing excessive accumulation.

However, Zone 3 still exhibited RH levels between 60% and 70%, indicating weaker airflow exchange and limited ventilation-induced dehumidification. Unlike Scenario 1, where humidity accumulation was more widespread, Scenario 2 exhibited more balanced RH control due to increased thermal stratification effects, which helped disperse moisture more evenly across the space.

Scenario 3, representing summer conditions with an open window (Window 6) in Zone 1 and an AC unit operating at 19 °C in Zone 2, exhibited the most stable and well-balanced levels of RH in all zones. RH values at the occupant level ranged from 53% to 60%, remaining within the optimal ASHRAE comfort range. At levels above the occupant, HR remained consistently stable around 60%, demonstrating effective humidity regulation throughout the office space.

The success of Scenario 3 in maintaining RH levels can be attributed to the synergy between NV and AC. The open window in Zone 1 facilitated fresh air exchange, preventing excessive moisture buildup, while the AC unit in Zone 2 introduced conditioned air, reducing latent heat and promoting uniform moisture distribution. Furthermore, Zone 3, which previously exhibited excessive RH in Scenarios 1 and 2, benefited from improved ventilation dynamics, ensuring better moisture dilution and limiting localised humidity accumulation.

In summary, the findings highlight that while mixed-mode ventilation significantly improves RH control compared to NV alone, achieving a uniform humidity balance remains a challenge.

#### 7.6.5 Scenario Performance Evaluation

The performance of the mixed ventilation system was analysed based on air velocity, temperature, RH, TC indices (PMV and PPD), and overall ventilation effectiveness. These key parameters were evaluated to determine how well the integration of AC and NV maintained IAQ and TC under different environmental conditions.

Scenario 1, conducted under winter conditions (outdoor temperature = 4°C), exhibited the worst overall performance among the three cases. The air velocity remained extremely low (< 0.1 m/s) in most zones, leading to stagnant airflow and the formation of dead zones, particularly in Zones 2 and 3. In Zone 1, where an open window and an AC unit were present, air movement was slightly better but high air velocities near the open window (> 0.4 m/s) caused discomfort for occupants seated close to the airflow source.

The temperature distribution highlighted excessive cooling because the open window introduced unregulated cold air (4 °C), drastically lowering indoor temperatures. Due to the absence of a heating mechanism and the inability of AC to counterbalance these thermal losses, widespread discomfort was recorded. RH levels exceeded 80% in some areas, particularly in Zones 2 and 3, creating moisturerelated discomfort and increasing the potential for the stability of airborne viruses, a known risk in high-humidity environments.

TC analysis showed that PMV values ranged from -2 to -3, confirming that the occupants experienced severe cold discomfort. The PPD values exceeded 50%, with Zones 2 and 3 experiencing over 90% PPD, meaning that most of the occupants found the environment thermally unsatisfactory. These results indicate that mixed ventilation under extremely cold outdoor conditions was ineffective because NV introduced excessive cooling, while AC was insufficient to maintain thermal balance.

Scenario 2, conducted under milder winter conditions (outdoor temperature =  $13^{\circ}$ C), showed significant improvements compared to Scenario 1. The air velocity was better distributed (0 m/s to 0.2 m/s), the temperature regulation was more stable (20 °C to 22 °C), and the TC metrics improved significantly (PMV: -1 to 0.8). The improved balance between AC and NV reduced excessive cooling while allowing adequate air exchange.

However, RH levels remained elevated (65% to 75%), particularly in Zones 2 and 3, suggesting that while mixed ventilation improved airflow, it was insufficient for full humidity regulation. High humidity in enclosed spaces is associated with increased airborne virus stability and reduced occupant comfort, highlighting the need for dehumidification strategies or improved air mixing.

PPD values significantly improved (<10%), suggesting that most of the occupants were within acceptable comfort limits. However, above occupant height, PMV values slightly exceeded 0.6 and PPD increased beyond 20%, suggesting localised heat accumulation near the ceiling. These results imply that further adjustments to ventilation rates or enhanced air circulation mechanisms could reduce stratification and improve overall IAQ performance.

#### 7.6 Discussion

Scenario 3, conducted under summer conditions (outdoor temperature =  $19^{\circ}$ C), exhibited the best overall performance, with:

- Balanced air velocity (0.1 m/s to 0.2 m/s),
- Stable indoor temperatures (20 °C to 22 °C) within ASHRAE-recommended levels, and
- Optimal RH levels (53% to 60%), ensuring improved IAQ and reduced airborne virus stability.

PMV values remained near 0, confirming optimal TC, while the PPD values were consistently < 10%, indicating high occupant satisfaction. The coordinated operation of the AC unit in Zone 2 and the open window in Zone 1 facilitated stable temperature regulation, improved airflow distribution, and improved humidity control.

Unlike Scenario 1, where unregulated cold air infiltration led to excessive cooling and Scenario 2, where humidity levels remained problematic, Scenario 3 provided a well-balanced ventilation strategy. The findings emphasise that a well-designed mixed ventilation system, with controlled window placement and properly calibrated AC operation, is essential to achieve optimal IAQ and TC.

Scenario	Air Velocity	Temperature	RH	$\mathbf{PMV}$	PPD
1	< 0.02 - 0.4 m/s	$< 18 ^{\circ}\mathrm{C}$	> 70%	-2 to -3	> 70%
2	0.05 - 0.4 m/s	20 °C - 22 °C	65%-75%	-2 to 1.2	> 50%
3	0.05 - 0.4 m/s	21 °C - 23 °C	50% - 65%	-0.5 - 1	10% - 20%

 Table 7.3: Comparative summary of scenario performance

# 7.7 Recommendations for Optimising Mixed-mode Ventilation Performance

Mixed-mode ventilation, which integrates both NV and AC, demonstrated superior performance in this study using the strengths of both systems while mitigating their individual weaknesses. However, to fully optimise its performance, several enhancements should be considered in terms of system control, airflow management, and humidity regulation.

One of the key advantages of mixed-mode ventilation is its ability to dynamically switch between NV and AC modes based on real-time environmental conditions. To maximise this benefit, it is recommended to implement an intelligent ventilation control system that monitors indoor temperature, humidity, CO2 levels, and occupancy patterns. These systems can automatically adjust window openings and humidity control mechanisms to maintain optimal IAQ and TC, while minimising virus transmission.

Airflow balance is crucial in mixed-mode systems, as improper coordination between NV and AC components can lead to excessive ventilation in some zones, while leaving others under-ventilated. This study found that areas near open windows sometimes experienced excessive air velocity (> 0.4 m/s), creating localised discomfort and increasing the risk of viral dispersion. To prevent this, a well-designed airflow distribution strategy should be employed, ensuring that the fresh air intake from NV complements, rather than competes with, the AC air supply. Adjusting the placement of the AC unit and the direction of airflow to work in tandem with open windows can enhance air mixing and prevent thermal stratification.

Humidity control was another challenge observed in mixed-mode ventilation scenarios. Although AC can regulate moisture levels, NV can introduce excess humidity when outdoor RH is high. This study found that RH levels in certain zones occasionally exceeded 70%, which could contribute to the increased stability of the virus and the discomfort of the occupants. To mitigate this, it is recommended to integrate dehumidification units in areas prone to high moisture levels. In addition, automated window controls that close openings when outdoor RH exceeds indoor thresholds can help maintain balanced humidity levels.

Finally, occupant engagement and adaptability are key to maximising the effectiveness of mixed-mode ventilation. Providing building users with real-time feedback on ventilation performance through IAQ monitoring displays or smart building interfaces can encourage optimal window operation and ventilation behaviour. Educating occupants about when to rely on NV versus AC assistance can further enhance the overall performance of the system.

By implementing these recommendations, mixed-mode ventilation can provide an adaptive, efficient, and health-conscious approach to IAQ management, ensuring optimal airflow, stable humidity levels, and superior occupant comfort under diverse environmental conditions.

### 7.8 Conclusion

This chapter evaluated the effectiveness of mixed-mode ventilation in optimising IAQ, TC, and airborne infection control through CFD simulations. This study examined airflow patterns, temperature regulation, humidity control, TC of the occupant (PMV and PPD), and aerosol dispersion in three distinct hybrid ventilation scenarios. The findings demonstrated that strategically integrating NV and AC can improve air circulation, maintain thermal stability, and reduce airborne virus transmission risks compared to fully AC or fully NV.

The limitations identified in the previous chapters were a key motivation for this study. Chapter 5 (AC system) revealed that fully AC was insufficient in optimising IAQ in all zones because it resulted in stagnation areas and uneven airflow distribution. In contrast, Chapter 6 (NV) showed that fully NV was ineffective in maintaining stable indoor temperatures, leading to thermal discomfort and increased airborne particle transmission. These findings necessitated an investigation into hybrid ventilation systems as a more balanced alternative.

The results indicate that mixed ventilation significantly improved airflow uniformity, reducing stagnation zones while maintaining controlled air circulation. The introduction of AC helped stabilise IAQ and minimise excessive temperature fluctuations that were prevalent in purely NV scenarios. Among the three cases studied, Scenario 3 (summer conditions with an AC in Zone 2 and a window in Zone 1) demonstrated the most balanced performance, providing optimal air distribution, IAQ, and thermal stability. Scenario 2 (winter conditions with moderate outdoor temperatures) performed well but exhibited humidity fluctuations that require further refinement. In contrast, Scenario 1 (cold winter conditions) was the least effective because cold air infiltration and insufficient heating resulted in substantial thermal discomfort.

Regarding virus dispersion, mixed ventilation minimised airborne virus transmission risks by improving air circulation, reducing aerosol retention time, and ensuring that particles were effectively diluted and removed. Compared to fully NV, the combination of AC and natural airflow reduced localised viral accumulation, mitigating the risk of prolonged exposure of the occupant. However, the findings also highlight the importance of ventilation placement because an asymmetric airflow distribution may lead to microclimates within the space, affecting occupant TC and localised IAQ.

This study contributes to existing knowledge by demonstrating that mixed ventilation can provide a more adaptable and effective solution to maintain IAQ and TC in open-plan offices. Previous research has analysed hybrid ventilation in controlled environments, but this study systematically evaluates its performance under real-world seasonal variations. The findings reinforce that adaptive control mechanisms and airflow balancing strategies are essential to maximise the benefits of hybrid ventilation.

Although this study provides a comprehensive evaluation of mixed ventilation performance, certain limitations should be addressed in future research. First, real-time occupant interactions and their impact on airflow dynamics were not included in the CFD simulations, which could introduce additional variability in practical applications. Second, while airflow and IAQ parameters were analysed, energy consumption comparisons were not explicitly evaluated, which remains a crucial consideration for large-scale implementation. Finally, further investigations should explore advanced ventilation control strategies, such as demandcontrolled hybrid ventilation, to optimise IAQ while mitigating or minimising virus transmission.

In direct response to Research Question 3 (RQ3), the results demonstrate that mixed-mode ventilation provides an effective solution for improving indoor air quality, ensuring thermal comfort, and reducing airborne infection risks. By combining AC system and natural ventilation, the system overcomes limitations observed in standalone AC and NV strategies such as stagnation zones, thermal discomfort, and uncontrolled aerosol dispersion.

**Regarding Research Question 4 (RQ4)**, this chapter highlights how hybrid ventilation can be configured and adapted under varying seasonal and environmental conditions to optimise indoor environmental quality. Specifically, the configuration used in Scenario 3 (summer conditions with AC and window ventilation) was found to be the most effective, offering a balanced approach to airflow regulation, temperature control, and aerosol dilution, thereby supporting a more resilient and adaptable ventilation framework.

This chapter contributes to the growing body of research on hybrid ventilation by providing a detailed computational evaluation of mixed-mode strategies in a realworld office context. Through CFD simulations under seasonal conditions, the chapter identifies the optimal configurations that balance airflow, temperature regulation, humidity control, and airborne infection risk mitigation. The findings extend the understanding of how hybrid systems can address the limitations of standalone AC and NV solutions, offering practical guidance for designing energyefficient, health-conscious ventilation strategies in modern workspaces.

In conclusion, Scenario 3 demonstrated the best overall performance, confirming that a well-designed mixed-ventilation strategy can significantly improve IAQ, reduce airborne virus transmission risks, and ensure occupant TC. These findings provide valuable information for the design and optimisation of hybrid ventilation systems, supporting the development of healthier and more efficient indoor environments. Future research should focus on adaptive ventilation strategies that dynamically respond to environmental conditions and occupancy levels, further enhancing the practicality and efficiency of mixed-mode ventilation systems.

# Chapter 8

# Conclusion

This thesis has systematically investigated the effectiveness of different ventilation strategies (i.e., air conditioning (AC), natural ventilation (NV), and mixedmode ventilation) in controlling indoor air quality (IAQ), thermal comfort (TC), and airborne infection risks in an open-plan office environment. By integrating Computational Fluid Dynamics (CFD) simulations with real-world environmental monitoring, this research has provided critical insights into airflow dynamics, pollutant dispersion, and thermal regulation under varying ventilation conditions. The findings offer practical guidance for designing healthier indoor environments, particularly in the context of post-pandemic ventilation strategies.

# 8.1 Revisiting Research Questions and Key Findings

This research aimed to answer four key research questions that guided the methodological approach and analysis of the study, particularly with respect to ventilation performance, thermal comfort, and the risk of airborne virus transmission. Each research question is addressed in detail below, linking the findings of the relevant chapters and ensuring the implications for viral infection mitigation are clearly articulated.

# 8.1.1 Research Question 1: How do viral particles transmit within indoor environments and what factors influence their transmission dynamics?

The first research question was explored in Chapter 2 through a systematic review of the literature, which highlighted the key mechanisms influencing airborne transmission of respiratory viruses. The findings underscore the complexity of airborne virus transmission in indoor environments, with aerosol dispersion being influenced by factors such as air circulation, humidity levels, temperature, and ventilation effectiveness.

The review demonstrated that respiratory viruses can remain viable in airborne particles for extended periods, with transmission risks increasing in poorly ventilated spaces. Evidence suggests that inadequate ventilation leads to higher aerosol retention times, increasing the risk of exposure to occupants. Computational studies further revealed that airflow patterns play a critical role in viral dispersion, with high-velocity air currents potentially exacerbating the spread of infectious aerosols. The findings confirmed the necessity of strategic airflow management to prevent stagnation zones, where viral aerosols may accumulate and persist.

This study highlights the importance of integrating ventilation strategies that balance air distribution with effective pollutant removal. The use of CFD simulations has proven valuable in identifying high-risk areas for viral accumulation, reinforcing the role of computational modelling in optimising IAQ management strategies. These insights underscore the necessity of airflow optimisation not only for comfort but also for reducing airborne infection risk, highlighting that viral transmission must be considered a central factor in indoor ventilation design.

# 8.1.2 Research Question 2: Can CO2 levels serve as reliable indicators to assess IAQ and signal the potential risk of airborne infections in office environments?

Chapter 4 addressed the second research question by evaluating the correlation between CO2 levels and IAQ parameters in the teaching office environment. The analysis revealed that CO2 concentrations consistently exceeded the recommended threshold of 700 ppm during peak occupancy periods, aligning with increased risks of airborne infection due to poor ventilation.

The study confirmed that CO2 serves as a useful proxy for ventilation effectiveness, as high concentrations indicate inadequate air exchange. This finding supports the adoption of real-time CO2 monitoring as an early warning system for ventilation deficiencies. Since elevated CO2 levels often correlate with exhaled breath accumulation, they can serve as indirect indicators of increased viral particle concentration, further reinforcing their value in infection risk monitoring.

However, the study also acknowledged the limitations in using CO2 as an independent IAQ indicator. While elevated CO2 levels suggest insufficient ventilation, they do not directly measure pathogen concentrations or pollutant accumulation. As a result, a multiparameter approach is recommended, incorporating temperature, humidity, and particulate matter monitoring, to provide a comprehensive IAQ assessment.

## 8.1.3 Research Question 3: What ventilation strategies can be implemented to minimise the risk of airborne infections in office environments?

This question was comprehensively addressed in Chapters 5, 6, and 7, where three ventilation systems AC, NV, and mixed-mode ventilation were comparatively

evaluated.

Chapter 5 established the baseline performance of a typical four-way ceiling cassette AC system. Although maintaining acceptable temperature ranges and moderate airflow in the core zones, it also revealed significant limitations, including stagnant zone formation, uneven air distribution, and indoor air recirculation, which elevated the risk of aerosol accumulation and cross-contamination.

Chapter 6 expanded the analysis to NV systems. While NV particularly CV proved highly effective in pollutant dilution and increasing air change rates under favourable conditions, it was hindered by outdoor environmental variability. Low air exchange during stagnant weather undermined its reliability. Nonetheless, NV emerged as a viable infection mitigation tool when conditions permitted, especially in buildings with operable windows and adaptive use protocols.

Finally, Chapter 7 introduced mixed-mode (hybrid) ventilation as a balanced solution. By combining the strengths of NV and AC, this strategy demonstrated improved airflow uniformity, enhanced pollutant removal, and greater control over temperature and humidity. Scenario 3 (AC in Zone 2 + open window in Zone 1) emerged as the most effective configuration in terms of IAQ, thermal comfort, and virus mitigation metrics. This chapter confirmed that integrated ventilation strategies, rather than isolated systems, are better suited to support occupant health in dynamic indoor settings. The comparative analysis confirmed that engineering solutions must prioritise not only air distribution and comfort but also targeted virus mitigation strategies to reduce transmission risk in real-world conditions.

# 8.1.4 Research Question 4: How can ventilation systems be configured to maintain a good IAQ, ensure TC, and reduce virus dispersion in the office environment?

Research Question 4 focusses on the configuration and optimisation of ventilation systems and was explored primarily through Chapters 6 and 7. These chapters went beyond evaluation by simulating different configurations, airflow placements, and seasonal variations to identify actionable setup strategies.

Chapter 6 illustrated how NV strategies such as cross-ventilation can be spatially and temporally adjusted to meet seasonal IAQ needs. The analysis demonstrated that strategic window opening configurations, adjusted WWR, and placement of occupants relative to openings significantly impacted both IAQ and thermal satisfaction. However, it also highlighted that relying solely on NV introduces uncontrolled variability, which limits its standalone applicability.

Chapter 7 addressed these shortcomings by configuring mixed-mode systems to deliver controlled and stable performance in both summer and winter conditions. Among the three setups tested, Scenario 3 was the most effective due to its balance of air distribution, thermal regulation, and humidity control. These findings provide strong evidence that hybrid systems, when properly configured, offer the most promising pathway to achieve both infection risk reduction and occupant comfort.

The chapter concluded with design recommendations that include optimising airflow directions, integrating smart control systems, and continuously monitoring IAQ metrics to dynamically respond to occupancy and weather changes. Ultimately, the optimal configuration of ventilation systems must integrate virus control measures, such as minimising particle suspension and enhancing pollutant removal pathways, particularly in shared spaces with fluctuating occupancy. These configuration strategies directly respond to the requirements of Research Question 4.

# 8.2 Implications for Building Design and Operation

The findings of this research have critical implications for current and future practices in the Architecture, Engineering, and Construction (AEC) sector. The increasing demand for resilient, health-focused, and energy-efficient indoor environments—especially in the post-pandemic era—calls for scalable and adaptable ventilation strategies. This study demonstrates that mixed-mode ventilation offers a robust solution by combining the control of mechanical systems with the flexibility of natural ventilation. Although this research focused on a teaching office at Cardiff University, the underlying methodology and simulation framework are readily transferable to other building typologies, including schools, hospitals, and commercial spaces, each with unique occupancy profiles and ventilation challenges.

For building designers, engineers, and facility managers, this research underscores the importance of prioritising flexibility in ventilation design. Strategies such as integrating operable windows, variable dampers, demand-controlled systems, and ceiling fans can help tailor airflow patterns to seasonal and occupancy variations. These configurations also support adaptability across diverse regional climates and architectural forms.

Moreover, real-time monitoring using advanced sensor networks for CO2, temperature, humidity, and particulate matter is essential for dynamically assessing IAQ and adjusting ventilation in response to actual conditions. The incorporation of smart, data-driven control systems enables buildings to respond more efficiently to occupant presence and environmental changes, paving the way for truly intelligent indoor environments.

From a human-centric perspective, the research highlights the integral role of occupant comfort, wellbeing, and behavioural adaptation in the success of ventilation strategies. Thermal satisfaction, perceived air quality, and trust in the system directly affect acceptance and use. Therefore, future implementations must go beyond technical performance and consider psychological and social dimensions—ensuring systems are intuitive, responsive, and user-friendly.

Finally, these findings support broader sustainability and public health goals. By promoting energy-efficient solutions that also reduce airborne infection risk, this research contributes to emerging standards for healthier indoor environments. It encourages the AEC sector to adopt holistic, resilient approaches that integrate environmental responsibility, socio-cultural responsiveness, and economic viability—fostering healthier, more inclusive, and adaptable buildings for diverse populations and future challenges.

# 8.3 Comparison of Mixed-mode Ventilation Against Natural Ventilation and Air Conditioning

Mixed-mode ventilation combines NV and AC to leverage the strengths of both systems, while mitigating their individual weaknesses. This approach aims to improve airflow distribution, stabilise temperature fluctuations, regulate humidity, and improve overall IAQ. The results of this study indicate that mixed-mode ventilation outperformed both NV and the AC system in achieving optimal indoor environmental conditions.

In terms of airflow distribution, mixed-mode ventilation provided a more balanced and effective air movement pattern, reducing stagnation zones that were prevalent in both NV and AC scenarios. In Scenario 3, which was conducted under summer conditions, air velocity remained stable between 0.1 m/s and 0.2 m/s, significantly minimising dead zones in Zones 2 and 3. In contrast, NV alone resulted in highly variable airflow patterns, with certain areas experiencing excessive air movement while others suffered from stagnation. While the AC system was more consistent, it showed stagnant zones in corners and areas blocked by furniture, limiting air mixing. The combination of AC and NV in mixed mode scenarios improved airflow uniformity and reduced localised ventilation deficiencies.

Temperature regulation was also more stable under mixed-mode ventilation. The combined use of NV and AC cooling prevented the overheating observed in NV scenarios while eliminating the over-cooling effects present in AC-only settings. Scenario 3 demonstrated the most balanced temperature range, maintaining indoor conditions between 20 °C and 23 °C, which is well within the recommended levels of ASHRAE. In comparison, NV often caused temperature variations in different zones, while AC created cold air pockets near air supply vents. The PMV values in mixed-mode ventilation remained close to 0, confirming an optimal balance of TC, whereas NV exhibited PMV values exceeding 0.5 in stagnant zones, and AC occasionally caused over-cooling discomfort.

Humidity control was another area where mixed-mode ventilation proved to be superior. Unlike NV, which allowed RH to fluctuate with outdoor conditions, and AC, which lacked sufficient dehumidification in certain areas, mixed-mode ventilation effectively regulated indoor moisture levels. In Scenario 3, RH remained between 53% and 60%, which is well within the comfort recommendations of ASHRAE. This was a significant improvement over NV, where RH frequently exceeded 65% and AC, sometimes resulting in elevated indoor moisture levels. The introduction of fresh air through NV helped dilute indoor contaminants, while AC contributed to stable humidity regulation.

In general, mixed-mode ventilation achieved a more balanced approach to IAQ management, outperforming both NV and AC in airflow distribution, temperature

regulation, and humidity control. It effectively mitigated the stagnation issues associated with NV while addressing the air recirculation concerns of AC. This hybrid approach ensures a healthier and more comfortable indoor environment by maintaining optimal ventilation effectiveness, stable thermal conditions, and regulated humidity levels.

To further optimise mixed-mode ventilation, adaptive control systems that adjust window openings and AC airflow based on real-time IAQ and TC metrics could be implemented. Furthermore, strategic placement of windows and vents could enhance cross-ventilation while preventing excessive airflow near occupants. Integrating dehumidification systems would further stabilise RH levels, particularly in humid climates. The use of air filtration, such as HEPA or MERV filters, within the AC system could also help reduce indoor pollutant concentrations, improving air quality while maintaining energy efficiency.

In conclusion, mixed-mode ventilation presents the most effective solution to achieve optimal IAQ and TC in indoor environments. By integrating NV and AC, this approach maximises the benefits of both systems, providing superior airflow distribution, temperature stability, and humidity control. Its ability to regulate fresh air intake while ensuring controlled air circulation makes it a highly suitable strategy for improving indoor environmental conditions and occupant well-being.

### 8.4 Limitations

While this research has produced valuable insights into the performance of AC, NV, and mixed-mode ventilation strategies in office settings, several limitations should be acknowledged:

• Static occupancy modelling: The CFD simulations assumed static occupants and fixed heat loads. In real-world scenarios, occupant movement, varied breathing rates, and behavioural diversity can significantly influence airflow and pollutant dispersion. This simplification may affect the accuracy of dynamic exposure assessments.

- Single building type and location: The study was confined to a teaching office at Cardiff University. While the seasonal scenarios enhance contextual realism, the findings may not directly generalise to other building typologies, such as schools, hospitals, or commercial centres, which have different ventilation demands and occupancy patterns.
- Limited validation scope: Although the model was validated against literature and monitoring data, the availability of high-resolution spatiotemporal field data limited deeper validation for transient airflow or virus concentration distributions.
- No direct pathogen modelling: The study relied on proxy indicators like CO<sub>2</sub> and PM, but did not simulate specific viral particle survivability or viability under varying humidity or UV light levels.
- Exclusion of energy consumption modelling: While thermal comfort and IAQ were addressed, the energy demands of ventilation configurations

   — especially mixed-mode systems — were not evaluated, limiting sustainability insights.

### 8.5 Future Research Directions

Building upon the limitations and findings of this thesis, several future research avenues are proposed:

• Dynamic occupancy and behaviour simulation: Future studies should incorporate dynamic occupancy schedules, metabolic rate variations, and

movement pathways using real-time sensors or agent-based models to improve the realism of IAQ predictions.

- Multi-building, multi-climate assessments: Expanding the methodology to diverse building types (e.g., classrooms, healthcare facilities) and climate zones will help generalise the findings and guide location-specific ventilation strategies.
- Energy and cost performance integration: Future work should assess the energy efficiency and lifecycle cost of hybrid ventilation systems using building energy simulation tools, allowing trade-off analyses between IAQ, comfort, and energy use.
- Integration of viral viability models: Coupling CFD with virologyinformed decay or inactivation models (e.g., based on RH, temperature, and UV exposure) can improve predictions of infection risk and pathogen persistence.
- Development of AI-enabled smart ventilation systems: Leveraging machine learning and IoT sensor networks can enable real-time optimisation of ventilation settings, adapting to occupancy, IAQ trends, and outdoor conditions dynamically.
- Field studies and human-centred validation: Real-world implementation of mixed-mode systems with occupant feedback, sensor logging, and health data would provide holistic validation and practical guidance for scalable deployment.
- Broader adaptability and impact assessments: Future research should explore environmental sustainability (e.g., carbon footprint and energy consumption of ventilation systems), socio-cultural influences (e.g., local behaviours and acceptance of natural ventilation), and economic implications (e.g., cost-benefit analyses of hybrid systems). These aspects are critical for

ensuring scalable, inclusive, and policy-relevant applications across varied regional and building contexts.

### 8.6 Final Reflections

Although this thesis focused on a specific office setting at Cardiff University, the methodology and simulation framework were designed to explore ventilation strategies, particularly natural and mixed-mode ventilation, across multiple seasonal conditions. By evaluating performance under both winter and summer scenarios, the research demonstrates how environmental context influences airflow dynamics, thermal comfort, and infection risk. While the results are locationspecific, the broader modelling approach and analytical methods are adaptable and can be applied to other building types, climates, or occupancy patterns. This adaptability makes the study a valuable reference for future ventilation research and for practitioners aiming to optimise IAQ and occupant comfort in varied settings.

Crucially, this research confirms that indoor air quality, thermal comfort, and airborne virus transmission are deeply interconnected. Poor ventilation not only degrades comfort but also increases the concentration and persistence of airborne pathogens. The integrated evaluation of these three factors is therefore essential for designing safe, comfortable, and health-resilient indoor environments.

This thesis makes several significant contributions to the field of indoor environmental quality and ventilation engineering. First, it provides a comprehensive CFD-based framework for assessing the impact of various ventilation strategies AC systems, natural ventilation, and mixed-mode ventilation on indoor air quality, thermal comfort, and airborne infection risk. Second, the study presents an integrated methodology that combines real-world monitoring with advanced numerical simulations to evaluate performance across realistic seasonal conditions. Third, it introduces scenario-based evaluations that highlight both the strengths and limitations of each strategy, offering practical insights for optimising ventilation design in office environments. Finally, the thesis contributes to current knowledge by demonstrating the effectiveness of hybrid ventilation systems in achieving a balanced approach to infection mitigation and occupant comfort, laying the groundwork for future smart, adaptive ventilation systems in postpandemic building design.

A comparative analysis across the chapters revealed that no single ventilation strategy is universally optimal. AC systems provide thermal stability but often lead to stagnant zones and recirculated air, reducing ventilation efficiency. NV enhances air change and pollutant dilution but lacks climate control and consistency. Mixed-mode ventilation emerged as the most balanced solution especially in Scenario 3, which is successfully combining natural airflow with AC to maintain stable IAQ, reduce airborne transmission risk, and support thermal comfort. The study also identified several optimisation strategies: controlling window opening ratios, maintaining air velocity below 0.2 m/s to limit aerosol spread, managing RH under 60%, and placing AC outlets to complement natural airflow paths. These findings form a practical framework for tailoring ventilation setups to diverse indoor environments.

This thesis has demonstrated that mixed-mode ventilation provides a viable and effective strategy to optimise IAQ, improve thermal comfort, and mitigate the risks of airborne infection in office environments. By integrating natural ventilation with AC, this research contributes to the ongoing discourse on sustainable and health-focused building design. The findings advocate for a paradigm shift in ventilation strategies, emphasising adaptability, real-time IAQ monitoring, and energy-efficient solutions.

As urban environments continue to evolve, future research should focus on developing intelligent and responsive ventilation systems that dynamically adjust
to environmental conditions. Through interdisciplinary collaboration, these insights can contribute to the development of next-generation building regulations and ventilation standards, ultimately shaping healthier and more resilient indoor environments for generations to come.

- Alicia Alonso, Jesús Llanos, Rocío Escandón, and Juan J Sendra. Effects of the covid-19 pandemic on indoor air quality and thermal comfort of primary schools in winter in a mediterranean climate. *Sustainability*, 13(5): 2699, 2021.
- [2] Yuanhui Zhang. Indoor air quality engineering. CRC press Boca Raton, FL, USA:, 2005.
- [3] Jan Sundell. On the history of indoor air quality and health. Indoor Air, 14, 2004.
- [4] Vannan Kandi Vijayan, Haralappa Paramesh, Sundeep Santosh Salvi, and Alpa Anil Kumar Dalal. Enhancing indoor air quality-the air filter advantage. Lung India, 32(5):473-479, 2015.
- [5] Nehul Agarwal, Chandan Swaroop Meena, Binju P Raj, Lohit Saini, Ashok Kumar, N Gopalakrishnan, Anuj Kumar, Nagesh Babu Balam, Tabish Alam, Nishant Raj Kapoor, et al. Indoor air quality improvement in covid-19 pandemic. Sustainable Cities and Society, 70:102942, 2021.
- [6] T Faria, M Almeida-Silva, Ana Dias, and SM Almeida. Indoor air quality in urban office buildings. *International Journal of Environmental Technology* and Management, 19(3-4):236–256, 2016.
- [7] Andreea Tepeneu, Dorina Chambre, Teodora Surdea-Blaga, Andreea Lupitu, Cristian Moisa, Dana Maria Copolovici, and Lucian Copolovici. The variability of indoor air pollutants in the office and their impact on the workers' health. *Polish Journal of Environmental Studies*, 2024.

- [8] Zahra Atarodi, Kamaladdin Karimyan, Vinod Kumar Gupta, Morteza Abbasi, and Masoud Moradi. Evaluation of indoor air quality and its symptoms in office building–a case study of mashhad, iran. *Data in Brief*, 20:74–79, 2018.
- [9] Fátima Felgueiras, Zenaida Mourão, André Moreira, and Marta Fonseca Gabriel. Indoor environmental quality in offices and risk of health and productivity complaints at work: A literature review. *Journal of Hazardous Materials Advances*, 10:100314, 2023.
- [10] Man Fan, Zheng Fu, Jia Wang, Zhaoying Wang, Hanxiao Suo, Xiangfei Kong, and Han Li. A review of different ventilation modes on thermal comfort, air quality and virus spread control. *Building and Environment*, 212:108831, 2022.
- [11] Lidia Morawska, Julian W Tang, William Bahnfleth, Philomena M Bluyssen, Atze Boerstra, Giorgio Buonanno, Junji Cao, Stephanie Dancer, Andres Floto, Francesco Franchimon, et al. How can airborne transmission of covid-19 indoors be minimised? *Environment International*, 142:105832, 2020.
- [12] Jacob E Kohlmeier and David L Woodland. Immunity to respiratory viruses. Annual Review of Immunology, 27:61–82, 2009.
- [14] Chiara Achangwa, Huikyung Park, Sukhyun Ryu, and Moo-Sik Lee. Collateral impact of public health and social measures on respiratory virus activity during the covid-19 pandemic 2020–2021. Viruses, 14(5):1071, 2022.
- [15] Parsa Hodjat, Paul A Christensen, Sishir Subedi, David W Bernard, Randall J Olsen, and S Wesley Long. The reemergence of seasonal respiratory viruses in houston, texas, after relaxing covid-19 restrictions. *Microbiology Spectrum*, 9(2):e00430–21, 2021.
- [16] Dana Weissberg, Jürg Böni, Silvana K Rampini, Verena Kufner, Maryam Zaheri, Peter W Schreiber, Irene A Abela, Michael Huber, Hugo Sax, and

Aline Wolfensberger. Does respiratory co-infection facilitate dispersal of sars-cov-2? investigation of a super-spreading event in an open-space office. *Antimicrobial Resistance & Infection Control*, 9:1–8, 2020.

- [17] Neeltje Van Doremalen, Trenton Bushmaker, Dylan H Morris, Myndi G Holbrook, Amandine Gamble, Brandi N Williamson, Azaibi Tamin, Jennifer L Harcourt, Natalie J Thornburg, Susan I Gerber, et al. Aerosol and surface stability of sars-cov-2 as compared with sars-cov-1. New England Journal of Medicine, 382(16):1564–1567, 2020.
- [18] Nancy HL Leung. Transmissibility and transmission of respiratory viruses. *Nature Reviews Microbiology*, 19(8):528–545, 2021.
- [19] Jasmin S Kutter, Monique I Spronken, Pieter L Fraaij, Ron AM Fouchier, and Sander Herfst. Transmission routes of respiratory viruses among humans. *Current Opinion in Virology*, 28:142–151, 2018.
- [20] Michael Klompas, Meghan A Baker, and Chanu Rhee. Airborne transmission of sars-cov-2: Theoretical considerations and available evidence. Jama, 324(5):441–442, 2020.
- [21] Sally Shahzad, John Kaiser Calautit, Angelo I Aquino, Diana SNM Nasir, and Ben Richard Hughes. A user-controlled thermal chair for an open plan workplace: Cfd and field studies of thermal comfort performance. *Applied Energy*, 207:283–293, 2017.
- [22] Ahmad Rasdan Ismail, Norfadzilah Jusoh, Nor Kamilah Makhtar, Raemy Md Zein, Ismail Abdul Rahman, Nurul Husna Che Hassan, and Darliana Mohamed. Experimental investigations and computational thermal simulation on human thermal comfort during performing office tasks. In *Journal of Physics: Conference Series*, volume 1793, page 012076. IOP Publishing, 2021.
- [23] PO Fanger. Thermal comfort: Analysis and applications in environmental engineering, 1970.
- [24] Giovanni Semprini, Aminhossein Jahanbin, Beatrice Pulvirenti, Paolo Guidorzi, et al. Evaluation of thermal comfort inside an office equipped

with a fan coil hvac system: A cfd approach. Future Cities and Environment, 5(1):1-10, 2019.

- [25] Naglaa A Megahed and Ehab M Ghoneim. Indoor air quality: Rethinking rules of building design strategies in post-pandemic architecture. *Environ*mental Research, 193:110471, 2021.
- [26] Hua Qian and Xiaohong Zheng. Ventilation control for airborne transmission of human exhaled bio-aerosols in buildings. *Journal of Thoracic Disease*, 10(Suppl 19):S2295, 2018.
- [27] Renyi Zhang, Yixin Li, Annie L Zhang, Yuan Wang, and Mario J Molina. Identifying airborne transmission as the dominant route for the spread of covid-19. Proceedings of the National Academy of Sciences, 117(26):14857– 14863, 2020.
- [28] Maosheng Yao, Lu Zhang, Jianxin Ma, and Lian Zhou. On airborne transmission and control of sars-cov-2. Science of The Total Environment, 731: 139178, 2020.
- [29] Otis Sloan Brittain, Hannah Wood, and Prashant Kumar. Prioritising indoor air quality in building design can mitigate future airborne viral outbreaks. *Cities & Health*, 5(sup1):S162–S165, 2021.
- [30] Yuguo Li, Hua Qian, Jian Hang, Xuguang Chen, Ling Hong, Peng Liang, Jiansen Li, Shenglan Xiao, Jianjian Wei, Li Liu, et al. Evidence for probable aerosol transmission of sars-cov-2 in a poorly ventilated restaurant. *MedRXiv*, 2020.
- [31] Leonardo Setti, Fabrizio Passarini, Gianluigi De Gennaro, Pierluigi Barbieri, Maria Grazia Perrone, Massimo Borelli, Jolanda Palmisani, Alessia Di Gilio, Valentina Torboli, Francesco Fontana, et al. Sars-cov-2rna found on particulate matter of bergamo in northern italy: First evidence. *Envi*ronmental Research, 188:109754, 2020.
- [32] Jovan Pantelic and Kwok Wai Tham. Adequacy of air change rate as the sole indicator of an air distribution system's effectiveness to mitigate airborne infectious disease transmission caused by a cough release in the room with

overhead mixing ventilation: A case study. *HVAC&R Research*, 19(8):947–961, 2013.

- [33] Yiping Li, Gabriel M Leung, JW Tang, Xiaozhan Yang, CY Chao, John Zhang Lin, JW Lu, Per Væggemose Nielsen, Jianlei Niu, H Qian, et al. Role of ventilation in airborne transmission of infectious agents in the built environment: A multidisciplinary systematic review. *Indoor Air*, 17(1):2–18, 2007.
- [34] Zhihong Pang, Pingfan Hu, Xing Lu, Qingsheng Wang, and Zheng O'Neill. A smart co2-based ventilation control framework to minimize the infection risk of covid-19 in public buildings 2. In *Building Simulation 2021 Confer*ence, 2021.
- [35] Kwok Wai Tham. Indoor air quality and its effects on humans: A review of challenges and developments in the last 30 years. *Energy and Buildings*, 130:637–650, 2016.
- [36] He Li, Xiao-Long Xu, Da-Wei Dai, Zhen-Yu Huang, Zhuang Ma, and Yan-Jun Guan. Air pollution and temperature are associated with increased covid-19 incidence: A time series study. *International journal of infectious* diseases, 97:278–282, 2020.
- [37] Kenichi Azuma, U Yanagi, Naoki Kagi, Hoon Kim, Masayuki Ogata, and Motoya Hayashi. Environmental factors involved in sars-cov-2 transmission: effect and role of indoor environmental quality in the strategy for covid-19 infection control. *Environmental Health and Preventive Medicine*, 25(1): 1–16, 2020.
- [38] Stefan Boschert and Roland Rosen. Digital twin—the simulation aspect. Mechatronic futures: Challenges and solutions for mechatronic systems and their designers, pages 59–74, 2016.
- [39] Fei Tao, Jiangfeng Cheng, Qinglin Qi, Meng Zhang, He Zhang, and Fangyuan Sui. Digital twin-driven product design, manufacturing and service with big data. *The International Journal of Advanced Manufacturing Technology*, 94:3563–3576, 2018.

- [40] Amir Gandomi and Murtaza Haider. Beyond the hype: Big data concepts, methods, and analytics. International Journal of Information Management, 35(2):137–144, 2015.
- [41] Abdul Karim Feroz, Hangjung Zo, and Ananth Chiravuri. Digital transformation and environmental sustainability: A review and research agenda. *Sustainability*, 13(3):1530, 2021.
- [42] Zheng Tao Ai and Arsen Krikor Melikov. Airborne spread of expiratory droplet nuclei between the occupants of indoor environments: A review. *Indoor Air*, 28(4):500–524, 2018.
- [43] Ruiqing Shen, Zeren Jiao, Trent Parker, Yue Sun, and Qingsheng Wang. Recent application of computational fluid dynamics (cfd) in process safety and loss prevention: A review. *Journal of Loss Prevention in the Process Industries*, 67:104252, 2020.
- [44] Hui Dai and Bin Zhao. Association of the infection probability of covid-19 with ventilation rates in confined spaces. In *Building Simulation*, volume 13, pages 1321–1327. Springer, 2020.
- [45] Lee Shiu Hung. The sars epidemic in hong kong: what lessons have we learned? Journal of the Royal Society of Medicine, 96(8):374–378, 2003.
- [46] Y Li, S Duan, IT Yu, and TW Wong. Multi-zone modeling of probable sars virus transmission by airflow between flats in block e, amoy gardens. *Indoor Air*, 15(2):96–111, 2005.
- [47] Jianlei Niu and TC Tung. On-site quantification of re-entry ratio of ventilation exhausts in multi-family residential buildings and implications. *Indoor* Air, 18(1):12–26, 2007.
- [48] SC Chen, CF Chang, and CM Liao. Predictive models of control strategies involved in containing indoor airborne infections. *Indoor Air*, 16(6):469– 481, 2006.
- [49] Yi Jiang, Bin Zhao, Xiaofeng Li, Xudong Yang, Zhiqin Zhang, and Yufeng Zhang. Investigating a safe ventilation rate for the prevention of indoor sars transmission: An attempt based on a simulation approach. In *Building Simulation*, volume 2, pages 281–289. Springer, 2009.

- [50] HC Yu, Kwok Wai Mui, Ling Tim Wong, and HS Chu. Ventilation of general hospital wards for mitigating infection risks of three kinds of viruses including middle east respiratory syndrome coronavirus. *Indoor and Built Environment*, 26(4):514–527, 2017.
- [51] World Health Organization et al. Pandemic h1n1 2009. Technical report, WHO Regional Office for South-East Asia, 2009.
- [52] S-C Chen and C-M Liao. Modelling control measures to reduce the impact of pandemic influenza among schoolchildren. *Epidemiology & Infection*, 136 (8):1035–1045, 2008.
- [53] Jovan Pantelic, Gin Nam Sze-To, Kwok Wai Tham, Christopher YH Chao, and Yong Chuan Mike Khoo. Personalized ventilation as a control measure for airborne transmissible disease spread. *Journal of the Royal Society Interface*, 6(suppl\_6):S715–S726, 2009.
- [54] Theodore A Myatt, Matthew H Kaufman, Joseph G Allen, David L Mac-Intosh, M Patricia Fabian, and James J McDevitt. Modeling the airborne survival of influenza virus in a residential setting: the impacts of home humidification. *Environmental Health*, 9(1):1–7, 2010.
- [55] Tyler H Koep, Felicity T Enders, Chris Pierret, Stephen C Ekker, Dale Krageschmidt, Kevin L Neff, Marc Lipsitch, Jeffrey Shaman, and W Charles Huskins. Predictors of indoor absolute humidity and estimated effects on influenza virus survival in grade schools. *BMC Infectious Diseases*, 13(1): 1–8, 2013.
- [56] John D Noti, Francoise M Blachere, Cynthia M McMillen, William G Lindsley, Michael L Kashon, Denzil R Slaughter, and Donald H Beezhold. High humidity leads to loss of infectious influenza virus from simulated coughs. *PloS One*, 8(2):e57485, 2013.
- [57] Parham Azimi and Brent Stephens. Hvac filtration for controlling infectious airborne disease transmission in indoor environments: Predicting risk reductions and operational costs. *Building and Environment*, 70:150–160, 2013.

- [58] Dimitrios Giannis, Ioannis A Ziogas, and Panagiota Gianni. Coagulation disorders in coronavirus infected patients: Covid-19, sars-cov-1, mers-cov and lessons from the past. *Journal of Clinical Virology*, 127:104362, 2020.
- [59] Minki Sung, Seongmin Jo, Sang-Eun Lee, Moran Ki, Bo Youl Choi, and JinKwan Hong. Airflow as a possible transmission route of middle east respiratory syndrome at an initial outbreak hospital in korea. *International Journal of Environmental Research and Public Health*, 15(12):2757, 2018.
- [60] S Jo, J Hong, SE Lee, M Ki, BY Choi, and M Sung. Airflow analysis of pyeongtaek st mary's hospital during hospitalization of the first middle east respiratory syndrome patient in korea, r. soc. open sci. 6 (2019), 181164.
- [61] Clive B Beggs, Simon J Shepherd, and Kevin G Kerr. Potential for airborne transmission of infection in the waiting areas of healthcare premises: stochastic analysis using a monte carlo model. BMC Infectious Diseases, 10:1–8, 2010.
- [62] Gwo-Hwa Wan, Chung-Guei Huang, Fen-Fang Chung, Tzou-Yien Lin, Kuo-Chien Tsao, and Yhu-Chering Huang. Detection of common respiratory viruses and mycoplasma pneumoniae in patient-occupied rooms in pediatric wards. *Medicine*, 95(14), 2016.
- [63] Jayme P Coyle, Raymond C Derk, William G Lindsley, Francoise M Blachere, Theresa Boots, Angela R Lemons, Stephen B Martin Jr, Kenneth R Mead, Steven A Fotta, Jeffrey S Reynolds, et al. Efficacy of ventilation, hepa air cleaners, universal masking, and physical distancing for reducing exposure to simulated exhaled aerosols in a meeting room. Viruses, 13(12):2536, 2021.
- [64] Priscilla Gomes da Silva, Maria São José Nascimento, Ruben RG Soares, Sofia IV Sousa, and João R Mesquita. Airborne spread of infectious sarscov-2: Moving forward using lessons from sars-cov and mers-cov. Science of the Total Environment, 764:142802, 2021.
- [65] Charifa Zemouri, SF Awad, CMC Volgenant, W Crielaard, AMGA Laheij, and JJ De Soet. Modeling of the transmission of coronaviruses, measles

virus, influenza virus, mycobacterium tuberculosis, and legionella pneumophila in dental clinics. *Journal of Dental Research*, 99(10):1192–1198, 2020.

- [66] Mohamed Ismail Farouk, Ahmed Fayez Nassar, and Mohamed Hassan Elgamal. Numerical study of the transmission of exhaled droplets between the instructor and students in a typical classroom. *Applied Sciences*, 11 (20):9767, 2021.
- [67] Benjamin Jones, Patrick Sharpe, Christopher Iddon, E Abigail Hathway, Catherine J Noakes, and Shaun Fitzgerald. Modelling uncertainty in the relative risk of exposure to the sars-cov-2 virus by airborne aerosol transmission in well mixed indoor air. *Building and Environment*, 191:107617, 2021.
- [68] Jin Li, Zhu Cheng, Yin Zhang, Ning Mao, Shurui Guo, Qingqin Wang, Li Zhao, and Enshen Long. Evaluation of infection risk for sars-cov-2 transmission on university campuses. *Science and Technology for the Built Environment*, 27(9):1165–1180, 2021.
- [69] Carolanne VM Vouriot, Henry C Burridge, Catherine J Noakes, and Paul F Linden. Seasonal variation in airborne infection risk in schools due to changes in ventilation inferred from monitored carbon dioxide. *Indoor Air*, 31(4):1154–1163, 2021.
- [70] Alex Mikszewski, Luca Stabile, Giorgio Buonanno, and Lidia Morawska. The vaccination threshold for sars-cov-2 depends on the indoor setting and room ventilation. BMC infectious diseases, 21(1):1–9, 2021.
- [71] Phil Dacunto, Dylan Moser, Andrew Ng, and Michael Benson. Classroom aerosol dispersion: Desk spacing and divider impacts. *International Journal* of Environmental Science and Technology, 19(2):1057–1070, 2022.
- [72] Brett C Singer, Haoran Zhao, Chelsea V Preble, William W Delp, Jovan Pantelic, Michael D Sohn, and Thomas W Kirchstetter. Measured influence of overhead hvac on exposure to airborne contaminants from simulated speaking in a meeting and a classroom. *Indoor Air*, 32(1):e12917, 2022.

- [73] Marc Garbey, Guillaume Joerger, and Shannon Furr. A systems approach to assess transport and diffusion of hazardous airborne particles in a large surgical suite: potential impacts on viral airborne transmission. *International Journal of Environmental Research and Public Health*, 17(15):5404, 2020.
- [74] Bernard Polednik. Exposure of staff to aerosols and bioaerosols in a dental office. Building and Environment, 187:107388, 2021.
- [75] Yu Zhou and Shen Ji. Experimental and numerical study on the transport of droplet aerosols generated by occupants in a fever clinic. *Building and Environment*, 187:107402, 2021.
- [76] Xiangdong Li, Daniel Lester, Gary Rosengarten, Craig Aboltins, Milan Patel, and Ivan Cole. A spatiotemporally resolved infection risk model for airborne transmission of covid-19 variants in indoor spaces. *Science of the Total Environment*, 812:152592, 2022.
- [77] Cyril Crawford, Emmanuel Vanoli, Baptiste Decorde, Maxime Lancelot, Camille Duprat, Christophe Josserand, Jonathan Jilesen, Lila Bouadma, and Jean-François Timsit. Modeling of aerosol transmission of airborne pathogens in icu rooms of covid-19 patients with acute respiratory failure. *Scientific Reports*, 11(1):1–12, 2021.
- [78] Joan O Grimalt, Helem Vílchez, Pablo A Fraile-Ribot, Esther Marco, Antoni Campins, Jaime Orfila, Barend L Van Drooge, and Francisco Fanjul. Spread of sars-cov-2 in hospital areas. *Environmental Research*, 204:112074, 2022.
- [79] Marc Beaussier, Emmanuel Vanoli, Frédéric Zadegan, Herve Peray, Elodie Bezian, Jonathan Jilesen, Géraldine Gandveau, and Jean-Michel Gayraud. Aerodynamic analysis of hospital ventilation according to seasonal variations: A simulation approach to prevent airborne viral transmission pathway during covid-19 pandemic. *Environment International*, 158:106872, 2022.
- [80] Emma Burgos-Ramos, Itziar R Urbieta, and Diana Rodríguez. Is hydrogen peroxide an effective mouthwash for reducing the viral load of sars-cov-2 in dental clinics? *The Saudi Dental Journal*, 34(3):237–242, 2022.

- [81] Benjamin L Augenbraun, Zack D Lasner, Debayan Mitra, Sridhar Prabhu, Shivam Raval, Hiromitsu Sawaoka, and John M Doyle. Assessment and mitigation of aerosol airborne sars-cov-2 transmission in laboratory and office environments. Journal of Occupational and Environmental Hygiene, 17(10):447–456, 2020.
- [82] Henry C Burridge, Shiwei Fan, Roderic L Jones, Catherine J Noakes, and PF Linden. Predictive and retrospective modelling of airborne infection risk using monitored carbon dioxide. *Indoor and Built Environment*, 31(5): 1363–1380, 2022.
- [83] Alessandro Cammarata and Giuliano Cammarata. Dynamic assessment of the risk of airborne viral infection. *Indoor Air*, 31(6):1759–1775, 2021.
- [84] Chen Ren, Chang Xi, Junqi Wang, Zhuangbo Feng, Fuzhan Nasiri, Shi-Jie Cao, and Fariborz Haghighat. Mitigating covid-19 infection disease transmission in indoor environment using physical barriers. Sustainable cities and society, 74:103175, 2021.
- [85] Michael Riediker, Christian Monn, et al. Simulation of sars-cov-2 aerosol emissions in the infected population and resulting airborne exposures in different indoor scenarios. Aerosol and Air Quality Research, 21(2):200531, 2021.
- [86] Prateek Shrestha, Jason W DeGraw, Mingkan Zhang, and Xiaobing Liu. Multizonal modeling of sars-cov-2 aerosol dispersion in a virtual office building. *Building and Environment*, 206:108347, 2021.
- [87] Hamid Reza Tamaddon Jahromi, Igor Sazonov, Jason Jones, Alberto Coccarelli, Samuel Rolland, Neeraj Kavan Chakshu, Hywel Thomas, and Perumal Nithiarasu. Predicting the airborne microbial transmission via human breath particles using a gated recurrent units neural network. International Journal of Numerical Methods for Heat & Fluid Flow, 2022.
- [88] Yidan Shang, Jingliang Dong, Lin Tian, Fajiang He, and Jiyuan Tu. An improved numerical model for epidemic transmission and infection risks assessment in indoor environment. *Journal of Aerosol Science*, 162:105943, 2022.

- [89] Amelia Harrichandra, A Michael Ierardi, and Brian Pavilonis. An estimation of airborne sars-cov-2 infection transmission risk in new york city nail salons. *Toxicology and industrial health*, 36(9):634–643, 2020.
- [90] Alaa Alaidroos, Ayad Almaimani, Ahmed Baik, Mohamed Al-Amodi, and Khan Rubayet Rahaman. Are historical buildings more adaptive to minimize the risks of airborne transmission of viruses and public health? a study of the hazzazi house in jeddah (saudi arabia). International Journal of Environmental Research and Public Health, 18(7):3601, 2021.
- [91] Sreekanth Bathula, Srinivasan Anand, Thaseem Thajudeen, Yelia Shankaranarayana Mayya, Probal Chaudhury, Chaturvedi Shashank, et al. Survival of expiratory aerosols in a room: study using a bicompartment and bi-component indoor air model. Aerosol and Air Quality Research, 21(5):200547, 2021.
- [92] Tareq Hussein, Jakob Löndahl, Sara Thuresson, Malin Alsved, Afnan Al-Hunaiti, Kalle Saksela, Hazem Aqel, Heikki Junninen, Alexander Mahura, and Markku Kulmala. Indoor model simulation for covid-19 transport and exposure. *International Journal of Environmental Research and Public Health*, 18(6):2927, 2021.
- [93] Jayaveera Muthusamy, Syed Haq, Saad Akhtar, Mahmoud A Alzoubi, Tariq Shamim, and Jorge Alvarado. Implication of coughing dynamics on safe social distancing in an indoor environment—a numerical perspective. *Building* and Environment, 206:108280, 2021.
- [94] Bert Blocken, Thijs van Druenen, Alessio Ricci, Luyang Kang, Twan van Hooff, Peng Qin, Lili Xia, C Alanis Ruiz, JH Arts, JFL Diepens, et al. Ventilation and air cleaning to limit aerosol particle concentrations in a gym during the covid-19 pandemic. *Building and Environment*, 193:107659, 2021.
- [95] Seo Eun Hwang, Je Hwan Chang, Bumjo Oh, and Jongho Heo. Possible aerosol transmission of covid-19 associated with an outbreak in an apartment in seoul, south korea, 2020. International Journal of Infectious Diseases, 104:73–76, 2021.

- [96] Guanyu Jiang, Can Wang, Lu Song, Xing Wang, Yangyang Zhou, Chunnan Fei, and He Liu. Aerosol transmission, an indispensable route of covid-19 spread: case study of a department-store cluster. Frontiers of Environmental Science & Engineering, 15(3):1–12, 2021.
- [97] Mingkan Zhang, Prateek Shrestha, Xiaobing Liu, Tugba Turnaoglu, Jason DeGraw, Dustin Schafer, and Nathan Love. Computational fluid dynamics simulation of sars-cov-2 aerosol dispersion inside a grocery store. *Building* and Environment, 209:108652, 2022.
- [98] Niklas Kappelt, Hugo Savill Russell, Szymon Kwiatkowski, Alireza Afshari, and Matthew Stanley Johnson. Correlation of respiratory aerosols and metabolic carbon dioxide. *Sustainability*, 13(21):12203, 2021.
- [99] Shelly L Miller, William W Nazaroff, Jose L Jimenez, Atze Boerstra, Giorgio Buonanno, Stephanie J Dancer, Jarek Kurnitski, Linsey C Marr, Lidia Morawska, and Catherine Noakes. Transmission of sars-cov-2 by inhalation of respiratory aerosol in the skagit valley chorale superspreading event. *Indoor Air*, 31(2):314–323, 2021.
- [100] NN Peng, KW Chow, and CH Liu. Computational study on the transmission of the sars-cov-2 virus through aerosol in an elevator cabin: Effect of the ventilation system. *Physics of Fluids*, 33(10):103325, 2021.
- [101] Talib Dbouk and Dimitris Drikakis. On airborne virus transmission in elevators and confined spaces. *Physics of Fluids*, 33(1):011905, 2021.
- [102] Wolfgang Schade, Vladislav Reimer, Martin Seipenbusch, and Ulrike Willer. Experimental investigation of aerosol and co2 dispersion for evaluation of covid-19 infection risk in a concert hall. International Journal of Environmental Research and Public Health, 18(6):3037, 2021.
- [103] Jesse H Schreck, Masoud Jahandar Lashaki, Javad Hashemi, Manhar Dhanak, and Siddhartha Verma. Aerosol generation in public restrooms. *Physics of Fluids*, 33(3):033320, 2021.
- [104] David Vernez, Sophie Schwarz, Jean-Jacques Sauvain, Christiane Petignat, and Guillaume Suarez. Probable aerosol transmission of sars-cov-2 in a poorly ventilated courtroom. *Indoor Air*, 31(6):1776–1785, 2021.

- [105] Jungyeon Yu, Chul Kim, Yun Gyu Lee, and Sanghwan Bae. Impact on airborne virus behavior by an electric heat pump (ehp) operation in a restaurant during winter season. *Building and Environment*, 200:107951, 2021.
- [106] Mikko Auvinen, Joel Kuula, Tiia Grönholm, Matthias Sühring, and Antti Hellsten. High-resolution large-eddy simulation of indoor turbulence and its effect on airborne transmission of respiratory pathogens—model validation and infection probability analysis. *Physics of Fluids*, 34(1):015124, 2022.
- [107] Esther Rivas, Jose Luis Santiago, Fernando Martín, and Alberto Martilli. Impact of natural ventilation on exposure to sars-cov 2 in indoor/semiindoor terraces using co2 concentrations as a proxy. *Journal of Building Engineering*, 46:103725, 2022.
- [108] Stefan Moritz, Cornelia Gottschick, Johannes Horn, Mario Popp, Susan Langer, Bianca Klee, Oliver Purschke, Michael Gekle, Angelika Ihling, Frank DL Zimmermann, et al. The risk of indoor sports and culture events for the transmission of covid-19. *Nature communications*, 12(1):1–9, 2021.
- [109] Jiawei Leng, Qi Wang, and Ke Liu. Sustainable design of courtyard environment: From the perspectives of airborne diseases control and human health. Sustainable Cities and Society, 62:102405, 2020.
- [110] Jie Wang, Haiting Feng, Sheng Zhang, Zuowei Ni, Lingmei Ni, Yu Chen, Lixin Zhuo, Zifeng Zhong, and Tingting Qu. Sars-cov-2 rna detection of hospital isolation wards hygiene monitoring during the coronavirus disease 2019 outbreak in a chinese hospital. *International Journal of Infectious Diseases*, 94:103–106, 2020.
- [111] Carla Viegas, Raquel Pimenta, Marta Dias, Bianca Gomes, Miguel Brito, Liliana Aranha Caetano, Elisabete Carolino, and Anita Quintal Gomes. Microbiological contamination assessment in higher education institutes. Atmosphere, 12(8):1079, 2021.
- [112] Katia Razzini, Marta Castrica, Laura Menchetti, Lorenzo Maggi, Lucia Negroni, Nicola V Orfeo, Alice Pizzoccheri, Matteo Stocco, Stefano Muttini, and Claudia M Balzaretti. Sars-cov-2 rna detection in the air and on surfaces in the covid-19 ward of a hospital in milan, italy. *Science of The Total Environment*, 742:140540, 2020.

- [113] Janina Krambrich, Dario Akaberi, Jiaxin Ling, Tove Hoffman, Lennart Svensson, Marie Hagbom, and Åke Lundkvist. Sars-cov-2 in hospital indoor environments is predominantly non-infectious. Virology journal, 18 (1):1–9, 2021.
- [114] Karolina Nissen, Janina Krambrich, Dario Akaberi, Tove Hoffman, Jiaxin Ling, Åke Lundkvist, Lennart Svensson, and Erik Salaneck. Long-distance airborne dispersal of sars-cov-2 in covid-19 wards. *Scientific Reports*, 10(1): 1–9, 2020.
- [115] Patrick F Horve, Leslie G Dietz, Mark Fretz, David A Constant, Andrew Wilkes, John M Townes, Robert G Martindale, William B Messer, and Kevin G Van Den Wymelenberg. Identification of sars-cov-2 rna in healthcare heating, ventilation, and air conditioning units. *Indoor Air*, 31(6): 1826–1832, 2021.
- [116] María Rodríguez, M Llanos Palop, Susana Seseña, and Ana Rodríguez. Are the portable air cleaners (pac) really effective to terminate airborne sars-cov-2? Science of The Total Environment, 785:147300, 2021.
- [117] Sripriya Nannu Shankar, Chiran T Witanachchi, Alyssa F Morea, John A Lednicky, Julia C Loeb, Md Mahbubul Alam, Z Hugh Fan, Arantzazu Eiguren-Fernandez, and Chang-Yu Wu. Sars-cov-2 in residential rooms of two self-isolating persons with covid-19. *Journal of aerosol science*, 159: 105870, 2022.
- [118] Hossein Masoumbeigi, Ghader Ghanizadeh, Reza Yousefi Arfaei, Soleyman Heydari, Hassan Goodarzi, Ruhollah Dorostkar Sari, and Mehdi Tat. Investigation of hospital indoor air quality for the presence of sars-cov-2. *Journal* of Environmental Health Science and Engineering, 18(2):1259–1263, 2020.
- [119] Sasan Faridi, Sadegh Niazi, Kaveh Sadeghi, Kazem Naddafi, Jila Yavarian, Mansour Shamsipour, Nazanin Zahra Shafiei Jandaghi, Khosro Sadeghniiat, Ramin Nabizadeh, Masud Yunesian, et al. A field indoor air measurement of sars-cov-2 in the patient rooms of the largest hospital in iran. Science of the Total Environment, 725:138401, 2020.
- [120] Mehdi Vosoughi, Chiman Karami, Abdollah Dargahi, Farhad Jeddi, Kamyar Mazloum Jalali, Aidin Hadisi, Somayeh Biparva Haghighi, Hadi Peeri

Dogahe, Zahra Noorimotlagh, and Seyyed Abbas Mirzaee. Investigation of sars-cov-2 in hospital indoor air of covid-19 patients' ward with impinger method. *Environmental Science and Pollution Research*, 28(36): 50480–50488, 2021.

- [121] Azra Kenarkoohi, Zahra Noorimotlagh, Shahab Falahi, Ali Amarloei, Seyyed Abbas Mirzaee, Iraj Pakzad, and Elham Bastani. Hospital indoor air quality monitoring for the detection of sars-cov-2 (covid-19) virus. Science of the Total Environment, 748:141324, 2020.
- [122] Zeynab Baboli, Niloofar Neisi, Ali Akbar Babaei, Mehdi Ahmadi, Armin Sorooshian, Yaser Tahmasebi Birgani, and Gholamreza Goudarzi. On the airborne transmission of sars-cov-2 and relationship with indoor conditions at a hospital. *Atmospheric Environment*, 261:118563, 2021.
- [123] Hamid Reza Ghaffari, Hossein Farshidi, Vali Alipour, Kavoos Dindarloo, Mehdi Hassani Azad, Marzieh Jamalidoust, Abdolhossein Madani, Teamour Aghamolaei, Yaser Hashemi, Mehdi Fazlzadeh, et al. Detection of sars-cov-2 in the indoor air of intensive care unit (icu) for severe covid-19 patients and its surroundings: considering the role of environmental conditions. Environmental Science and Pollution Research, pages 1–7, 2021.
- [124] Sara Hemati, Gholam Reza Mobini, Mohsen Heidari, Fereidoun Rahmani, Akbar Soleymani Babadi, Marzieh Farhadkhani, Heshmatollah Nourmoradi, Ahmad Raeisi, Ali Ahmadi, Abbas Khodabakhshi, et al. Simultaneous monitoring of sars-cov-2, bacteria, and fungi in indoor air of hospital: a study on hajar hospital in shahrekord, iran. *Environmental Science and Pollution Research*, 28(32):43792–43802, 2021.
- [125] N Habibi, S Uddin, F Al-Salameen, S Al-Amad, V Kumar, M Al-Otaibi, N Abdul Razzack, A Shajan, and F Shirshikar. Sars-cov-2, other respiratory viruses and bacteria in aerosols: Report from kuwait's hospitals. *Indoor Air*, 31(6):1815–1825, 2021.
- [126] Ricardo Gomes Passos, Marina Bicalho Silveira, and Jônatas Santos Abrahão. Exploratory assessment of the occurrence of sars-cov-2 in aerosols in hospital facilities and public spaces of a metropolitan center in brazil. *Environmental Research*, 195:110808, 2021.

- [127] Shahriyar Bazzazpour, Masoumeh Rahmatinia, Seyed Reza Mohebbi, Mostafa Hadei, Abbas Shahsavani, Philip K Hopke, Behzad Houshmand, Alireza Raeisi, Ahmad Jonidi Jafari, Maryam Yarahmadi, et al. The detection of sars-cov-2 rna in indoor air of dental clinics during the covid-19 pandemic. *Environmental Science and Pollution Research*, pages 1–9, 2021.
- [128] Sven G Gehrke, Claudia Förderer, Ralf Weiskirchen, and Wolfgang Stremmel. Cold traps as reliable devices for quantitative determination of sarscov-2 load in aerosols. *Environmental Monitoring and Assessment*, 193(12): 1–8, 2021.
- [129] Jianpeng Li, Yatai Men, Xinlei Liu, Zhihan Luo, Yaojie Li, Huizhong Shen, Yilin Chen, Hefa Cheng, Guofeng Shen, and Shu Tao. Field-based evidence of changes in household pm2. 5 and exposure during the 2020 national quarantine in china. *Environmental Research Letters*, 16(9):094020, 2021.
- [130] Hyungkeun Kim, Kyungmo Kang, and Taeyeon Kim. Effect of occupant activity on indoor particle concentrations in korean residential buildings. *Sustainability*, 12(21):9201, 2020.
- [131] Salem Algarni, Roohul Abad Khan, Nadeem Ahmad Khan, and Nabisab Mujawar Mubarak. Particulate matter concentration and health risk assessment for a residential building during covid-19 pandemic in abha, saudi arabia. *Environmental Science and Pollution Research*, 28(46):65822– 65831, 2021.
- [132] Eliani Ezani, Peter Brimblecombe, Zulfa Hanan Asha'ari, Amirul Aiman Fazil, Sharifah Norkhadijah Syed Ismail, Zamzam Tuah Ahmad Ramly, Md Firoz Khan, et al. Indoor and outdoor exposure to pm2. 5 during covid-19 lockdown in suburban malaysia. *Aerosol and Air Quality Research*, 21 (3):200476, 2021.
- [133] Howook Sean Chang, Bradford Capuozzo, Bendegul Okumus, and Meehee Cho. Why cleaning the invisible in restaurants is important during covid-19: A case study of indoor air quality of an open-kitchen restaurant. *International Journal of Hospitality Management*, 94:102854, 2021.
- [134] Laurentiu Predescu and Daniel Dunea. Performance evaluation of particulate matter and indoor microclimate monitors in university classrooms

under covid-19 restrictions. International journal of environmental research and public health, 18(14):7363, 2021.

- [135] Naveen Puttaswamy, V Sreekanth, Ajay Pillarisetti, Adithi R Upadhya, Sudhakar Saidam, Balachandar Veerappan, Krishnendu Mukhopadhyay, Sankar Sambandam, Ronak Sutaria, Kalpana Balakrishnan, et al. Indoor and ambient air pollution in chennai, india during covid-19 lockdown: An affordable sensors study. Aerosol and Air Quality Research, 22(1):210170, 2022.
- [136] Samuel Domínguez-Amarillo, Jesica Fernández-Agüera, Sonia Cesteros-García, and Roberto Alonso González-Lezcano. Bad air can also kill: Residential indoor air quality and pollutant exposure risk during the covid-19 crisis. International Journal of Environmental Research and Public Health, 17(19):7183, 2020.
- [137] Jessica Tryner, Mollie Phillips, Casey Quinn, Gabe Neymark, Ander Wilson, Shantanu H Jathar, Ellison Carter, and John Volckens. Design and testing of a low-cost sensor and sampling platform for indoor air quality. *Building and Environment*, 206:108398, 2021.
- [138] Maria Chiara Pietrogrande, Lucia Casari, Giorgia Demaria, and Mara Russo. Indoor air quality in domestic environments during periods close to italian covid-19 lockdown. *International journal of environmental research* and public health, 18(8):4060, 2021.
- [139] Taehyun Roh, Alejandro Moreno-Rangel, Juha Baek, Alexander Obeng, Nishat Tasnim Hasan, and Genny Carrillo. Indoor air quality and health outcomes in employees working from home during the covid-19 pandemic: A pilot study. Atmosphere, 12(12):1665, 2021.
- [140] Sara Zanni, Gabriella Motta, Matteo Mura, Mariolina Longo, and Davide Caiulo. The challenge of indoor air quality management: A case study in the hospitality industry at the time of the pandemic. *Atmosphere*, 12(7): 880, 2021.
- [141] Diana Rodríguez, Itziar R Urbieta, Ángel Velasco, Miguel Ángel Campano-Laborda, and Elena Jiménez. Assessment of indoor air quality and risk of

covid-19 infection in spanish secondary school and university classrooms. Building and Environment, 226:109717, 2022.

- [142] Alberto Meiss, Héctor Jimeno-Merino, Irene Poza-Casado, Alfredo Llorente-Álvarez, and Miguel Ángel Padilla-Marcos. Indoor air quality in naturally ventilated classrooms. lessons learned from a case study in a covid-19 scenario. Sustainability, 13(15):8446, 2021.
- [143] Paulo Henrique Peitl Gregorio, Alessandro Wasum Mariani, João Marcelo Lopes Toscano Brito, Bruno José Martini Santos, and Paulo Manuel Pêgo-Fernandes. Indoor air quality and environmental sampling as support tools to detect sars-cov-2 in the healthcare setting. *Journal of Occupational and Environmental Medicine*, 63(11):956, 2021.
- [144] Chul Kim, Doowon Choi, Yun Gyu Lee, and Kichul Kim. Diagnosis of indoor air contaminants in a daycare center using a long-term monitoring. *Building and Environment*, 204:108124, 2021.
- [145] Antonio J Aguilar, María L de la Hoz-Torres, M<sup>a</sup> Dolores Martínez-Aires, and Diego P Ruiz. Monitoring and assessment of indoor environmental conditions after the implementation of covid-19-based ventilation strategies in an educational building in southern spain. Sensors, 21(21):7223, 2021.
- [146] Yiran Lu, Yifan Li, Hao Zhou, Jinlan Lin, Zhuozhao Zheng, Huji Xu, Borong Lin, Minggui Lin, and Li Liu. Affordable measures to monitor and alarm nosocomial sars-cov-2 infection due to poor ventilation. *Indoor* Air, 31(6):1833–1842, 2021.
- [147] Farhang Tahmasebi, Yan Wang, Elizabeth Cooper, Daniel Godoy Shimizu, Samuel Stamp, and Dejan Mumovic. Window operation behaviour and indoor air quality during lockdown: A monitoring-based simulation-assisted study in london. Building Services Engineering Research and Technology, 43(1):5–21, 2022.
- [148] Carmen María Calama-González, Ángel Luis León-Rodríguez, and Rafael Suarez. Indoor environmental assessment: Comparing ventilation scenarios in pre-and post-retrofitted dwellings through test cells. *Journal of Building Engineering*, 43:103148, 2021.

- [149] Vesna Lovec, Miroslav Premrov, and Vesna Žegarac Leskovar. Practical impact of the covid-19 pandemic on indoor air quality and thermal comfort in kindergartens. a case study of slovenia. International Journal of Environmental Research and Public Health, 18(18):9712, 2021.
- [150] Alessia Di Gilio, Jolanda Palmisani, Manuela Pulimeno, Fabio Cerino, Mirko Cacace, Alessandro Miani, and Gianluigi de Gennaro. Co2 concentration monitoring inside educational buildings as a strategic tool to reduce the risk of sars-cov-2 airborne transmission. *Environmental Research*, 202: 111560, 2021.
- [151] Qirong Huang, Tamer Marzouk, Razvan Cirligeanu, Hans Malmstrom, Eli Eliav, and Y-F Ren. Ventilation assessment by carbon dioxide levels in dental treatment rooms. *Journal of Dental Research*, 100(8):810–816, 2021.
- [152] Rafael Fayos-Jordan, Jaume Segura-Garcia, Antonio Soriano-Asensi, Santiago Felici-Castell, Jose M Felisi, and Jose M Alcaraz-Calero. Ventqsys: Low-cost open iot system for

# $co_2$

co 2 monitoring in classrooms. Wireless Networks, 27(8):5313–5327, 2021.

- [153] Florentina Villanueva, Alberto Notario, Beatriz Cabañas, Pilar Martín, Sagrario Salgado, and Marta Fonseca Gabriel. Assessment of co2 and aerosol (pm2. 5, pm10, ufp) concentrations during the reopening of schools in the covid-19 pandemic: The case of a metropolitan area in central-southern spain. *Environmental Research*, 197:111092, 2021.
- [154] Giulia Ulpiani, Negin Nazarian, Fuyu Zhang, and Christopher J Pettit. Towards a living lab for enhanced thermal comfort and air quality: Analyses of standard occupancy, weather extremes, and covid-19 pandemic. *Frontiers* in Environmental Science, page 556, 2021.
- [155] Rafia Mumtaz, Syed Mohammad Hassan Zaidi, Muhammad Zeeshan Shakir, Uferah Shafi, Muhammad Moeez Malik, Ayesha Haque, Sadaf Mumtaz, and Syed Ali Raza Zaidi. Internet of things (iot) based indoor air quality sensing and predictive analytic—a covid-19 perspective. *Electronics*, 10(2): 184, 2021.

- [156] Praveen Kumar Sharma, Ananya Mondal, Shivam Jaiswal, Mousumi Saha, Subrata Nandi, Tanmay De, and Sujoy Saha. Indoairsense: A framework for indoor air quality estimation and forecasting. *Atmospheric Pollution Research*, 12(1):10–22, 2021.
- [157] Saman Taheri and Ali Razban. Learning-based co2 concentration prediction: Application to indoor air quality control using demand-controlled ventilation. *Building and Environment*, 205:108164, 2021.
- [158] Mohamed Marzouk and Mohamed Atef. Assessment of indoor air quality in academic buildings using iot and deep learning. *Sustainability*, 14(12): 7015, 2022.
- [159] Ajit Ahlawat, Alfred Wiedensohler, Sumit Kumar Mishra, et al. An overview on the role of relative humidity in airborne transmission of sarscov-2 in indoor environments. Aerosol and Air Quality Research, 20(9): 1856–1861, 2020.
- [160] Han June Park, Sung-Gwang Lee, Jeong Suk Oh, Minhyuk Nam, Steven Barrett, Soohyung Lee, and Wontae Hwang. The effects of indoor temperature and humidity on local transmission of covid-19 and how it relates to global trends. *PloS One*, 17(8):e0271760, 2022.
- [161] Kevin S Raines, Sebastian Doniach, and Gyan Bhanot. The transmission of sars-cov-2 is likely comodulated by temperature and by relative humidity. *PloS One*, 16(7):e0255212, 2021.
- [162] Anice C Lowen, Samira Mubareka, John Steel, and Peter Palese. Influenza virus transmission is dependent on relative humidity and temperature. *PLoS Pathogens*, 3(10):e151, 2007.
- [163] Anice C Lowen and John Steel. Roles of humidity and temperature in shaping influenza seasonality. *Journal of Virology*, 88(14):7692–7695, 2014.
- [164] Kwok-Hung Chan, JS Malik Peiris, SY Lam, LLM Poon, KY Yuen, and Wing Hong Seto. The effects of temperature and relative humidity on the viability of the sars coronavirus. *Advances in Virology*, 2011, 2011.

- [165] Yi-Hsien Cheng and Chung-Min Liao. Modeling control measure effects to reduce indoor transmission of pandemic h1n1 2009 virus. *Building and Environment*, 63:11–19, 2013.
- [166] N Van Doremalen, T Bushmaker, and VJ Munster. Stability of middle east respiratory syndrome coronavirus (mers-cov) under different environmental conditions. *Eurosurveillance*, 18(38):20590, 2013.
- [167] Oleg V Pyankov, Sergey A Bodnev, Olga G Pyankova, and Igor E Agranovski. Survival of aerosolized coronavirus in the ambient air. *Journal of Aerosol Science*, 115:158–163, 2018.
- [168] María L de la Hoz-Torres, Antonio J Aguilar, Diego P Ruiz, and María Dolores Martínez-Aires. Analysis of impact of natural ventilation strategies in ventilation rates and indoor environmental acoustics using sensor measurement data in educational buildings. *Sensors*, 21(18):6122, 2021.
- [169] Arminder K Deol, Danny Scarponi, Peter Beckwith, Tom A Yates, Aaron S Karat, Ada WC Yan, Kathy S Baisley, Alison D Grant, Richard G White, and Nicky McCreesh. Estimating ventilation rates in rooms with varying occupancy levels: Relevance for reducing transmission risk of airborne pathogens. *PloS One*, 16(6):e0253096, 2021.
- [170] Ester Lepore, Patricia Aguilera Benito, Carolina Piña Ramírez, and Giacomo Viccione. Indoors ventilation in times of confinement by sars-cov-2 epidemic: A comparative approach between spain and italy. *Sustainable Cities and Society*, 72:103051, 2021.
- [171] Yang Liu, Wuxuan Pan, and Zhengwei Long. Optimization of air supply parameters for stratum ventilation based on proper orthogonal decomposition. *Sustainable Cities and Society*, 75:103291, 2021.
- [172] Sowoo Park, Younhee Choi, Doosam Song, and Eun Kyung Kim. Natural ventilation strategy and related issues to prevent coronavirus disease 2019 (covid-19) airborne transmission in a school building. Science of the Total Environment, 789:147764, 2021.
- [173] Andres Nunez and Ana M García. Effect of the passive natural ventilation on the bioaerosol in a small room. *Building and Environment*, 207:108438, 2022.

- [174] Claudia C Vassella, Jeremy Koch, Alexander Henzi, Alexander Jordan, Roger Waeber, Reto Iannaccone, and Roland Charrière. From spontaneous to strategic natural window ventilation: Improving indoor air quality in swiss schools. International Journal of Hygiene and Environmental Health, 234:113746, 2021.
- [175] Fangyuan Zhang and Yuji Ryu. Simulation study on indoor air distribution and indoor humidity distribution of three ventilation patterns using computational fluid dynamics. *Sustainability*, 13(7):3630, 2021.
- [176] Luca Borro, Lorenzo Mazzei, Massimiliano Raponi, Prisco Piscitelli, Alessandro Miani, and Aurelio Secinaro. The role of air conditioning in the diffusion of sars-cov-2 in indoor environments: a first computational fluid dynamic model, based on investigations performed at the vatican state children's hospital. *Environmental Research*, 193:110343, 2021.
- [177] Weiqi Chen, Dong-Bin Kwak, Jonathon Anderson, Kaushik Kanna, Chenxing Pei, Qingfeng Cao, Qisheng Ou, Seong Chan Kim, Thomas H Kuehn, David YH Pui, et al. Study on droplet dispersion influenced by ventilation and source configuration in classroom settings using low-cost sensor network. Aerosol and Air Quality Research, 21:210232, 2021.
- [178] Xiangfei Kong, Chenli Guo, Zhang Lin, Shasha Duan, Junjie He, Yue Ren, and Jianlin Ren. Experimental study on the control effect of different ventilation systems on fine particles in a simulated hospital ward. *Sustainable Cities and Society*, 73:103102, 2021.
- [179] Chunying Li and Haida Tang. Study on ventilation rates and assessment of infection risks of covid-19 in an outpatient building. *Journal of Building Engineering*, 42:103090, 2021.
- [180] Dnyanesh Mirikar, Silambarasan Palanivel, and Venugopal Arumuru. Droplet fate, efficacy of face mask, and transmission of virus-laden droplets inside a conference room. *Physics of Fluids*, 33(6):065108, 2021.
- [181] Haohan Sha, Xin Zhang, and Dahai Qi. Optimal control of high-rise building mechanical ventilation system for achieving low risk of covid-19 transmission and ventilative cooling. *Sustainable cities and society*, 74:103256, 2021.

- [182] Hamid Motamedi, Mohammadreza Shirzadi, Yoshihide Tominaga, and Parham A Mirzaei. Cfd modeling of airborne pathogen transmission of covid-19 in confined spaces under different ventilation strategies. *Sustain-able Cities and Society*, 76:103397, 2022.
- [183] Ioannis Tzoutzas, Helena C Maltezou, Nikolaos Barmparesos, Panagiotis Tasios, Chrysanthi Efthymiou, Margarita Niki Assimakopoulos, Maria Tseroni, Rengina Vorou, Fotios Tzermpos, Maria Antoniadou, et al. Indoor air quality evaluation using mechanical ventilation and portable air purifiers in an academic dentistry clinic during the covid-19 pandemic in greece. International journal of environmental research and public health, 18(16): 8886, 2021.
- [184] Alex Vlachokostas, Carolyn A Burns, Timothy I Salsbury, Richard C Daniel, Daniel P James, Julia E Flaherty, Na Wang, Ronald M Underhill, Gourihar Kulkarni, and Leonard F Pease. Experimental evaluation of respiratory droplet spread to rooms connected by a central ventilation system. *Indoor Air*, 32(1):e12940, 2022.
- [185] Javier M Rey-Hernández, Julio F San José-Alonso, Eloy Velasco-Gómez, Charles Yousif, and Francisco J Rey-Martínez. Performance analysis of a hybrid ventilation system in a near zero energy building. *Building and Environment*, 185:107265, 2020.
- [186] Bruno Perazzo Pedroso Barbosa and Nisio de Carvalho Lobo Brum. Ventilation mode performance against airborne respiratory infections in small office spaces: Limits and rational improvements for covid-19. Journal of the Brazilian Society of Mechanical Sciences and Engineering, 43(6):1–19, 2021.
- [187] L Stabile, A Pacitto, A Mikszewski, L Morawska, and G Buonanno. Ventilation procedures to minimize the airborne transmission of viruses in classrooms. *Building and Environment*, 202:108042, 2021.
- [188] Ruichen He, Wanjiao Liu, John Elson, Rainer Vogt, Clay Maranville, and Jiarong Hong. Airborne transmission of covid-19 and mitigation using box fan air cleaners in a poorly ventilated classroom. *Physics of Fluids*, 33(5): 057107, 2021.

- [189] Sai Ranjeet Narayanan and Suo Yang. Airborne transmission of virus-laden aerosols inside a music classroom: Effects of portable purifiers and aerosol injection rates. *Physics of Fluids*, 33(3):033307, 2021.
- [190] Yan-Fang Ren, Qirong Huang, Tamer Marzouk, Ray Richard, Karen Pembroke, Pat Martone, Tom Venner, Hans Malmstrom, and Eli Eliav. Effects of mechanical ventilation and portable air cleaner on aerosol removal from dental treatment rooms. *Journal of dentistry*, 105:103576, 2021.
- [191] Philomena M Bluyssen, Marco Ortiz, and Dadi Zhang. The effect of a mobile hepa filter system on 'infectious' aerosols, sound and air velocity in the senselab. *Building and Environment*, 188:107475, 2021.
- [192] Jos Lelieveld, Frank Helleis, Stephan Borrmann, Yafang Cheng, Frank Drewnick, Gerald Haug, Thomas Klimach, Jean Sciare, Hang Su, and Ulrich Pöschl. Model calculations of aerosol transmission and infection risk of covid-19 in indoor environments. *International Journal of Environmental Research and Public Health*, 17(21):8114, 2020.
- [193] Yu Feng, Jianan Zhao, Michele Spinolo, Kyle Lane, Darren Leung, Dave Marshall, Peter Mlinaric, et al. Assessing the filtration effectiveness of a portable ultraviolet air cleaner on airborne sars-cov-2 laden droplets in a patient room: A numerical study. Aerosol and Air Quality Research, 21(5): 200608, 2021.
- [194] Shubham Srivastava, Xingwang Zhao, Ati Manay, and Qingyan Chen. Effective ventilation and air disinfection system for reducing coronavirus disease 2019 (covid-19) infection risk in office buildings. Sustainable Cities and Society, 75:103408, 2021.
- [195] Kevin Kahn and Richard M Mariita. Quantifying the impact of uvc in reducing airborne pathogen transmission and improving energy efficiency for healthy buildings: Kahn-mariita equivalent ventilation model. *Frontiers* in Built Environment, page 121, 2021.
- [196] Susana Oliveira de Souza, Antônio Américo Cardoso Jr, Aquiles Sales Craveiro Sarmento, and Francesco d'Errico. Effectiveness of a uvc air disinfection system for the hvac of an icu. The European Physical Journal Plus, 137(1):37, 2022.

- [197] Jialei Shen, Meng Kong, Bing Dong, Michael J Birnkrant, and Jianshun Zhang. A systematic approach to estimating the effectiveness of multiscale iaq strategies for reducing the risk of airborne infection of sars-cov-2. Building and Environment, 200:107926, 2021.
- [198] Ali Ghoroghi, Yacine Rezgui, and Ruth Wallace. Impact of ventilation and avoidance measures on sars-cov-2 risk of infection in public indoor environments. Science of the Total Environment, 838:156518, 2022.
- [199] ZM Razlan, NAA Razali, MI Izham, I Ibrahim, H Desa, NN Zulkepli, MA Azizan, et al. A numerical study on indoor air quality (iaq) designs for hospital waiting room in the context of covid-19 for hot and warm humidity climate. In *Journal of Physics: Conference Series*, volume 2071, page 012056. IOP Publishing, 2021.
- [200] Parham A Mirzaei, Mohammad Moshfeghi, Hamid Motamedi, Yahya Sheikhnejad, and Hadi Bordbar. A simplified tempo-spatial model to predict airborne pathogen release risk in enclosed spaces: An eulerianlagrangian cfd approach. *Building and Environment*, 207:108428, 2022.
- [201] Yixin Dong, Li Zhu, Sui Li, and Martin Wollensak. Optimal design of building openings to reduce the risk of indoor respiratory epidemic infections. In *Building Simulation*, volume 15, pages 871–884. Springer, 2022.
- [202] AR Sarhan, P Naser, and J Naser. Covid-19 aerodynamic evaluation of social distancing in indoor environments, a numerical study. *Journal of Environmental Health Science and Engineering*, 19(2):1969–1978, 2021.
- [203] Michael J Risbeck, Martin Z Bazant, Zhanhong Jiang, Young M Lee, Kirk H Drees, and Jonathan D Douglas. Modeling and multiobjective optimization of indoor airborne disease transmission risk and associated energy consumption for building hvac systems. *Energy and Buildings*, 253:111497, 2021.
- [204] Luigi Schibuola and Chiara Tambani. High energy efficiency ventilation to limit covid-19 contagion in school environments. *Energy and Buildings*, 240: 110882, 2021.
- [205] Luigi Schibuola and Chiara Tambani. Performance comparison of heat recovery systems to reduce viral contagion in indoor environments. Applied Thermal Engineering, 190:116843, 2021.

- [206] Pranaynil Saikia, Dibakar Rakshit, Ramesh Narayanaswamy, Fujen Wang, et al. Energy performance and indoor airflow analysis of a healthcare ward designed with resource conservation objectives. *Journal of Building Engineering*, 44:103296, 2021.
- [207] Fabrizio Ascione, Rosa Francesca De Masi, Margherita Mastellone, and Giuseppe Peter Vanoli. The design of safe classrooms of educational buildings for facing contagions and transmission of diseases: A novel approach combining audits, calibrated energy models, building performance (bps) and computational fluid dynamic (cfd) simulations. *Energy and Buildings*, 230:110533, 2021.
- [208] Junqi Wang, Jingjing Huang, Zhuangbo Feng, Shi-Jie Cao, and Fariborz Haghighat. Occupant-density-detection based energy efficient ventilation system: Prevention of infection transmission. *Energy and Buildings*, 240: 110883, 2021.
- [209] Muhammad Saidu Aliero, Muhammad Fermi Pasha, Adel N Toosi, and Imran Ghani. The covid-19 impact on air condition usage: a shift towards residential energy saving. *Environmental Science and Pollution Research*, pages 1–15, 2022.
- [210] Dorit Aviv, Kian Wee Chen, Eric Teitelbaum, Denon Sheppard, Jovan Pantelic, Adam Rysanek, and Forrest Meggers. A fresh (air) look at ventilation for covid-19: Estimating the global energy savings potential of coupling natural ventilation with novel radiant cooling strategies. *Applied Energy*, 292:116848, 2021.
- [211] Fernando de Frutos, Teresa Cuerdo-Vilches, Carmen Alonso, Fernando Martín-Consuegra, Borja Frutos, Ignacio Oteiza, and Miguel Ángel Navas-Martín. Indoor environmental quality and consumption patterns before and during the covid-19 lockdown in twelve social dwellings in madrid, spain. Sustainability, 13(14):7700, 2021.
- [212] Amit Kaushik, Mohammed Arif, Prasad Tumula, and Obas John Ebohon. Effect of thermal comfort on occupant productivity in office buildings: Response surface analysis. *Building and Environment*, 180:107021, 2020.

- [213] Frederick H Rohles Jr and Ward V Wells. Interior design, comfort and thermal sensitivity. Journal of Interior Design Education and Research, 2 (2):36–44, 1976.
- [214] Ezgi Kocaman, Merve Kuru, and Gulben Calis. Do thermal comfort standards ensure occupant satisfaction? learning from occupants' thermal complaints. In *Creative Construction Conference 2019*, pages 7–11. Budapest University of Technology and Economics, 2019.
- [215] Salvatore Carlucci, Lujian Bai, Richard de Dear, and Liu Yang. Review of adaptive thermal comfort models in built environmental regulatory documents. *Building and Environment*, 137:73–89, 2018.
- [216] American Society of Heating, Refrigerating, Air-Conditioning Engineers, and American National Standards Institute. *Thermal environmental conditions for human occupancy*, volume 55. American Society of Heating, Refrigerating and Air-Conditioning Engineers, 2004.
- [217] Carlos Rubio-Bellido, Alexis Perez-Fargallo, Jesus A Pulido-Arcas, and Maureen Trebilcock. Application of adaptive comfort behaviors in chilean social housing standards under the influence of climate change. In *Building Simulation*, volume 10, pages 933–947. Springer, 2017.
- [218] Elena Barbadilla-Martín, José Manuel Salmerón Lissén, José Guadix Martín, Pablo Aparicio-Ruiz, and Luisa Brotas. Field study on adaptive thermal comfort in mixed mode office buildings in southwestern area of spain. *Building and Environment*, 123:163–175, 2017.
- [219] ASHRAE Standard. Thermal environmental conditions for human occupancy. ANSI/ASHRAE, 55, 5, 1992.
- [220] AK Mishra, MGLC Loomans, and Jan LM Hensen. Thermal comfort of heterogeneous and dynamic indoor conditions—an overview. Building and Environment, 109:82–100, 2016.
- [221] Kathryn J McCartney and J Fergus Nicol. Developing an adaptive control algorithm for europe. *Energy and Buildings*, 34(6):623–635, 2002.

- [222] David Bienvenido-Huertas, Carlos Rubio-Bellido, David Bienvenido-Huertas, and Carlos Rubio-Bellido. Adaptive Thermal Comfort Models for Buildings. Springer, 2021.
- [223] Bjarne W Olesen and KC Parsons. Introduction to thermal comfort standards and to the proposed new version of en iso 7730. Energy and Buildings, 34(6):537–548, 2002.
- [224] Bjarne W Olesen. Indoor environmental input parameters for the design and assessment of energy performance of buildings. *Rehva Journal*, pages 17–23, 2015.
- [225] J Fergus Nicol and Mike Wilson. A critique of european standard en 15251: strengths, weaknesses and lessons for future standards. *Building Research & Information*, 39(2):183–193, 2011.
- [226] Arup Bhattacharya, Ali Ghahramani, and Ehsan Mousavi. The effect of door opening on air-mixing in a positively pressurized room: Implications for operating room air management during the covid outbreak. *Journal of Building Engineering*, 44:102900, 2021.
- [227] Hongyun Si, Lin Shen, Wenchao Liu, and Guangdong Wu. Uncovering people's mask-saving intentions and behaviors in the post-covid-19 period: Evidence from china. Sustainable Cities and Society, 65:102626, 2021.
- [228] Ying Zou, Xingwang Zhao, and Qingyan Chen. Comparison of star-ccm+ and ansys fluent for simulating indoor airflows. In *Building Simulation*, volume 11, pages 165–174. Springer, 2018.
- [229] Savannah B Bell, Andrew C Blair, Lauren N Wagner, Vincent Zou, Ali R Buendia, and Farhad Ashrafzadeh. Survey of computational fluid dynamics software for rotational purposes. In ASME International Mechanical Engineering Congress and Exposition, volume 59421, page V005T07A014. American Society of Mechanical Engineers, 2019.
- [230] Hao Xu, Chonghuai Yan, Qingyan Fu, Kai Xiao, Yamei Yu, Deming Han, Wenhua Wang, and Jinping Cheng. Possible environmental effects on the spread of covid-19 in china. *Science of the Total Environment*, 731:139211, 2020.

- [231] Donatella Sarti, Tamara Campanelli, Tommaso Rondina, and Beatrice Gasperini. Covid-19 in workplaces: secondary transmission. Annals of Work Exposures and Health, 65(9):1145–1151, 2021.
- [232] Cătălin George Popovici. Hvac system functionality simulation using ansysfluent. *Energy Procedia*, 112:360–365, 2017.
- [233] Mehrdad Rabani, Habtamu Bayera Madessa, and Natasa Nord. Cfd study on the effect of archimedes number and heating rate on the thermal stratification of aventilated office. 2018.
- [234] Qingyan Chen. Comparison of different k-ε models for indoor air flow computations. Numerical Heat Transfer, Part B Fundamentals, 28(3):353–369, 1995.
- [235] Peter Vilhelm Nielsen. Computational fluid dynamics and room air movement. Indoor Air, 14(Supplement 7):134–143, 2004.
- [236] Zheshu Ma and Shaozhe Sun. Numerical assessment of the effect of cigarette smoking on indoor pm2. 5 distribution and study of ventilation strategies. *Indoor and Built Environment*, 27(3):369–379, 2018.
- [237] Aliyu M Aliyu, Dharminder Singh, Chino Uzoka, and Rakesh Mishra. Dispersion of virus-laden droplets in ventilated rooms: Effect of homemade facemasks. *Journal of Building Engineering*, 44:102933, 2021.
- [238] H Kotb and EE Khalil. Sneeze and cough pathogens migration inside aircraft cabins. *Energy*, 2(4), 2020.
- [239] Zhongquan C Zheng, Z Wei, James S Bennett, and Xiaofan Yang. Simulation and comparison of particle injection in an indoor environment using the species transport and discrete phase models. In *Fluids Engineering Division Summer Meeting*, volume 44755, pages 559–568. American Society of Mechanical Engineers, 2012.
- [240] Behrouz Pirouz, Stefania Anna Palermo, Seyed Navid Naghib, Domenico Mazzeo, Michele Turco, and Patrizia Piro. The role of hvac design and windows on the indoor airflow pattern and ach. *Sustainability*, 13(14):7931, 2021.

- [241] Christina Kakoulli, Alexis Kyriacou, and Michalis P Michaelides. A review of field measurement studies on thermal comfort, indoor air quality and virus risk. *Atmosphere*, 13(2):191, 2022.
- [242] Sani Dimitroulopoulou, Marzenna R Dudzińska, Lars Gunnarsen, Linda Hägerhed, Henna Maula, Raja Singh, Oluyemi Toyinbo, and Ulla Haverinen-Shaughnessy. Indoor air quality guidelines from across the world: An appraisal considering energy saving, health, productivity, and comfort. *Environment International*, 178:108127, 2023.
- [243] Sepideh Sadat Korsavi, Azadeh Montazami, and Dejan Mumovic. Perceived indoor air quality in naturally ventilated primary schools in the uk: Impact of environmental variables and thermal sensation. *Indoor Air*, 31(2):480– 501, 2021.
- [244] Bingyun Liu and Stephanie Gauthier. Empirical study: investigating the effect of co2 concentration on reported thermal comfort. In MC2014 Masters Conference People and Buildings, 2014.
- [245] PTBS Branco, MCM Alvim-Ferraz, FG Martins, and SIV Sousa. Children's exposure to indoor air in urban nurseries-part i: Co2 and comfort assessment. *Environmental Research*, 140:1–9, 2015.
- [246] Shuqing Cui, Michaël Cohen, Pascal Stabat, and Dominique Marchio. Co2 tracer gas concentration decay method for measuring air change rate. *Build*ing and Environment, 84:162–169, 2015.
- [247] Andrew K Persily. Evaluating building iaq and ventilation with indoor carbon dioxide. 1997.
- [248] Zhe Peng and Jose L Jimenez. Exhaled co2 as a covid-19 infection risk proxy for different indoor environments and activities. *Environmental Science & Technology Letters*, 8(5):392–397, 2021.
- [249] EMG SAGE and BM EMG. Role of ventilation in controlling sars-cov-2 transmission, 2020.
- [250] Angela Eykelbosh. Indoor co2 sensors for covid-19 risk mitigation: Current guidance and limitations. National Collaborating Centre for Environmental Health: Vancouver, BC, Canada, pages 1–16, 2021.

- [251] Peder Wolkoff. Indoor air humidity revisited: Impact on acute symptoms, work productivity, and risk of influenza and covid-19 infection. International Journal of Hygiene and Environmental Health, 256:114313, 2024.
- [252] Simon de Crane D'Heysselaer, Gianni Parisi, Maxime Lisson, Olivier Bruyère, Anne-Francoise Donneau, Sebastien Fontaine, Laurent Gillet, Fabrice Bureau, Gilles Darcis, Etienne Thiry, et al. Systematic review of the key factors influencing the indoor airborne spread of sars-cov-2. *Pathogens*, 12(3):382, 2023.
- [253] Gabriella Guarnieri, Bianca Olivieri, Gianenrico Senna, and Andrea Vianello. Relative humidity and its impact on the immune system and infections. *International Journal of Molecular Sciences*, 24(11):9456, 2023.
- [254] Anthony V Arundel, Elia M Sterling, Judith H Biggin, and Theodor D Sterling. Indirect health effects of relative humidity in indoor environments. *Environmental Health Perspectives*, 65:351–361, 1986.
- [255] Leila Allermann, Jan Pejtersen, L Gunnarsen, and Otto Melchior Poulsen. Building-related symptoms and inflammatory potency of dust from office buildings. *Indoor Air*, 17(6), 2007.
- [256] Corinne Mandin, Marilena Trantallidi, Andrea Cattaneo, Nuno Canha, Victor G Mihucz, Tamás Szigeti, Rosanna Mabilia, Erica Perreca, Andrea Spinazzè, Serena Fossati, et al. Assessment of indoor air quality in office buildings across europe-the officair study. *Science of the Total Environment*, 579:169–178, 2017.
- [257] L Guo, Z Yang, L Zhang, S Wang, T Bai, Y Xiang, and E Long. Systematic review of the effects of environmental factors on virus inactivation: implications for coronavirus disease 2019. International Journal of Environmental Science and Technology, 18:2865–2878, 2021.
- [258] Wan Yang and Linsey C Marr. Dynamics of airborne influenza a viruses indoors and dependence on humidity. *PloS One*, 6(6):e21481, 2011.
- [259] Kenichi Azuma, Naoki Kagi, U Yanagi, Hoon Kim, and Haruki Osawa. A longitudinal study on the effects of hygro-thermal conditions and indoor air

pollutants on building-related symptoms in office buildings. *Indoor Air*, 32 (11):e13164, 2022.

- [260] Peder Wolkoff, Kenichi Azuma, and Paolo Carrer. Health, work performance, and risk of infection in office-like environments: The role of indoor temperature, air humidity, and ventilation. *International Journal of Hygiene and Environmental Health*, 233:113709, 2021.
- [261] Karen Head, Mike Clarke, Meghan Bailey, Alicia Livinski, Ramona Ludolph, Ambrish Singh, World Health Organization, et al. Who housing and health guidelines: web annex d: report of the systematic review on the effect of indoor heat on health. 2018.
- [262] Mihail Solomonovic Goromosov, World Health Organization, et al. The physiological basis of health standards for dwellings. World Health Organization, 1968.
- [263] Stephen S Cheung, Jason KW Lee, and Juha Oksa. Thermal stress, human performance, and physical employment standards. *Applied Physiology*, *Nutrition, and Metabolism*, 41(6):S148–S164, 2016.
- [264] Fan Zhang, Richard de Dear, and Peter Hancock. Effects of moderate thermal environments on cognitive performance: A multidisciplinary review. *Applied energy*, 236:760–777, 2019.
- [265] Pawel Wargocki and David P Wyon. Ten questions concerning thermal and indoor air quality effects on the performance of office work and schoolwork. *Building and Environment*, 112:359–366, 2017.
- [266] Shi-Jie Cao and Hua-Yan Deng. Investigation of temperature regulation effects on indoor thermal comfort, air quality, and energy savings toward green residential buildings. *Science and Technology for the Built Environment*, 25(3):309–321, 2019.
- [267] Bjarne W Olesen. Indoor environment-health-comfort and productivity. Proceedings of Clima, 2005.
- [268] Ioannis Nezis, George Biskos, Konstantinos Eleftheriadis, and Olga-Ioanna Kalantzi. Particulate matter and health effects in offices-a review. *Building* and Environment, 156:62–73, 2019.

- [269] Chiara Copat, Antonio Cristaldi, Maria Fiore, Alfina Grasso, Pietro Zuccarello, Salvatore Santo Signorelli, Gea Oliveri Conti, and Margherita Ferrante. The role of air pollution (pm and no2) in covid-19 spread and lethality: a systematic review. *Environmental Research*, 191:110129, 2020.
- [270] Simona Kirešováa, Milan Guzana, and Volodymyr Rusynb. Particulate matter pm2. 5 and pm10 and its impact on air quality in urban and rural areas. In CEUR Workshop Proceedings, volume 3309, pages 329–337, 2022.
- [271] Mohd Shahrul Mohd Nadzir, Maggie Chel Gee Ooi, Kemal Maulana Alhasa, Mohd Aftar Abu Bakar, Anis Asma Ahmad Mohtar, Mohd Fadzil Firdzaus Mohd Nor, Mohd Talib Latif, Haris Hafizal Abd Hamid, Sawal Hamid Md Ali, Noratiqah Mohd Ariff, et al. The impact of movement control order (mco) during pandemic covid-19 on local air quality in an urban area of klang valley, malaysia. Aerosol and Air Quality Research, 20(6): 1237–1248, 2020.
- [272] Grazia M Marcazzan, Stefano Vaccaro, Gianluigi Valli, and Roberta Vecchi. Characterisation of pm10 and pm2. 5 particulate matter in the ambient air of milan (italy). Atmospheric Environment, 35(27):4639–4650, 2001.
- [273] Yan-Lin Zhang and Fang Cao. Is it time to tackle pm2. 5 air pollutions in china from biomass-burning emissions? *Environmental Pollution*, 202: 217–219, 2015.
- [274] Piet Eichholtz, Nils Kok, and Xudong Sun. The effect of post-covid-19 ventilation measures on indoor air quality in primary schools. *PNAS Nexus*, 3(1):pgad429, 2024.
- [275] Ljiljana Stošić and Dušica Stojanović. The importance of monitoring of suspended particles in the ambient air of the city of niš. Acta Facultatis Medicae Naissensis, 36(3):229–234, 2019.
- [276] Alberto Izzotti, Paola Spatera, Zumama Khalid, and Alessandra Pulliero. Importance of punctual monitoring to evaluate the health effects of airborne particulate matter. International Journal of Environmental Research and Public Health, 19(17):10587, 2022.

- [277] Akanksha Priyadarshini, Narendra Dehury, and Amiya Kumar Samantaray. A real time portable embedded system design for particulate matter monitoring. In 2015 IEEE Bombay Section Symposium (IBSS), pages 1–5. IEEE, 2015.
- [278] Xingyue Fang, Ruidong Chang, Yanquan Zhang, Jian Zuo, Yang Zou, and Yilong Han. Monitoring airborne particulate matter from building construction: A systematic review. *Journal of Building Engineering*, page 108708, 2024.
- [279] Fátima Felgueiras, Zenaida Mourão, André Moreira, and Marta Fonseca Gabriel. A systematic review of ventilation conditions and airborne particulate matter levels in urban offices. *Indoor Air*, 32(11):e13148, 2022.
- [280] Sumei Liu, Mike Koupriyanov, Dale Paskaruk, Graham Fediuk, and Qingyan Chen. Investigation of airborne particle exposure in an office with mixing and displacement ventilation. *Sustainable Cities and Society*, 79: 103718, 2022.
- [281] Francesco Chirico, Angelo Sacco, Nicola Luigi Bragazzi, and Nicola Magnavita. Can air-conditioning systems contribute to the spread of sars/mers/covid-19 infection? insights from a rapid review of the literature. International Journal of Environmental Research and Public Health, 17(17):6052, 2020.
- [282] Elizabeth C Lee, Nikolas I Wada, M Kate Grabowski, Emily S Gurley, and Justin Lessler. The engines of sars-cov-2 spread. *Science*, 370(6515): 406–407, 2020.
- [283] Valentyn Stadnytskyi, Christina E Bax, Adriaan Bax, and Philip Anfinrud. The airborne lifetime of small speech droplets and their potential importance in sars-cov-2 transmission. *Proceedings of the National Academy of Sciences*, 117(22):11875–11877, 2020.
- [284] Jianyun Lu, Jieni Gu, Kuibiao Li, Conghui Xu, Wenzhe Su, Zhisheng Lai, Deqian Zhou, Chao Yu, Bin Xu, and Zhicong Yang. Covid-19 outbreak associated with air conditioning in restaurant, guangzhou, china, 2020. Emerging Infectious Diseases, 26(7):1628, 2020.

- [285] Alyssa C Fears, William B Klimstra, Paul Duprex, Amy Hartman, Scott C Weaver, Kenneth S Plante, Divya Mirchandani, Jessica Ann Plante, Patricia V Aguilar, Diana Fernández, et al. Persistence of severe acute respiratory syndrome coronavirus 2 in aerosol suspensions. *Emerging Infectious Diseases*, 26(9):2168, 2020.
- [286] Sheng Zhang, MengYuan Diao, Wenbo Yu, Lei Pei, Zhaofen Lin, and Dechang Chen. Estimation of the reproductive number of novel coronavirus (covid-19) and the probable outbreak size on the diamond princess cruise ship: A data-driven analysis. *International Journal of Infectious Diseases*, 93:201–204, 2020.
- [287] Pengcheng Xu, Wei Jia, Hua Qian, Shenglan Xiao, Te Miao, Hui-Ling Yen, Hongwei Tan, Min Kang, Benjamin J Cowling, and Yuguo Li. Lack of cross-transmission of sars-cov-2 between passenger's cabins on the diamond princess cruise ship. *Building and Environment*, 198:107839, 2021.
- [288] Yuguo Li, Xiuchang Huang, Ignatius TS Yu, Tze W Wong, and Hua Qian. Role of air distribution in sars transmission during the largest nosocomial outbreak in hong kong. *Indoor Air*, 15(2), 2005.
- [289] Sung-Han Kim, So Young Chang, Minki Sung, Ji Hoon Park, Hong Bin Kim, Heeyoung Lee, Jae-Phil Choi, Won Suk Choi, and Ji-Young Min. Extensive viable middle east respiratory syndrome (mers) coronavirus contamination in air and surrounding environment in mers isolation wards. *Reviews of Infectious Diseases*, 63(3):363–369, 2016.
- [290] Marcel Loomans, Jos De Leeuw, Gertjan Middendorf, and Peter V Nielsen. Covid-19 and recirculation. *REHVA Journal*, (5):5–9, 2020.
- [291] P Jacob Bueno de Mesquita, William W Delp, Wanyu R Chan, William P Bahnfleth, and Brett C Singer. Control of airborne infectious disease in buildings: Evidence and research priorities. *Indoor Air*, 32(1):e12965, 2022.
- [292] Yahya Sheikhnejad, Reihaneh Aghamolaei, Marzieh Fallahpour, Hamid Motamedi, Mohammad Moshfeghi, Parham A Mirzaei, and Hadi Bordbar. Airborne and aerosol pathogen transmission modeling of respiratory events in buildings: An overview of computational fluid dynamics. *Sustainable Cities and Society*, 79:103704, 2022.

- [293] Hamed Arjmandi, Reza Amini, Marzieh Fallahpour, et al. Minimizing the respiratory pathogen transmission: Numerical study and multi-objective optimization of ventilation systems in a classroom. *Thermal Science and Engineering Progress*, 28:101052, 2022.
- [294] Chang Heon Cheong, Beungyong Park, and Seong Ryong Ryu. Effect of under-floor air distribution system to prevent the spread of airborne pathogens in classrooms. *Case Studies in Thermal Engineering*, 28:101641, 2021.
- [295] Sungwan Son and Choon-Man Jang. Effects of internal airflow on iaq and cross-infection of infectious diseases between students in classrooms. Atmospheric Environment, 279:119112, 2022.
- [296] Mahdi Ahmadzadeh, Emad Farokhi, and Mehrzad Shams. Investigating the effect of air conditioning on the distribution and transmission of covid-19 virus particles. *Journal of Cleaner Production*, 316:128147, 2021.
- [297] Chen Ren, Shi-Jie Cao, and Fariborz Haghighat. A practical approach for preventing dispersion of infection disease in naturally ventilated room. *Journal of Building Engineering*, 48:103921, 2022.
- [298] Gerhard K Rencken, Emma K Rutherford, Nikhilesh Ghanta, John Kongoletos, and Leon Glicksman. Patterns of sars-cov-2 aerosol spread in typical classrooms. *Building and Environment*, 204:108167, 2021.
- [299] Mahshid Mirzaie, Esmail Lakzian, Afrasyab Khan, Majid Ebrahimi Warkiani, Omid Mahian, and Goodarz Ahmadi. Covid-19 spread in a classroom equipped with partition–a cfd approach. *Journal of Hazardous Materials*, 420:126587, 2021.
- [300] Mohamed Abuhegazy, Khaled Talaat, Osman Anderoglu, and Svetlana V Poroseva. Numerical investigation of aerosol transport in a classroom with relevance to covid-19. *Physics of Fluids*, 32(10), 2020.
- [301] Sebastian Burgmann and Uwe Janoske. Transmission and reduction of aerosols in classrooms using air purifier systems. *Physics of Fluids*, 33(3), 2021.

- [302] William Chung and Iris MH Yeung. A study of energy consumption of secondary school buildings in hong kong. *Energy and Buildings*, 226:110388, 2020.
- [303] Sourabh P Bhat, BV Kumar, Shainath Ramesh Kalamkar, Vinay Kumar, Sudhir Pathak, and Walter Schneider. Modeling and simulation of the potential indoor airborne transmission of sars-cov-2 virus through respiratory droplets. *Physics of Fluids*, 34(3), 2022.
- [304] Yuguo Li, Hua Qian, Jian Hang, Xuguang Chen, Pan Cheng, Hong Ling, Shengqi Wang, Peng Liang, Jiansen Li, Shenglan Xiao, et al. Probable airborne transmission of sars-cov-2 in a poorly ventilated restaurant. *Building* and Environment, 196:107788, 2021.
- [305] Sihwan Lee, Juyoun Lee, and Shinsuke Kato. Influence of vane angle on the effectiveness of air conditioning of wall-mounted split-type air conditioners in residential buildings. *Science and Technology for the Built Environment*, 23(5):761–775, 2017.
- [306] Weiwei Liu, Zhiwei Lian, and Ye Yao. Optimization on indoor air diffusion of floor-standing type room air-conditioners. *Energy and Buildings*, 40(2): 59–70, 2008.
- [307] Kwang-Chul Noh, Jae-Soo Jang, and Myung-Do Oh. Thermal comfort and indoor air quality in the lecture room with 4-way cassette air-conditioner and mixing ventilation system. *Building and Environment*, 42(2):689–698, 2007.
- [308] Y Cheng and Z Lin. Experimental investigation into the interaction between the human body and room airflow and its effect on thermal comfort under stratum ventilation. *Indoor Air*, 26(2):274–285, 2016.
- [309] Rafaela Mateus, José MC Pereira, and Armando Pinto. Natural ventilation of large air masses: Experimental and numerical techniques review. *Energy* and Buildings, 291:113120, 2023.
- [310] Ricardo MSF Almeida, Manuel Pinto, Paulo G Pinho, and Luís T de Lemos. Natural ventilation and indoor air quality in educational buildings: Experimental assessment and improvement strategies. *Energy Efficiency*, 10: 839–854, 2017.

- [311] Lin-Rui Jia, Jie Han, Xi Chen, Qing-Yun Li, Chi-Chung Lee, and Yat-Hei Fung. Interaction between thermal comfort, indoor air quality and ventilation energy consumption of educational buildings: A comprehensive review. *Buildings*, 11(12):591, 2021.
- [312] Gail Brager and Lindsay Baker. Occupant satisfaction in mixed-mode buildings. Building Research & Information, 37(4):369–380, 2009.
- [313] Adams Rackes, Ana Paula Melo, and Roberto Lamberts. Naturally comfortable and sustainable: Informed design guidance and performance labeling for passive commercial buildings in hot climates. *Applied Energy*, 174: 256–274, 2016.
- [314] Ihab Jabbar Al Rikabi, Mohamed Ali Abuelnour, Adil AM Omara, Najib Aminu Ismail, Farid Nasir Ani, and Abuelnuor AA Abuelnuor. A numerical study of the effect of façade opening on nighttime ventilation of a low-energy building. In 2022 8th International Engineering Conference on Sustainable Technology and Development (IEC), pages 129–135. IEEE, 2022.
- [315] Hardi K Abdullah and Halil Z Alibaba. Open-plan office design for improved natural ventilation and reduced mixed mode supplementary loads. *Indoor* and Built Environment, 31(8):2145–2167, 2022.
- [316] Xiang Deng and Zijing Tan. Numerical analysis of local thermal comfort in a plan office under natural ventilation. *Indoor and Built Environment*, 29 (7):972–986, 2020.
- [317] Yabin Jia, Xu Zheng, Jun Guan, Xue Tan, Shuqin Chen, Bing Qi, and Huiyan Zheng. Investigation of indoor total volatile organic compound concentrations in densely occupied university buildings under natural ventilation: Temporal variation, correlation and source contribution. *Indoor* and Built Environment, 30(6):838–850, 2021.
- [318] Yan Wu, Naiping Gao, Jianlei Niu, Jianbin Zang, and Qiang Cao. Numerical study on natural ventilation of the wind tower: Effects of combining with different window configurations in a low-rise house. Building and Environment, 188:107450, 2021.

- [319] Zhiqiang Zhai. Facial mask: A necessity to beat covid-19. Building and Environment, 175:106827, 2020.
- [320] Alessandro Zivelonghi and Massimo Lai. Mitigating aerosol infection risk in school buildings: the role of natural ventilation, volume, occupancy and co2 monitoring. *Building and Environment*, 204:108139, 2021.
- [321] WHO Scientific Brief. Transmission of sars-cov-2: implications for infection prevention precautions. World Health Organization, 2020.
- [322] YW Dai, Cheuk Ming Mak, and ZT Ai. Computational fluid dynamics simulation of wind-driven inter-unit dispersion around multi-storey buildings: Upstream building effect. *Indoor and Built Environment*, 28(2):217–234, 2019.
- [323] Wei Liu and Xue-Yi You. Transportation and risk analysis of influenza indoor and outdoor transportation and exposure risk analysis of influenza aerosol. *Indoor and Built Environment*, 21(5):614–621, 2012.
- [324] Lidia Morawska and Junji Cao. Airborne transmission of sars-cov-2: The world should face the reality. *Environment International*, 139:105730, 2020.
- [325] James W Axley and Steven J Emmerich. A method to assess the suitability of a climate for natural ventilation of commercial buildings. In *Proceedings: Indoor Air (2002), Monterey, CA*, 2002.
- [326] Haihua Zhang, Dong Yang, Vivian WY Tam, Yao Tao, Guomin Zhang, Sujeeva Setunge, and Long Shi. A critical review of combined natural ventilation techniques in sustainable buildings. *Renewable and Sustainable Energy Reviews*, 141:110795, 2021.
- [327] Tom Lipinski, Darem Ahmad, Nicolas Serey, and Hussam Jouhara. Review of ventilation strategies to reduce the risk of disease transmission in high occupancy buildings. *International Journal of Thermofluids*, 7:100045, 2020.
- [328] Hannah L Gough, Janet F Barlow, Zhiwen Luo, M-F King, Christos H Halios, and CSB Grimmond. Evaluating single-sided natural ventilation models against full-scale idealised measurements: Impact of wind direction and turbulence. *Building and Environment*, 170:106556, 2020.

- [329] Huai-Yu Zhong, Yang Sun, Jin Shang, Fu-Ping Qian, Fu-Yun Zhao, Hideki Kikumoto, Carlos Jimenez-Bescos, and Xiaochen Liu. Single-sided natural ventilation in buildings: a critical literature review. *Building and Environ*ment, 212:108797, 2022.
- [330] PJ Irga and FR Torpy. Indoor air pollutants in occupational buildings in a sub-tropical climate: Comparison among ventilation types. *Building and Environment*, 98:190–199, 2016.
- [331] Günsu Merin Abbas and Ipek Gursel Dino. The impact of natural ventilation on airborne biocontaminants: a study on covid-19 dispersion in an open office. Engineering, Construction and Architectural Management, 29 (4):1609–1641, 2022.
- [332] Liangquan Hu, Yong-Feng Ma, Farzad Pourfattah, Weiwei Deng, and Lian-Ping Wang. Numerical study of cough droplet transmission in an indoor environment. *Physics of Fluids*, 35(11), 2023.
- [333] Francesca Avella, Paolo Bonato, Annamaria Belleri, and Francesco Babich. Ventilation of residential buildings in alpine region: a comparison between natural, mechanical, and mixed-mode strategies. In *Building Simulation Applications BSA 2022: 5th IBPSA-Italy conference Bozen-Bolzano; 29th June-1st July 2022*, pages 313–320, 2022.
- [334] James F Montgomery, Stefan Storey, and Karen Bartlett. Comparison of the indoor air quality in an office operating with natural or mechanical ventilation using short-term intensive pollutant monitoring. *Indoor and Built Environment*, 24(6):777–787, 2015.
- [335] Leslie Kojo Norvihoho, Jing Yin, Xin-Gang Zhu, Shu-Qin Xue, Hai-Tao Yu, and Zhi-Fu Zhou. The effects of ventilation layout on cough droplet dynamics relating to seasonal influenza. *Physics of Fluids*, 36(11), 2024.
- [336] Atefeh Abbaspour, Ali Bahadori-Jahromi, Alan Janbey, Paulina B Godfrey, and Shiva Amirkhani. Enhancing indoor air quality and regulatory compliance: An in-depth comparative study on ventilation strategies and their impact on sars-cov-2 transmission risk. *Sustainability*, 16(1):271, 2024.
- [337] David S Fishman and Stephen L Pimbert. The thermal environment in offices. *Energy and Buildings*, 5(2):109–116, 1982.

- [338] Zi-Meng Kong, Harpal Singh Sandhu, Lu Qiu, Jicheng Wu, Wen-Jun Tian, Xiao-Jing Chi, Zhi Tao, Chi-Fu Jeffrey Yang, and Xiao-Jia Wang. Virus dynamics and decay in evaporating human saliva droplets on fomites. *Environmental Science & Technology*, 57(46):17737–17750, 2022.
- [339] Darrah K Sleeth and James H Vincent. Inhalability for aerosols at ultra-low windspeeds. In *Journal of Physics: Conference Series*, volume 151, page 012062. IOP Publishing, 2009.